# Differential object marking in Hungarian and the morphosyntax of case and agreement

András Bárány

Downing College, University of Cambridge

November 2015

This dissertation is submitted for the degree of Doctor of Philosophy.

Voor oESCC

De	clara	tion	ix
Ac	know	ledgements	xi
Ab	brevi	ations	xiii
Lis	t of T	ables	xv
Lis	t of F	igures 2	xvii
1	DOM	, case and agreement	1
	1.1	Introduction	1
	1.2	Differential object marking	2
	1.3	Person features and hierarchies	5
		1.3.1 Hierarchies and functional approaches to DOM	9
	1.4	Case and agreement	10
	1.5	Theoretical assumptions	14
		1.5.1 Cyclic Agree	14
		1.5.2 Agree can fail	17
		1.5.3 Syntax and morphology	18
	1.6	The sample of languages	21
Ра	rt I	Differential object marking in Hungarian	23
2	DOM	in Hungarian	25
	2.1	Introduction: Hungarian object agreement	25

2.2.1       Direct objects and subject agreement         2.2.2       Direct objects that trigger object agreement         2.3       "Unexpected" object agreement         2.3       Summary         2.3       Summary         2.3       Summary         3       A hybrid analysis of object agreement: syntactic structure and         3.1       Introduction         3.2       Towards an analysis         3.2.1       Problems for semantic approaches         3.2.2       Problems for syntactic approaches         3.2.3       Syntactic structure and person features         3.3       Evidence from possessive noun phrases in Hungarian         3.3.1       Types of possessors: nominative, dative, pronomina         3.3.2       Non-specific possessives and dative possessors         3.3.3       Possessed noun phrases and object agreement         3.3.4       The structure of possessed noun phrases         3.3.5       Possessed direct objects in Standard Hungarian         3.4.1       Object agreement and person features         3.4.2       Deriving agreement         3.4.3       Putting the pieces together: agreement in the clause         3.4.4       Interim summary         3.5       Conclusions         3.5		2.2	The di	stribution of object agreement	27
2.2.2       Direct objects that trigger object agreement         2.3       "Unexpected" object agreement         2.3       Summary         2.3       Summary         3       A hybrid analysis of object agreement: syntactic structure and         3.1       Introduction         3.2       Towards an analysis         3.2.1       Problems for semantic approaches         3.2.2       Problems for syntactic approaches         3.2.3       Syntactic structure and person features         3.3       Evidence from possessive noun phrases in Hungarian         3.3.1       Types of possessors: nominative, dative, pronomina         3.3.2       Non-specific possessives and dative possessors         3.3.3       Possessed noun phrases and object agreement         3.3.4       The structure of possessed noun phrases         3.3.5       Possessed direct objects in Standard Hungarian         3.3.6       Conclusions: syntactic structure and person features         3.4       In terim summary         3.4.1       Object agreement and person features         3.4.2       Deriving agreement         3.4.3       Putting the pieces together: agreement in the clause         3.4.4       Interim summary         3.5       Conclusions			2.2.1	Direct objects and subject agreement	28
2.2.3       "Unexpected" object agreement         2.3       Summary         2.3       Summary         3       A hybrid analysis of object agreement: syntactic structure and         3.1       Introduction         3.2       Towards an analysis         3.2.1       Problems for semantic approaches         3.2.2       Problems for syntactic approaches         3.2.3       Syntactic structure and person features         3.3       Evidence from possessive noun phrases in Hungarian         3.3.1       Types of possessors: nominative, dative, pronomina         3.3.2       Non-specific possessives and dative possessors         3.3.3       Possessed noun phrases and object agreement         3.3.4       The structure of possessed noun phrases         3.3.5       Possessed direct objects in Standard Hungarian         3.3.6       Conclusions: syntactic structure and person features         3.4       The structure of possessed noun phrases         3.4.1       Object agreement and person features         3.4.2       Deriving agreement         3.4.3       Putting the pieces together: agreement in the clause         3.4.4       Interim summary         3.5       Conclusions         3.6       Conclusions <td< th=""><th></th><th></th><th>2.2.2</th><th>Direct objects that trigger object agreement</th><th>33</th></td<>			2.2.2	Direct objects that trigger object agreement	33
<ul> <li>2.3 Summary</li></ul>			2.2.3	"Unexpected" object agreement	43
<ul> <li>3 A hybrid analysis of object agreement: syntactic structure and <ol> <li>Introduction</li> <li>Towards an analysis</li> <li>Towards an analysis</li> <li>Problems for semantic approaches</li> <li>3.2.1 Problems for syntactic approaches</li> <li>3.2.2 Problems for syntactic approaches</li> <li>3.2.3 Syntactic structure and person features</li> <li>3.3 Evidence from possessive noun phrases in Hungarian</li> <li>3.3.1 Types of possessors: nominative, dative, pronomina</li> <li>3.3.2 Non-specific possessives and dative possessors</li> <li>3.3 Possessed noun phrases and object agreement</li> <li>3.4 The structure of possessed noun phrases</li> <li>3.5 Possessed direct objects in Standard Hungarian</li> <li>3.6 Conclusions: syntactic structure and person features</li> <li>3.4.1 Object agreement and person features</li> <li>3.4.2 Deriving agreement</li> <li>3.4.3 Putting the pieces together: agreement in the clause</li> <li>3.4.4 Interim summary</li> <li>3.5 Conclusions</li> <li>4.1 Introduction</li> <li>4.2 Object agreement with personal pronouns</li> <li>4.1 Introduction</li> <li>4.2.1 The distribution of object agreement with personal pronouns</li> <li>4.3 Deriving inverse agreement in Hungarian</li> <li>4.3.1 Arguments for covert agreement</li> </ol></li></ul>		2.3	Summ	ary	45
<ul> <li>3.1 Introduction</li></ul>	3	A hy	brid an	alysis of object agreement: syntactic structure and $\pi$ -features	47
<ul> <li>3.2 Towards an analysis</li></ul>		3.1	Introd	uction	47
3.2.1       Problems for semantic approaches         3.2.2       Problems for syntactic approaches         3.2.3       Syntactic structure and person features         3.3       Evidence from possessive noun phrases in Hungarian         3.3.1       Types of possessors: nominative, dative, pronomina         3.3.2       Non-specific possessives and dative possessors         3.3.3       Possessed noun phrases and object agreement         3.3.4       The structure of possessed noun phrases         3.3.5       Possessed direct objects in Standard Hungarian         3.3.6       Conclusions: syntactic structure and person features         3.4       An implementation of object agreement         3.4.1       Object agreement and person features         3.4.2       Deriving agreement         3.4.3       Putting the pieces together: agreement in the clause         3.4.4       Interim summary         3.5       Conclusions         3.6       Conclusions         3.7       Putting the pieces together: agreement in the clause         3.4.2       Deriving agreement         3.4.3       Putting the pieces together: agreement in the clause         3.4.4       Interim summary         3.5       Conclusions         4.1       Introduction		3.2	Towar	ds an analysis	48
3.2.2       Problems for syntactic approaches         3.2.3       Syntactic structure and person features         3.3       Evidence from possessive noun phrases in Hungarian         3.3.1       Types of possessors: nominative, dative, pronomina         3.3.2       Non-specific possessives and dative possessors         3.3.3       Possessed noun phrases and object agreement         3.3.4       The structure of possessed noun phrases         3.3.5       Possessed direct objects in Standard Hungarian         3.3.6       Conclusions: syntactic structure and person features         3.4       An implementation of object agreement         3.4.1       Object agreement and person features         3.4.2       Deriving agreement         3.4.3       Putting the pieces together: agreement in the clause         3.4.4       Interim summary         3.5       Conclusions         3.6       Conclusions         3.7       Postestont         3.8       Putting the pieces together: agreement in the clause         3.4.4       Interim summary         3.5       Conclusions         4       Agreement with personal pronouns         4.1       Introduction         4.2       Object agreement with personal pronouns			3.2.1	Problems for semantic approaches	48
<ul> <li>3.2.3 Syntactic structure and person features</li></ul>			3.2.2	Problems for syntactic approaches	50
<ul> <li>3.3 Evidence from possessive noun phrases in Hungarian 3.3.1 Types of possessors: nominative, dative, pronomina 3.3.2 Non-specific possessives and dative possessors 3.3.3 Possessed noun phrases and object agreement 3.3.4 The structure of possessed noun phrases 3.3.5 Possessed direct objects in Standard Hungarian</li></ul>			3.2.3	Syntactic structure and person features	53
3.3.1       Types of possessors: nominative, dative, pronominal         3.3.2       Non-specific possessives and dative possessors         3.3.3       Possessed noun phrases and object agreement         3.3.4       The structure of possessed noun phrases         3.3.5       Possessed direct objects in Standard Hungarian		3.3	Evider	nce from possessive noun phrases in Hungarian	58
<ul> <li>3.3.2 Non-specific possessives and dative possessors 3.3.3 Possessed noun phrases and object agreement 3.3.4 The structure of possessed noun phrases 3.3.5 Possessed direct objects in Standard Hungarian 3.3.6 Conclusions: syntactic structure and person features 3.4.1 Object agreement and person features</li></ul>			3.3.1	Types of possessors: nominative, dative, pronominal	58
<ul> <li>3.3.3 Possessed noun phrases and object agreement</li></ul>			3.3.2	Non-specific possessives and dative possessors	61
<ul> <li>3.3.4 The structure of possessed noun phrases</li></ul>			3.3.3	Possessed noun phrases and object agreement	63
<ul> <li>3.3.5 Possessed direct objects in Standard Hungarian</li></ul>			3.3.4	The structure of possessed noun phrases	65
<ul> <li>3.3.6 Conclusions: syntactic structure and person features</li> <li>3.4 An implementation of object agreement</li></ul>			3.3.5	Possessed direct objects in Standard Hungarian	69
<ul> <li>3.4 An implementation of object agreement</li></ul>			3.3.6	Conclusions: syntactic structure and person features	75
<ul> <li>3.4.1 Object agreement and person features</li></ul>		3.4	An im	plementation of object agreement	76
<ul> <li>3.4.2 Deriving agreement</li></ul>			3.4.1	Object agreement and person features	76
<ul> <li>3.4.3 Putting the pieces together: agreement in the clause 3.4.4 Interim summary</li></ul>			3.4.2	Deriving agreement	79
3.4.4       Interim summary         3.5       Conclusions         4       Agreement with personal pronouns         4.1       Introduction         4.2       Object agreement with personal pronouns         4.2.1       The distribution of object agreement with personal p         4.3       Deriving inverse agreement in Hungarian         4.3.1       Arguments for covert agreement         4.3.2       Third person arguments			3.4.3	Putting the pieces together: agreement in the clause	91
<ul> <li>3.5 Conclusions</li></ul>			3.4.4	Interim summary	93
<ul> <li>4 Agreement with personal pronouns</li> <li>4.1 Introduction</li></ul>		3.5	Conclu	usions	94
<ul> <li>4.1 Introduction</li></ul>	4	Agre	ement	with personal pronouns	97
<ul> <li>4.2 Object agreement with personal pronouns</li></ul>		4.1	Introd	uction	97
<ul> <li>4.2.1 The distribution of object agreement with personal p</li> <li>4.3 Deriving inverse agreement in Hungarian</li></ul>		4.2	Object	agreement with personal pronouns	98
<ul> <li>4.3 Deriving inverse agreement in Hungarian</li></ul>			4.2.1	The distribution of object agreement with personal pronouns .	101
<ul><li>4.3.1 Arguments for covert agreement</li></ul>		4.3	Derivi	ng inverse agreement in Hungarian	106
4.3.2 Third person arguments			4.3.1	Arguments for covert agreement	115
1 8			4.3.2	Third person arguments	119

		4.3.3	Interim summary	121
	4.4	Spellir	ng out agreement	122
		4.4.1	Hungarian verb morphology and spell-out rules	123
		4.4.2	Revised spell-out rules	130
		4.4.3	Verbal agreement and possessive suffixes	132
		4.4.4	Further issues in verb morphology	136
		4.4.5	Interim summary	139
	4.5	Compa	arison to other analyses of object agreement	139
		4.5.1	Agreement with third person objects	140
		4.5.2	Agreement with first, second, and third person objects $\ldots$ .	141
		4.5.3	Inclusive reference, again	146
		4.5.4	Interim summary	151
	4.6	Summ	ary and conclusions	151
	4.A	Appen	dix: Full derivations	152
		4.A.1	Direct configuration	152
		4.A.2	Inverse configuration	154
		4.A.3	Two third person arguments	155
Pa	rt II	Case	and agreement in the grammar	159
Ра	rt II	Case	and agreement in the grammar	159
Pa 5	rt II Inve	Case a	and agreement in the grammar eement and global case splits	159 161
Pa 5	rt II Inve 5.1	Case a rse agre Introd	and agreement in the grammar eement and global case splits uction	<b>159</b> <b>161</b> 161
Pa 5	rt II Inve 5.1 5.2	Case a rse agre Introd Case s	and agreement in the grammar eement and global case splits uction	<b>159</b> <b>161</b> 161 164
Pa 5	rt II Inve 5.1 5.2	Case a rse agro Introd Case s 5.2.1	and agreement in the grammar eement and global case splits uction	<b>159</b> <b>161</b> 161 164
Pa 5	rt II Inve 5.1 5.2	Case a rse agro Introd Case s 5.2.1 5.2.2	and agreement in the grammar eement and global case splits uction	<b>159</b> <b>161</b> 164 164 164
Pa 5	rt II Inve 5.1 5.2 5.3	Case a rse agro Introd Case s 5.2.1 5.2.2 Case a	and agreement in the grammar eement and global case splits uction	<b>159</b> <b>161</b> 164 164 167 168
Pa 5	rt II Inve 5.1 5.2 5.3	Case a rse agre Introd Case s 5.2.1 5.2.2 Case a 5.3.1	and agreement in the grammar         eement and global case splits         uction	<b>159</b> <b>161</b> 164 164 167 168 174
Pa 5	rt II Inve 5.1 5.2 5.3 5.4	Case a rse agro Introd Case s 5.2.1 5.2.2 Case a 5.3.1 Analys	and agreement in the grammar         eement and global case splits         uction         tudies: Inverse agreement and global case splits         Kashmiri         Sahaptin	<b>159</b> <b>161</b> 164 164 167 168 174 177
Pa 5	rt II Inve 5.1 5.2 5.3 5.4	Case a rse agro Introd Case s 5.2.1 5.2.2 Case a 5.3.1 Analys 5.4.1	and agreement in the grammar         eement and global case splits         uction         tudies: Inverse agreement and global case splits         Kashmiri         Sahaptin         Sainaptin         CASE features, Case assignment and impoverishment         sis         Global case splits on subject and object	159 161 164 164 164 167 168 174 177 178
Pa 5	rt II Inve 5.1 5.2 5.3 5.4	Case a rse agro Introd Case s 5.2.1 5.2.2 Case a 5.3.1 Analys 5.4.1 5.4.2	and agreement in the grammar         eement and global case splits         uction         tudies: Inverse agreement and global case splits         Kashmiri         Sahaptin         Sahaptin         CASE features, Case assignment and impoverishment         sis         Global case splits on subject and object         Global case splits and animacy	159 161 164 164 164 167 168 174 177 178 189
Pa 5	rt II Inve 5.1 5.2 5.3 5.4	Case a rse agre Introd Case s 5.2.1 5.2.2 Case a 5.3.1 Analys 5.4.1 5.4.2 5.4.3	and agreement in the grammar         eement and global case splits         uction	159 161 164 164 167 168 174 177 178 189 199
Pa 5	rt II Inve 5.1 5.2 5.3 5.4 5.5	Case a rse agro Introd Case s 5.2.1 5.2.2 Case a 5.3.1 Analys 5.4.1 5.4.2 5.4.3 Discus	and agreement in the grammar         eement and global case splits         uction         tudies: Inverse agreement and global case splits         Kashmiri         Sahaptin         ssignment and cyclicity         CASE features, Case assignment and impoverishment         sis         Global case splits on subject and object         Global case splits and animacy         Crossing the distinction between direct and inverse	159 161 164 164 167 168 174 177 178 189 199 211
Pa 5	rt II Inve 5.1 5.2 5.3 5.4 5.5	Case a rse agro Introd Case s 5.2.1 5.2.2 Case a 5.3.1 Analys 5.4.1 5.4.2 5.4.3 Discus 5.5.1	and agreement in the grammar         eement and global case splits         uction         tudies: Inverse agreement and global case splits         Kashmiri         Sahaptin         Sahaptin         Asse features, Case assignment and impoverishment         sis         Global case splits on subject and object         Crossing the distinction between direct and inverse         sion and conclusions         Other approaches to global case splits	159 161 164 164 167 168 174 177 178 189 199 211 211

		5.5.2	Functional heads and probes	215
		5.5.3	Conclusions	216
6	Case	e and ag	greement in the grammar	217
	6.1	Introd	uction	217
	6.2	Case, a	agreement and Bobaljik's generalisations	218
		6.2.1	Hindi and Nepali	219
		6.2.2	Bobalijk's generalisations	220
		6.2.3	Marathi	223
	6.3	Analy	sis	225
		6.3.1	Theoretical assumptions	226
		6.3.2	Case and CASE features	227
		6.3.3	Deriving agreement in Hindi, Nepali and Marathi	230
		6.3.4	Languages with two $\phi\text{-}probes$ $\hdots$	243
		6.3.5	Splits between one and two probes	252
		6.3.6	A language with three $\varphi$ -probes: Kashmiri $\ldots \ldots \ldots$	257
	6.4	Conclu	usions	262
7	Para	meters	of DOM	265
	7.1	A hier	archical approach to parameters	265
	7.2	Cross-	linguistic variation in DOM	267
		7.2.1	Modelling dependencies	269
	7.3	Conclu	usions – Parameters of DOM?	272
8	Con	clusion	S	275
Bil	bliogr	aphy		277
Au	thor i	index		297

## Declaration

This dissertation is the result of my own work and includes nothing which is the outcome of work done in collaboration except as declared in the Preface and specified in the text.

It is not substantially the same as any that I have submitted, or, is being concurrently submitted for a degree or diploma or other qualification at the University of Cambridge or any other University or similar institution except as declared in the Preface and specified in the text. I further state that no substantial part of my dissertation has already been submitted, or, is being concurrently submitted for any such degree, diploma or other qualification at the University of Cambridge or any other University of similar institution except as declared in the Preface and specified in the text.

It does not exceed the prescribed word limit for the relevant Degree Committee.

## Acknowledgements

The research reported in this dissertation was funded by the European Research Council Advanced Grant No. 269752 "Rethinking Comparative Syntax" (ReCoS). My deepest gratitude goes to my fellow members on this project. First, to my supervisors Ian Roberts, Theresa Biberauer and Michelle Sheehan. I have learned an incredible amount from each one of you and I am very grateful for everything — I could not have had a better team of supervisors.

Other members of ReCoS might not have read, commented on and re-read all parts of this dissertation, but I have received a great amount of support from everyone. Thanks to Jenneke van der Wal, for always being helpful, uplifting, motivating, fun and a little bit crazy. Thanks to Georg Höhn for many, many conversations and for listening to and commenting on many of my half-baked ideas. Thanks to Alison Biggs and Tim Bazalgette for sharing their wisdom when I started in Cambridge and for being excellent role-models in many ways. And thanks to Anders Holmberg for further inspiration, and bringing another Finno-Ugric language to ReCoS. I am very happy to have been able to work with all of you over the past three years.

Dora Alexopoulou and Adam Ledgeway helped me get on track in my first year in Cambridge, and Adam has been an encouraging and motivating advisor since. Visitors and part-time members of ReCoS whose company and input over the years I appreciated a lot are Sten Vikner, Maia Duguine, Aritz Irurtzun, and Gereon Müller, who were guests at one time or another (or at one time *and* another, like Sten).

ReCoS also generously allowed me to spend time away from Cambridge, travelling to conferences and summer schools. I want to thank Katalin É. Kiss, Anna Szabolcsi, Laura Kalin, and Jaklin Kornfilt for valuable discussions at various places. I am also grateful to Klaus von Heusinger for welcoming me at the University of Cologne for a few months, and to Martine Grice, who let me work in her department. And thanks to Umut Özge, who was a great office mate.

#### Acknowledgements

But what's writing a dissertation without the company of other PhD students? I am grateful to Eleni Savva for countless linguistic and non-linguistic conversations and for organising a Christmas party with me that some would call legendary (others would call it something else), and to Anna Jespersen, because we talked about the things that really matter, like rodent vocalisation, hangovers and cats in party hats doing Praat. I'll miss you guys! (And no, I haven't considered prosody.)

Others in the faculty I've had great times with are Luca Sbordone, Rodanthi Christofaki, Damar Hoogland, Jessica Brown, Norma Schifano, Giusi Silvestri, Lucy Hosker, Iain Mobbs, and Tom Rowley. In Vienna, at the LSA Institute in Ann Arbor, at WCCFL 32 at USC, and at ESSLLI 2014 in Tübingen, I had lots of fun and pizza with Ulli Steindl and Thomas Borer, both of whom played a role in getting me to where I am. Danke! Thanks to Ádám Szalontai for our (first) fruitful collaboration.

Outside of linguistics, thanks to Girish Venkata Nivarti for great conversations about Hindi and turbulence in coffee. A big bear hug to Bill Wood and MPhil Sitte, thanks to Camilla d'Angelo and Krit Sitathani for cosy times in Eden Street, and to Claudia Bedin and Romain Guion for a regular dose of real life.

Other people in the real world I am grateful to are Bernhard Romstorfer and Felix Leditzky — thanks for always providing me with distraction and encouragement along the way. Thanks to my family for judging Hungarian data and for being the clever people that they are. The older and wiser I grow, they keep having an edge on me. Thanks to my father in particular for his unconditional support over many years.

Finally, dankjewel, Anna, for being around for all of it, all the time. This is for you!

# Abbreviations

1	first person		
2	second person	ELA	elative
3	third person	ERG	ergative
А	agent-like argument of a	F	feminine
	canonical transitive verb	FOC	focus
ABL	ablative	FUT	future
ABS	absolutive		
ACC	accusative	GEN	genitive
ADV	adverbial		
ANTIP	antipassive	HAB	habitual aspect
APPL	applicative		
AUX	auxiliary	ILL	illative
		IND	indicative
CAUS	causative	INE	inessive
CL	clitic	INF	infinitive
СОМ	comitative	INS	instrumental
COND	conditional	INV	inverse
		ΙΟ	indirect object
DAT	dative	IPFV	imperfective aspect
DEF	definite		
DEM	demonstrative	LAT	lative

#### Abbreviations

locative	CDI	subject		
locative	SBJ	subject		
	SBJV	subjunctive		
masculine	SEC	secundative		
modal	SG	singular		
	SUP	superessive		
neuter				
negative	Т	theme- or patient-like		
nominative		internal argument of a		
non-past		ditransitive verb		
	TR	transitive		
object				
oblique	VM	verbal modifier		
obviative				
patient-like argument				
of a canonical transitive				
verb				
passive				
- perfective				
plural				
possessive				
perfect				
past				
recipient-like internal ar-				
gument of a ditransitive				
verb				
relative				
single argument of a ca-				
	notative masculine modal neuter negative nominative nominative non-past object oblique obviative patient-like argument of a canonical transitive verb passive perfective plural possessive perfect past recipient-like internal ar- gument of a ditransitive verb relative	Interative stars masculine sec modal sc sup neuter negative T nominative non-past TR object oblique vM obviative vM obviative vM obviative vM obviative verb passive perfective plural possessive perfect past recipient-like internal ar- gument of a ditransitive verb relative single argument of a ca-		

nonical intransitive verb

# List of Tables

1.1	Representation of person in a feature geometry
1.2	Sources of original data 22
2.1	Present indicative forms of <i>lát</i> 'to see' 26
2.2	Classes of noun phrases triggering subject and object agreement 29
4.1	Transitive singular agreement paradigm with personal pronouns in
	Hungarian
4.2	Distribution of person features in Hungarian direct and inverse contexts 124
4.3	Present and past indicative forms of <i>lát</i> 'to see'
4.4	<i>-j-</i> and <i>-t-</i> in present and past tense forms of <i>lát</i> 'to see'
5.1	Distribution of inverse dative in Kashmiri
5.2	Distribution of the inverse ergative with singular subjects in Sahaptin 169
5.3	Distribution of the inverse verb markers for singular subjects in Sahaptin 169
5.4	Distribution of person features on $v$
5.5	Distribution of object case-marking in Awtuw
5.6	Distribution of person features in Awtuw
5.7	Distribution of the SAP in Chukchi
5.8	Objectiv case suffixes in Yukaghir
6.1	Alignment and $\varphi$ -probes of languages discussed in this section 243
6.2	Alignment and $\phi\text{-}probes$ of languages discussed in this section 251
6.3	Alignment and $\phi\mbox{-}probes$ of languages discussed in this chapter 262

# List of Figures

1.1	Standard view of the order of syntax and morphology
1.2	Keine's proposed order of syntax and morphology
1.3	Geographical distribution of languages in this dissertation
5.1 5.2	Standard view of the order of syntax and morphology
7.1	Parameters of functional heads and $\varphi$ -probes

### 1.1 Introduction

This dissertation discusses patterns of differential object (and subject) marking in a number of different languages. In Part I (Chapters 2 to 4), I discuss differential object agreement in Hungarian in great detail and I provide a novel syntactic analysis of the distribution of agreement in this language. In Part II (Chapters 5 to 7), I discuss languages unrelated to Hungarian and I show that the analysis proposed in Part I can be extended to account for similar patterns of case-marking and agreement across and within languages. This cross-linguistic comparison leads me to develop an analysis of the interaction of case and agreement in the grammar which is able to capture regularities in subject and object agreement and case-marking across languages in a systematic and predictive way.

The structure of the dissertation is as follows. In this introductory chapter, I will briefly lay out the main claims of this dissertation before providing some theoretical background for the following chapters.

In Chapters 2 to 4, I argue for a novel analysis of object agreement in Hungarian. I concur with much previous work that definiteness is a good, but not a perfect predictor of object agreement in Hungarian. I propose instead that *person features* trigger object agreement. I argue that Hungarian, like other languages, makes a distinction between nominals that completely lack person features and nominals that have person features (following e.g. M. Richards 2008). Only the latter, I claim, trigger object agreement in Hungarian. In Chapter 4, I extend this analysis to first and second person objects. These are often neglected in studying Hungarian object agreement, since they do not trigger (overt) agreement on the verb. I argue, however, that all personal pronouns trigger object agreement and I provide an analysis of agreement that takes into

account the person of both the subject and the object to determine whether agreement is expressed morphologically or not.

In Chapter 5, I apply the analysis of the distribution of agreement with Hungarian personal pronouns to a number of unrelated languages: Sahaptin, Kashmiri, Awtuw and Fore, among others (see Section 1.6 for the sample of languages discussed in this dissertation). I show that all these languages provide further evidence for treating person features as formal features that grammaticalise properties like specificity or definiteness (as in Hungarian to some degree) or animacy (as in Awtuw and Fore). The patterns of case-marking and agreement in these languages show the same kind of interaction of the person of the subject and the object that is found in Hungarian.

In Chapter 6, I discuss the interaction of case and agreement in syntax and argue for their dissociation in order to account for the cross-linguistic variation of differential argument marking. While continuing to treat agreement as a syntactic operation (rather than a post-syntactic one), I show that cross-linguistic generalisations about the alignment of case-marking and verbal agreement suggested by Bobaljik (2008) naturally fit into the proposed architecture.

Finally, in Chapter 7, I model the variation I discuss in Chapters 2 to 6 as a parameter hierarchy, building on a recent approach to comparative syntax that includes insights from both language acquisition and typological variation (Roberts and Holmberg 2010; Roberts 2012; Biberauer *et al.* 2014; Biberauer, Roberts and Sheehan 2014; Sheehan 2014b; Bazalgette 2015).

The rest of this chapter is structured as follows. In Section 1.2, I provide a brief overview of the phenomenon of differential object marking. In Section 1.3, I discuss one of the main hypotheses I explore in this dissertation, namely that person features can grammaticalise referential and semantic properties like definiteness or animacy. Section 1.4 relates theoretical aspects of case and agreement to DOM. I summarise the theoretical assumptions I make in Section 1.5 and I discuss the sample of languages in this dissertation in Section 1.6.

### 1.2 Differential object marking

Differential object marking, or DOM, is a widespread phenomenon in the languages of the world which I define as in (1).

#### (1) Differential object marking

A proper subset of direct objects in a language is marked differently from the complementary subset of direct objects. The marked subset of direct objects is generally more definite, more animate, or more topical than the complementary subset.

(1) does not specify the means by which a proper subset of direct objects is marked — this underspecification is introduced on purpose, as I take DOM in case-marking and verbal agreement to be two possible expressions of the same phenomenon. (2) shows an example of differential object marking in verb morphology, from Hungarian.

- (2) a. *Én lát-ok egy kutyá-t.* I see-1sg.sbj a dog-Acc 'I see a dog.'
  - b. Én lát-om a kutyá-t. I see-1sg.obj the dog-Acc 'I see the dog.'

In these Hungarian examples, the indefinite object in (2a) appears with a verb form glossed as sBJ, indicating that it only shows subject agreement. The definite object in (2b) requires the verb form glossed as oBJ, indicating that a certain property of the direct object also influences verbal agreement, in addition to subject agreement (shown by the 1sG gloss in both examples). Note that the verb never only agrees with the object in Hungarian, object agreement always appears together with subject agreement. While definiteness is generally a good predictor of the presence of object agreement in Hungarian, I will argue that it is in fact the person features of the object that determine whether the verb agrees with it. I provide evidence for this claim Chapters 2 to 4.

Other than being underspecified with respect to the expression of DOM, (1) does not include reference to the function(s) of differential object marking either. I will mostly refrain from discussing such questions, i.e. *why* DOM is used the way it is, but see Section 1.3.1 below for a brief discussion (for general discussion, see Silverstein 1976; Bossong 1985; Comrie 1989; Bossong 1991; Lazard 2001; Newmeyer 2002a,b; Aissen 2003; de Hoop and Malchukov 2007; P. de Swart 2007; Haspelmath 2008; Malchukov 2008; Haspelmath 2009; Iemmolo 2012; with respect to Hungarian, see Bárány 2012a,b).

Instead, I will explore the consequences of differential object marking and differential object agreement for a theory of case and agreement in a (mostly) Minimalist framework (Chomsky 1995, 2000, 2001) with some significant modifications to the general architecture of the grammar.

A crucial hypothesis I discuss in the following chapters is based on a proposal by M. Richards (2004, 2008). M. Richards argues for a close connection between the notion of person on the one hand, and animacy and definiteness on the other hand: first and second person nominals are generally assumed to be both definite and animate, only third person nominals can be indefinite (and inanimate). He therefore suggests analysing person as "a (syntactic) property of definite and animate nominals only: a person specification on indefinites and inanimates is redundant and thus plausibly left unspecified." (M. Richards 2008: 140). I adopt and test this hypothesis throughout this dissertation and I argue that we find exactly those patterns of case-marking and agreement that this view of person predicts. Related proposals have been made by Longobardi (1994, 2005, 2008) who suggests that the syntactic expression of person features is associated with the head D, a position also taken up by M. Richards (2008).

I will also adopt this suggestion and argue that object agreement in Hungarian is in fact sensitive to person features expressed on D. Rather than suggesting that the direct objects in (2a) and (2b) trigger object agreement based on their definiteness, I suggest that it is the formal specification of these noun phrases that determines whether they agree or not. Concretely, I will argue in Chapters 2 to 4 that only the object in (2b) has a person feature, but not the object in (2a).

An example of a language in which animacy behaves in a similar way is Awtuw (Feldman 1986, discussed in Chapter 5). In this language, some animate direct objects always require an object case marker, but others only do when the subject is less or equally animate. This pattern is illustrated in (3). In (3a), *tey tale-re* 'the woman' is case-marked and interpreted as the direct object. If it lacks the case-marker, as in (3b), it is automatically interpreted as the subject, because *woman* as a human entity counts as more animate than *pig*, a non-human entity.<sup>1</sup>

<sup>&</sup>lt;sup>1</sup> Feldman (1986) uses FA for "factive".

1.3 Person features and hierarchies

- (3) a. *tey tale-re yaw d-æl-i* 3.F.SG woman-OBJ pig FA-bite-PST 'The pig bit the woman.'
  - b. tey tale yaw(-re) d-æl-i
    3.F.SG woman pig(-OBJ) FA-bite-PST
    'The woman bit the pig.', \*'The pig bit the woman.'

(Feldman 1986: 110, glosses adapted)

I analyse this pattern by suggesting that animacy is grammaticalised as person in Awtuw. In Chapters 4 and 5, I link differential object marking based on animacy in Awtuw to differential object agreement in person features in Hungarian, as well as to other languages showing similar phenomena. In the following section, I elaborate on some theoretical aspects of this approach to person features and its relation to differential object marking.

### 1.3 Person features and hierarchies

One of the consequences of adopting M. Richards's (2008) approach to person features is that it leads to a possible answer to the question of whether "third" person is in fact a grammatical property like first and second person, or whether it is the absence of the property person (see e.g. Benveniste 1971 for such a suggestion; see Nevins 2007 for a different view). I will argue that both approaches are true: at least in some languages, there is a distinction between "third" person that carries a person feature, and "third" person that does not. The former is definite and/or animate, whereas the latter is not. I formalise this by assuming that there is a generic person feature  $\pi$  which can grammaticalise different properties in different languages.<sup>2</sup> Following Harley and Ritter (2002), McGinnis (2005), Adger and Harbour (2007) and Béjar and Rezac (2009), I will assume that there are certain entailment relations among person features. Adger and Harbour (2007), for example, argue that a [Participant] feature on a nominal implies that it is animate.

I will also assume that first person arguments entail the animacy and definiteness of their referent semantically, but with a proviso: neither of these properties have to

<sup>&</sup>lt;sup>2</sup> Throughout this thesis, I am using  $\pi$  to refer to person features following Béjar (2003) and Béjar and Rezac (2003, 2009). Similarly, # will refer to number features, and  $\varphi$  will cover both, as well as gender.

be *grammaticalised* in a language and therefore neither of them have to be reflected in syntax. Gender, for example, while recoverable for interpretation in Hungarian, is not grammaticalised in the language and is never involved in determining agreement. If a language does grammaticalise animacy as person, however, we expect first and second person pronouns to behave like third person *animates* as well, as all personal pronouns share these features. This approach to the role of features like animacy and definiteness in the grammar also follows Lyons (1999). Lyons (1999: 275ff.) argues that definiteness can but does not have to be grammaticalised: while all languages can have semantic-pragmatic concepts of what it is to be definite (identifiability, uniqueness, etc.), these may be grammaticalised differently (or not at all) in a given language. Treating the relevant features as privative, i.e. as either present or absent in a given language, can capture part of this intuition.

Entailment relations among referential and semantic properties are one of the hallmarks of differential object marking and are reflected in what have been called "animacy" or "definiteness" or "Hale/Silverstein hierarchies" (see e.g. Givón 1976; Silverstein 1976; Comrie 1989; Aissen 1999, 2003; Haspelmath 2008; Keine and Müller 2008; Haspelmath 2009; see also the discussion in Chapter 5). Examples of such hierarchies are shown in (4):

- (4) a. 1 > 2 > 3
  - b. Personal pronouns > Proper names > definite NP > specific indefinite NP
     > non-specific indefinite NP
  - c. 1, 2 > Human > Non-human animate > Inanimate

In general, if any element on one of the scales in (4) triggers case-marking in a language, all higher elements will also trigger case-marking (Aissen 2003; Haspelmath 2009). There are certain counterexamples or seeming counterexamples to Silverstein's (1976) (and Haspelmath's) generalisation that types of objects that trigger marking always form continuous stretches in hierarchies like (4) (cf. Filimonova 2005 and Chapter 4). But hierarchies like (4) are helpful for characterising the set of direct objects in a language that triggers differential object marking, be it in case-marking or verb morphology and have been adopted very productively in the analysis of differential marking and related phenomena (see also Croft 2003: Ch. 5). Any approach to DOM should be able to capture these hierarchical effects (as well as exceptions). In the definition in (1), I stated that the marked subset of direct objects in languages with DOM is generally more definite and more animate than the unmarked subset.

In this thesis, I will assume a specific implementation of such hierarchies that is based on the idea that each person is composed of several person features which form sets (see Harley and Ritter 2002; Béjar 2003; Béjar and Rezac 2003, 2009 for versions of this approach). Each of these sets is what is usually called a "person". Entailment relations are expressed by subset/superset relations among such features. I will discuss three related types of such hierarchies. First, I assume that representing person features as sets provides a natural way of capturing hierarchies (see immediately below). This is closely related to the second type of hierarchy, the ones shown in (4). The reason why DOM is sensitive to such hierarchies is that if a certain person feature triggers agreement or case-marking, "higher" elements will include that feature because they are proper supersets. Finally, in Chapter 6, I will adopt a similar view with respect to the representation of case (following Caha 2009).

Turning to the discussion of person features, Table 1.1 shows the representation of person features as feature geometries, following Harley and Ritter (2002) and McGinnis (2005) and (5) shows them as sets. I am ignoring the content of the INDIVIDUATION (or number) branch for now, focusing on the left branch which indicates person. First person is the most complex in that it includes a "Speaker" node as the daughter of PARTICIPANT, while second and third person lack part of this structure. Third person is indicated as the mere presence of the "Referring Expression" root node (Harley and Ritter 2002 suggest that "referring expressions" are pronouns; their feature specifications are thus arguably naturally embedded under D in the syntax).

In a similar vein, Béjar and Rezac (2009) use the features  $\pi$ , PARTICIPANT and SPEAKER which (roughly) correspond to the nodes Referring Expression, PARTICIPANT and Speaker in (5), respectively. In this thesis, I adopt Béjar and Rezac's (2009) approach and I will generally take first, second, and third person to be represented as sets of different cardinalities whose members are the features  $\pi$ , PARTICIPANT and SPEAKER. This alternative is shown in (5).

First	Second	l person	Third person	
Referring	Expression	F	Æ	RE
		/		
PARTICIPANT	INDIVIDUATION	PART.	INDIV.	INDIV.
	$\bigtriangleup$		$\bigtriangleup$	$\bigtriangleup$
Speaker				

 Table 1.1
 Representation of person in a feature geometry

(5) 
$$[1] = \begin{cases} \text{SPEAKER,} \\ \text{PARTICIPANT,} \\ \pi \end{cases} \quad [2] = \begin{cases} \text{PARTICIPANT,} \\ \pi \end{cases} \quad [3] = \{\pi\} \end{cases}$$

Representing person features as sets allows us to formulate hierarchies like the one in (4a) as relations among these sets. What makes first person more "prominent" than second person, for example, is that the set of features representing first person is a proper superset of the features representing second person.

(6) 
$$[1] = \begin{cases} \text{SPEAKER,} \\ \text{PARTICIPANT,} \\ \pi \end{cases} \supset [2] = \begin{cases} \text{PARTICIPANT,} \\ \pi \end{cases} \supset [3] = \{\pi\} \end{cases}$$

The more complex hierarchies in (4b,c) can be captured in similar ways. I assume, again following M. Richards (2008), that  $\pi$  can grammaticalise different properties across languages: it can express animacy or definiteness, for example. If  $\pi$  grammaticalises animacy in a given language, second and first person will show the same grammatical behaviour with respect to animacy as third person, since all contain  $\pi$ . In such a language, I assume that inanimates will simply be represented as an empty set of features. Inanimate and animate third person thus differ in whether they have  $\pi$  or not. In Chapters 2 to 5, this distinction will play a crucial role.

This approach to hierarchies also addresses the question of the role of such hierarchies in grammar. M. Richards (2008: 139) argues that the "syntactic status of prominence hierarchies is dubious" (see also Newmeyer 2002a,b; Jelinek and Carnie 2003; Carnie 2005a,b; den Dikken 2006 for similar views). Since person features are generally thought of as part of syntax, however, analysing them as sets of different cardinalities allows us to incorporate some hierarchical effects into the grammar (see also Béjar and Rezac 2009). I will come back to these hierarchical effects in Chapters 4 and 5.

#### 1.3.1 Hierarchies and functional approaches to DOM

Hierarchies play an important role in functional explanations for differential object marking. They are often linked to *markedness* and a notion of what kinds of nominals are *typical* subjects and objects (see also Chapter 5).

Aissen's (2003) well-known analysis of differential object marking in Optimality Theory (OT, Prince and Smolensky 2004) uses hierarchies to motivate constraint hierarchies that determine patterns of differential object marking. Aissen argues that subjects (or agents) are typically high in definiteness and animacy, whereas objects (or patients) are typically low (Silverstein 1976; Comrie 1989). If arguments in a clause diverge from this norm, for example when a direct object is definite and animate, morphological marking indicates the general markedness of that structure. This approach, while influential, has also been criticised for various reasons: see Næss (2004), Haspelmath (2006), Dalrymple and Nikolaeva (2011) and Iemmolo (2012) for a critique of the role of markedness; cf. also the above discussion of the role of hierarchies in the grammar.

While Aissen (2003) takes hierarchies to represent cross-linguistic tendencies for the marking of arguments, it is not clear where these tendencies come from. Aissen's (2003) OT account suggests that constraints determining DOM are universally valid across languages and thus part of UG, a position criticised by Haspelmath (2008: 79) who suggests that "Aissen does not consider the possibility that [a] functional explanation makes an explanation in terms of UG superfluous". Haspelmath claims that scales "can generally be interpreted in terms of processing costs" (Haspelmath 2009: 458), suggesting in particular that accusative case on elements higher on (6), e.g. pronouns, is less redundant than on lower elements, for example non-specific NPs. This kind of functional reasoning is also hinted at in Silverstein (1976: 113) who mentions the "semantic naturalness" of case-marking along hierarchies of noun phrases. See also Seržant and Witzlack-Makarevich (2015) for a recent overview of functional approaches to differential object (and subject) marking.

As the main focus of this thesis is on analysing differential object marking in Hungarian and other languages in a Minimalist framework, I will mostly leave functional questions aside — not because they are not relevant to the topic in general, but because I believe that the formal modelling of the phenomenon is a prerequisite for answering functional questions.

#### 1.4 Case and agreement

Differential object marking raises questions about the interaction of case and agreement which I will discuss in detail in Chapters 5 and 6. Particularly relevant are the timing of case assignment and agreement in the syntactic derivation, and the ways in which case and agreement influence and determine each other.

In Chomsky's (2000, 2001) framework, case and agreement are two aspects of a single syntactic operation called Agree. Continuing the analysis of case from so-called Government and Binding theory (Chomsky 1980, 1981, 1982; Vergnaud 2008 [1977]), the theory distinguishes between abstract Case (written with a capital "C") and morphological case.

This traditional approach to case suggests that even if a language lacks morphological case, noun phrases in the grammatical functions subject, direct object, etc. can bear abstract Case. In English, for example, most nominals do not show any morphological reflex of Case. It is often assumed, however, that English transitive verbs nevertheless assign accusative to their direct objects. And indeed, personal pronouns in English do show different morphological cases: cf. *she* in "subject" case vs. *her* in "object" case.

Chomsky (1981) further distinguishes structural from inherent Cases. The former are assigned in a given syntactic context, whereas the latter are determined by lexical properties of a Case assigner (see also Woolford 2006). Since Chomsky (2000, 2001), assigning structural Case is not thought of as involving a particular syntactic *position*, but assigned by a particular *head* in syntax: while it used to be assumed that nominative is assigned to an NP governed by INFL, it is now generally thought that the head T enters into an Agree relation (see below) with an accessible noun phrase. This will often be the subject in Spec vP (but it does not have to be). Similarly, accusative is assigned to an accessible NP by v.

In Government and Binding theory, structural Case was already linked to agreement (see e.g. Chomsky 1981: 172, Chomsky and Lasnik 1995: 121), but this connection has been further strengthened in recent Minimalist approaches to syntax. Chomsky (2000, 2001) suggests that the operation *Agree* underlies both Case assignment and agreement. (7) illustrates this. In (7a) *v* acts as a *probe*. It has unvalued  $\varphi$ -features which it aims to value in the course of the syntactic derivation. Accessible *goals* for an Agree relation are nominals that have an unvalued CASE feature, which renders them active. In (7a), *v* and the direct object (DO), respectively, fulfill these criteria and an Agree relation is established between them.



The goal values the probe's unvalued  $\varphi$ -features, and the probe values the goal's unvalued CASE feature, as shown in (7b). In this scenario, the direct object is assigned accusative Case (ACC) and the probe is valued as [3, PL]. Whether any of these features are spelled-out is not determined at this point of the derivation.

In addition to abstract Case, agreement can therefore also be abstract. Case and agreement as Agree have an important role in driving the derivation: probes need to value their  $\varphi$ -features, and nominals need to value their CASE features, otherwise the derivation crashes. A nominal that has not entered into an Agree relation with

a Case-assigning probe will not get Case and it will therefore violate the Case Filter, which rules out overt nominals appearing without Case (Chomsky 1981). When an NP's Case has been valued, that NP becomes *inactive* and will no longer be visible to probes. Agree thus links Case and agreement and determines the licensing of noun phrases: a noun phrase that has no Case at the end of the derivation will lead to a crash.

This conception of Case and agreement also makes clear predictions about the cooccurrence of morphological case and morphological agreement. Since both are determined by the same operation, we do not expect mismatches between case and agreement. Yet these do, in fact, exist. Consider first the Hungarian examples seen above, repeated here:

- (8) a. Én lát-ok egy kutyá-t. I see.1sG.SBJ a dog-ACC 'I see a dog.'
  b. Én lát-om a kutyá-t.
  - I see the dog.'

The direct object *kutyá-t* 'dog-ACC' has case (and thus Case) in both (8a) and (8b), but only in (8b) is there object agreement. There are several ways in which this mismatch could arise. On the "classical" view of Agree of person, both direct objects in (8) have  $\varphi$ -features. In this scenario, both agree with the verb, value its  $\varphi$ -features and the Case features of both are valued as accusative. However, only the  $\varphi$ -features in (8b) are spelled out in morphology. This is compatible with Agree covering both Case and agreement simultaneously and the presence of object agreement is determined independently of syntax.

An alternative that is arguably not compatible with the standard conception of Agree is that the verb only agrees with the direct object in (8b), but not in (8a). On this view, *v* attempts to enter an Agree relation with the direct object, but Agree fails fails, because the direct object is lacking the relevant feature. *v* receives an empty value and object agreement on the verb is not spelled out, but the direct object is still assigned accusative case. This alternative raises questions about the origin of this accusative, as well as why a failed Agree relation does not lead to a crash of the derivation. Notwithstanding these problems, I will argue that this alternative view of Agree, on which Case and agreement do not go hand in hand, is more suitable to explain the distribution of agreement in Hungarian and the other languages discussed in this thesis, like Hindi and Amharic. These languages also show DOM and mismatches between case and agreement; I will illustrate with Hindi (see Chapter 6 for further discussion; see Baker 2012 for the same argument, based on Amharic data).

Hindi is a split-ergative language, in which the subject bears the ergative suffix *-ne* in perfective aspect, as shown in (9b). Hindi also shows DOM: some direct objects bear the suffix *-ko*, as shown in (10). Agreement in Hindi crucially does not follow case-marking: the verb agrees with the structurally highest argument *without* case-marking (see e.g. Bhatt 2005).

In (9a), this is the subject *Rahul*. The verb shows masculine singular agreement. In (9b), where the subject is case-marked, the verb agrees with the direct object *kitaab* and shows feminine singular agreement.

(9)	a.	Rahul kitaab paṛh-taa thaa.	
		R.м book.f read-нав.м.sg be.pst.м.sg	
		'Rahul used to read a/the book.'	
	b.	Rahul-ne kitaab paṛh-ii thii.	
		R-ERG book.F read-PFV.F be.F.SG	
		'Rahul had read the book.'	(Bhatt 2005: 759)

When both arguments in a monotransitive are case-marked, the verb shows default agreement as in (10).

(10)	Mona-ne	is	kitaab-ko paṛh-aa.	
	Mona.F-ER	G this.	Эвь book.f-acc read-pfv.m.sg	
	'Mona had	read t	nis book.'	(Bhatt 2005: 768)

These data from Hindi show a more serious mismatch between case and agreement than the Hungarian data above: agreement on the verb can be controlled by two different arguments, and rather than co-occurring with each other, case and agreement are in complementary distribution. Indeed, Bhatt (2005) suggests that case and agreement are dissociated in Hindi. I will also adopt this view (for Hindi and in general) and I will argue for a dissociation of case and agreement in more detail in Chapters 5 and 6.

### 1.5 Theoretical assumptions

As I depart from some fairly common assumptions in Minimalist work following Chomsky (2000, 2001), I will briefly lay out some of the assumptions I adopt. I first discuss the nature of Agree before turning to the general architecture of the grammar.

#### 1.5.1 Cyclic Agree

Following Béjar and Rezac (2009), but modifying their proposal, I will argue that Agree can proceed cyclically, i.e. a probe does not stop agreeing after a single Agree relation (see Nevins 2007, 2011 about how this approach relates to *Multiple Agree*). Whether a probe continues to probe or not depends on whether it has unvalued sets of features left.

Recall the structure in (7) above, repeated here as (11). v has an unvalued  $\varphi$ -feature which takes on the value [3, PL] after agreeing with the direct object.



Béjar and Rezac (2009) argue that probes can be "articulated". Rather than assuming that a  $\varphi$ -probe has a single unvalued person feature, the person probe has an unvalued feature of each person (and independent number probes; see also Béjar 2003; Preminger 2009, 2014). Such a probe is illustrated in (12). I suggest that each of elements in [u1, u2, u3] corresponds to a set of features. When valued by a third person argument, as in (11), not all of the probe's sets of  $\varphi$ -features are valued, but only the one that matches the goal's set of  $\varphi$ -features. In the example in (11), this is [3]. The remaining unvalued sets of features (in bold) can continue to probe.<sup>3</sup>

 $<sup>^3</sup>$  See Béjar and Rezac (2009) for discussion about how the probe's uninterpretable  $\varphi$ -features are deleted in the course of the derivation.

(12) 
$$\begin{bmatrix} u1\\ u2\\ u3 \end{bmatrix} \xrightarrow{(A)} Agree \begin{bmatrix} u1\\ u2\\ 3 \end{bmatrix}$$

In (12), after the Agree relation (A), there are unvalued sets of features left on the probe. If probes can continue to probe as long as they have unvalued (sets of) features, the probe in (12) can continue. This is shown schematically in (13) and illustrated in more detail in the structure in (14). In (13), the second Agree relation in step (B) values the probe as [1] (and entails [2], in parentheses). The set of features [2] is entailed as it forms a proper subset of the set [1]. In general, when a set of features values a set of features on a probe, it will entail all sets of features that are a proper subset of it. The previously valued set [3] is shown in grey.

(13) 
$$\begin{bmatrix} u1\\ u2\\ u3 \end{bmatrix} \xrightarrow{(A) \text{ Agree}} \begin{bmatrix} u1\\ u2\\ 3 \end{bmatrix} \xrightarrow{(B) \text{ Agree}} \begin{bmatrix} 1\\ (2)\\ 3 \end{bmatrix}$$

(14) illustrates the repeated steps of Agree as part of the syntactic structure. Here I only indicate the values of features that a probe receives in an Agree relation (and I ignore Case for ease of exposition). The derivation proceeds as follows: (A) shows an Agree relation between v and the direct object. The direct object values v's unvalued [3] feature set. But v has unvalued sets of features left and it moves to T in (B) from where it probes again, in step (C). This time, the subject's [1] feature set values v's unvalued set [1].



As Béjar and Rezac (2009) suggest, this cyclicity of Agree rules out certain combinations of features on v: if the direct object has a [1] feature set, it will fully value v and it will not be able to probe again. While I adopt this approach, my analysis diverges from Béjar and Rezac's in that I do not assume that in these scenarios a second probe is added as a repair strategy. Rather, if v cannot agree any more, the derivation simply continues (see Section 1.5.2 below). I will discuss this type of derivation in much more detail in Chapters 4 and 5.

When showing derivations in the following chapters, I will adopt the following conventions. I will indicate which feature sets are valued on a probe in a given step as in (15). The notation A/C indicates that the values [1, 3] are the results of steps A and C of the derivation. Movement will be indicated by solid lines with a single arrowhead ( $\longrightarrow$ ), Agree relations by dashed ones with two arrowheads ( $\leftarrow - \rightarrow$ ).

(15)



I will further assume that the derivation does not crash when a probe has not (fully) valued its sets of unvalued  $\varphi$ -features by entering Agree relations with a goal. An

approach of this kind has been proposed by Preminger (2011, 2014) and I turn to its discussion now.

#### 1.5.2 Agree can fail

Preminger (2011, 2014) refers to Chomsky's (2000, 2001) version of Agree as a "derivational time-bomb", suggesting that failed Agree relations lead to a crash of the derivation. Preminger argues for an alternative analysis of Agree: it is a *fallible* operation. A probe, say T, will attempt to value its  $\varphi$ -features but if it fails to do so, the derivation will not crash and the probe can get a default value. We have already seen two languages which provide some evidence for an approach along these lines, namely Hindi and Hungarian.

Recall that in Hindi, the verb agrees with arguments that do not bear overt case. Bhatt (2005) suggests that there is a single  $\varphi$ -probe in this language, on T, which can agree with either the subject or the object. If both the external and the internal argument are overtly case-marked, T will not find any goal to agree with. But the derivation does not crash: instead, the verb is spelled out with default (third singular masculine) agreement. This can be seen as evidence of a "failed" Agree relation that leads to a default (see also Preminger 2009).

In Chapters 4 and 5, I combine this approach to Agree with the cyclicity just discussed. This means that a probe, say *v*, can agree with the internal argument, and agree again with the external argument if it has unvalued feature sets left. The probe will attempt to agree as long as it has unvalued sets of features and it will stop when it has run out of features or goals to agree with. This will play an important role in my analysis of differential agreement in Hungarian, and other languages like Kashmiri and Sahaptin. In Hungarian, for example, *v* stops probing if it fails to encounter an argument that carries person features. The default value it receives will be an empty set of person features.

This approach has some important consequences for the role of Agree in the syntactic derivation: Agree no longer fulfills everything it does in Chomsky's (2000, 2001) system, since it is not (necessarily) involved in licensing nominals. The particular implementation of Agree I will adopt is shown in (16).

(16)  $\operatorname{FIND}_{\varphi}(f)$ :

Given an unvalued feature f on a head  $H^0$ , look for an XP bearing a valued instance of f. Upon finding such an XP, check whether its case is acceptable with respect to case discrimination:

a. yes 
$$\rightarrow$$
 assign the value of f found on XP to H<sup>0</sup>

b. *no*  $\rightarrow$  abort FIND<sub> $\omega$ </sub>(*f*) and continue with derivation

(Preminger 2014: 159)

The relevant points are the following. First, the Hindi data discussed above illustrate the idea behind "case discrimination". In Hindi, DPs with overt case-marking are *not* accessible for Agree, but only unmarked DPs are. Second, unvalued features f, say  $\varphi$ -features, on a head probe for matching features on an XP. If they find such an XP, valuation will take place, but if they do not, as shown in (16b), the derivation does not crash.

#### 1.5.3 Syntax and morphology

I will assume that the features that are transmitted from arguments to the verb in syntax are spelled out post-syntactically, as is standardly assumed in Distributed Morphology (or DM; see Halle and Marantz 1993, 1994; Harley and Noyer 2003; Embick and Noyer 2007; Siddiqi 2010; see also den Dikken 2013). In standard DM, bundles of features at terminal nodes in the syntax are spelled out by vocabulary items (VIs) which link a certain morphophonological form to a bundle of features. If several vocabulary items compete for insertion in a particular node, the most specific one (i.e. the one with the highest number of features) will win. The least specific VI is also referred to as an *elsewhere* case: it will be inserted if no more specific VI can be inserted in a terminal node (this gives rise to default agreement in Hindi, for example, and the lack of overt case in a language, see Chapters 5 and 6).

To illustrate this, let us assume that, in Hungarian, T agrees with a first person plural subject and is valued as [1, PL]. Assuming that T is spelled-out as agreement on the verb, (17) shows VI items for a verb like *lát* 'to see' in Hungarian.
(17) Vocabulary insertion rules

a.  $[1, PL] \leftrightarrow -unk$ b.  $[2, PL] \leftrightarrow -tok$ c  $[3, PL] \leftrightarrow -nak$ 

The terminal with the features [1, PL] is therefore spelled-out with the VI in (17a) and gives rise to the form in (18):

(18)  $l\acute{a}t$ -[1, PL]  $\leftrightarrow$   $l\acute{a}t$ -unk

Feature bundles that come about via Agree relations can also be modified by (postsyntactic) rules, among others so-called *impoverishment rules*. As Embick and Noyer (2007: 311) note, impoverishment rules "allow for the expression of [systematic syncretisms]." Imagine there were a syncretic suffix for plural verbs in Hungarian, as shown in (17). It is less specific than the VIs in (17), and is therefore not considered when a terminal node is specified for person as well as number.

(17) d. [PL] 
$$\leftrightarrow$$
 -ekk

The suffix in (17) is only inserted if something leads to a loss of a person feature, like the hypothetical impoverishment rule in (19a). Assume that the made-up verb *nét* triggers this rule, which deletes the person features of a terminal node in a certain domain. The result is illustrated in (19b), where the feature bundle on T is impoverished.

(19)	a.	$[\{1, 2, 3\}] \rightarrow \emptyset / n\acute{e}t-\_$	
	b.	$n\acute{e}t$ -[1, pl] $\rightarrow$ $n\acute{e}t$ -[pl]	by (19a)
	c.	nét-[pL] ↔ nét-ekk	insertion of (17d)

Another operation that I will assume is *fusion* (see e.g. Siddiqi 2010: 533). Under certain conditions, I will assume that two feature bundles can *fuse* and result in a single bundle that is spelled out by a single VI. Concretely, in Chapter 4, I suggest that when v moves to T in Hungarian and the strongest, i.e. the most specified, sets of person features on v and T match, the two feature bundles can fuse, as shown in (20). This allows the two heads to be spelled out as by a single VI.

#### 1 DOM, case and agreement

(20) T: 
$$[1, s_G], v: [1, 2, s_G] \rightarrow v + T: [1, 2, s_G]$$

A less standard assumption about Distributed Morphology and syntax that I adopt refers to the timing of impoverishment rules. Usually it is assumed that impoverishment rules only apply after the syntactic derivation (in accordance with the "notampering condition" in Chomsky 2005). However, it has also been proposed that certain "morphological" operations can apply during the syntactic derivation (Müller 2005, 2006, 2007; Keine and Müller 2008; Keine 2010). In these analyses, impoverishment rules can apply as soon as the conditions for a given rule are met, not just after the syntactic derivation has finished. Therefore impoverished feature structures can serve as the input to subsequent syntactic processes, like Agree. Figure 1.1 shows the standard view of impoverishment and Figure 1.2 shows the modified architecture suggested by Keine (2010). I will discuss the need for this architecture of the grammar in more detail in Chapters 5 and 6.



Figure 1.1 Standard view of the order of syntax and morphology (Keine 2010: 1)



Figure 1.2 Keine's proposed order of syntax and morphology (Keine 2010: 2)

# 1.6 The sample of languages

The following list shows the languages and language families I discuss in this dissertation; Table 1.2 shows the sources of my data and Figure 1.3 indicates where these languages are spoken (coordinates from http://www.wals.info/).

Chukotko-Kamchatkan Chukchi (Chapter 5)

Uralic Hungarian (Chapters 2 to 4, 6); Northern Khanty, Eastern Mansi (Chapter 6)
Indo-European Kashmiri (Chapters 5 and 6), Nepali, Marathi, Hindi (Chapters 6)
Sahaptian Sahaptin (Chapter 5)
Sepik Awtuw (Chapter 5)
Semitic Amharic (Chapter 6)
Trans-New Guinea Fore (Chapter 5)
Yukaghir Kolyma Yukaghir (Chapter 5)

The choice of languages in this dissertation reflects what Baker (2010, 2015) calls the "middle way" between typological and generative studies. While the sample size is relatively low, it allows for a certain analytical depth. The selection is also motivated by the availability of literature, my own competence in Hungarian as well as the existence of previous analyses of the relevant phenomena in some of the languages, which allows for a comparison of the present analysis to other approaches.

#### 1 DOM, case and agreement

Language	Source
Amharic	Baker (2012, 2015)
Awtuw	Feldman (1986)
Chukchi	Comrie (1980) and Bobaljik and Branigan (2006)
Eastern Mansi	Virtanen (2012, 2014, 2015)
Fore	Scott (1978)
Hindi	Bhatt (2005), informants
Hungarian	own data, Bartos (1999) and É. Kiss (2002)
Kashmiri	Wali and Koul (1997)
Kolyma Yukaghir	Maslova (2003)
Marathi	Pandharipande (1997) (via Legate 2008; Keine 2010)
Nepali	Bickel and Yādava (2000)
Northern Khanty	Nikolaeva (1999a,b, 2001) and Dalrymple and Nikolaeva (2011)
Sahaptin	Rigsby and Rude (1996)

Table 1.2Sources of original data



Figure 1.3 Geographical distribution of languages in this dissertation

# Part I

# Differential object marking in Hungarian

# 2.1 Introduction: Hungarian object agreement

Example (2) from Chapter 1, repeated as (1), shows an instance of differential object agreement in Hungarian.

- (1) a. Én lát-ok egy kutyá-t. I see-1sg.sbj a dog-ACC 'I see a dog.'
  - b. *Én lát-om a kutyá-t.* I see-1sg.obj the dog-ACC 'I see the dog.'

There are a number differences between (1a) and (1b). First, the direct object in (1a) is an indefinite noun phrase, whereas the object in (1b) is definite, which is reflected by the choice of determiner: egy 'one, a' in (1a), and *a* 'the' in (1b). Second, this difference in the definiteness of the direct object correlates with a difference in verb morphology. In (1a), the verb shows only subject agreement. The same verb form appears in an intransitive context, shown in (2) for a first and a third person subject.<sup>1</sup>

(2) a. Én lát-ok. I see-1sg.sBJ 'I see.'
b. Mari lát. Mari see.3sg.sBJ

'Mari sees.'

<sup>&</sup>lt;sup>1</sup> Because personal pronoun objects can be dropped, (2b) can also mean 'Mari sees you' or 'Mari sees me', in the right context, but never 'Mari sees him/her/it'. The point that this is the intransitive form still holds, however. I return to this in Section 4.3.1.

In (1b), on the other hand, the verb form references not just the subject, but the object as well. In other words, in (1b), the verb agrees with the subject as well as the object. Traditionally, as well as in grammars, these different verb forms have been referred to as two distinct *paradigms* or *conjugations* (see e.g. Rounds 2002; Törkenczy 2002; Pete 2006; H. Varga 2010). The verb form in (1a) would be part of the *subjective* or *indefinite* conjugation, whereas the form in (1b) is part of the *objective* or *definite* conjugation. The present indicative forms of the verb *lát* are shown in Table 2.1.

	Subject agreement only	Subject and object agreement
1sg	lát-ok	lát-om
2sg	lát-sz	lát-od
3sg	lát-∅	lát-ja
1pl	lát-unk	lát-juk
2pl	lát-tok	lát-játok
3pl	lát-nak	lát-ják

 Table 2.1
 Present indicative forms of lát 'to see'

The idea behind these notions is that one set of verbal suffixes only references the subject (*subjective conjugation*). This set appears with indefinite objects (as in (1b), *indefinite conjugation*), whereas the other set of suffixes appears when the object is definite (*objective* or *definite conjugation*). Throughout the dissertation, I will gloss the former as SBJ and the latter as OBJ; the  $\varphi$ -features preceding OBJ in (1b), for example, indicate the person and number of the subject.

In modern work on Hungarian, the verb forms in (1b) have been referred to as *object agreement*, and I adopt this more appropriate terminology (as do Szabolcsi 1994; Bartos 1997; Dalmi 1998; Bartos 1999; Koopman and Szabolcsi 2000; Bartos 2001; É. Kiss 2002; den Dikken 2004, 2006; Coppock and Wechsler 2012; Coppock 2013; Ortmann and Gerland 2014). These analyses mostly discuss Hungarian object agreement independently of other languages, but Iemmolo (2010), Dalrymple and Nikolaeva (2011), Bárány (2012b) and Ortmann and Gerland (2014) explicitly mention Hungarian as a language with differential object marking in a cross-linguistic context. In the current

and the following two chapters, I will deal with Hungarian data only, but I will adopt the analysis proposed here for related phenomena in Chapters 5 to 7.

In line with the hypothesis introduced in Chapter 1, I will argue that Hungarian object agreement is sensitive to person: the verb agrees with the formal features of the direct object, including its person ( $\pi$ ), but not its number features, as well as the person and number ( $\varphi$ ) features of the subject (in line with É. Kiss 2003, 2005, 2013; Bárány 2015a,b; see Coppock and Wechsler 2012; Coppock 2013 for the opposing view that the person of the object does not play a role in object agreement).

In the next section, I will discuss aspects of the morphological structure of Hungarian object agreement and its distribution, i.e. which types direct objects require object agreement.

# 2.2 The distribution of object agreement

The characterisation of Hungarian agreement suffixes as belonging to two distinct paradigms is useful for expository purposes. Using the verb *lát* 'to see' from above, Table 2.1 above illustrates the suffixes added to the root *lát* in the present tense indicative (without going into any morphological detail; but see Section 4.4 below).<sup>2</sup>

What is often only implicit in characterisations of Hungarian verb paradigms is the fact that the forms that agree with the object in Table 2.1 (the rightmost column) require *third person* objects. In fact, object agreement in Hungarian is often characterised as only being triggered by such objects, leading some researchers to suggest that object agreement is independent of  $\varphi$ -features in general (see Coppock and Wechsler 2010, 2012). There are some cases in which the direct object in a transitive clause is not third person, of course. (3a,b) show only subject agreement and subject as well as object agreement, respectively. When the subject is first person singular, and the object is second person, the verb has the suffix *-lak*. Crucially, this suffix is distinct from both the non-agreeing and the agreeing forms in (3a,b).

<sup>&</sup>lt;sup>2</sup> Note that Hungarian has vowel harmony and several morphophonological processes influence the exact form of the suffixes; I will not be concerned with these processes much, but see Rebrus (2000), Rounds (2002), Törkenczy (2002), Rebrus (2005) and Trommer (2005) for some discussion. See Section 4.4 for detailed discussion of the morphological structure of the suffixes.

- b. Én lát-om ő-t. I see-1sg.obj s/he-ACC 'I see him/her.'
- c. Én lát-lak téged. I see-1sG>2 you.sG 'I see you (sg.).'

While the distribution of object agreement with objects other than third person is not straightforward, I will nevertheless suggest that object agreement in Hungarian is sensitive to person ( $\pi$ ) features. I will discuss this extensively in Section 3.4 and Chapter 4.

Nevertheless, third person objects are a useful starting point to illustrate the distribution of object agreement in general, so I will start with them.

#### 2.2.1 Direct objects and subject agreement

In general, verbs are inflected for subject agreement only when they are intransitive and with a certain class of direct objects. Verbs show both subject and object agreement when they are transitive and the direct object belongs to a mostly complementary class of direct objects. This distribution is shown in Table 2.2. While the two columns in Table 2.2 roughly correlate with indefiniteness and definiteness, respectively, there are some mismatches between agreement and definiteness, which I will focus on in this chapter and the next one. These types of objects are marked with an \* in Table 2.2.<sup>3</sup>

Among these mismatches are first and second person pronouns which are generally taken to be definite, but which do not (always) trigger object agreement. On the other hand, there are possessed noun phrases that are not definite, yet still require object agreement. Before discussing these mismatches, I will illustrate examples from most of the cells in Table 2.2, starting with the left column.

In some grammars, like Rounds (2002) and Törkenczy (2002), as well as in some work on Finno-Ugric languages possibly influenced by traditional Hungarian grammar writing like Rédei (1965) and Gulya (1966), authors use the terms *indefinite* and *definite* conjugations. These authors have to specify what "counts" as indefinite and definite, respectively, like first personal pronouns "counting" as indefinite in Hungarian. This usage of the terms indefinite and definite is misleading and circular, as it is determined by what actually triggers agreement and what does not. Nevertheless, the property triggering object agreement is often generalised as being, roughly, definiteness.

Subject agreement only	Object agreement	
Det	erminers	
bare objects		
numerals		
indefinite article <i>egy</i> 'a, one'	definite article $a(z)$	
	demonstratives* ez a 'this', az a 'that'	
	possessed noun phrases*	
	proper names	
valami 'something', valaki 'someone'	valamelyik 'one or the other, one of	
	them', semelyik 'none (of them)'	
wh-e	xpressions	
ki 'who', mi 'what'	melyik 'which'	
Qu	antifiers	
néhány 'some', sok 'many'	a legtöbb 'most', az összes 'all'	
minden 'every'	valamennyi* 'each', mindegyik 'each'	
Pr	onouns	
first and second person pronouns*	third person pronouns	
	reflexive and reciprocal pronouns	
	Other	
	complement clauses with <i>hogy</i> 'that'	

Table 2.2Classes of noun phrases that do not require object agreement (left) and that do<br/>require object agreement (right); cells indicated with a \* show some variation and<br/>are discussed extensively below

#### 2.2.1.1 Bare objects, indefinite determiners and numerals

(4) illustrates that bare plural direct objects appear with verb forms inflected for subject agreement only, i.e. these objects do not trigger object agreement. Bare plural objects have been argued to be incorporated (Szabolcsi 1986; Farkas and H. de Swart 2003): they do not introduce discourse referents and their interpretation is neutral with respect to number. This is particularly clear in (4b), which need not mean that Mari bought a single flower — in fact, it is usually interpreted as meaning she bought

more than one. As for their syntax, Farkas and H. de Swart (2003: 95) point out that incorporated nominals cannot take an article, in contrast to full DPs, which cannot be incorporated.

# (4) Bare objects:

- Mari bicikli-t keres.
   Mari bicycle look for.3sg.sвј
   'Mari is looking for a bicycle / bicycles.'
- b. *Mari virág-ot ve-tt.* Mari flower-ACC buy-PST.3SG.SBJ 'Mari bought flowers.'

The indefinite determiner egy 'a' (also a numeral meaning 'one') does not trigger object agreement; neither do numerals on their own, (5b,c).

(5) Indefinite determiner: *egy* 'a' and numerals: *két* 'two', *ötven* 'fifty'

- a. *Péter lát egy lov-at.* Péter see.3sg.sbj a horse-ACC 'Péter sees a horse.'
- b. *Rozi két újság-ot olvas.* Rozi two newspaper-ACC read.3sg.sbj 'Rozi is reading two newspapers.'
- c. Rozi ötven sajtburger-t ev-ett.
   Rozi fifty cheeseburger-Acc eat-Pst.3sg.sbj
   'Rozi ate fifty cheeseburgers.'

# 2.2.1.2 Wh-words

Wh-words like *mi* 'what' and *ki* 'who' do not co-occur with object agreement; the same holds for the relative pronouns *ami* 'what.REL' and *aki* 'who.REL'.

- (6) Interrogative and relative pronouns: *mi* 'what', *ki* 'who', *aki* 'who'
  - a. *Kati mi-t mond-ott?* Kati what-ACC say-PST.3SG.SBJ 'What did Kati say?'
  - b. *Kati ki-t lát?* Kati who-ACC see.3sG.SBJ 'Whom does Kati see?'

c. *Szép a fiú, aki-t Kati lát-ott?* beautiful the boy, who.REL-ACC Kati see-PST.3SG.SBJ 'Is the boy whom Kati saw handsome?'

#### 2.2.1.3 Weak quantifiers and minden 'every'

Weak quantifiers, as well as the universal quantifier *minden* 'every' do not trigger object agreement, as shown in (7a,b).

- (7) Quantifiers: *néhány* 'some', *minden* 'every'
  - a. Zsuzsa varr néhány ing-et. Zsuzsa sew.3sg.sbj some shirt-Acc 'Zsuzsa is sewing some shirts.'
  - b. *Zsuzsa minden könyv-et elolvas-ott.* Zsuzsa every book-ACC read-PST.3SG.SBJ 'Zsuzsa read every book.'

It is surprising that *minden* 'every' patterns with weak quantifiers with respect to object agreement, as *minden* and weak quantifiers do not generally show the same syntactic behaviour. For example, in contexts corresponding to the English existential *there*-construction, weak quantifiers are allowed, but *minden* is not — here it patterns with other universal quantifiers like *mindegyik* and *valamennyi* 'each':

- (8) a. Van az asztal-on néhány alma.
   is the table-sup some apple
   'There are some apples on the table.'
  - b. *\*Van az asztal-on minden / mindegyik alma.* is the table-sup every each apple intended: 'There is every/each apple on the table.'

This kind of definiteness effect is well-known from other languages and is often discussed as a matter of the syntax-semantics interface (see Szabolcsi 1986; É. Kiss 1995; Kálmán 1995; Maleczki 2001; É. Kiss 2002; Piñón 2006 for discussion relating to Hungarian where a much larger class of predicates show similar effects; see Diesing 1992; Keenan 2003; Hallman 2004 for general discussion). The fact that weak quantifiers and *minden* do not pattern alike with respect to object agreement in (7) and the definiteness effect in (8) is a first indication that the distribution of object agreement is not directly

influenced by the semantic properties of the direct object - I will argue below that the syntactic properties of the direct object play a more important role.

#### 2.2.1.4 First and second person pronouns

Finally, first and second person pronouns cannot co-occur with object agreement when the subject is third person, even though they are definite:

- (9) First and second person pronouns:
  - a. Balázs lát engem. Balázs see.3sg.sbj I.Acc 'Balázs sees me.'
  - b. Balázs lát téged. Balázs see.3sg.sbj you.Acc 'Balázs sees you.'

As I discuss at length in Chapter 4, the distribution of agreement with personal pronouns is more complicated than hinted at in (9). While first person (singular) objects never show object agreement, second person pronouns only do in very restricted contexts. (9b) shows that with a third person subject, the verb does not agree with the object. However, when the subject is first person, a different suffix appears (see also (3) above):

(10) (Én) lát-lak téged. I see-1sG>2 you.sG 'I see you (sg.).'

The suffix *-lak/-lek* is sometimes treated as exceptional with respect to the rest of the object agreement system in Hungarian. Bartos (1997) refers to it as the only case of object agreement in person; in Bartos (1999) the agreeing form in (10) is derived differently from the other forms seen so far. Similarly, *-lak/-lek* is treated as an exception by Coppock and Wechsler (2012) and Coppock (2013).

However, it is also possible to take (10) as a regular instantiation of object agreement in Hungarian and treat the forms in (9) as the exception. This entails that Hungarian object agreement takes the direct object's  $\varphi$ -features into account — and necessitates an explanation for the seeming lack of agreement in (9). This is the strategy I adopt in this thesis, in line with much research on similar case and agreements splits across languages (see Chapter 4 as well as É. Kiss 2003, 2005, 2013; Bárány 2015b on Hungarian and Silverstein 1976; Béjar and Rezac 2009; Keine 2010; Georgi 2012 on a number of other, unrelated languages, some of which I will discuss in Chapters 5 and 6).

# 2.2.2 Direct objects that trigger object agreement

It is easier to define the class of noun phrases that do require object agreement extensionally than intensionally. I will therefore first go through some examples of direct objects triggering object agreement on the verb (following Table 2.2 on p. 29) before discussing the common property that is responsible for object agreement.

# 2.2.2.1 The definite determiner and demonstratives

(11) shows that noun phrases with the definite determiner a(z) 'the'<sup>4</sup> require object agreement. The definite determiner is used with common nouns and can be used with proper names as well.

(11) Definite determiner a(z) 'the', proper names:

- a. *Mari lát-ja a ház-at.* Mari see-3sg.obj the house-ACC 'Mari sees the house.'
- b. *Mari keres-i* (*a*) *Péter-t*. Mari look for-3sg.obj the Péter-Acc 'Mari is looking for Péter.'

Direct objects with the demonstratives ez a(z) 'this' and az a(z) generally trigger object agreement as well, as shown in (12). When the demonstrative substitutes a whole noun phrase and it appears as ez 'this' or az 'that', it can sometimes co-occur with subject agreement. I will discuss these unexpected uses of demonstratives in Section 2.2.3 below.

 $<sup>^4</sup>$  *z* appears when the word following the determiner starts with a vowel; cf. English *a*(*n*).

- (12) Demonstratives ez a(z) 'this', az a(z) 'that':
  - a. *Péter ez-t* a kenyer-et szeret-i. Péter this-ACC the bread-ACC like-3sg.obj 'Péter likes this bread.'
  - b. *Péter az-t a könyv-et olvas-t-a*. Péter that-ACC the book-ACC read-PST-3SG.OBJ 'Péter read that book.'

### 2.2.2.2 Determiners ending in -ik

There is a group of determiners ending in *-ik*: *melyik* 'which', *semelyik* 'none', *valame-lyik* 'one of them', *hányadik* 'which number', *mindegyik* 'each'. All of these trigger object agreement, both when replacing a full NP, (13a), or when used as a determiner, (13b). Note that these determiners generally do not appear adjacent to the definite determiner a(z) (apart from *hányadik*), as shown in (13d).

- (13) Determiners ending in *-ik*:
  - a. *Melyik-et kér-ed?* which-Acc want-2sg.oBJ 'Which one do you want?'
  - b. *Melyik bicikli-t kér-ed?* which bicycle-ACC want-2sg.oBJ 'Which bicycle do you want?'
  - c. *Valamelyik-et lát-t-a.* one of them-ACC see-PST-3SG.OBJ 'S/he saw one of them.'
  - d. \*A melyik-et kér-ed? the which-Acc want-2sg.овј intended: 'Which one do you want?'

Apart from their morphosyntactic similarity (the *-ik* suffix), these determiners are semantically similar. They are all presuppositional in that they refer to elements of a context-salient set. The difference between *mi* 'what' and *melyik* 'which' in Hungarian parallels the one between their English counterparts (see e.g. Pesetsky 1987 on *which* and "D-linking").

#### 2.2.2.3 Possessive constructions and object agreement

Possessed nouns will be the topic of Section 3.3, so I will not go into a lot of detail here (but see Szabolcsi 1983, 1987, 1994; Bartos 1999; den Dikken 1999; Dékány 2011; É. Kiss 2014; Dékány 2015 for extensive overviews). In brief, in Standard Hungarian, possessed direct objects trigger object agreement. Possessed nouns in Hungarian carry a possessive suffix which indicates the person and number of the possessor, *-jé-* in (14a). The possessor itself can be covert as in (14a), or carry nominative case (unmarked or 'null') or dative case, as in (14b) and (14c), respectively. Nominative and dative possessors differ in their position in the noun phrase and their syntactic mobility. Finally, (14d) shows a pronominal possessor with nominative (unmarked) case. The following examples illustrate the different types of possessors and their co-occurrence with object agreement.

- (14) Possessive constructions:
  - a. Péter keres-i a bickli-jé-t.
     Péter look for-3sg.овј the bicycle-3sg.poss-Acc
     'Péter<sub>i</sub> is looking for his<sub>i/i</sub> bicycle.'
  - b. *Mari szeret-i Kati palacsintá-já-t.* Mari like-3sg.овј Kati.nom pancake-3sg.poss-Acc 'Mari likes Kati's pancakes.'
  - c. *Kati imád-ja en-nek az író-nak a könyv-e-i-t.* Kati adore-3sg.овј this-dat the writer-dat the book-3sg.poss-pl-acc 'Kati adores this writer's books.'
  - d. Az én ház-am piros. the I house-1sg.poss red. 'My house is red.'

Syntactically, the structures in (14) differ in the position of the possessors. While nominative possessors like (14b) cannot co-occur with a definite determiner, pronominal possessors like (14d) *must* do. In (14c), the dative possessor *en-nek az író-nak* 'this-DAT the writer-DAT' also precedes a definite determiner. The distribution of the possessors with respect to the definite determiner indicates that there are different positions for possessors.

Note, however, that the complementarity of the determiners ending in -ik and the definite determiner, on the one hand, and the nominative possessor and the definite

determiner, on the other hand, does not mean that nominative possessors and determiners ending in *-ik* are in complementary distribution as well.

The examples in (15) illustrate this. While both nominative possessors and -ik-determiners are incompatible with the definite determiner a(z), as shown in (15a,b), a nominative and an -ik-determiner *can* co-occur. This is shown in (15c).

- (15) a. \**Kati a palacsintá-ja* Kati the pancake-3sg.poss intended: 'Kati's pancake'
  - b. \**a mindegyik bicikli* the each bicycle intended: 'each bicycle'
  - c. *Kati mindegyik palacsintá-ja* Kati each pancake-3sg.Poss 'each of Kati's pancakes'

To account for the incompatibility of the definite determiner and certain determiners in the Hungarian noun phrase, Szabolcsi (1994) suggests a rule of haplology that rules out co-occurrences like (15b) by deleting the definite determiner on the surface. Given that nominative possessors are generally assumed to appear in DP, the fact that they can appear with -ik-determiners indicates that these determiners are lower in the structure of the noun phrase.

Nominative possessors and the definite determiner are in actual complementary distribution, then, whereas *-ik*-determiners and the definite determiner do not co-occur because of an additional rule, like Szabolcsi's (1994) haplology rule or movement of the *-ik*-determiners to D, as suggested by É. Kiss (2000).

While the possessed objects shown above were all definite, this is not generally true of possessive constructions. Hungarian lacks a verb meaning 'to have' and it expresses this possession relation by the combination of, first, an optional NP in dative, second, the copula, and third, a possessed noun. This construction shows that the possessive suffix itself does not contribute a specific or definite interpretation. Evidence for this comes from the fact that it appears freely in the Hungarian *mihi est*-construction (lit. 'to me is x'), as shown in (16), even though this is a definiteness effect-context (Szabolcsi 1986, 1994).

#### 2.2 The distribution of object agreement

- (16) a. *Mari-nak van macská-ja.* Mari-DAT is cat-3sg.poss 'Mari has a cat.'
  - b. *Mari-nak nincs macská-ja.* Mari-DAT is.NEG cat-3sG.POSS 'Mari does not have a cat.'
  - \*Mari van macská-ja. Mari.nom is a cat-3sg.poss intended: 'Mari has cat.'

(16a) illustrates how a *have*-relation is expressed in Hungarian. This context is sensitive to the definiteness and specificity of the complement, i.e. the possessed noun in this case. The possessed noun is a non-specific argument in Hungarian as well. (16b) illustrates this with a negated copula: the possessed noun *macská-ja* 'his/her cat' does not carry an existence presupposition and scopes below negation. Finally, (16c), in contrast to (16a), shows that the possessor in this construction has to be a dative, and cannot be a nominative (see also Szabolcsi 1986, 1994).

Possessed direct objects can be indefinite, then, but only when the possessor is not an overt nominative (cf. English pre-nominal and post-nominal possessors; the former are sometimes said to induce definiteness, as opposed to the latter, see e.g. Woisetschlaeger 1983). In Standard Hungarian, such noun phrases also require object agreement, as shown in (17), even if they are indefinite.<sup>5</sup>

(17) Csak egy diák-nak két dolgozat-át talál-t-a jutalomra méltónak a only one student-DAT two paper-ACC find-PST-3SG.OBJ of prize worthy the zsűri.
jury
'The jury found only one student's two papers worthy of a prize.'

(É. Kiss 2002: 173)

#### 2.2.2.4 Quantifiers and object agreement

As mentioned in Section 2.2.1 above, the universal quantifier *minden* 'every' does not trigger object agreement. Its counterpart *mindegyik* 'each', another *-ik*-determiner,

<sup>&</sup>lt;sup>5</sup> É. Kiss's translation of (17) has a definite flavour to it, but the Hungarian sentence does not entail that there are two unique papers of a student. In addition, the student in question does not need to be known to the speaker.

does trigger object agreement. The same holds for the universal quantifier valamennyi 'each'.<sup>6</sup>

(18)	a.	<i>Péter fel-hív-t-a mindegyik gyerek-et.</i> Péter vм-call-psт-3sg.овј each child-Acc 'Péter call each child.'
	b.	* <i>Péter fel-hív-ott mindegyik gyerek-et.</i> Péter vм-call-pst-3sg.sbJ each child-Acc intended: 'Péter call each child.'
	c.	<i>Péter fel-hív-ott minden gyerek-et.</i> Péter vм-call-рsт-3sg.sвJ every child-Acc 'Péter call every child.'
	d.	* <i>Péter fel-hív-t-a minden gyerek-et.</i> Péter vм-call-рsт-3sg.овј every child-асс intended: 'Péter call every child.'

What is striking about the differences between *mindegyik* 'each' and *minden* 'every' is that the overt syntactic distribution of the two quantifiers seems to be identical in the noun phrase, yet they behave differently with respect to object agreement (Szabolcsi 1994; É. Kiss 2000; Coppock 2013; see Szabolcsi 1997; Brody and Szabolcsi 2003 on the distribution of these quantifiers in the clause). This might indicate that a semantic treatment of the difference between these quantifiers is necessary (see e.g. Coppock and Wechsler 2012; Coppock 2013 who generalise this idea). I will return to this question in Section 3.2.

# 2.2.2.5 Third person and reflexive pronouns

In contrast to first and second person pronouns, third person pronouns behave uniformly with respect to object agreement: third person pronoun objects always require object agreement, as shown in (19) (note also that the number of the object is irrelevant for determining object agreement).

<sup>&</sup>lt;sup>6</sup> The quantifier *valamennyi* can also mean 'some' — on that reading, it does not trigger object agreement. Because of this ambiguity, I will continue to discuss *mindegyik* rather than *valamennyi*.

#### 2.2 The distribution of object agreement

- (19) Third person pronouns:
  - a. *Mari lát-ja ő-t.* Mari see-3sg.obj s/he-ACC 'Mari sees him/her.'
  - b. *Mari fel-hív-t-a* ők-et. Mari vm-call-PST-3sG.OBJ they-ACC 'Péter called them.'

Reflexive pronouns of all persons also require object agreement — recall, however, that non-reflexive first and second person pronouns do not. Arguably, the reason for this is that reflexive pronouns historically derive from possessed nouns, which would be third person. They resemble the noun *mag* 'core, kernel' followed by a possessive suffix (see also den Dikken 2006; Coppock and Wechsler 2012; Coppock 2013; Rocquet 2013).<sup>7</sup>

- (20) Reflexive pronouns:
  - a. *Mi lát-juk magunk-at.* we see-1PL.OBJ ourself-ACC 'We see ourselves.'
  - b. *Mari lát-ja magá-t.* Mari see-3sg.OBJ herself-ACC 'Mari sees herself.'

There are several arguments for the third person status of reflexive pronouns. First, as Rounds (2002: 126) points out, they inflect like nouns, not like personal pronouns. Second, reflexive pronouns can control third person agreement, for example as possessors of another possessed noun, as in the following examples ((21a) is from the Hungarian National Corpus, via Rákosi 2014).

(21)	a.	El-mond-aná-m a magam eset-é-t.	
		vм-say-cond-1sg the myself.noм case-3sg.poss-Acc	
		'I would tell you about my own case.'	(Rákosi 2014: 550)

<sup>&</sup>lt;sup>7</sup> Note, however, that the possessive suffixes do not match their synchronic forms, cf. the reflexive *mag-a* 'him/her-self' and *mag-ja* 'his/her core'. I take this as an argument for the grammaticalisation of a nominal form as a pronoun — it is interesting, however, that even first and second person reflexive pronouns retain some formal third person properties; see the discussion in the text.

b. *Mindez-t a mag-am ere-jé-ből értem el.* all this-ACC the self-1sg power-3sg.POSS-ELA achieved VM 'I achieved all this with my own powers.'<sup>8</sup>

The reciprocal pronoun *egymás* shows the same behaviour: it inflects like a common noun and triggers third person possessive agreement, even with non-third person antecedents.

# 2.2.2.6 Complement clauses and object agreement

Hungarian complement clauses introduced by the complementiser *hogy* 'that' also trigger object agreement (see Kenesei 1994 for detailed discussion). (22) shows an example.

(22) Kati az-t képzel-i [<sub>CP</sub> hogy a gép el-roml-ott]. Kati that-ACC believe-3sg.овј that the machine vм-break-pst.3sg.sвј 'Kati imagines that the engine's broken down.'

(Kenesei 1994: 309)

Note that in (22), there is a demonstrative *az-t* 'that-ACC' in the matrix clause that is associated with the CP. Kenesei (1994) suggests that this is an expletive element that shows the case-marking the verb assigns, as the complement CP cannot be case-marked. We have seen above that demonstratives require object agreement, so the fact that complement clauses appear with object agreement might thus be due to the presence of the demonstrative. There are, however, some questions about the structure of such DP-CP constructs. If all *hogy*-CPs come with a DP layer, movement out of the CP should not be possible as such a DP-CP structure forms a complex noun phrase and is thus an island for movement (cf. Coppock and Wechsler 2012 for this argument; Gervain 2004, 2009 argues for other strategies of escaping CPs to which I return below). It is however possible to focus elements from the embedded clause in the matrix clause, when the demonstrative is not present.

(23) a. Anna  $[_{DP} az-t_j]$  akar-ja  $[_{CP} hogy meg-látogas-s-am Péter-t]_j$ . Anna it-ACC want-3sg.obj that vM-visit-sbjv.1sg.obj Péter-ACC 'What Anna wants is for me to visit Péter.'

(Kenesei 1994: 314)

<sup>&</sup>lt;sup>8</sup> From Hungarian author István Örkény's story *Férfiarckép*, part of the collection *Egyperces novellák*.

2.2 The distribution of object agreement

b. Anna  $[_{DP}$  Péter-t<sub>i</sub>] akar-ja (\*az-t)  $[_{CP}$  hogy meg-látogas-s-am  $e_i$ ]. Anna Péter-ACC want-3sG.OBJ it-ACC that vM-visit-sBJv.1sG.OBJ 'It is Péter that Anna wants me to visit.'

(Kenesei 1994: 314)

In (23a), as indicated by the index *j*, the demonstrative behaves like an expletive in focus position that is associated with the complement clause. However, elements from the embedded clause can appear in the focus position as well, as shown in (23b). In addition, (23b) does not allow spelling out the expletive demonstrative in any other position in the matrix clause, and the displacement of a constituent of the embedded clause suggests that we are not dealing with a DP-CP structure, as there does not seem to be a CNPC violation in (23b). The focused element interacts with agreement on the verb, as is shown by the following example, in which there is no object agreement, because the element in the matrix focus position is not of the type that triggers object agreement, e.g. an indefinite noun phrase in (24).

(24)	Csak [ <sub>F</sub> két dolg-ot <sub>i</sub> ] akar-ok / *akar-om [ <sub>CP</sub> hogy [ <sub>VP</sub>
	only two thing-ACC want-1sg.sbj want-1sg.obj that
	el-mond-j-ál $e_i$ ]]
	vm-say-sbjv-2sg.sbj
	'There's only two things that I want you to say.' (Kenesei 1994: 317)

The focused element can also be an adverb, i.e. non-argument. In that case, interestingly, the matrix verb still shows object agreement - with the complement clause, seemingly.

(25) Anna  $[F_{\rm gyorsan_j}]$  akar-ja / \*akar  $[C_{\rm P} e_j \text{ hogy meg-javít-s-am}]$ Anna fast want-3sg.obj want.3sg.sbj that vm-repair-sbjv.1sg.obj az autó-t  $e_j$ ] the car-ACC 'Anna wants me to repair the car FAST.' (Kenesei 1994: 316)

É. Kiss (2002: 252) points out that not only focus movement is possible from embedded clauses, but also movement to the topic position of the matrix clause, i.e. A'-movement in general. Object agreement behaves as with focused phrases. In (26a), the matrix verb shows object agreement even though there is no eligible DP: the moved topic *János-sal* 'János-com' bears comitative case and comitative DPs do not trigger object agreement

in Hungarian. The verb (26a) seems to agree with the complement clause. In (26b), the topic *Mari-t* 'Mari-Acc' is a proper name in accusative and thus triggers agreement.

(26)	a.	$[_{TopP} János-sal_i nem akar-om [_{CP} hogy meg-beszél-j-ünk]$
		János-сом not want-1sg.овј that vм-discuss-sbjv-1pl.sbj
		bármit is <b>e</b> <sub>i</sub> ]]
		'With John, I don't want us to discuss anything.'
	b.	$ \begin{bmatrix} T_{\text{TopP}} & \text{Mari-t}_i & [F & \text{Péter}] & \text{igér-t-e} & \text{meg} & [CP & \text{hogy fel-hív-ja} \\ Mari-ACC & Péter & \text{promise-PST-3SG.OBJ VM} & \text{that VM-call-3SG.OBJ} \end{bmatrix} $
		e,]]

'Mary, Péter promised to call up.'

(É. Kiss 2002: 252)

Kenesei (1994) points out a problem for his analysis of these constructions. As briefly mentioned above, he argues that the expletive is the element that receives the Case the verb has to assign because the CP cannot. However, he points out that while his approach

can account for the obligatory absence of the expletive and case change of the moved item, it has no natural explanation to offer for the properties of conjugation in case oblique arguments or adjuncts are moved—in fact, no proposal to our knowledge has been successful in this respect. If an oblique noun phrase or adjunct is raised, the matrix verb has definite conjugation, whether the phrase is definite or indefinite. (Kenesei 1994: 318)

It seems that Kenesei (1994) does not entertain the possibility that it is the CP itself that triggers object agreement in certain cases, namely when there is no accusative nominal in the matrix clause. This has been proposed by den Dikken (2012), for example, and is compatible with proposals about the referentiality of CPs discussed by de Cuba and Ürögdi (2009, 2010). I will return to this claim in Section 3.4.2.5.

To sum up this brief overview of complement clauses, the matrix verb in a complex Hungarian sentence can agree with, first, a demonstrative expletive that is associated with a CP argument, second, with focused or topicalised elements that have been either A'-moved out of these clauses or base-generated in the matrix clause, or, third, with the CP itself if there is no other suitable phrase to agree with.

This state of affairs is not unattested otherwise: in Turkish, a language with differential object marking, certain complement clauses have an accusative suffix, i.e. they undergo DOM; in Japanese, complement clauses can also have overt accusative case (see Rocquet 2013: 192ff. for discussion). In analogy, Rocquet argues that the same happens in Hungarian: the CP triggers agreement just like definite noun phrases do because it has the same kind of definiteness marker (see also Jánosi *et al.* 2014 and den Dikken 2012).

#### 2.2.3 "Unexpected" object agreement

In this section, I will discuss data that seems to contradict the previous discussion. I show some examples of demonstratives that do not trigger object agreement, and other, indefinite demonstratives that *do* trigger agreement, as well as proper nouns and non-specific definite noun phrases. Given the correlation between object agreement and the definiteness of the direct object, it is interesting to point out certain mismatches between the interpretation of the direct object and its ability to trigger object agreement. In all cases, I will argue that these examples provide evidence that the syntactic structure of the direct object influences the presence of object agreement.

Demonstratives like *ez* 'this' and *az* 'that' do not always trigger agreement, but there is a clear semantic difference between occurrences of demonstratives that do and those that do not. (27a,b) illustrate the relevant contrast: when a demonstrative direct object does not trigger object agreement, it does not refer to an individual (Bartos 1999; H. Varga 2010).

(27)	a.	<i>Az-t (az almá-t) kér-em.</i> that-ACC the apple-ACC want-1sg.OBJ 'I want that one (/that apple).'	
	b.	Az-t kér-ek. that-ACC want-1sg.sbj 'I want some of that.'	
	c.	<i>Az-t e-het-sz, ami-t csak akar-sz.</i> that-Acc eat-мод-3sg.sвj rel-Acc only want-2sg.sвj 'You can eat whatever you want.'	(Bartos 1999: 116)

Note that there are also syntactic differences between the two types: whereas the object in (27a) refers to an entity and can co-occur with the NP expressing that entity, (27b,c) cannot: these demonstratives have to occur by themselves in order not to trigger object agreement.

There is another type of demonstrative that is indefinite, as shown by the fact that it can appear in contexts like the English *there*-construction and an equivalent construction in Hungarian (see e.g. Ionin 2006; Abbott 2010; von Heusinger 2011; Ebert and Hinterwimmer 2013; Ionin 2013 for discussion).

(28) a. There was this dishy logician at Mary's party tonight. (Abbott 2010: 154)
b. Van ez a barát-om. is this the friend-1sG.Poss 'There is this friend of mine.'

The contexts in (28) do not generally allow definite noun phrases, suggesting that the demonstratives appearing in (28a,b) are not definite. These indefinite demonstratives can also occur as direct objects. Such constructions are obviously ambiguous between an indexical demonstrative reading and the specific, indefinite demonstrative reading shown in (28).

(29) Sétál-t az utcá-n és hirtelen lát-t-a ez-t az autó-t. walk-PST.3SG.SBJ the street-SUP and suddenly see-PST-3SG.OBJ this-ACC the car-ACC 'S/he was walking on the street and suddenly s/he saw this car.'

While (29) is ambiguous between the definite and the indefinite readings of the demonstrative, (29) can straightforwardly form part of a narrative discourse in which *ezt az autó-t* 'this car-Acc' has not yet been mentioned: the demonstrative here is cataphoric. But note that the direct object triggers object agreement: given that it is a demonstrative, this is not unexpected. What could be surprising, however, is that the semantic difference between the two readings of the direct object in (29) does not correlate with a change in agreement. (29) thus provides an argument that object agreement is sensitive to the syntactic structure of a direct object, even when it gives rise to different interpretations. The syntactic structure of the object remains constant while its interpretation changes.

(29) is also remarkable because specificity does not give rise to object agreement in Hungarian *per se* (Bartos 1999; É. Kiss 2002; Coppock and Wechsler 2012; Coppock 2013): the syntactic structure of the direct object (or its formal properties) can take priority over its semantic properties with respect to object agreement.

In the previous section, we saw that proper names trigger object agreement - but again, there are uses of proper names that do not allow object agreement and these correlate with certain semantic and syntactic properties. Consider (30).

(30) Petőfi-t olvas-ok. Petőfi-Acc read-1sg.sbj 'I read Petőfi.'

The proper name in (30) refers to works by the author Petőfi, i.e. it is a metonymic use of the proper name. What the object expresses in this context is roughly *works by Petőfi.* Given this property-like interpretation, it is no surprise that the proper name in (30) can appear with different determiners which can give rise to object agreement. This is shown in (31).

- (31) a. *Mari ve-tt egy Monet-t.* Mari buy-pst.3sg.sbj a Monet-Acc 'Mari bought a (painting by) Monet.'
  - b. Én mindegyik Petőfi-t el-olvas-om. I each Petőfi-Acc vм-read-1sg.овј 'I read each Petőfi.'

The proper names in (30) and (31) behave like common nouns, semantically and syntactically. I take this to mean that this metonymic use of a proper name has a different syntactic representation than the referential use discussed in the previous section and its behaviour with respect to object agreement is therefore not exceptional.

# 2.3 Summary

In this brief chapter, I provided an overview of the distribution of object agreement in Hungarian, i.e. which types of noun phrases require object agreement and which types do not allow it. We saw that object agreement in Hungarian is clearly not optional: there are barely any minimal pairs that only differ in the presence of agreement, but determining whether there is object agreement or not is generally fairly clear-cut.

In Section 2.2.1, I showed that bare and indefinite nouns can only appear with subject agreement, and that this also holds for weak quantifiers, and somewhat surprisingly,

for the universal quantifier *minden* 'every'. On the other hand, noun phrases with a definite or a demonstrative determiner, proper names and third person pronouns trigger object agreement, as does the universal quantifier *mindegyik* 'each'.

While these two classes of direct objects suggest that it is definiteness that determines whether there is object agreement or not, we have also seen that there are certain mismatches between the definiteness of the direct object and its potential to trigger agreement: possessive constructions trigger agreement, even if they are indefinite, as do the (specific) determiners *melyik* 'which' and *valameyik* 'one of them'. On the other hand, some definites, like first person pronouns, are incompatible with object agreement.

That such mismatches exist is not a new observation. Szabolcsi (1994: 227), discussing possessive constructions, points out that object agreement cannot be used as a semantic "litmus test" in Hungarian. I mentioned above that in Hungarian grammars these mismatches are often glossed over: first and second person pronouns, for example, are said to "count as indefinite" (Törkenczy 2002: 70; see also Rounds 2002: 23f.). While helpful for a learner, such an extensional characterisation of what gives rise to object agreement is clearly not a strong linguistic generalisation.

Thus the property of certain direct objects that determines whether they require object agreement or not is neither definiteness nor specificity. Two recent attempts to derive intensional generalisations about Hungarian object agreement are Bartos's (1999) syntactic approach, which aims to derive object agreement from the syntactic structure of the direct object alone, and a lexically oriented approach suggested by Coppock and Wechsler (2012). While Bartos argues that (all and only) DPs trigger object agreement, Coppock and Wechsler suggest a formal feature [DEF] that triggers agreement. Coppock (2013) extends this approach by providing a semantic explanation for distribution of the feature [DEF].

In the following chapter, I will argue for a "hybrid" approach to the analysis of object agreement in Hungarian. I will follow Bartos (1999) to a certain degree in assuming that the syntactic structure of the direct object plays a crucial role in determining object agreement, but I will incorporate insights by Coppock and Wechsler (2012) and Coppock (2013) that show that this cannot be the whole story (see also Bárány 2014b for an analysis along these lines).

# 3 A hybrid analysis of object agreement: syntactic structure and π-features

# 3.1 Introduction

In this chapter, I will discuss why object agreement in Hungarian is neither determined in a purely syntactic way nor in a purely semantic way, and I will provide an analysis that combines the best of both worlds to account for the data shown in the previous chapter.

The main proposal of this chapter is that the direct object has to project a DP and have a particular  $\varphi$ -feature specification in order to trigger object agreement. I present evidence for this approach from several sources: first, I show that noun phrases that trigger object agreement behave like DPs. I will argue for this by examining the syntactic distribution of determiners inside the noun phrase and its "size".

Second, cross-linguistic evidence provides support for the idea that it is a formal feature, namely the object's person feature, that triggers object agreement. Adger and Harbour (2007) and M. Richards (2008), for example, argue that the presence of  $\varphi$ -features distinguishes animate and definite nominals from inanimate and indefinite ones, respectively. While first and second person are always animate and definite, third person nominals do not have to be. I analyse Hungarian along these lines and argue that there are in fact four "persons": first, second, and third which are specified for  $\varphi$ -features, and a fourth one which lacks person features altogether (but can be specified for number). The latter two classes show distinct behaviour: while subject agreement treats both of them as "third" person, object agreement is only sensitive to one type of "third" person, namely the one with person features.

Third, treating object agreement as sensitive to  $\varphi$ -features fits well with the requirement of a DP projection. In much recent work, D has been argued to be the head

#### 3 A hybrid analysis of object agreement: syntactic structure and $\pi$ -features

expressing the formal feature PERSON (Bernstein 2008; Longobardi 2008; Danon 2011, see also Höhn 2015; van der Wal 2015; following Béjar 2003; Béjar and Rezac 2003, 2009, I will adopt  $\pi$  as representing this formal feature). Combining these insights will shed some light on subject/object asymmetries with respect to agreement and provide the basis for the analysis of the distribution of object agreement with personal pronouns proposed in Chapter 4.

This chapter is structured as follows. In the following section, I briefly outline why neither a purely syntactic nor a purely semantic account of object agreement in Hungarian is sufficient. In Section 3.3, I discuss the structure of possessive constructions and argue that they support the idea that object agreement is sensitive to the syntactic structure of the object. In Section 3.4, I propose an analysis to object agreement that accounts for the data discussed in Chapter 2 and here and sets the stage for the discussion in Chapter 4. Section 3.5 concludes.

# 3.2 Towards an analysis

#### 3.2.1 Problems for semantic approaches

I have already briefly illustrated some mismatches between certain semantic properties and the occurrence of object agreement. Possessed noun phrases can be indefinite, yet trigger object agreement; first and second person pronouns are also definite, but cannot appear with object agreement.<sup>1</sup> I will now discuss some evidence that shows that specificity and indexicality do not predict the correct distribution either and I conclude that semantic properties do not account for the distribution of object agreement.

Bartos (1999) and Coppock and Wechsler (2012) provide evidence that specificity does not suffice to explain the distribution of object agreement (this is true for any kind of specificity; see von Heusinger 2011). The following example from Coppock and Wechsler (2012) illustrates a specific direct object, referring to a particular individual identified in the following utterance, which cannot trigger object agreement:

<sup>&</sup>lt;sup>1</sup> I will argue in Chapter 4 that these pronouns do in fact trigger object agreement, it is just not spelled out in all cases.

(1) Minden nap egy görög énekes-t hallgat-t-ak / \*hallgat-t-ák. Máriá-nak every day a Greek singer-ACC listen-PST-3PL.SBJ listen-PST-3PL.OBJ Máriá-DAT hív-ják.
 call-3PL.OBJ
 'Every day, they listened to a Greek singer. Her name is Maria.'

(Coppock and Wechsler 2012: 715)

Building on Coppock and Wechsler (2012), Coppock (2013) suggests that object agreement is triggered by the presence of a formal feature on the direct object that she calls [DEF]. She suggests that the presence of this feature on a given lexical item is determined by its anaphoricity, i.e. whether is has an antecedent in the discourse. Coppock suggests that this straightforwardly accounts for the distribution of agreement with personal pronouns.

In particular, she argues that because first and second person pronouns are not anaphoric but indexical, they do not have the feature [DEF] and therefore do not trigger agreement (see also Bartos 1999 for a similar argument; cf. de Groot 2009 for a similar idea in terms of "identifiability").

But this referential explanation does not explain all the facts: the polite personal pronouns *ön* and *maga* refer to the addressee, i.e. they are indexical and not anaphoric, yet they behave like third person pronouns when it comes to agreement (shown for the pronoun *ön*).

- (2) a. Ön / ő szeret utaz-ni? you s/he like.3sg.sbj travel-inf 'Do you like to travel?' / 'Does s/he like to travel?'
  b. Lát-om ön-t / ő-t az utcá-n.
  - b. Latom on-t/ o-t az atca-h. see-1sg.obj you-Acc s/he-Acc the street-sup 'I see you / him/her on the street.'

(2a) shows that both  $\ddot{o}n$  and the third person pronoun  $\ddot{o}$  trigger third person subject agreement. (2b) shows that both pronouns also trigger object agreement: the indexicality of the pronoun  $\ddot{o}n$  therefore does not determine object agreement.

It is thus difficult to provide a uniform semantics for noun phrases that trigger object agreement. It is, however, equally difficult to provide a straightforward *syntactic* generalisation that characterises the types of noun phrases triggering object agreement.

#### 3 A hybrid analysis of object agreement: syntactic structure and $\pi$ -features

#### 3.2.2 Problems for syntactic approaches

Bartos (1997, 1999, 2001) suggests a syntactic approach along these lines: he suggests that object agreement is determined by the syntactic structure of the direct objects. If and only if the direct object projects a DP, the verb shows object agreement. The intuitive appeal of this analysis follows from the distribution of determiners in the noun phrase. As shown in Section 2.2.1, indefinite determiners and numerals do not trigger object agreement. Bartos suggests that this is because they only project a NumP but not a full DP (due to projectional economy).

(3)



Since the nominal in (3) does not project a DP, it does not trigger object agreement. If it were to project a DP, it would trigger object agreement. Since a phrase like *egy biciklit* is compatible with a definite determiner, the following structure is also licit and the phrase *az egy biciklit* does trigger object agreement.





Since, as we have seen, there is no strict one-to-one mapping between object agreement and semantic properties, by extension this means that if Bartos (1999) is correct, there cannot be a strict one-to-one mapping between the syntactic structure of a noun phrase and its semantic properties either. The reason for this is that, for Bartos, even the indefinite noun phrases that require object agreement, like possessed direct objects, have to project a DP. If true, this means that the D head in Hungarian does not (always) give rise to a definite interpretation of the noun phrase. As we will see below, there is ample evidence that suggests that possessed noun phrases project more syntactic structure than an indefinite noun phrase like the one in (3). First, however, let me point out some more general problems for Bartos's (1999) approach (see also Coppock and Wechsler 2012; Coppock 2013 for extensive criticism).

One potential problem with respect to Bartos's analysis arises from the distribution of agreement with personal pronouns. Given the strong correlation between the presence of D and object agreement, first and second person pronouns have to differ in their syntactic structure from third person pronouns. This is in fact what Bartos (1999: 65ff.) suggests: because of referential differences between first and second person pronouns and third person pronouns, only the latter project a DP. The former, being indexical, do not need D to get their reference.<sup>2</sup> Bartos (1999) therefore suggests that first and second person pronouns are mere NumPs, but not DPs.

There are some problems with this assumption, however. First, to the best of my knowledge, there is no syntactic evidence showing that first and second person pronouns behave differently from third person pronouns in Hungarian apart from the matter at hand, object agreement. Second, while I have mentioned that the mapping between syntax and semantic structure might not be straightforward in Hungarian, NumPs are generally not indexical — they denote properties, but not indexicals (É. Kiss 2000; Déchaine and Wiltschko 2002; É. Kiss 2002).

This leads to a third problem: we have seen above that indexicality does not determine object agreement (see (2) above). For Bartos, however, the indexicality of the formal pronoun *ön* 'you (sg.)' would mean that it projects a DP, since it triggers agreement, while other indexical pronouns do not. Bartos's (1999) account for why first and second person pronouns should be NumPs therefore does not carry over to *ön*.

Finally, arguing that personal pronouns have different structures implies that the suffix *-lak/-lek* is not part of the general system of object agreement in Hungarian since second person would be a NumP. (3), repeated here, shows that *-lak/-lek* appears

 $<sup>^2</sup>$  See Bartos (2001: 322) for a sketch of a different approach.

3 A hybrid analysis of object agreement: syntactic structure and  $\pi$ -features

with second person objects and behaves like other object agreement suffixes (see the discussion in Section 2.2).

- (5) a. *Én lát-ok valaki-t.* I see-1sg.sbj someone-ACC 'I see someone.'
  - b. Én lát-om ő-t. I see-1sg.obj s/he-ACC 'I see him/her.'
  - c. Én lát-lak téged. I see-1sG>2 you.sG 'I see you (sg.).'

An approach to object agreement that treats this suffix as part of the regular agreement paradigm seems to be stronger. I will defer further discussion to Chapter 4, where I implement an analysis that overcomes the problems just mentioned.

A second problem for Bartos's DP hypothesis has been pointed out by Coppock and Wechsler (2012) and Coppock (2013) and relates to the universal quantifiers *mindegyik* and *valamennyi* 'each' and *minden* 'every' (and by extension to the determiners and quantifiers ending in *-ik*, discussed in Section 2.2.2.2, p. 34). The problem that arises for a syntactic approach to object agreement is that the quantifiers *minden* and *mindegyik* are in the same position in the noun phrase but only *mindegyik* 'each' triggers object agreement (see also Szabolcsi 1994; É. Kiss 2000). (6) illustrates this contrast in agreement.

(6)	a.	El-olvas-ok /	*el-olvas-om	minden	könyv-et.
		vм-read-1sg.sвj 'I read every bo	у vм-read-1sg.овј ok.'	every	book-ACC
	b.	El-olvas-om /	*el-olvas-ok	mindeg	yik könyv-et
		vм-read-1sg.sbj	ј vм-read-1sg.овј	r each	book-ACC

'I read each book.'

For a syntactic approach, a solution would be to suggest that the two quantifiers are in different positions in the syntactic structure of the noun phrase, e.g. *mindegyik* projecting a DP, but *minden* only a NumP. É. Kiss (2000, 2002) suggests that this is the case: *-ik* quantifiers move to DP covertly, whereas *minden* is lower. Coppock and

Wechsler (2012) point out, however, that *minden* is higher than NumP, as it can take NumP complements (see also Kornai 1989).

(7) *Hardver minden két év-ben meg-dupláz-za kapacitás-á-t.* hardware every two year-INE VM-double-3sg.OBJ capacity-3sg.Poss-ACC 'Hardware doubles its capacity every two years.'

(Coppock and Wechsler 2012: 726)<sup>3</sup>

Coppock and Wechsler (2012) and Coppock (2013) interpret this to mean that while *mindegyik* is specified as [DEF], *minden* is not. This hypothesis is supported by semantic differences between the two quantifiers. While Hungarian *minden* and English *every* are felicitous in contexts in which the domain they quantify over is undefined, *mindegyik* and *each* are less so (see also Beghelli and Stowell 1997 for differences between *every* and *each* in English and Bárány 2012b: Ch. 4).

- (8) a. Every/?each girl likes to dance.
  - b. *Minden / ??valamennyi lány szeret táncol-ni.* every each girl like.3sg.sbJ dance-INF 'Every/each girl likes to dance.'

(Coppock 2013: 360, (59)-(60))

Since *minden* and *mindegyik/valamennyi* appear in the same position inside the noun phrase and neither can appear with the definite determiner, Coppock and Wechsler (2012) and Coppock (2013) argue that there is no evidence that a syntactic difference between the two quantifiers leads to their different behaviour with respect to object agreement. Coppock (2013) suggests that because *mindegyik* and *each* need a salient set in the discourse context over whose members they quantify, *mindegyik* is anaphoric in the same way that other lexical items specified as [DEF] are as well.

# 3.2.3 Syntactic structure and person features

Given the problems just mentioned, it seems that neither a syntactic nor a semantic generalisation can account for the distribution of object agreement. Coppock's (2013) approach is actually not purely semantic in the sense that object agreement is triggered

<sup>&</sup>lt;sup>3</sup> Cf. http://www.liska.hu/fliska/bgates.htm

#### 3 A hybrid analysis of object agreement: syntactic structure and $\pi$ -features

by a formal feature, i.e. a feature visible for syntax - it is the distribution of this feature that is determined by semantic properties.

I will argue for such a combined approach here: the formal feature specification of the direct object noun phrase is what triggers object agreement, but the distribution of the features is determined by the syntactic structure of the noun phrase, rather than its interpretation. This approach has advantages in the analysis of possessive constructions, for example, because it links their shared syntactic structure (involving D) to the syntactic properties of other kinds of objects that trigger object agreement, even though possessive constructions can show a wider range of interpretations, from indefinite to definite. The same logic holds for personal pronoun objects, as I will show in detail in Chapter 4.

The ingredients of this kind of hybrid approach to object agreement are the syntactic structure of the (direct object) noun phrase, and a formal feature. As I have hinted at before, I take this formal feature to be a person feature. Recall that I assume that (in Hungarian) the inventory of person features consists of  $\pi$ , PARTICIPANT and SPEAKER, where  $\pi$  is a generic person feature (cf. Béjar and Rezac 2009, and Harley and Ritter's (2002) notion of "referring expression"). As discussed in Section 1.3, these features form sets of person features which correspond to [1], or first person, and [2] and [3], for second and third person, respectively. [3] is an abbreviation for a set of person features that only includes  $\pi$ , i.e. { $\pi$ }.

Third person objects which trigger object agreement are specified as [3], whereas direct objects that do not trigger agreement lack this feature. I will argue that  $\pi$  in Hungarian grammaticalises presuppositionality. This means that direct objects triggering agreement are presuppositional, but it does not entail that all presuppositional noun phrases trigger agreement: an epistemically specific object with the indefinite determiner *egy* 'a' can be presuppositional, for example, but not by virtue of its feature content, but rather because of the discourse environment or the speaker's knowledge.

I further assume that person features in Hungarian are not expressed on all heads in the noun phrase, but rather that they are a property of D (Bernstein 2008; Longobardi 2008; M. Richards 2008; Danon 2011). When a noun phrase projects a DP, the D head introduces the possibility that the noun phrase is specified for  $\varphi$ -features expressing person (number being expressed lower, in NumP, see É. Kiss 2002; Dékány 2015). I argue that projecting DP is necessary to trigger object agreement but it does not suffice:
this resembles Bartos's (1999) hypothesis that DP structure triggers object agreement, but it is weaker, because Bartos claims that projecting DP is a necessary and sufficient condition for object agreement. D can be present but lack person features — in this case, it does not give rise to object agreement.

While weaker, this approach allows for an empirically more adequate analysis. I will follow Coppock and Wechsler (2012) and Coppock (2013) in assuming that certain lexical items have person features and that this correlates both with their syntactic and their referential properties. Again, this means that D can be present without a person specification, in which case a DP does not trigger object agreement.

For example, this approach makes it possible to represent the semantic difference between *mindegyik* and *minden* as the presence vs. absence of a person feature. As Coppock (2013) suggests, *mindegyik* is anaphoric or D-linked and presupposes the domain it quantifies over, whereas *minden* is not. In addition, it is necessary that this property be specified for certain (functional) lexical items, because we have seen above that specificity (and presuppositionality) that arises through the semantic composition and the influence of the discourse alone does not suffice to trigger object agreement (see (1), p. 49, for example). Some lexical items are therefore lexically specified for person features on D, while others are not.

Other determiners ending in *-ik*, as mentioned in Section 2.2.2.2 above, behave like *mindegyik* with respect to object agreement. In addition, the definite determiner, as well as demonstratives, personal pronouns and proper names also have a person feature. Non-referential uses of demonstratives and proper nouns, which were discussed in Section 2.2.3, lack this feature, which accounts for their behaviour with respect to object agreement (and they arguably have a different syntactic structure, as only non-referential proper nouns can appear with indefinite determiners, cf. (11) below).

By relying on syntactic structure (the presence or absence of D) and a formal feature (the presence or absence of person features), it is possible to account for subtle syntactic and semantic differences in the behaviour of noun phrases with respect to object agreement. This presents a way of capturing some of the "unexpected" data presented in Section 2.2.3. Recall that demonstrative pronouns sometimes do not require object agreement ((9) is repeated from (27), p. 43):

(9)	a.	Az-t (az almá-t) kér-em.					
		that-Acc the apple-Acc want-1sg.овј					
		'I want that one (/that apple).'					
	b.	Az-t kér-ek.					
		that-ACC want-1sg.sbj					
		'I want some of that.'					
	c.	Az-t e-het-sz, ami-t csak akar-sz.					
		that-ACC eat-MOD-2SG.SBJ REL-ACC only want-2SG.SBJ					
		'You can eat whatever you want.'	(Bartos 1999: 116)				

Since the demonstratives in (9b,c) only occur with subject agreement if they cannot take a complement, it is plausible to assign them a different syntactic structure: they are not DPs. Irrespective of their person features, they will not trigger object agreement, as they lack D. The demonstrative in (9a) *can* take a complement including a definite determiner, however, and crucially refers to the same entity with or without an overt nominal complement. Whether the complement *az almát* 'the apple' is present or not depends on the context; but there is no reason to assume that its feature specification differs, depending on whether its complement is overt or not. It behaves like a DP and is referential, which I take to be an effect of its person feature.

The second indefinite use of demonstratives mentioned above can be analysed along these lines as well ((10) is repeated from (29), p. 44):

(10) Sétál-t az utcá-n és hírtelen lát-t-a ez-t az autó-t. walk-PST.3SG.SBJ the street-SUP and suddenly see-PST-3SG.OBJ this-ACC the car-ACC 'He was walking on the street yesterday and suddenly he saw this car.'

On their definite interpretations, demonstratives are plausibly still DPs, as their syntactic behaviour remains constant. They also refer in the way "regular" demonstratives do, so I take them to carry a person feature. Recall that (10) is ambiguous between a definite and an indefinite reading, but there is no change in the syntactic behaviour of the demonstrative.

The metonymic use of proper names can be characterised as lacking a DP layer, since such proper names can take modifiers and indefinite determines like other common nouns: (11) Egy unalmas Coelhó-t olvas-ott.
 a boring Coelho-ACC read-PST.3SG.SBJ
 'S/he read a boring (book by) Coelho.'

These examples illustrate the logic behind this "hybrid" approach: one aspect relates to syntactic structure, since person features are only visible for agreement in D. In brief, I propose that object agreement is triggered when a noun phrase projects a DP and the D head is specified for person features. The importance of the syntactic structure is shown by the example in (10).

On its indefinite, cataphoric reading, the specific demonstrative in (10) is still referential and it triggers object agreement. This means that even though its interpretation with respect to definite, indexical demonstratives can change, there is no reason to assume that its structure differs from that of definite demonstratives. Because person features are formal features, they are immune to changes in interpretation induced by the context alone, i.e. whether an item with the same syntactic structure is interpreted in one way or another.

This also holds for indefinite determiners like egy 'a' and numerals. These items do not carry person features, but they can have specific interpretations. Nevertheless, since they are note lexically specified for person features, specificity alone will not give rise to object agreement (see also the discussion of (1), p. 49).<sup>4</sup>

Note that while I agree with Coppock and Wechsler (2012) that elements that trigger object agreement are lexically specified for a formal feature, I do not merely re-label Coppock and Wechsler (2012) and Coppock's (2013) [DEF] feature: I suggest that it is expressed only in a particular syntactic position, namely D. This is line with much cross-linguistic research on the role of D as the syntactic locus of referential properties (Longobardi 1994, 2001, 2005; Bernstein 2008; Longobardi 2008; M. Richards 2008). In addition, by making a connection to person features I make the strong claim that all personal pronouns trigger object agreement (*contra* Coppock and Wechsler 2012; Coppock 2013). I will argue that this is indeed the case in Chapter 4.

<sup>&</sup>lt;sup>4</sup> I leave the question of how such specific interpretations arise open here. López (2012) suggests an account of DOM in Spanish, Persian, and other languages in terms of choice functions (see also Chung and Ladusaw 2004 who argue for different rules of semantic composition). In any case, the difference in specificity is not attributed to a difference in the formal feature content of the determiner.

First, however, I will provide further evidence for the role of both D and a formal feature in determining object agreement, by discussing the nature of Hungarian possessive constructions.

# 3.3 Evidence from possessive noun phrases in Hungarian

## 3.3.1 Types of possessors: nominative, dative, pronominal

In this section, I relate the structure of Hungarian possessive constructions to object agreement. I will only discuss those aspects of the nature of possessives that are relevant for my present purposes. For further and more general discussion of possessed noun phrases in Hungarian, see Szabolcsi (1983, 1987, 1994), Bartos (1999), den Dikken (1999), É. Kiss (2000, 2002, 2014) and Dékány (2015).

First, recall that in Standard Hungarian, possessed direct objects require object agreement, even when the noun phrase as a whole is indefinite. This is shown in (12), repeated from (17) above, including the structure that É. Kiss assigns to the example.

(12)	Csak [ <sub>DP</sub> egy diák-nak     [ <sub>DP</sub> [ <sub>NumP</sub> két dolgozat-át ]]] talál-t-a							
	only one student-DAT two paper-ACC find-PST-3SG.OBJ							
	jutalomra méltónak a zsűri.							
	of prize worthy the jury							
	'The jury found only one student's two papers worthy of a prize.'							
	(É. Kiss 2002: 173)							

The structure in (12) shows a shell of two DPs on the direct object. The outer shell which includes the dative possessor *egy diák-nak* 'a student-DAT' is adjoined to the DP including the possessed noun *két dolgozat-á-t* 'two paper-3sG-POSS'. Evidence for this position comes from the fact that a dative possessor can precede a definite determiner:

(13) [<sub>DP</sub> egy diák-nak [<sub>DP</sub> a [<sub>NumP</sub> két dolgozat-a ]]] one student-DAT the two paper-3sg.poss 'a student's two papers' This contrasts with unmarked possessors<sup>5</sup> which are in complementary distribution with the definite determiner (see also Section 2.2.2.3).

(14) [<sub>DP</sub> egy diák [<sub>NumP</sub> két dolgozat-a ]] a student two paper-3sg.poss 'a student's two papers'

The two structures in (13) and (14) are repeated in (15a,b) for convenience.



The presence of the unmarked possessor *egy diák* gives rise to a definite reading in (14) and (15b): the two papers in question are understood to be the unique two papers that stand in a possession relation with the student (though the nature of this possession

<sup>&</sup>lt;sup>5</sup> É. Kiss (2000, 2002, 2014) and Dékány (2015) argue that unmarked possessors are in fact caseless and not nominative. This question is tangential to my present concerns, and my use of "unmarked", "nominative" or "caseless" merely serves to indicate the contrast to dative possessors.

relation is underspecified and determined by the context). Since unmarked possessors generally have this semantic contribution to a possessive noun phrase, and given that they are in complementary distribution with the definite determiner, I analyse possessive DPs with such possessors as projecting a DP introducing a person feature, just like the definite determiner.

Since the unmarked possessor can be a phrase (as in (14) and (15a)), it is generally assumed to be in the specifier of D, and not in D itself. In principle, then, it should be possible to spell out a determiner in D: but this is not allowed. A possible way to account for this is that there is an allomorph of the definite determiner a(z) in D, whose zero spell-out is conditioned by the possessor in the same phrase. A different type of unmarked possessor appears when the possessor is a pronoun, as shown in (16).

(16) \*(*az*) én két dolgozat-om the I two paper-1sg.poss 'my two papers'

The pronominal possessor in (16) appears *following* the definite determiner, showing the need for an additional position for possessors in the left periphery of the noun phrase. The relevant parts of (16) are shown in (17) (cf. Dékány 2015: 8, (31); I abstract away from the syntactic representation of possessive morphology).

(17)



As Szabolcsi (1994) points out, personal pronouns cannot appear with definite determiners in general which suggests that az in (17) fills the D position of the whole possessive noun phrase (this is compatible with pronouns being DPs; cf. Déchaine and Wiltschko 2002). Given the similarities in meaning between (16) and (17), this is another argument for the existence of a null allomorph of the definite determiner when the possessor is an unmarked noun phrase rather than a personal pronoun.

In brief, possessed noun phrases with overt unmarked phrasal and pronominal possessors pattern with noun phrases that include a definite determiner: they have a definite interpretation and they trigger object agreement.

### 3.3.2 Non-specific possessives and dative possessors

The syntax of dative possessors is more complex than that of unmarked possessors. As we have seen above, dative possessors are usually analysed as being in a higher position in the possessed noun phrase than unmarked and pronominal possessors (for the nature and the origin of this dative case, see Szabolcsi 1994; Bartos 1999; den Dikken 1999; É. Kiss 2000, 2002, 2014).

In addition, it is possible for dative possessors to appear in a position that is not adjacent to other elements in the possessed noun phrase, as shown in (18). This is impossible for unmarked possessors.

(18)	a.	Péter-nek olvas-t-a	(a) dolgozat-á-t.
		Péter-dat read-pst-3sg	.овј the paper-3sg.poss-Acc
		'It was Péter's paper s/h	e read.'
	հ	*Déter / " aluga + a	a delsezat á t

b. *Peter / o olvas-t-a a dolgozat-a-t.* Péter s/he read-PST-3SG.OBJ the paper-3SG.POSS-ACC

In fact, as shown by Szabolcsi (1994) dative possessors *must* sometimes extract out of the possessed noun phrase (see also É. Kiss 2002: Ch. 7). Crucially, this notion of extraction refers to constituency, and not merely the surface word order. The constituency of possessed direct objects can be diagnosed by certain syntactic tests. Consider again the way Hungarian expresses the relation *to have* (repeated from (16), p. 36):

- (19) a. *Mari-nak van egy macská-ja.* Mari-DAT is a cat-3sg.Poss 'Mari has a cat.'
  - Mari-nak nincs macská-ja.
     Mari-DAT is.NEG cat-3sg.Poss
     'Mari does not have a cat.'

 \*Mari van macská-ja. Mari.NOM is cat-3sg.Poss intended: 'Mari has a cat.'

Hungarian uses a dative noun phrase to express the possessor, a copula (with positive or negative polarity) and a possessed noun instead of a verb meaning 'to have'. (19a,b) illustrate that the dative possessor does not have to be adjacent to the possessed noun. (19c) shows that only dative possessors can appear in this construction.

That the dative possessor is actually extracted from the constituent of the possessed noun is shown by the test in (20). Focus particles like *csak* force a single constituent into the pre-verbal focus position (Szabolcsi 1994; É. Kiss 2002).

(20)	a.	Csak egy diák-nak két dolgozat-á-t olvas-t-	uk.
		only one student-dat two paper-3sg.poss-acc read-pst	'-1pl.obj
		'We only read one student's two papers.'	

- b. \**Csak Mari-nak macská-ja van.* only Mari-DAT cat-3sg.poss is
- c. \**Csak Mari macská-ja van.* only Mari cat-3sg.poss

(20a) shows that a dative possessor can form a constituent with its possessed noun (cf. also (17)). (20b) shows that when expressing the *have*-relation, the dative possessor cannot form a constituent with the possessed noun; (20c) shows that the unmarked possessor is also ungrammatical in this case. Szabolcsi (1994) suggests that this is because (20b,c) constitute definiteness effect contexts which can only take non-specific arguments (cf. also Szabolcsi 1986). Because the structures in (20), in which the possessors are not extracted, are ungrammatical, Szabolcsi concludes that non-specific possessed nouns that can appear in such contexts *cannot* form a constituent with their possessor:

- (21) a. When the possessor is inside DP (in the nominative or in the dative), DP is specific (potentially also definite).
  - b. For DP to be non-specific, it must have the possessor extracted (in addition to not containing any specific determiner, of course).

(Szabolcsi 1994: 226)

## 3.3.3 Possessed noun phrases and object agreement

Let us now consider how this conclusion relates to object agreement in Hungarian. For ease of exposition, I will first consider data from non-standard varieties of Hungarian. As mentioned above, in Standard Hungarian (and according to some grammars, e.g. Rounds 2002; Törkenczy 2002) possessed direct objects always trigger object agreement. There are, however, speakers of Hungarian who allow possessed direct objects to appear with verbs that agree only with the subject (see e.g. Rácz 1968; Szabolcsi 1994; Bartos 1999; É. Kiss 2000; Kiefer 2003; H. Varga 2010; Bárány 2014b).<sup>6</sup> These varieties are interesting as they show a clear correlation between the syntactic structure and the interpretation of possessed direct objects: only possessed objects whose possessor has been extracted can appear without object agreement.

(22) shows an example in which the direct object is a possessed noun phrase and the verb only agrees with the subject.

(22) Liszt Ferenc-nek talál-t-ak ismeretlen kézirat-a-i-t.
 Liszt Ferenc-DAT find-PST-3PL.SBJ unknown manuscript-3sG.POSS-PL.POSS-ACC
 'Unknown manuscripts by Ferenc Liszt were found.'

(É. Kiss 2000: 142)

É. Kiss (2000) points out that the same restrictions hold for the object in (22) as for non-specific possessives discussed in the previous section: the possessor cannot form a constituent with the possessed object, so (23) is impossible.

(23) \**Csak L. F.-nek ismeretlen kézirat-a-i-t* talal-t-ak. only L. F.-DAT unknown manuscript-3sg.poss-pl.poss-Acc find-pst-3pl.sbj (É. Kiss 2000: 142)

<sup>&</sup>lt;sup>6</sup> Even though much of the literature refers to examples of subject agreement with possessed direct objects like (22), it is not very clear which speakers of Hungarian actually accept these constructions. Most authors merely refer to non-standard, "minority" (Szabolcsi 1994) or "archaic" (É. Kiss 2000) varieties, but not much more about this or these varieties is known.

It should also be noted that in a series of online surveys, Bárány and Szalontai (2015) have found no evidence for the dialectal variation reported in the literature. The data presented here therefore reflect judgments by the authors reporting them (Szabolcsi, É. Kiss, Bartos; see H. Varga 2010; Bárány 2014b for some literary data). I take the data to be found on the internet to be representative as well, as they are unlikely to arise from typos.

In addition, the predicate *talál* 'to find' requires a non-specific complement. The following examples, with unmarked possessors, are also ungrammatical.

(24)	a.	* Tálal-t-ak	L. F. ismeretlen	kézirat-a-i-t.
		find-pst-3pl.sbj	L. F. unknown	manuscript-3sg.poss-pl.poss-acc
	b.	*Talál-t-ák	L. F. ismeretlen	kézirat-a-i-t.
		find-pst-3pl.obj	L. F. unknown	manuscript-3sg.poss-pl.poss-ACC
				(É. Kiss 2000: 142, (51a,b))

Szabolcsi (1994) cites similar examples, arguing that in (25) the possessed direct object scopes under the negation and receives the non-specific interpretation shown in the English translation.

(25) Chomsky-nak nem olvas-t-ál vers-é-t. Chomsky-DAT not read-PST-2SG.SBJ poem-3SG.POSS-ACC 'You haven't read any poem of Chomsky's.' (Szabolcsi 1994: 227)

Analogous examples can also be found outside of the linguistic literature, although often without an overt possessor, and sometimes including overt modifiers like *néhány* 'some' or *egy vagy több* 'one or more' as in the following examples.

(26)	a.	Ha valaki meghív-ott	egy vag	y több barát-já-t				
		if someone invite-PST.3SG.S	BJ one or	more friend-3sg.ро	SS-ACC			
		'If someone invited one or more of their friends' <sup>7</sup>						
	b.	Legalább tisztáz-ná-l	vele	néhány problé	má-d-at			
		at least clarify-cond-2sg.s	вJ with hin	n/her some problem	m-2sg.poss-acc			
		'At least you would clarify so	me of your	problems with him/	her' <sup>8</sup>			

. . .

Given the existence of such structures, Szabolcsi (1994) and É. Kiss (2000) argue that like in contexts expressing the *have*-relation, possessed direct objects can only appear with subject agreement if the dative possessor is extracted (in the relevant varieties). This rules out non-agreeing direct objects with unmarked possessors<sup>9</sup> and those with

<sup>&</sup>lt;sup>7</sup> http://hu.pokerstrategy.com/forum/thread.php?threadid=116626&page=6,7/10/2015.

<sup>&</sup>lt;sup>8</sup> http://www.nlcafe.hu/forum/?fid=441&topicid=159911&step=2&page=755, 7/10/2015.

<sup>&</sup>lt;sup>9</sup> At least this seems to be the general consensus. Kornai (1989) suggests that some speakers of Hungarian even accept unmarked possessors of direct objects that do not require subject agreement. I do not agree with this judgment and I have not been able to find such examples.

dative possessors that form a constituent with the possessed noun, like (23) above (I will come back to the nature of covert possessor below).

The restriction on the distribution of possessors is a syntactic restriction on the structure of possessed noun phrases that has a correlate in object agreement.

# 3.3.4 The structure of possessed noun phrases

Having established the existence of a syntactic restriction on the distribution of dative possessors and the interpretation of possessed noun phrases, I now turn to the discussion of the structure of possessed noun phrases in more detail. I have argued above that object agreement is only triggered by noun phrases that are DPs and have person features represented on D. This proposal can be carried over to the analysis of the structure of Hungarian possessed nouns.

The basic idea is as follows: possessive constructions that trigger object agreement have person features by virtue of projecting a DP layer with a potentially null determiner, which carries person features. Non-specific noun phrases lack this layer and therefore do not give rise to object agreement (or a referential interpretation). This could explain why their possessor has to be extracted, too: there is no place for it in the specifier of DP. This analysis closely follows Bartos (1999: 106) who suggests that possessor extraction can go ahead from a position below D (see also Bartos 2001: 319).<sup>10</sup>

In what follows, I will only discuss the structure of possessive constructions insofar as they are relevant for present purposes and I will basically adopt the structures suggested in Dékány (2015), which are in turn based on Szabolcsi (1994), Bartos (1999) and É. Kiss (2002) (for further overviews see Szabolcsi 1983, 1987; den Dikken 1999; É. Kiss 2000; Dékány 2011; see also Chisarik and Payne 2001; Chisarik 2002; Laczkó 2007).

<sup>&</sup>lt;sup>10</sup> A comment is in order here: recall that it is only true for possessed objects in non-standard varieties that they do not trigger object agreement. Szabolcsi (1994) suggests that non-specific possessives in Standard Hungarian do trigger agreement, like all other possessives.

However, indefinite possessed objects that can be interpreted as non-specific are generally ambiguous and can also be interpreted as specific (or referential). On any analysis of object agreement with possessed direct objects, non-specific direct objects are the only non-referential type of noun phrases triggering object agreement, an interesting irregularity with respect to object agreement in general (also noted by Szabolcsi 1994; see also Bárány 2012b). Further research about how readily speakers allow non-specific readings of possessed direct objects is necessary to shed light on this question.

The structures include the following functional projections: NumP, which indicates the number of the possessum; two PossPs, which are necessary to account for some complex possessive suffixes.<sup>11</sup> The lowest position for a possessor is the specifier of Poss2P, just below DP. The following examples illustrate this with unmarked phrasal and pronominal possessors, as well as dative possessors. I will use the lexical material shown in (27) in the trees below (cf. (25) above):

(27) Chomsky / (az én) vers-e-i-(m).
 Chomsky the I poem-3sg.Poss-PL.Poss-(1sg.Poss)
 'Chomsky's / my poems.'

**Unmarked pronominal possessors** These possessors appear with the definite determiner preceding the possessor, providing evidence for a possessor position below DP. They cannot be extracted and form a constituent with the possessed noun.



- <sup>11</sup>Dékány (2015) identifies the suffixes *-e-i-m* in (27) as separate heads. Both *-e-* and *-m* indicate possession, but *-i-* is a plural suffix and intervenes between the two possessive morphemes. What is relevant for the present discussion is that the pronominal possessor follows the definite determiner and precedes numerals in the specifier of NumP:
  - (i) *az én két vers-em* the I two poem-1sg.poss 'my two poems'

See Dékány (2015) for further discussion.

**Unmarked phrasal possessors** These possessors are in SpecDP and cannot appear with a definite determiner. They cannot be extracted and therefore form a constituent with the possessed noun. The possessor arguably moves to SpecDP from SpecPossP (indicated by the dashed arrow).

(29)



**Dative pronominal/phrasal possessors** Dative possessors can appear inside the possessed noun phrase, forming a constituent with it, but they can also be extracted, and *have to* be extracted in order for the possessed noun to be non-specific. Phrasal and pronominal possessors behave alike. Again, the possessor arguably moves to its final position from a SpecPoss2P.<sup>12</sup>

<sup>&</sup>lt;sup>12</sup> The origin of the dative on these possessors is not directly relevant for the present discussion. I simplify these structures by leaving open whether the lower copy of the possessor in (30) already has dative or not; see Szabolcsi (1994), Bartos (1999), den Dikken (1999) and É. Kiss (2002) on different proposals about the origin of the dative; cf. also Georgi (2014: 198ff.).



**Extracted possessors** Following Bartos (1999) and É. Kiss (2000, 2002), I suggest the following structure for the remnant of an extracted possessor that gives rise to a non-specific interpretation.



68

Several questions arise at this point with respect to the position of the extracted dative possessor in (31) and what triggers its movement; in any case, the dative possessor has to move to a position that is not part of the extended projection of the possessive construction. See É. Kiss (2014) for a recent analysis.

I suggest that the reason why the structure in (31), as opposed to the one in (30), for example, does not trigger object agreement is that it lacks the D head (again, basically following Bartos 1999). This implies that other possessive constructions (those that are not non-specific) project a DP and include a D with person features: in these constructions the possessors form a constituent with the noun and we have seen evidence above that they all project DP. For dative possessors, it is possible to spell out this overt D. For nominative possessors, it is not, but they are in complementary distribution with the definite determiner. Finally, for pronominal possessors, the definite determiner is obligatory. There is thus reason to assume that when D is projected in possessive constructions, the whole noun phrase is specified as having person features (see Danon 2011 for an analysis of DP-internal feature sharing).

Note that given the feature system I introduced above, this proposal requires another ingredient, namely the presence of an element in the noun phrase that introduces a person feature. This is straightforward with pronominal possessors, which require D to spell out overtly, but less clear with nominative and dative possessors. Given that there is no *a priori* reason why nominative possessors should rule out the presence of a definite determiner, it is possible that a null allomorph of the definite determiner can be spelled out when DP's specifier is filled.

To summarise briefly: in the varieties of Hungarian that do not require object agreement with possessed direct objects, there is a clear correlation between syntactic structure and interpretation. The possessed noun can only be interpreted as non-specific when the possessor does not form a constituent with it. I have suggested that this can be illustrated by the structure in (31), having left open some properties of the dative possessor for now.

## 3.3.5 Possessed direct objects in Standard Hungarian

The discussion in the previous section focused on a non-standard variety of Hungarian. We saw that the syntax-semantics mapping in this variety seems to be more 'regular'

in that object agreement does seem to have a certain interpretational correlate because only specific possessed direct objects require agreement (cf. the data in Section 3.3.3). In addition, most accounts of object agreement in Hungarian aim to account for both these and the standard variety. I turn to the latter now.

The facts are described easily: whenever a direct object is a possessed noun phrase, it triggers object agreement, even when it is non-specific (Szabolcsi 1994). Bartos (1999) suggests that this results from the fact that the option of extracting the possessor from below DP is not available in Standard Hungarian (i.e. the structure in (31) is never derived). This is technically possible and I will tentatively adopt this solution.

I argued above that both syntactic structure (the presence or absence of D) and formal features (the presence or absence of person features) determine object agreement. One way of characterising the difference between the non-standard varieties of Hungarian and the standard-varieties of Hungarian is the choice of D head that is part of possessive constructions. Since possessive constructions with phrasal nominative possessors are always definite, I have assumed that they include a null D head that carries person features. The same is true of possessive constructions with pronominal nominative possessors: these spell out the definite determiner overtly.

Constructions with dative possessors differ across the varieties, however. As Szabolcsi (1994) points out, non-standard varieties are more regular in this respect, since their syntactic structure, their interpretation, and behaviour with respect to object agreement pattern together. Bartos (1999) suggests that these lack D and therefore do not trigger agreement. He suggests that in the standard variety, it is simply not possible to derive possessive constructions without a D head, and therefore all trigger object agreement.

If person features and syntactic structure determine agreement together, it is possible that possessive constructions have the same syntactic structure across the varieties, but that non-standard varieties allow for the D head to surface without a person feature. Standard varieties, however, might lack this option: here, all D heads in possessive constructions come with a person feature and therefore trigger object agreement. This would mean that there is a mismatch between the presence of a person feature on D and the existence of non-specific readings of such possessive noun phrases in standard Hungarian (see also footnote 10 above). In sum, there is clear evidence that the syntactic structure of the possessive constructions influences object agreement across all varieties of Hungarian, since nominative possessors do not show any variation. Constructions with dative possessors pose a more difficult analytical challenge, but D seems to play a role in these constructions, too. Bárány and Szalontai (2015) find that speakers generally prefer possessive constructions with a dative possessor with an overt determiner over those that lack an overt determiner (both for definite and indefinite determiners). Note that Bárány and Szalontai's (2015) surveys do not test for the interpretation of possessed noun phrases: it is not clear how common non-specific possessed DOs are. I conclude that while there is evidence for the role of syntax in determining object agreement with possessive constructions, further research is needed in this domain.

Before concluding this section, I turn to an alternative proposed by Coppock and Wechsler (2012) and Coppock (2013), namely that the nature of the possessor is irrelevant, and that it is the possessive suffix that determines object agreement.

# 3.3.5.1 A potential alternative: the possessive suffix as the trigger

I have argued above that there is a syntactic restriction on the distribution of object agreement with possessed noun phrases in some varieties of Hungarian: in order for subject agreement to be grammatical, the possessor has to be extracted from the noun phrase. Following Bartos (1999), I suggested that this can be modelled syntactically by assuming that the lack of a D projection leads to subject agreement. In this section, I consider an alternative approach to deriving object agreement, in particular with respect to possessive constructions and argue that the more syntactic approach suggested above covers a wider range of data (see also Bárány 2014b).

Coppock and Wechsler (2012) and Coppock (2013) propose that certain lexical items come specified for a feature called [DEF] (note again the connection to person features in the present analysis). Because possessed direct objects often trigger object agreement, these authors suggest that the possessive suffix is one of the lexical items that is specified as [+DEF] and therefore possessive constructions give rise to object agreement. Note that Coppock (2013) only considers varieties of Hungarian in which possessed direct objects can co-occur with subject agreement, i.e. non-standard varieties.

The system works as follows: lexical items can be specified as [+DEF], [-DEF] or lack a specification for the feature. A principle of percolation ensures that the feature on an embedded head is projected upwards in the noun phrase. Since different lexical items can have conflicting feature specifications, this system predicts that there can be feature clashes in the noun phrase, for example when one lexical item introduces a feature [-DEF] and another one introduces [+DEF]. This is a welcome consequence for Coppock (2013), however. We have seen that there *is* variation with respect to object agreement with possessed direct objects. Coppock suggests that this is due to the conflicting feature specification. The following example illustrates this reasoning.

(32)

[-DEF]/[+DEF]				
$\left[-\text{DEF}\right]$	[+DEF]			
néhány	titk-od			
'some'	'secret-2sg.poss'			

(Coppock 2013: 368)

Coppock suggests that in (32) both [-DEF] and [+DEF] are present on the noun phrase, and either feature can determine agreement. She suggests that this "predicts [that] there is variation and uncertainty in the judgments about the subjective vs. objective conjugation in this case" (Coppock 2013: 368).

There are some problems with this account: first, it ignores the fact that the syntactic nature of the possessor influences the choice of agreement (as discussed above). Second, relying on both [+DEF] and [-DEF] leads to false predictions about the distribution of object agreement. I will expand on this argument in connection with (33). Third, the semantic analysis of [+DEF] in Coppock (2013) predicts that there are existential presuppositions where there are in fact none. I elaborate on this argument in connection with (34).

Let us consider the syntactic argument first. The reasoning for the variation in object agreement with respect to (32) was the presence of both features in the noun phrase. Imagine a situation in which there is no determiner that introduces [-DEF]. The prediction is that in such cases the [+DEF] will always give rise to object agreement. This,

in turn, predicts that unmodified possessed direct objects always trigger object agreement. This prediction is false, however. Consider (33), repeated from (25) above.

(33) Chomsky-nak nem olvas-t-ál vers-é-t. Chomsky-DAT not read-PST-2SG.SBJ poem-3SG.POSS-ACC
'You haven't read any poem of Chomsky's.' (Szabolcsi 1994: 227)

In (33), the possessed noun does not have a modifier, so there is no element that could introduce [-DEF] into the structure in (33): there is no feature conflict and if the possessive suffix is indeed specified for [+DEF], the system outlined above fails to account for the lack of object agreement.

There is also a semantic problem with Coppock's (2013) analysis of the possessive suffix. As we have just seen, it is argued to be specified as [+DEF] and Coppock treats the suffix as a presupposition trigger. We have already seen, however, that possessed nouns can appear as complements to predicates that require a non-specific argument. In such cases, the argument does not carry an existential presupposition. Consider the following example, repeated from (16b) above.

(34) Mari-nak nincs macská-ja. Mari-DAT is.NEG cat-3sG.POSS 'Mari does not have a cat.'

(34) asserts that it is not the case that there is a cat-individual that stands in a possession relation to *Mary* and crucially, (34) does not presuppose the existence of any such cat-individual. Recall also that according to Coppock (2013) the feature [DEF] is what determines the difference in presuppositionality between *minden* and *mindegyik* (see (8) on p. 53 and the discussion there). The feature [DEF] cannot be present on the possessive suffix, however, since the interpretation of (34) is not presuppositional.

I take these facts to show that the possessive suffix is not the trigger of object agreement. This assumption makes false predictions about the distribution of object agreement and the interpretation of possessed noun phrases in non-specific contexts.

### 3.3.5.2 Interim summary

In this section, I have illustrated several aspects of Hungarian possessive constructions. First, they can appear with different kinds of possessors: they can be phrasal or pro-

nominal, and both types can be unmarked or in dative case. I have discussed evidence that certain predicates require non-specific arguments. These predicates also require the possessor to be extracted from the noun phrase.

The same syntactic requirement holds when a possessed noun phrase is a direct object. In the varieties of Hungarian that allow subject agreement with possessed direct objects, the possessor of these noun phrases *must* be extracted, according to the literature.

I have argued that this provides evidence for locating the trigger of object agreement in the D layer of the noun phrase: possessors are located in the periphery of the noun phrase. Since removing a possessor from this periphery bleeds object agreement, associating the D layer with object agreement is a sensible hypothesis. To strengthen this syntactic argument, I have pointed out some shortcomings of an analysis that treats the possessive suffix as the trigger of object agreement.

Questions remain, however. To the best of my knowledge, it is not known which varieties of Hungarian allow for variation in object agreement with possessed direct objects (Bárány and Szalontai 2015). Given that the present (and other analyses) focus on these non-standard varieties to a large degree, it is important for future research to get a clearer picture of the relation between the non-standard varieties and Standard Hungarian. Otherwise, it is not clear which grammatical system linguists are actually describing.

In a nutshell, the problem is the following: because possessors have to be extracted in both standard and non-standard varieties when a possessed noun is non-specific, there is a parallel syntactic requirement on such structures. However, while in nonstandard varieties the syntactic process of possessor extraction correlates with interpretation and the absence of object agreement, in the standard variety, we only see a correlation in interpretation. More research on these structures and their exact properties is therefore needed to settle these questions definitely. Part of this research involves collecting more data to determine the reality and the properties of the varieties in question; another part is to provide further evidence for the internal structure of noun phrases and the relation of external possessors to the possessed noun (see den Dikken 1999; É. Kiss 2014 for such attempts).

## 3.3.6 Conclusions: syntactic structure and person features

In this section, I have argued for a "hybrid" trigger of object agreement in Hungarian: object agreement is not triggered by a purely syntactic or a purely semantic property alone, but by syntactic structure and the presence of person features. Object agreement is only triggered by a D head that is specified for person features. If D lacks person features, it does not trigger agreement.

This analysis bears some resemblance to work by Coppock and Wechsler (2012) and Coppock (2013) who suggest that the trigger of object agreement is determined lexically (see also Farkas 1990 who argues for the need of a *morphological* feature [def]). But the present approach incorporates insights from Bartos (1999) who argues that the mere presence of D determines object agreement, independently of its feature content.

I pointed out certain problems for both syntax-only and semantics-only approaches. For the former, the syntactically identical behaviour of the universal quantifiers *minden* and *mindegyik* is problematic, while for the latter, the wide range of possible interpretations of noun phrases triggering object agreement is a problem.

I showed that specificity does not *per se* trigger agreement (following *i.a.* Coppock and Wechsler 2012) but that certain classes of lexical items that *are* specific do, namely the *-ik*-determiners discussed in Section 2.2.2.2. I take this to mean that there is a formal feature that indicates that a lexical item contributes a presupposition to a noun phrase. I further suggest that this formal feature is grammaticalised as  $\pi$  in Hungarian, and is therefore present on any direct object with a person specification, including first and second person noun phrases, discussed in Chapter 4.

The fact that this feature is a formal feature of lexical items distinguishes its semantic effect from specificity that arises via the semantic composition directly, i.e. from cases of ambiguity between specific and non-specific readings with determiners like *néhány* 'some' or the indefinite determiner egy 'a'. I further argued (and will argue below) that this formal feature is a person feature.

The syntactic position of this person feature is also relevant, however. I followed Bartos (1999) and É. Kiss (2000, 2002) in suggesting that the class of noun phrases triggering object agreement behave alike syntactically and project the same structure, namely a DP. Object agreement in Hungarian is therefore not triggered by definiteness, specificity or other semantic properties, but by the interaction between person features and syntactic structure.

# 3.4 An implementation of object agreement

In the next chapter, I will show that the distribution of object agreement with personal pronouns provides a good argument in favour of treating Hungarian object agreement as the spell-out of an Agree relation between v and the direct object. In this section, I will specify how this Agree relation works and how the differential character of object agreement can be modelled.

As mentioned before, the main claim of this chapter is that in Hungarian, the verb agrees with the object when the object has person features. I will refer to these as  $\pi$ -features (Béjar 2003; Béjar and Rezac 2009) to distinguish them from  $\varphi$ -features proper which also include number (and gender, in languages where it is grammaticalised). Since the verb does not agree with its object in number, it suffices to refer to  $\pi$ -features (I will use both " $\pi$ " and "person" equivalently).

If the verb only agrees with objects that have person features, there must be noun phrases that lack person features. I follow Bernstein (2008), Longobardi (2008) and M. Richards (2008) in assuming that definiteness, animacy and person are connected and that they are specified in the D layer of the noun phrase. Noun phrases that lack this D layer lack visible person features and are thus not available for agreement with v. At the same time, it is possible to project a D layer but lack person features, as in the case of the quantifier *minden* 'every', discussed below. The argument for object agreement in person will be presented in more detail in Chapter 4, but here I focus on implementing the general approach to object agreement in Hungarian. I will first discuss the role of person features before illustrating the way these features interact with Agree in the syntactic derivation.

# 3.4.1 Object agreement and person features

I have argued above that both the presence of formal features and the projection of DP are necessary. I treat D as the locus where  $\varphi$ -features, and therefore person features, are expressed (see Bernstein 2008; Longobardi 2008; M. Richards 2008; Danon 2011 for arguments of D as a person head; see also Höhn 2015; van der Wal 2015).

I interpret this proposal to mean that there are four, rather than three persons: first and second person remain the same, but what is traditionally referred to as "third" person can be divided into two sub-classes: definite and indefinite, animate and inanimate, or obviative and non-obviative, for example. On M. Richards's (2008) analysis, definites and animates carry a person feature, but indefinites and inanimates do not. Person is thus tied to referential and semantic properties. Given that definites and indefinites can have different kinds of semantic referents (individuals vs. properties; cf. Longobardi 1994, 2005, 2008 for a similar view), this is a welcome consequence. Note also that a common noun lacking a person feature be provided with one by combining with a definite determiner, which carries a person feature and is merged in D. I discuss a similar case of an animate determiner in Section 5.4.2.2.

M. Richards (2008) in particular proposes that since third person noun phrases are the only ones that can be indefinite (and inanimate), it is possible to treat definiteness (and animacy) as person features: he suggests that indefinites can be treated as lacking a person feature and thus definiteness (cf. Adger and Harbour 2007 for a similar proposal with respect to animacy). These views also fit with Bartos's (1999) proposal of linking object agreement to DP, as well as É. Kiss's (2000, 2002) suggestions that DPs refer to individuals and smaller projections (like NumPs) refer to properties.

My perspective is slightly different. While person features are expressed on D, the set of person features of a noun phrase can be empty. This makes the present system more flexible which is empirically necessary as object agreement in Hungarian does not simply correlate with definiteness. The exact semantic contribution of person features remains an open question, but I propose to link it to presuppositionality in the sense that noun phrases that have person features presuppose the existence of an individual they refer to while noun phrases lacking it do not (this can be a salient set in the discourse as with *-ik* determiners). As M. Richards (2008: 139) puts it, person is the syntactic representation of specificity, definiteness or animacy. I suggest that in Hungarian, the person feature  $\pi$  grammaticalises referentiality.

This allows me to capture the behaviour of definite determiners, demonstratives, personal pronouns (I will return to these below and in Chapter 4), as well as the class of *-ik*-determiners that are D-linked and presuppose a referent or a domain restriction to quantify over. This characterisation differs from Coppock's (2013) proposal that noun phrases with the feature [DEF] have to be anaphoric; her proposal specifically

excludes first and second person pronouns from the triggers of object agreement. My proposal includes them as triggers of object agreement. Possessive constructions are a further difference: Coppock argues that the possessive suffix is a presupposition trigger. We have seen in Section 3.3.5.1 that this analysis makes wrong predictions about the distribution of presuppositions. Locating person features inducing referentiality in D avoids this. We have seen evidence from non-standard varieties of Hungarian that object agreement with possessed direct objects correlates with their syntactic structure: only when the possessor is extracted can the verb fail to agree with the object (see the discussion in Section 3.3).

In the spirit of M. Richards (2008) and related work, I thus propose that the features triggering differential agreement in Hungarian are person features. Hungarian nominals can then be split into four classes: first, second, and third person, as well as a person-less class of noun phrases that include indefinites and other non-referential noun phrases. In Hungarian, this latter class consists of those noun phrases that do not trigger object agreement. Other languages have been argued to make similar distinctions, too. Lochbihler (2008, 2012) discusses the Algonquian language Ojibwe, which distinguishes different "third" persons for "proximate" and "obviative" categories (see also Zúñiga 2006 for other Algonquian languages). See Mithun (1999: Sect. 3.1) for discussion of types of "fourth" person.

I will assume that these four persons in Hungarian are represented as shown in (35) (cf. (5) in Chapter 1).

Subject agreement appears with any kind of noun phrase, but object agreement is only sensitive to noun phrases with person features. This, of course, is differential object marking. However, it also highlights an asymmetry between subject and object agreement: why can subjects that lack person features trigger agreement? I propose that the difference lies in the fact that subject agreement is sensitive to number, in addition to person, whereas object agreement is not. A subject will always agree in number with T, and I suggest that this number agreement always entails agreement in person, too (see also Adger and Harbour 2007 and the discussion in N. Richards 2010: 81; I will return to the question of such asymmetries in Chapter 6).<sup>13</sup>

## 3.4.2 Deriving agreement

The asymmetry just discussed will be reflected in the following derivations by a difference in the feature content of T and v. While T has a full  $\varphi$ -probe, including person and number, v only ever agrees in person, i.e. it lacks a number probe. In (36), and below, this is represented as u $\varphi$  and u $\pi$ , respectively. (36) shows a schematic derivation that illustrates the order of Agree and Case valuation as well as movement.



The order of operations in (36) is as follows:

- (A) V is merged with the direct object; it subsequently moves to v.
- (B) Agree between v and DO, valuation of  $\pi$  and CASE features.
- (c) Movement of v to T.
- (D) Agree between v+T and SUBJ, valuation of  $\varphi$  and CASE features.

<sup>&</sup>lt;sup>13</sup>Note that this does not mean that all subjects are DPs. This rather suggests that a number probe can see different layers in a noun phrase; if it encounters a number feature, present on all nominals, it will automatically value a person feature as well — the asymmetry is thus not reflected in the structure of the arguments themselves, but in the structure of the probes.

Having established that person features trigger agreement and the basic idea behind the syntactic derivation, I turn to the discussion of how person features are represented in the Hungarian noun phrase. This section combines the data introduced in Sections 2.2 and 3.1 with the implementation of object agreement suggested in this section.

In order for object agreement to arise, v has to be valued by a person feature. I propose that this only happens if the direct object is a DP with a non-empty set of person features. The mechanism valuing the person features on v is Agree (Chomsky 2000, 2001), in the specific implementation of Preminger (2011, 2014). As briefly discussed in Section 1.5.2, I assume that if Agree does not find a goal that can value its person features, the derivation can continue.

In Hungarian, this works as follows. If v does not encounter an object with person features, it will abort probing and the derivation will continue with v no longer active and receiving an empty default value. I assume that a probe only continues to probe after a successful Agree relation. Preminger's (2014) formulation of Agree is shown in (37).<sup>14</sup>

(37)  $\operatorname{FIND}_{\omega}(f)$ :

Given an unvalued feature f on a head H<sup>0</sup>, look for an XP bearing a valued instance of f. Upon finding such an XP, check whether its case is acceptable with respect to case discrimination:

- a. *yes*  $\rightarrow$  assign the value of *f* found on XP to H<sup>0</sup>
- b. *no*  $\rightarrow$  abort FIND<sub> $\varphi$ </sub>(*f*) and continue with derivation

(Preminger 2014: 159)

## 3.4.2.1 Common nouns

Common nouns do not come with a person feature that expresses referentiality, but it can be introduced by a definite determiner or a quantifier, etc. This is shown in (38), for the direct object *a két bicikli-t* 'the two bicycles'. In the following representations,

<sup>&</sup>lt;sup>14</sup> I will discuss the notion of *case discrimination* in (37) in detail in Chapter 6. In brief, it refers to whether arguments bearing certain cases can agree or not — recall from Chapter 1 that in Hindi, the verb only agrees with arguments that are not overtly case-marked.

I am abstracting away from  $\varphi$ -features other than person and case until considering the role of these noun phrases in the clause below.



# 3.4.2.2 Universal quantifiers

The following structures will illustrate the difference between noun phrases that trigger object agreement and those that do not. First, let us consider the universal quantifiers *minden* 'every' and *mindegyik* 'each'. As Coppock and Wechsler (2012) and Coppock (2013) show, these quantifiers can take NumP complements and they do not occur adjacent to the definite determiner a(z) (Szabolcsi 1994). One way to account for this is to argue that the quantifiers *minden* and *mindegyik* move to and spell-out the D position in certain cases. The differences in object agreement come about through the different feature specification of these two items. This is shown in (39) and (40).

Following Szabolcsi (1994), I represent the quantifiers as being introduced as Det heads. The elements that are (externally) merged in the D head are the definite determiner a(z) and a null allomorph that appears with certain possessive constructions (see examples (17), p. 37, (18a), p. 61, and Section 3.3, p. 58). Since *minden* lacks a person feature, the noun phrase does not have person features either.



In (40), the quantifier *mindegyik* does have a person feature, [3], and the noun phrase triggers object agreement. Again, this approach resembles the one in Coppock and Wechsler (2012) and Coppock (2013) with the difference that D is the relevant position of the person feature involved.





# 3.4.2.3 Specific adjectives

A lexical specification giving rise to a referential interpretation can arguably also be introduced lower in the structure than in D. This could be the case with lexical items like *bizonyos* 'certain' which have a specific interpretation. When these co-occur with a definite determiner or demonstratives (in (41)), D is merged with a [3] feature and the noun phrase triggers object agreement. If D is not projected, however, the noun phrase will not have a D head with a set of  $\varphi$ -features.

(41)



When D is absent, the noun phrase does not trigger object agreement, as shown in (42). The syntactic restriction on object agreement is implemented by assuming that D is necessary to house the person features of the direct object.



# 3.4.2.4 Possessed noun phrases

Recall the discussion in Section 3.3.3 above about how to capture the distribution of agreement with possessive constructions. In non-standard varieties, possessed direct objects with a non-specific interpretation do not trigger object agreement. We have seen evidence from Szabolcsi (1994) and É. Kiss (2000, 2002) that these constructions must involve a non-local dative possessor. Following Bartos (1999), I suggest that the extraction of the dative possessor is represented structurally as in (43), in which there is no D, and therefore no head to house the person features of the whole noun phrase (cf. the arguments presented in Section 3.3.5.1 against treating the possessive suffix as the location of [3]).

### 3.4 An implementation of object agreement



The structures of possessive noun phrases that do trigger object agreement are as shown in (44) and (45), for nominative and dative possessors, respectively.<sup>15</sup>

<sup>&</sup>lt;sup>15</sup> Theresa Biberauer (p.c.) points out that possessed nouns do seem to carry  $\varphi$ -features. Why would they not trigger agreement? Note that while the possessive suffix on a noun phrase indicates the person and number of its possessor, the possessed noun itself is always third person, independently of the person of its possessor. This suggests that the  $\varphi$ -features of the possessor do not interfere with the  $\varphi$ -features of the possessed noun phrase itself.



### 3.4.2.5 Complement clauses with hogy

I discussed the behaviour of complement clauses introduced by the complementiser *hogy* with respect to object agreement in Section 2.2.2.6. The main facts can be summarised as follows.

First, complement clauses can be associated with an accusative demonstrative pronoun in the matrix clause. This is shown in (46) (these examples are repeated from Section 2.2.2.6).

(46) Kati az-t képzel-i [<sub>CP</sub> hogy a gép el-roml-ott]. Kati that-Acc believe-3sg.овј that the machine vм-break-psт.3sg.sbj 'Kati imagines that the engine's broken down.'

(Kenesei 1994: 309)

In (46), the verb *képzeli* can agree with the demonstrative *az-t* 'that-OBJ'. Coppock and Wechsler (2012) argue that movement out of the CP should not be possible in all circumstance, however.

Such movement is possible in many cases, as shown in (47). In (47a), *Péter-t* 'Péter-ACC' is moved out of the *subject* position of the CP, but it is nevertheless assigned accusative in the matrix clause. Since it is a referential proper name direct object, it triggers object agreement.

(47)	a.	Anna [ <sub>DP</sub>	Péter-t <sub>i</sub> ]	akar-ja	(*az-t) [ <sub>CP</sub>	hogy	meg-látogas-s-am	$\mathbf{e}_{i}].$
		Anna	Péter-ACC	want-3sg.obj	it-ACC	that	vм-visit-sbjv-1sg.obj	
		'It is Péter	that Anna	wants me to	visit.'			

(Kenesei 1994: 314)

b. Anna  $[Fgyorsan_j]$  akar-ja / \*akar  $[CP e_j hogy$ Anna fast want-3sg.obj want.3sg.sbj that meg-javít-s-am az autó-t  $e_j$ ] vM-repair-sbjv-1sg.obj the car-ACC 'Anna wants me to repair the car FAST.' (Kenesei 1994: 316)

(47b) is different, however, in that the moved item *gyorsan* 'quickly' is not the type of noun phrase to trigger object agreement.

There are two alternative ways of explaining the pattern of agreement with such CP objects. First, it is possible to retain the idea of a DP-CP association under certain

assumptions. As Gervain (2004, 2009) shows, what seems to be the subject of the embedded clause can be base-generated as an object in the matrix clause. Evidence for this comes from the subject agreement patterns of such base-generated objects. First, (48) shows that a phrase like *all the girls* triggers singular agreement in Hungarian.

(48) Az összes lány jön / \*jön-nek. the all girl come.3sg.sbj come-3pl.sbj 'All the girls come.'

But when the subject of (48) is focus-"raised", subject agreement in the embedded clause changes to third person plural. Gervain (2009) argues that a resumptive pronoun is responsible for this (see also Jánosi *et al.* 2014).

(49) Az összes lány-t mond-t-ad, hogy jön-nek / ?jön. the all girl-ACC say-PST-2SG.OBJ that come-3PL.SBJ come.3SG.SBJ 'You said that ALL THE GIRLS would come.'

(Gervain 2009: 692)

Since the focused element in the matrix clause can be base-generated there, it is possible to maintain that there is a demonstrative pronoun bearing a person feature in the matrix clause, which is the expletive associate of the complement clause and triggers agreement.

Alternatively, it is possible that the verb agrees with the CP directly. This has been suggested for Tagalog by Rackowski and N. Richards (2005) and for Hungarian by den Dikken (2006, 2012) and Rocquet (2013: 192ff.), for example (see also Section 2.2.2.6).

Given the previous discussion and my proposal that object agreement is triggered by D with person features, the question is why CPs can also trigger agreement. De Cuba and Ürögdi (2009, 2010) and Coppock (2013) argue independently of each other that CP complements can be referential; Coppock (2013: 368) suggests that the complementiser *hogy* could be a quantifier over possibilities, analogous to the definite determiner being a quantifier over individuals. De Cuba and Ürögdi (2009, 2010) further claim that the demonstrative expletive appears when the CP is not referential. If true, it is possible that not only D can carry person features in Hungarian, but also C, and that therefore CPs can directly trigger object agreement. Note also that Szabolcsi (1994) stresses the parallelism between DPs and CPs, including their status as arguments and treating

"articles as complementisers" (Szabolcsi 1994: 179). See also Takahashi (2010), Sheehan (2011), Sheehan and Hinzen (2011) and Moulton (2015) for further discussion of similarities and differences between DPs and CPs.

## 3.4.2.6 Properties of T and v

The structures in (38)–(45) show the feature specifications of different kinds of noun phrases. I have argued that when the set of  $\varphi$ -features located on D contains a set of person features, e.g. [3], and the verb enters an Agree relation with the noun phrase, object agreement appears. Subsuming this feature under D accounts for the 'hybrid' aspect of Hungarian object agreement: it relies on the formal features of the nominal projection but also its syntactic structure. As mentioned above, being specific is not enough: the adjective *bizonyos* that gives rise to a specific interpretation of the nominal it modifies, does not project D and therefore cannot carry a person feature. Accordingly, it does not trigger object agreement.

I suggested that the properties of T and v can capture the subject-object asymmetry with respect to person agreement. Recall that finite verbs in Hungarian Agree with any type of subject, independently of its syntactic structure. For objects, agreement is restricted to noun phrases that carry a person feature. I will discuss this asymmetry and how it is implemented here in more detail.

It is plausible that the cause of this asymmetry lies in the nature of the probes responsible for subject and object agreement, respectively, i.e. T and v. T carries a full set of (sets of) unvalued  $\varphi$ -features (person and number, as gender is not grammaticalised in Hungarian), while v only has unvalued (sets of) person features. This is shown in (50).

(50)



Recall that I assume that Agree can fail and that case and agreement are dissociated (see Section 1.5.2 and Chapter 6). Given these assumptions, v behaves as shown in

(51). In (51a), we see a successful Agree relation in which the direct object's [3] feature values v's person features. v also assigns accusative to the direct object.



(51b), on the other hand, shows a failed Agree relation. v assigns case to the direct object, but its person features remain unvalued, as the direct object does not have a person feature. As v does not have a number probe, the direct object's number feature is not involved in the Agree relation. Case is assigned nevertheless.



(51)



T, on the other hand, does have a full set of  $\varphi$ -features, including number ( $\mathbf{x}$  in a feature matrix indicates that the relevant features are valued at step x). This can account for the subject-object asymmetry under the assumption that an Agree-relation that values T's number feature also values T's person features. This matches the empirical fact that all subjects Agree in number with T, but there is no object agreement in number. The relevant kind of Agree relation is illustrated in (52).
## 3.4 An implementation of object agreement



## 3.4.3 Putting the pieces together: agreement in the clause

We are now in a position to illustrate the derivation of transitive clauses with indefinite and definite objects, respectively. (53a) shows a transitive sentence with the indefinite object *egy bicikli-t* 'a bicycle-ACC'. This object does not trigger agreement, and the result is that the verb only agrees with the subject. This derivation is illustrated in (53b).<sup>16</sup>

(53) a. *Péter lát egy bicikli-t.* Péter see.3sg.sbj a bicycle-ACC 'Péter sees a bicycle.'

 $<sup>^{16}</sup>$ I include fully inflected verb forms in this tree for the sake of presentation only – I will discuss the spell-out of verb forms in Chapter 4.

3 A hybrid analysis of object agreement: syntactic structure and  $\pi\mbox{-}features$ 



The derivation of (53) proceeds as follows:

- (A) Agree between v and DO. The direct object egy bicikli-t lacks person features so v's person features are not valued. The object's uCASE feature is valued as ACC.
- (B)  $\nu$  moves to T.
- (c) Agree between v+T and the subject *Péter*; the subject values the features of v+T and in turn its uCASE features is valued as NOM.

(54a,b) illustrate an example that involves object agreement. The difference to (53b) is the syntactic structure of the noun phrase and its features.

(54) a. *Péter lát-ja a bicikli-t.* Péter see-3sg.obj the bicycle-Acc 'Péter sees the bicycle.'

#### 3.4 An implementation of object agreement



The relevant steps of the derivation of (54b) are the following:

- (A) Agree between v and DO. The direct object *a bicikli-t* has [3] among its  $\varphi$ -features and values the probe on v, which in turn values the direct object's uCASE as ACC.
- (B)  $\nu$  moves to T.
- (c) Agree between v+T and the subject *Péter*; the subject values the features of v+T and in turn its uCASE features is valued as NOM.

## 3.4.4 Interim summary

In this section, I showed how the interaction of  $\varphi$ -features and syntactic structure argued for above can be implemented in an Agree-based framework. I provided structural representations for two kinds of noun phrases: those that do trigger object agreement, and those that do not. I argued that the difference lies in their syntactic structure and their  $\varphi$ -feature specification.

Syntactic structure takes priority over the semantic properties of the nominal projection. This is because no argument lacking a DP projection will trigger object agreement. But I suggest that projecting DP is not a sufficient condition (*contra* Bartos 1999). DP is necessary to make the sets of person features of the noun phrase visible to Agree and only sets of person features on D can value v: a D head without person features does not give rise to object agreement.

## 3.5 Conclusions

In this chapter, I implemented an analysis of object agreement in Hungarian that relies on the syntactic structure of the direct object and its  $\varphi$ -feature specification. I called this approach "hybrid" because it is inspired by Bartos's (1999) suggestion that DP structure is necessary for object agreement to arise, as well as Coppock and Wechsler's (2012) and Coppock's (2013) approach to object agreement using formal features.

The analysis proposed here differs from both of these approaches in that it covers more empirical ground, however. In particular, I suggested that the formal features involved in triggering agreement are person features, and that they are expressed on D. There are several advantages to this approach.

First, D has been argued to encode referential properties of noun phrases in a number of languages (Longobardi 1994; É. Kiss 2000, 2002; Longobardi 2005; Danon 2006; Bernstein 2008; Longobardi 2008; M. Richards 2008; Danon 2011). Second, by linking person features to D and its referential properties, it is possible to capture the generalisation that first and second person are always definite and animate, while third person can be definite or indefinite, as well as animate or inanimate. Third, this approach makes strong predictions, namely that Hungarian direct objects of all persons should trigger object agreement. At first glance, this prediction seems to be wrong, since first person pronouns never trigger object agreement. In the next chapter, I will argue that personal pronouns actually all do trigger agreement, but that there are independent restrictions on when agreement is spelled out. In Chapter 5, I provide additional evidence that this is the right approach by discussing agreement and case-marking in languages which grammaticalise person features in different ways from Hungarian, but can be analysed along the same lines.

The analysis proposed in this chapter gives rise to the following generalisations about object agreement with third person objects in Hungarian. For (55) and (56), I take *definite* to mean *referring to a unique, existing individual in a given situation*.

- (55) Object agreement in HungarianIf a direct object is definite, it triggers object agreement.
- (56) Definiteness and noun phrase structure in HungarianIf a direct object is definite, it projects a DP.

These two generalisations are not necessarily new when restricted to third person arguments. One of the goals of the next chapter will be to extend these generalisations to first and second person objects as well, however, and therefore extend the coverage of (55) and (56).

# 4.1 Introduction

The goal of this chapter is to provide an analysis of object agreement with personal pronouns in Hungarian. The basic distribution of agreement is described as follows: first person pronoun objects never trigger object agreement, second person pronoun objects only do with first person subjects, and third person personal pronoun objects always do.

I will argue that the surface distribution of object agreement does not reflect the underlying agreement relations: the position taken in this chapter is that all personal pronouns trigger object agreement, but the interaction of syntax and morphology gives rise to "gaps" in object agreement. One advantage of this analysis is that agreement with second person objects, sometimes treated as exceptional in the literature, is a fully regular part of the Hungarian agreement system.

I believe that this result holds independently of the specific formalism chosen in this chapter. Part of the reason for this is that the contexts in which object agreement is expected, but not seen on the surface, match so-called *inverse* contexts in other languages that give rise to special verbal morphology and case-marking in several languages (this will be the topic of Chapter 5). I suggest that this cross-linguistic similarity provides further evidence for the analysis proposed here. Another reason to assume that all personal pronouns Agree is that they behave alike when it comes to independent syntactic properties like object drop and licensing a secondary predicate. I discuss this in Section 4.3.1.

This chapter is structured as follows. In the following section, I introduce the distribution of object agreement with personal pronouns and the puzzle it poses for an analysis of object agreement in Hungarian. In Section 4.3, I present the analysis of the syntactic derivations that distinguish contexts in which object agreement surfaces from those in which it does not. I discuss morphological aspects of object agreement and provide spell-out rules in Section 4.4. The present analysis is compared to other approaches in Section 4.5. Section 4.6 concludes.

## 4.2 Object agreement with personal pronouns

I ended the previous chapter by stating two generalisations about object agreement in Hungarian that linked definiteness and DP structure to the presence of object agreement (see (55) and (56), p. 94). The behaviour of personal pronouns with respect to object agreement, however, seems to provide a clear counterexample to the claim that definite noun phrases trigger object agreement.

Consider again the following examples, illustrating how the person of the object determines whether a verb shows object agreement or not. In (1), with a third person subject and a third person personal pronoun as the object, object agreement is obligatory. In (2), on the other hand, when the direct object is first person, object agreement is ruled out.

- (1) Mari lát-ja / \*lát ő-t. Mari see-3sg.obj see.3sg.sbj s/he-ACC 'Mari sees him/her.'
- (2) Mari lát / \*lát-ja engem. Mari see.3sg.sbj see-3sg.obj I.Acc 'Mari sees me.'

The same pattern holds when the person of the subject changes: the personal pronoun  $\delta$ -*t* 's/he-ACC' always triggers object agreement, while the first person pronoun *engem* 'I.ACC' never does.<sup>1</sup>

However, the behaviour of second person pronouns suggests that this is not the whole story. In (3), with a third person subject and a second person object, the verb shows subject agreement only (cf. (2)). In (4), on the other hand, with a first person subject, neither the verb form showing only subject agreement, nor the one showing agreement with third person objects is grammatical, but we see the suffix *-lak*.

<sup>&</sup>lt;sup>1</sup> In general, plural pronouns behave the same way, but there are some complications to be discussed in Section 4.4.

#### 4.2 Object agreement with personal pronouns

- (3) Mari lát / \*lát-ja téged. Mari see.3sg.sbJ see-3sg.obJ you.sg.Acc 'Mari sees you (sg.).'
- (4) (Én) lát-lak / \*lát-ok / \*lát-om téged.
  I see-1sg>2 see-1sg.sBJ see-1sg.oBJ you.sg.ACC
  'I see you (sg.).'

These data seem to violate generalisation (55), which states that definite noun phrases trigger object agreement. In this chapter, I argue that the first and second person pronouns in the examples just illustrated *do* trigger object agreement in principle, but this agreement is not spelled out.

One crucial piece of evidence for this idea is the contrast between (3) and (4). The *-lak/-lek*-suffix is sometimes assumed to be an element distinct from regular object agreement because it appears with second person objects.<sup>2</sup> Considering the distribution of agreement in (4), however, it makes perfect sense to treat *-lak/-lek* as a suffix expressing the first person singular features of the subject and the second person feature of the object, just like other object agreement suffixes express the features of the subject and the *third* person feature of the object. Below, I provide an analysis of object agreement that accounts for object agreement with third person pronouns in the same way as for agreement with second person pronouns, as well as the gaps in agreement. This allows for an economical analysis of object agreement, which is favourable to an analysis that stipulates that the *-lak/-lek* suffix lies outside the general agreement paradigm (Coppock 2013; Rocquet 2013).

The fact that this suffix appears exactly when the subject is first person also provides the main clue for the analysis I propose in this chapter: when looking into the distribution of object agreement with personal pronouns in more detail, it becomes clear that object agreement (including the *-lak/-lek* suffix) appears when the person of the subject and the person of the object are in a certain relation to each other.

<sup>&</sup>lt;sup>2</sup> Recall that the vowel alternation is due to vowel harmony.

É. Kiss (2003, 2005, 2013) suggests that the distribution can be captured on a hierarchy like the one in (5) (I will discuss the role of number in Section 4.4).<sup>3</sup>

(5) 1SG > 1PL/2 > 3 (É. Kiss 2013: 8)

É. Kiss (2013) suggests the generalisation in (6) to account for the distribution of agreement drawing on (5). She refers to this hierarchy as an "animacy" hierarchy in a broad sense of the term: what is relevant for Hungarian object agreement is simply person.

## (6) Inverse agreement constraint for Hungarian

An object agreeing with a verb must be lower in the animacy hierarchy [in (5)] than the subject agreeing with the same verb, unless both the subject and the object represent the lowest level of the animacy hierarchy [in (5)].

(É. Kiss 2013: 8)

Whenever the person of the subject is higher than the person of the object on the hierarchy in (6), we see object agreement. In addition, (6) states that if the subject and the object are both on the lowest level, e.g. third person, we also find object agreement. On the other hand, when the person of the direct object is higher than the person of the subject, the verb shows subject agreement only.

I build on É. Kiss's insights, but I derive the distribution of agreement with personal pronouns in an Agree-based system that implements the effects of the scale in (5) in syntax (based on Béjar and Rezac 2009). This proposal is fully compatible with the approach to object agreement and person features that I proposed in the previous chapter. Before turning to the concrete analysis in Section 4.3, I discuss the distribution of object agreement with personal pronouns in more detail.

(i) 1SG > 1PL > 2SG > 2PL > 3SG > 3PL

(É. Kiss 2013: 7)

<sup>&</sup>lt;sup>3</sup> É. Kiss (2013) suggests that (5) is a specific form a more general animacy (or person) hierarchy shown in (i). She argues that Hungarian collapses certain levels to give rise to (5). I discuss her approach and potential shortcomings in Section 4.5 below.

## 4.2.1 The distribution of object agreement with personal pronouns

The following examples illustrate the distribution of object agreement with personal pronouns. First, we see examples in which the person of the subject and the person of the object are not identical. As mentioned in Chapter 3, reflexives behave like third person pronouns and they are therefore distinct from first or second person subjects. I will return to further combinations of the same person in Sections 4.4 and 4.5.

The examples in (7)-(9) show third person pronoun objects in singular and plural varying the person of the subject. These data show that personal pronouns in the third person trigger object agreement. Note that whenever object agreement is grammatical, no other verb form is allowed — object agreement is not optional.

## (7) $3sg \rightarrow 3sg/pl: obj$

a. *Lát-ja ő-t / ők-et.* see-3sg.obj s/he-Acc they-Acc 'S/he sees him/her / them.'

 $3SG \rightarrow 3SG/PL$ : \*SBJ

b. \**Lát ő-t / ők-et.* see.3sg.sbj s/he-ACC they-ACC intended: 'S/he sees him/her / them.'

```
(8) 2SG \rightarrow 3SG/PL: OBJ
```

a. *Lát-od ő-t / ők-et.* see-2sg.obj s/he-Acc they-Acc 'You (sg.) see him/her / them.'

 $2SG \rightarrow 3SG/PL$ : \*SBJ

b. \**Lát-sz ő-t / ők-et.* see-2sG.SBJ s/he-ACC they-ACC intended: 'You (sg.) see him/her / them.'

(9)  $1SG \rightarrow 3SG/PL: OBJ$ 

a. *Lát-om ő-t / ők-et.* see-1sg.obj s/he-Acc they-Acc 'I see him/her / them.'

```
1sG → 3sG/PL: *sBJ
b. *Lát-ok ő-t / ők-et.
see-1sG.SBJ s/he-ACC they-ACC
intended: 'I see him/her / them.'
```

(7)–(9) also illustrate that the number of the direct object does not influence object agreement. The verb does agree with the subject in person and number, however, as shown in the following examples.

- (10)  $3PL \rightarrow 3SG/PL: OBJ$ 
  - a. *Lát-ják ő-t / ők-et.* see-3PL.OBJ s/he-ACC they-ACC 'They<sub>i</sub> see him/her / them<sub>j</sub>.'

 $3PL \rightarrow 3SG/PL$ : \*SBJ

b. \**Lát-nak ő-t / ők-et.* see-3PL.SBJ s/he-ACC they-ACC intended: 'They<sub>i</sub> see him/her / them<sub>i</sub>.'

(11)  $2PL \rightarrow 3SG/PL: OBJ$ 

a. *Lát-játok ő-t / ők-et.* see-2PL.OBJ s/he-ACC they-ACC 'You (pl.) see him/her / them.'

 $2PL \rightarrow 3SG/PL$ : \*SBJ

b. \**Lát-tok ő-t / ők-et.* see-2pl.sBJ s/he-ACC they-ACC intended: 'You (pl.) see him/her / them.'

(12)  $1PL \rightarrow 3SG/PL: OBJ$ 

a. *Lát-juk ő-t / ők-et.* see-1pl.obj s/he-Acc they-Acc 'We see him/her / them.'

 $1PL \rightarrow 3SG/PL: *SBJ$ 

b. \**Lát-unk ő-t / ők-et.* see-1PL.SBJ s/he-ACC they-ACC intended: 'We see him/her / them.' When the direct object is first person, the facts are equally clear: object agreement is ungrammatical and the verb only agrees with the subject. Again, the number of the object does not influence the choice of agreement. (13)–(14) show this.

```
(13) 3sg \rightarrow 1sg/pl: sbj
```

a. *Lát* engem / minket. see.3sg.sbj I.Acc we.Acc 'S/he sees me / us.'

 $3SG \rightarrow 1SG/PL$ : \*OBJ

b. \**Lát-ja engem / minket.* see-3sg.OBJ I.ACC we.ACC intended: 'S/he sees me / us.'

(14)  $2sG \rightarrow 1sG/PL: sBJ$ 

a. *Lát-sz* engem / minket. see-2sg.sbj I.Acc we.Acc 'You (sg.) see me / us.'

 $2SG \rightarrow 1SG/PL$ : \*OBJ

b. \**Lát-od engem / minket.* see-2sg.obj I.Acc we.Acc intended: 'You (sg.) see me / us.'

(7)-(12) show that third person pronouns require object agreement while (13)-(14) show that with first person objects, the verb only agrees with the subject. Considering analogous structures with second person objects, the results are less clear.

In the examples above, when the object's person is kept constant, the person of the subject only influences the grammatical form of the verb with respect to its number and person agreement. It crucially does not influence whether object agreement is grammatical or not. Consider now the examples in (15)–(16). While in (15) only subject agreement is grammatical, in (16) neither subject agreement nor the forms I have identified as "object agreement" so far are grammatical.

- 4 Agreement with personal pronouns
- (15)  $3SG \rightarrow 2SG/PL: SBJ$ 
  - a. Lát téged / titeket. see.3sg.sbj you.sg.acc you.pl.acc 'S/he sees you (sg./pl.).'

 $3SG \rightarrow 2SG/PL: *OBJ$ 

- b. \**Lát-ja téged / titeket.* see-3sg.obj you.sg.Acc you.pL.Acc int.: 'S/he sees you (sg./pl.)'
- (16)  $1SG \rightarrow 2SG/PL$ : \*SBJ
  - a. \**Lát-ok téged / titeket.* see-1sg.sbJ you.sg.Acc you.pl.Acc int.: 'I see you (sg./pl.).'

 $1SG \rightarrow 2SG/PL: *OBJ$ 

b. \**Lát-om téged / titeket.* see-1sg.obj you.sg.acc you.pl.acc int.: 'I see you (sg./pl.).'

 $1SG \rightarrow 2SG/PL: ?$ 

c. Lát-lak téged / titeket. see-1sg>2 you.sg.acc you.pl.acc 'I see you (sg./pl.).'

What is the status of the suffix *-lak/-lek*, then? It is clear that it references properties of both the subject and the object, as it only appears with first person singular subjects and second person objects. As (16a) and (16b) show, no other form is grammatical with such arguments. The suffix *-lak/-lek* also behaves like other object agreement suffixes in that it is not sensitive to the number of the object, as (16c) shows.<sup>4</sup> This is also the case for the suffixes in (7)–(9) where verbal agreement is not sensitive to the number of the object. I conclude that *-lak/-lek* 

<sup>&</sup>lt;sup>4</sup> What makes this suffix seem somewhat exceptional is arguably the lack of a corresponding object agreement suffix when the subject is first person plural and the object is second person (cf. (5)). I return to this in Section 4.4.

EA→IA	1	2	3
1	_	<i>Lát-lak téged.</i> see-1sG>2 you.sG.ACC 'I see you (sg.).'	<i>Lát-om ő-t.</i> see-1sg.овј s/he-Acc 'I see him.'
2	<i>Lát-sz engem.</i> see-2sg.sbj I.acc 'You see me.'	_	<i>Lát-od ő-t.</i> see-2sg.овј s/he-Acc 'You see him.'
3	<i>Lát engem.</i> see.3sg.sbJ I.Acc 'S/he sees me.'	<i>Lát téged.</i> see.3sg.sbJ you.sg.Acc 'S/he sees you (sg.).'	<i>Lát-ja ő-t.</i> see-3sg.obj s/he-ACC 'S/he sees him/her.'

indicates object agreement (as well as subject agreement).<sup>5</sup> The distribution of object agreement with personal pronouns is summarised in Table 4.1.

Table 4.1Transitive singular agreement paradigm with personal pronouns in Hungarian:<br/>shaded cells show inverse contexts with subject agreement only

As the data above show, the distribution of object agreement is not influenced by the number of the subject with third person objects. There is one outlier, however: when the subject is first person plural and the object is second person, the verb shows subject agreement only, as in (17).

- (17) a. Lát-unk téged / titeket. see-1PL.SBJ you.SG.ACC you.PL.ACC 'We see you (sg./pl.).'
  - b. \**Lát-juk téged / titeket.* see-1PL.OBJ you.sG.ACC you.PL.ACC intended: 'We see you (sg./pl.).'

<sup>&</sup>lt;sup>5</sup> A note about the morphological structure of this suffix here is in order here. The terms "subjective" and "objective" paradigm are often used to refer only to the sets of suffixes that appear with third person objects (see den Dikken 2006: 11, Coppock and Wechsler 2012: 702) and some authors also suggest that *-lak/-lek* can be segmented into the second person singular agreement suffix *-l* and the first person subject agreement suffix *-k* (Bartos 1999; den Dikken 2006; É. Kiss 2013).

This gives rise to the question of whether *-lak/-lek* belongs to the subjective or the objective paradigm, morphologically. As should be clear from the text, I think that the distribution of the suffix clearly shows that it is an object agreement suffix and that this is independent of which "paradigm" it belongs to (and more consequential for present purposes). I discuss the vocabulary items and spell-out rules that give rise to object agreement in Section 4.4.

This empirical fact leads É. Kiss (2013) to suggest that 1PL and 2 are on the same level on her hierarchy in (5), since in this configuration her constraint does not predict object agreement.

I return to this pattern in Section 4.4. After this overview of the relevant data, in the following section I introduce an analysis that derives the distribution of object agreement with personal pronouns in Hungarian.

## 4.3 Deriving inverse agreement in Hungarian

In Chapter 3, I argued that it is possible to analyse Hungarian as making four instead of three distinctions of person. In particular, I argued that "third" person noun phrases can be argued to consist of two classes of noun phrases: those with a person feature and those lacking one. I argued that this inventory of person features makes it possible to capture the asymmetry between subject and object agreement: verbs agree with all subject noun phrases in person and number, whereas they only agree with a proper subset of direct objects. The latter class is the one that has a set of person features. Here, I will describe the role of syntax in determining the distribution of object agreement. I adopt aspects of Béjar and Rezac's (2009) approach to Agree, namely that Agree is cyclic, i.e. a single probe can agree more than once if it has unvalued sets of features left after a previous cycle of Agree.

The basic algorithm for determining whether a certain configuration of subject and direct object shows object agreement or not is simple. I am assuming that the inventory of person (or  $\pi$ -) features consists of the sets [1], [2], and [3]. I will indicate by [] that a nominal lacks person features. Recall from Chapters 1 and 3 that [1] etc. are a shorthand for the following sets as shown in (18) (repeated from (5), p. 7):

In addition, there are entailment relations between the person features such that [1] entails the presence [2] and [3] (Harley and Ritter 2002; Béjar and Rezac 2009; Lochbihler 2012). I model these as subset/superset-relations among the respective sets. A simple person hierarchy therefore shows the following entailment relations (repeated from (6), p. 8):

(19) 
$$[1] = \begin{cases} \text{SPEAKER,} \\ \text{PARTICIPANT,} \\ \pi \end{cases} \supset [2] = \begin{cases} \text{PARTICIPANT,} \\ \pi \end{cases} \supset [3] = \{\pi\} \end{cases}$$

*v* in Hungarian has unvalued features corresponding to the sets [1], [2], and [3]: *v*[u1, u2, u3]. Each of these unvalued sets can be valued by the corresponding set [1], [2], or [3], respectively, or be valued by entailment if an argument provides a superset of features. In the latter case, if an argument provides [2], the probe's unvalued set [3] will be entailed. By the same logic, when *v* agrees with a direct object that has a set of person features [1], this will also value the sets of features [2] and [3] on the probe, since they are proper subsets of [1]. These sets of features cover all personal pronouns, and those third person noun phrases that were argued to trigger object agreement in Chapters 2 and 3. On this analysis, a first person noun phrase has the set [1] of person features, and thus {PARTICIPANT,  $\pi$ }. A second person noun phrase has the set [2], and thus {PARTICIPANT,  $\pi$ }, etc. I will mostly use the shorthand [1] and [2] for ease of exposition. Noun phrases that lack a D layer and person features do not value any person features on *v*, as they are not visible for *v*.

Cyclic Agree and sets of person features on v interact as follows: v probes and agrees with the object if the object has a set of person features. After an Agree relation with the object, v can probe again if it has unvalued sets of features left. The remaining sets of features, however, can only be valued if the subject's features are a proper superset of the object's. Therefore v will have two sets of person features only if the subject's person features are a proper superset of the object's person features. Such configurations give rise to object agreement.

(20) shows a first example of this system. The scenario involves a first person singular subject and a second person object: the subject has the set of features [1] and the object has [2]. As just shown, the subject's set of person features is therefore a proper superset of the object's person features. The probe, v, agrees with the direct object first. In this case, v is valued as [2]. I follow Béjar and Rezac (2009) in assuming that a probe with unvalued sets of features can probe again. In Hungarian, v moves to T, and its unvalued sets of features probe again. In the given scenario, v will enter an

Agree relation with the subject and receive a value of [1], *in addition* to its previous value of [2]. v is therefore valued as [1, 2]. This is shown in (20) where — indicates valuation of a feature set on the probe by a feature set on an argument and the grey feature indicates entailment.<sup>6</sup>

(20)	$1 \rightarrow 2: 0$	1 → 2: obj		
	SUBJ	v	DO	
	[1]-	-[1]		
		[2]-	-[2]	
		[3]		

Why is the result of [1, 2] on  $\nu$  valid? As just said, a probe can try to enter Agree relations as long as it has unvalued sets of features. The first cycle of Agree, with a second person object, does not fully value the probe, so it can attempt to enter another Agree relation. This relation will only succeed, however, if the probe finds an argument that has a *stronger* set of person features, i.e. a set of person features that is a proper superset of the sets of features present on the probe. In (20), this is clearly the case, since {SPEAKER, PARTICIPANT,  $\pi$ }  $\supset$  {PARTICIPANT,  $\pi$ }. I will, from now on, write that [1] entails [2] etc. to express that the set [1] is a proper superset of [2]. In the same sense, [1] is stronger than [2] which in turn is stronger than [3]. (Lochbihler 2012: 44ff. argues for a similar approach to valuation and entailment).

*v* can therefore only have the values [1, 2] if the stronger set of features, [1] in this case, comes from the *second* argument the verb agrees with, i.e. the subject. If the probe encounters a first person argument on its first cycle of Agree, it will be fully valued and it will not be able to probe again, i.e. *v*: [1]. The reasoning extends straightforwardly to all other configurations of person. This allows for a first (tentative) characterisation of object agreement in these syntactic terms.

<sup>&</sup>lt;sup>6</sup> Note that on the current approach v agrees with two arguments even though it only carries a single  $\varphi$ -probe, rather than two probes, one each for the subject and the object.

Alternatively, it might also be possible to assume that v comes with separate probes for the person features of the subject and the object, respectively (see Keine 2010; Georgi 2012 for such an approach). Making use of the entailment relations between features arguably provides a simpler mechanism for deriving the Hungarian pattern, and other languages. I will return to this in Chapters 5 and 6 where I discuss the distribution of  $\varphi$ -probes on functional heads across languages.

#### (21) Object agreement in Hungarian and *v*

When v is valued twice, the verb shows object agreement.

Such double valuation is ruled out when the direct object's set of person features is a superset of the person features of the subject. v will not able to agree on a second cycle with any argument that has a weaker set of features than the argument on the first cycle. The values of features on v therefore reflect the relative "position" of the subject and the object on a person hierarchy like 1 > 2 > 3. When v has two values, the subject is higher. This approach to Agree and valuation of feature sets therefore builds in hierarchical effects into syntax proper.

At this point I diverge from Béjar and Rezac's (2009) proposal substantially. Béjar and Rezac (2009) argue that if v fails to enter a second Agree relation with the subject, the subject is not licensed (which would lead to a crash of the derivation). They suggest that in such cases, some languages utilise a *repair strategy*, by which a new probe can be activated on v which can agree with (and license) the subject (cf. also the similar notion of "proxy category" in Nash and Rouveret 2002). This probe is then spelled out as an inverse marker in some languages, for example (see Béjar and Rezac 2009 for a number of examples).

If v does not enter an Agree relation with the subject, T's set of person features probes and finds the external argument. This argument values T's features and T can assign Case to it. This approach arguably is simpler than Béjar and Rezac's (2009) by avoiding the need of adding features or probes at a given stage of the derivation.

This is the kind of derivation in which the verb only shows subject agreement with personal pronoun objects in Hungarian. A first person direct object will always fully value v and v will not be able to agree with the external argument again, therefore violating the generalisation in (21) (which will be formalised below). Similarly, when the direct object is second person, a third person subject will not be able to value v in a second cycle. If, however, the subject is first person and the object is second or third person, v will receive two sets of person features.

The following structures illustrate this process visually. First, (22) repeats (20) and shows again how two arguments can value v. Valuation is indicated by "—" linking a feature set on an argument and a feature set on the probe.



(23) illustrates this process as part of a derivation. As above, (23) only shows the result of an Agree relation, i.e. v gets the [2] feature after agreeing with the direct object. In (23a), v agrees with the direct object, gets a [2] feature, and assigns accusative to the object in step (A). Next, v moves to T and probes again, since it is not fully valued. It finds the subject and gets an additional set of person features, [1] in step (C).



The derivation continues with T probing as well in step (D). A final step that I have not discussed yet is the fusion of v and T. As briefly mentioned in Chapter 1, I assume that there is a post-syntactic operation that can *fuse* two sister nodes into one. Note that I do not wish to claim that fusion takes place in the syntax, it is only included in (23b) for expository purposes.

## 4.3 Deriving inverse agreement in Hungarian



(24) shows the fusion operation indicated in (23b). When v and T are sisters and their strongest sets of features match ([1] in (23)), their feature structures fuse into one. No information is lost (or added), since no features overwrite each other (I ignore CASE in (24), as it is no longer of importance at this stage of the derivation).

## (24) Fusion of v and T



I assume that fusion takes place because of the morphological structure of Hungarian verb forms. I will argue in Section 4.4 that there is only one agreement suffix in Hungarian. When the verb shows both subject and object agreement, this is spelled-out by portmanteau morphemes rather than separate agreement suffixes. However, as stated above, I assume fusion only to take place when v and T match in their strongest feature set. This restriction accounts for the distribution of agreement with two third person arguments, discussed in Section 4.3.2.

The derivation just shown involves a first person subject and a second person object and results in v+T having two sets of person features: [1, 2]. I have suggested above

that this is a kind of derivation in which v+T will be spelled out as object agreement. In the particular case of (23), this will be the suffix *-lak/-lek*.

Consider now a clause in which the subject is third person and the object is second person. In this scenario, the person features of the direct object are a proper superset of the person features of the subject. When v enters an Agree relation with the direct object, the object will provide the set [2]. v will be able to probe again, as in (23), but the subject's features will not be able to value it. T will probe and enter an Agree relation with the subject by virtue of T's number probe. The strongest set of person features on v and T will not match and therefore they will not fuse. Since there is a single agreement slot, only one of the heads will be spelled out. I assume that T is spelled out in such cases since it has a full set of  $\varphi$ -features, whereas v only has person features.

In this case, then, no probe will be valued by two arguments. This differs from the previous scenario, in which v was valued by the external *and* the internal argument. A sample derivation is illustrated in (25) and (26).

(25)	3 → 2: SBJ		
	SUBJ	Τν	DO
		[2]-	-[2]
	[3]—	<b>-[3]</b> [3]	

In (25), v cannot enter a second Agree relation with the subject since the subject's [3] feature set is a proper subset of the direct object's feature set. The probes v and T therefore agree with one argument each. Since their strongest feature sets do not match, the two heads will not be able to undergo fusion. The following structures illustrate this derivation. In (26a), v agrees with the direct object and receives a [2] value. v moves to T and attempts to probe again, since it has unvalued sets of features left. v does not find a goal, however, since the subject's [3] set is not strong enough, and therefore this instance of Agree fails in step (c).

## 4.3 Deriving inverse agreement in Hungarian



(27b) shows how the derivation continues. T probes and finds the subject, and receives the values [3, sG]. The subject is assigned NOM. Note again that no fusion takes place since the strongest feature sets on v and T do not match.



(23) and (26) illustrate the two syntactic possibilities that this system allows. During the syntactic derivation, v can be valued by a single argument or by two arguments. This distinction reflects the effects of a person hierarchy. When the subject's set of person features is a superset of the object's set of person features, v can be valued by both the subject and the object, but not otherwise. This corresponds to *direct* configurations, in which the subject is higher on a hierarchy than the object.

The other case, exemplified by (26), illustrates an *inverse* configuration. Here, the object's set of person features is a superset of the subject's person features and therefore v is only valued by the object. The subset/superset relations among person features therefore again model the hierarchical effects. This insight is due to Béjar and Rezac (2009), although my particular implementation diverges from theirs.

One of the differences lies in the role of fusion. Recall that I have argued that v and T fuse if their strongest sets of person features match. This will give rise to a complex head v+T in direct configurations; but fusion does not take place in inverse configurations. The restriction on fusion allows for one more configuration to count as direct, namely when both the subject and the object are third person. In such cases, we see object agreement, even though v is not valued by the two arguments in syntax. Fusion, however, can take place, since both v and T share the value [3]. Syntactically, a configuration involving two third person arguments should therefore pattern with inverse configurations, but the post-syntactic process of fusion makes it appear like a direct configuration. This is not just a stipulation: we will see in Chapter 5 that configurations in which both arguments involve [3] generally count as inverse across languages - the fact that they do not in Hungarian follows from an independent morphological reason, namely that heads undergo fusion when their strongest person features match. If a language does not trigger fusion at this point, a configuration involving two third person arguments counts as an inverse. I illustrate derivations of clauses involving two third person arguments in Section 4.3.2.

Another important property of the derivations just shown is that all personal pronouns trigger object agreement, i.e. they all value v. Since all persons trigger object agreement, no stipulation is necessary about how the suffix *-lak/-lek* arises. It is simply a suffix expressing subject and object agreement. I have also proposed a reason why object agreement is not always visible: it only surfaces when the two heads v and T share their strongest set of person features and undergo fusion. This provides a straightforward account of why second person objects sometimes show agreement and sometimes do not. I now turn to the discussion of independent evidence for this kind of "covert" agreement.

#### 4.3.1 Arguments for covert agreement

I have argued that first and second person pronouns in Hungarian trigger agreement even if this agreement is not reflected in verb morphology. Why should this be the case? First, it is cross-linguistically unusual for languages with DOM to skip first and second person for differential case-marking or agreement (Silverstein 1976; Comrie 1989; Aissen 2003; Haspelmath 2008, 2009; see Filimonova 2005 for some exceptions). Second, as we will see in Chapter 5, case-marking or agreement patterns that determine the presence or absence of agreement based on the properties of *both* the subject and the object are common. Such splits in case-marking and agreement are called *global* (Silverstein 1976; Malchukov 2008; Keine 2010; Georgi 2012) as opposed to *local*. In global splits, the properties of several arguments have to be taken into account rather than just the properties of a single one.

One way of deriving the Hungarian pattern of agreement is to assume that first and second person pronouns always trigger agreement, like third person pronouns, and that there is an independent reason that blocks the spell-out of this agreement. An advantage of such an approach is that pronouns form a natural class syntactically and semantically, i.e. they can all be analysed as definite and referential DPs (*contra* Bartos 1999 for example; recall also that Coppock 2013 suggests that differences in indexicality determine the agreement pattern with pronouns). I have argued in Chapter 3 that personal pronouns do in fact behave like a natural class with respect to syntax (apart from agreement) and that differences in indexicality cannot account for the distribution of agreement: something else is needed to derive the distribution of agreement.

Here, I discuss evidence that further supports treating pronouns as a natural class syntactically and thus with respect to their potential to trigger agreement. If true, then this strengthens my proposal that all personal pronouns in Hungarian trigger agreement in syntax and that the absence of agreement in morphology is a surfaceeffect.

I have already proposed a first argument for why at least second person pronouns should trigger agreement, namely the distribution of the suffix *-lak/-lek*. This suffix is similar to other suffixes expressing object agreement in that the number of the object does not matter, and it obligatorily appears in a certain syntactic configuration. If some

second person objects can agree, it is plausible to assume that second person pronouns do not rule out agreement *per se*.

A second argument for the existence of covert agreement comes from the distribution of dropped objects.<sup>7</sup> Verb forms that show object agreement allow dropping of the direct object. Object agreement morphology is explicit about the person of the dropped object and its reference is straightforwardly recovered (cf. *pro*-drop in languages with so-called "rich inflection"; see Rizzi 1982, 1986; Müller 2005; Roberts 2010; Roberts and Holmberg 2010; Holmberg and Roberts 2013). This is illustrated in (27).

(27) a. Lát-om. see-1sG.OBJ 'I see it/him/her.'
b. Lát-lak. see-1sG>2 'I see you.'

Object drop is also possible with exactly those verb forms that appear to be intransitive but that I have argued to agree with a pronominal object in the course of the derivation. Obviously, there is no object agreement morphology in these cases, so the forms in (28) are ambiguous with respect to the person of the dropped object:

(28) a. Lát. see.3sG.SBJ 'S/he sees.' or 'S/he sees me/you.'
b. Lát-sz. see-2sG.SBJ

'You see.' or 'You see me.'

Note that while (28a) can either have a dropped first or second person object when understood as transitive, (28b) can only be understood to have a dropped first person object on its transitive reading. The forms in (28) are therefore identically restricted in their interpretation to the forms in (27), in that they can only have dropped objects of a certain person.

This is a synchronic argument. É. Kiss (2012) discusses the development of object drop in Hungarian and points out that the connection between agreement and object drop has only arisen in the modern stages of Hungarian.

The argument for covert agreement is the following. Overt objects can be modified by depictive secondary predicates like *drunk*. Covert objects can be modified by such predicates too, but only when the verb agrees with them. Consider the following examples. In (29a), the indefinite object *someone* can be modified by the secondary predicate *részegen* 'drunk'. This is not possible in (29b). In (29c) and (29d), however, with an agreeing form, the dropped object can be modified by the secondary predicate.<sup>8</sup>

- (29) a. Lát-ok valaki-t részeg-en. see-1sg.sbj someone-ACC drunk-ADV ' $I_i$  see someone<sub>i</sub> drunk<sub>i/i</sub>.
  - b. Lát-ok részeg-en. see-1sg.sbj drunk-ADV 'I<sub>i</sub> see drunk $_{i/*j}$ .'
  - c. Lát-om részeg-en. see-1sg.obj drunk-ADV 'I<sub>i</sub> see her<sub>j</sub> drunk<sub>i/j</sub>.'
  - d. Lát-lak részeg-en. see-1sG>2 drunk-ADV 'I<sub>i</sub> see you<sub>i</sub> drunk<sub>i/i</sub>.'

Verb forms with first person subjects are a good baseline for determining the correlation between agreement, object drop, and modification by secondary predicates because they are never ambiguous with respect to the person of the object (as opposed to the examples in (28)).

Given this set of data, we can thus formulate the following hypothesis, to be tested below:

## (30) Object agreement, object drop, and secondary predication

Object agreement allows for object drop and modification of the dropped object by a secondary predicate.

The hypothesis in (30) makes a strong prediction: we should find object drop and secondary predicates controlled by the object with verb forms that do not show object agreement, but only if they have ambiguous derivations. Examples for such ambigu-

<sup>&</sup>lt;sup>8</sup> See Roberts (1988), Schultze-Berndt and Himmelmann (2004) and Irimia (2005) for some general aspects of the syntax and semantics of secondary predicates.

ous derivations are shown in (28) where the form of the verb does not indicate the possible types of objects.

This means that if we modify the sentences in (28) by adding a secondary predicate, that predicate should only be able to be controlled by objects that are compatible with that verb form. This is true.

In (31a), a third person pronoun object would trigger object agreement, and therefore the form *lát* 'see.3sG.SBJ' is ambiguous between an intransitive reading, a transitive reading with a first person object, and a transitive reading with a second person object. The secondary predicate can only refer to either the subject or a (covert) first or second person object. In (31b), the form *lát-sz* 'see-2sG.SBJ' is ambiguous between an intransitive reading and a transitive reading with a first person object. The secondary predicate can be controlled by the subject or by the (covert) first person object.

(31)	a.	Lát	részeg-en.
		see.3sg.si	bJ drunk-ADV
		'S/he <sub>i</sub> see	s drunk <sub>i</sub> .' or 'S/he <sub>i</sub> sees me <sub>j</sub> /you <sub>k</sub> drunk <sub>i/j/k</sub> .
	b.	Lát-sz	részeg-en.
		see-2sg.s	вј drunk-ADV
		'You <sub>i</sub> see	drunk <sub>i</sub> .' or 'You <sub>i</sub> see me <sub>i</sub> drunk <sub>i/i</sub> .'

I believe that these data provide evidence in favour of the hypothesis in (30). The approach to agreement suggested in the previous section exactly predicts which types of objects can be dropped and are able to control a secondary predicate. That these objects form a natural class with respect to these properties is shown particularly clearly by examples with second person pronouns like (29d) and (31a), repeated here. While object agreement is only overt in (29d), both dropped second person objects behave alike with respect to secondary predication.

- (29d) Lát-lak részeg-en. see-1sG>2 drunk-ADV 'I<sub>i</sub> see you<sub>i</sub> drunk<sub>i/i</sub>.'
- (31a) Lát részeg-en. see.3sg.sbj drunk-ADV 'S/he<sub>i</sub> sees drunk<sub>i</sub>.' or 'S/he<sub>i</sub> sees me<sub>j</sub>/you<sub>k</sub> drunk<sub>i/j/k</sub>.'

Again, this contrasts with the behaviour of indefinite objects which can only control secondary predication when they are overt. With respect to dropping indefinite objects, there is one seeming exception to this pattern. In answers, even *lát-ok* 'see-1sg.sBJ' can be understood to have an elided object:

(32) Lát-sz valaki-t? – Lát-ok.
see-2sg.sBJ someone-Acc see-1sg.sBJ
'Do you see someone?' – lit. 'I see.' meaning 'I do (see someone).'

But there are reasons to believe that the elided object in (32) differs from actual dropped objects. First, such examples need a very specific discourse context like the question-answer pair in (32). Second, such elided objects cannot control secondary predication, unlike dropped objects that trigger agreement. This is shown in (33).

(33) Lát-sz valaki-t? - #Lát-ok részeg-en. see-2sg.sbj someone-Acc see-1sg.sbj 'Do you see someone?' - intended: 'I<sub>i</sub> do see someone<sub>i</sub> drunk<sub>i/\*j</sub>.'

I conclude that there is independent syntactic evidence that the verb agrees with all personal pronoun direct objects in the course of the derivation, even if agreement is not always visible on the surface. When the object enters an Agree relation with the verb, it can be dropped and dropped objects can control depictive secondary predicates.

## 4.3.2 Third person arguments

The analysis of agreement in Hungarian introduced above works for the configurations  $1\rightarrow 2$ ,  $1\rightarrow 3$ ,  $2\rightarrow 3$  "out of the box" but I have not yet discussed examples in which both the subject and the object are third person in detail. The analytical challenge for such scenarios is that *v* cannot be valued by two arguments that have the same feature sets of [3] and therefore such configurations do not seem to conform to the generalisation that object agreement appears when *v* is valued by two arguments.

In this section, I argue that because v and T both have a [3] feature set in such derivations, fusion can apply after the syntactic derivation and the resulting terminal v+T will be specified as [3, 3]. As I briefly mentioned above, this analysis explains why [3, 3] patterns with direct configurations in Hungarian rather than with inverse configurations. We will see in Chapter 5 that this is not always the case cross-linguistically - I

suggest that the difference lies in the application of fusion that is specific to Hungarian (recall also the caveat referring to third person arguments in É. Kiss's 2013 hierarchy in (6), p. 100).

(34) illustrates a relevant example with a (dropped) third person subject and third person personal pronoun direct object.

(34) Lát-ja ő-t. see-3sg.obj s/he-ACC 'S/he sees him/her.'

In the derivation of (34), both v and T agree with a single argument bearing [3]. The structures in (36) illustrate the derivations.

(35) 
$$3 \rightarrow 3: \text{ OBJ}$$
  

$$\frac{\text{SUBJ T } \nu \text{ DO}}{[3]-[3][3]-[3]}$$

In (36a), v agrees with the direct object before moving to T. The second attempt at agreeing fails since the subject's [3] feature cannot value v again.



The derivation continues with T agreeing with the subject and receiving a [3] value as well in step (D). Since the strongest sets of person features of *v* and T now match,

the two heads can undergo fusion. Note again that I do not suggest that fusion is a syntactic operation; I merely include it in the derivation for the sake of exposition. Fusion is shown in (37).



## 4.3.3 Interim summary

In this section, I have argued that all personal pronouns in Hungarian trigger object agreement and I have implemented an analysis that can account for this. I provided independent evidence that agreement takes place even when it is not spelled out by showing that all arguments that I hypothesise to trigger agreement behave as a natural class with respect to object drop and licensing secondary predication.

The difference between whether agreement is spelled out or not lies in how v is valued by the arguments in a transitive clause. If v is valued by both arguments, object agreement arises. In direct configurations, v is straightforwardly valued by both the subject and the object since the subject's person features are a proper superset of the direct object's person features. In inverse configurations, v is valued only by the direct

object's features, since they are a superset of the subject's person features. However, when both arguments are third person, the feature sets of v and T can still be unified by post-syntactic fusion, giving rise to a result that is analogous to direct configurations.

By adopting parts of Béjar and Rezac's (2009) approach, the current analysis incorporates the effects of a person hierarchy into the syntax and provides more evidence for the treatment of the trigger of object agreement in Hungarian as a person feature, as argued in Chapter 3.

I discuss the morphological structure and the spell-out of Hungarian verb forms in the following section.

# 4.4 Spelling out agreement

So far, the current approach succeeds in predicting in which configurations object agreement appears with pronominal direct objects, but I have left questions about the spell-out of agreement with these pronouns open. On the analysis presented above, object agreement is spelled out when v and T fuse to form a complex head. These terminals then carry properties of both the subject (its person and number) and the object (its person only). This information is spelled out in verb forms that show subject and object agreement.

I have further argued that all personal pronoun objects agree with v but that v is not always part of what is spelled out. When v and T agree with only the internal and the external argument, respectively, they do not fuse to become a complex head and only T is spelled out (as it has a full set of  $\varphi$ -features).

In this section, I discuss the particular vocabulary items, impoverishment and spellout rules that determine the form of the inflectional suffixes on the Hungarian verb. In particular, I will argue that Hungarian only spells out a single syntactic terminal on the verb. Throughout this section, I discuss evidence that supports the view that subject and object agreement in Hungarian are expressed as portmanteau morphemes.

I adopt Distributed Morphology (DM; Halle and Marantz 1993, 1994; Harley and Noyer 2003; Embick and Noyer 2007) to model verbal and nominal morphology. After the syntactic derivation, illustrated in the previous section, Vocabulary Insertion (VI) takes place and matches features represented on terminals in the syntactic structure with lexical items (cf. also Chapter 1 and further discussion in Chapter 5). The relevant vocabulary items under discussion in this section are subject and object agreement suffixes.

## 4.4.1 Hungarian verb morphology and spell-out rules

I adopt a view of Hungarian verb morphology that mostly corresponds to the one in Törkenczy (2002) (*contra* some more elaborate approaches to be discussed below).

(38) stem + TENSE/MOOD +  $\varphi$  (Törkenczy 2002: 68)

I will argue that the template in (38) suffices to spell-out the verb forms under discussion so far. In Section 4.5 below, I relate this structure to other proposals in the literature. I will mostly set aside tense and mood morphology as it is not directly relevant to the question of agreement morphology ( $\varphi$  in (38)).

As argued above, syntax derives two kinds of configurations in transitive clauses with personal pronouns. One possibility is that v and T agree with the internal and the external argument, respectively. In the other case, v agrees with both the internal and the external argument (in this order), and T agrees with the external argument as well. I suggested that following the syntactic derivation the two heads can fuse, but only if their strongest sets of person features match. If that is the case, v and T fuse to form a complex head v+T which carries the features of both arguments (see the derivations in (24) and (37) above).

In both scenarios, then, there is a single head that has a full set of  $\varphi$ -features, namely person and number (recall that gender is not grammaticalised in Hungarian). I assume that this head, T or *v*+T if fusion has taken place, is targeted by spell-out rules. Note that, as mentioned previously, the object never agrees with the verb in number. In direct cases, therefore, while the person of both arguments is specified on *v*+T, only the subject's number is.

Table 4.2 shows the distribution of person features before spell-out, i.e. after the syntactic derivation has finished and after fusion has taken place (cf. also Table 4.1 above). This clearly distinguishes direct configurations and configurations with two third person arguments from inverse configurations. The gaps at  $1\rightarrow 1$  and  $2\rightarrow 2$  in

EA→IA	1	2	3
1	_	v+T: [1, 2]	v+T: [1, 3]
2	v: [1], T: [2]	_	v+T: [1, 2]
3	v: [1], T: [3]	v: [2], T: [3]	v+T: [3, 3]

Table 4.2 refer to reflexive pronouns which are third person syntactically and always trigger object agreement (I return to non-reflexive cases of  $1 \rightarrow 1$  and  $2 \rightarrow 2$  below).<sup>9</sup>

 Table 4.2
 Distribution of person features in Hungarian direct and inverse contexts (shaded)

The following (preliminary) spell-out rules refer to the cells of Table 4.2 and also take into account the number of the subject. The notation used is as follows. Capital letters indicate variables over vowels in a given vowel harmony alternation: *A* is realised as either a back or a front open vowel: /v/vs.  $/\epsilon/$ . *O* allows for an additional contrast in roundedness for front vowels, leading to /ce/ and  $/\epsilon/$ ; its back variant is /o/. *U* stands in for the closed rounded back and front vowels /v/ and /y/, respectively. Double capitals indicate long vowels. I am ignoring tense/mood features unless relevant. The two variants of the second person singular subject agreeing suffix in (40) depend on the morphophonology of the verb: roots ending in non-sibilant consonants get the *-sz* suffix. I will return to the role of the *-l* suffix below.

## (39) Vocabulary items involving first person subjects (provisional)

/ <i>-lAk</i> /	$\leftrightarrow$	[1, 2, sg]
/-Om/	$\leftrightarrow$	[1, 3, sg]
/ <i>-Uk</i> /	$\leftrightarrow$	[1, 3, pl]
/-Ok/	$\leftrightarrow$	[1, sG]
/-Unk/	$\leftrightarrow$	[1, pl]

<sup>&</sup>lt;sup>9</sup> The reader might wonder why *ν* can have two sets of φ-features while T only has one in Table 4.2. In principle, T can agree several times as well and be valued repeatedly. Nothing would change with respect to the analysis proposed here. It is possible, however, that accusative on the direct object renders the direct object inactive for agreement with T after it has agreed with *ν*. See Chapters 5 and 6 for discussion.

(40) Vocabulary items involving second person subjects (provisional)

/-Od/	$\leftrightarrow$	[2, 3, sg]
/-AAtOK/	$\leftrightarrow$	[2, 3, pl]
/-Ol/-sz/	$\leftrightarrow$	[2, sg]
/-tOk/	$\leftrightarrow$	[2, pl]

(41) Vocabulary items involving third person subjects

/-a/-i/-e/	$\leftrightarrow$	[3, 3, sG]
/ <i>-AAk/-ik/</i>	$\leftrightarrow$	[3, 3, pl]
/-Ø/	$\leftrightarrow$	[3, sg]
/-nAk/	$\leftrightarrow$	[3, pl]

The rules in (39)–(41) mostly suffice to spell-out present tense verb forms. There are, however, a few syncretic forms that are not yet covered by the rules shown here that warrant further discussion.

## 4.4.1.1 Modifications

First person singular suffixes in the past tense neutralise the distinction between subject agreement and subject and object agreement. This is shown in (42).

- (42) a. *Lát-t-am egy fiú-t.* see-PST-1SG a boy-ACC 'I saw a boy.'
  - b. Lát-t-am a fiú-t. see-PST-1SG the boy-ACC 'I saw the boy.'

The fact that the "agreeing" form is resorted to in this case casts doubt on the analysis of the suffix *-Om* as shown in (39). The reason for this is that the rule in (39) is too specific for the suffix *-am* to be inserted in (42a), with an object that does not trigger object agreement (see Trommer 2005 for this argument and further discussion). In addition, a form of *-Om* appears on nominals indicating a first person singular possessor.<sup>10</sup> Again, in a context without an object, an "object agreement" suffix appears.

<sup>&</sup>lt;sup>10</sup> In the past tense and possessive morphology the back variant of -Om and -Od is realised as /v/ rather than /v/; I nevertheless treat them as the same suffixes here and I will continue to refer to the -Om and -Od.

(43) *ház-am* house-1sg.poss 'my house'

It is of course possible that there are several homophones that are all spelled out as *-Om*, but there might a be a missing generalisation looming here as there is consistent overlap between verbal morphology and possessive morphology in Hungarian (Szabolcsi 1994; Trommer 2005).

Szabolcsi characterises this overlap as follows: possessive suffixes indicating a singular possessor show an overlap with verbal morphology expressing object agreement. Possessive suffixes indicating a plural possessor overlap with verbal morphology expressing only subject agreement. Some of these parallels are shown in (44), with the spell-out rules from above.

(44)	a.	<i>lát-om</i> see-1sg.obj 'I see (it)', 'r	<i>ház-am</i> house-1sg.poss ny house'	/- <i>Om</i> / ↔ [1, 3, sG]
	b.	<i>lát-od</i> see-2sg.овј 'you (sg.) se	<i>ház-ad</i> house-2sg.poss ee (it)', 'your (sg.)	$/-Od/ \leftrightarrow [2, 3, sG]$ ) house'
	c.	<i>lát-unk</i> see-1pl.sBJ 'we see', 'ou	<i>ház-unk</i> house-1pl.poss ır house'	$/-Unk/ \leftrightarrow [1, pl]$
	d.	<i>lát-tok</i> see-2sg.овј 'you (pl.) se	<i>ház-atok</i> house-2sg.poss re', 'your (pl.) hou	$/-tOk/ \leftrightarrow [2, PL]$ use'

These forms raise two questions: first, why does an arguably non-relational noun like  $h\dot{a}z$  'house' carry the same suffix as a verb agreeing with an object. Second, why is the specification of the vocabulary items in (45) different in singular and plural? In this section, I will attempt to provide answers to both questions.<sup>11</sup>

<sup>&</sup>lt;sup>11</sup> It might be possible to argue that possessed nouns like *ház-am* 'house-1sg.poss' in (43) are relational nouns that take two arguments and therefore do behave like transitive verb forms that spell out the features of both the subject and the object; however, I do not provide an analysis of agreement inside the noun phrases and I will therefore assume that possessive suffixes are only sensitive to the person of the possessor. Not much hinges on this question.
Trommer (2005) argues that the distribution of the suffix(es) -*Om* provides evidence against analysing it as a portmanteau suffix, i.e. a suffix that reflects properties of both a first person argument and a third person argument (I will return to the question of whether these suffixes are portmanteaus below). He suggests instead that -*Om* only spells-out a first person feature, but not a third person feature.

In addition to verb forms with third person objects, and nominals, *-Om* also appears in examples like (45), showing so-called *inclusive reference*. This means that the referent of the subject is included in the referent of the object, as both have the same person. As argued by den Dikken *et al.* (2001), some speakers of Hungarian show a preference for the "agreeing" form with configurations of the type X.SG  $\rightarrow$  X.PL for X  $\in$  {1,2} (see also É. Kiss 2013). Note that (45b) is merely more acceptable than (45a), but not generally seen as fully grammatical.

- (45) a. \*Én mink-et választ-ok meg. I we-ACC choose-1sg.sbj vm intended: 'I choose us.'
  - b. %Én minket választ-om meg. I we-ACC choose-1sg.овј vм 'I choose us.'

(den Dikken et al. 2001: 140f.)

Since the distribution of the suffix -Om which expresses subject and object agreement is less restricted than the distribution of the suffix -Ok which expresses only subject agreement, Trommer suggests deriving the context for inserting the suffix -Om by feature deletion. Concretely, he suggests that the vocabulary item -Ok is only inserted in the context of a feature [+V], as shown in (46).<sup>12</sup> This feature reflects the category of the terminals into which the suffix is inserted: since -Om can also appear in a nonverbal context, however, it can also be inserted in other contexts (predicting correctly that it can appear as a nominal suffix).

This suggestion makes the subject agreement suffix -Ok more specific than the alternative -Om. In order for the suffix -Om to be inserted on a terminal such as v+T, the feature [+V] is deleted (I adopt the gist of Trommer's approach in what follows, but see his paper for a different implementation).

 $<sup>^{12}</sup>$  Trommer (2005) uses [+v], but I will refer to [+V] in order not to confuse this with the syntactic head v in the derivations above.

## (46) Vocabulary items involving first person singular subjects a. /-Ok/ ↔ [1, sG] / [+V] b. /-Om/ ↔ [1, sG]

To derive the distribution of -Om, [+V] is deleted in several contexts which feeds the vocabulary insertion rule for /-Om/ in (46). The contexts in which impoverishment takes place involve nominals as in (43), past tense forms as in (42), as well as transitive clauses with objects that require object agreement, i.e. objects that have person features. These impoverishment rules are sketched in (47).<sup>13</sup>

#### (47) Impoverishment rules for first person singular subjects

- a.  $[+V] \rightarrow \emptyset / [+N]$
- b.  $[+V] \rightarrow \emptyset / [1, +PST]$
- c.  $[+V] \rightarrow \emptyset / [1, 3, sG]$

The rule in (47a) deletes [+V] in the context of a nominal; this allows for insertion of the VI -*Om*, given (46). (47b) specifies that [+V] is deleted in the past tense, again feeding the insertion rule in (46b). Finally, when features of both the first person subject and the third person object are specified on *v* in the course of the syntactic derivation, (47c) deletes [+V] and -*Om* is inserted.

To give a more concrete example, consider the structures in (48). Here, v+T is specified as [1, 3] and carries a [+V] feature (its syntactic category). This matches the context for the rule in (47c) to apply. This leads to the deletion of [+V]. Vocabulary insertion rules for first person subjects are shown in (46), repeated from above. After impoverishment applies, the form in (46a) cannot be inserted because it is specified for [+V] contexts and *-Om* is inserted instead.

<sup>&</sup>lt;sup>13</sup> When inserted as a possessive suffix, the feature [+V] is arguably not present in the first place and does not have to be deleted: *-Om* would be the only suitable candidate. The rule in (47a) might therefore be dispensable.

#### (48) Impoverishment

$$\begin{array}{c} \nu + T & \nu + T \\ \left[ \begin{array}{c} + V & & \\ u & u \pi & 1, 3 \\ u & u \# & sG \end{array} \right] \xrightarrow{\text{Impoverishment: (47c)}} \left[ \begin{array}{c} + V & & \\ u & u \pi & 1, 3 \\ u & u \# & sG \end{array} \right]$$

Vocabulary items involving first person singular subjects
a. /-Ok/ ↔ [1, sG] / [+V]
b. /-Om/ ↔ [1, sG]

Instead of a suffix that expresses properties of both a first person subject and a third person object, a suffix that is underspecified for the person of the object and the category of its terminal is inserted: this makes it possible that the suffix *-Om* has a wider distribution than the subject agreement-only suffix *-Ok*.

There are two further contexts that require deletion rules. First, a large number of verbs in the class of *-ik*-verbs (prescriptively) neutralises the difference between subject agreement and subject and object agreement and always bears the *-Om* suffix in first person singular.<sup>14</sup>

Second, in cases of inclusive reference as shown in (45) above, -Om is reported to be more acceptable than -Ok. This is reflected in the spell-out rules below.<sup>15</sup>

#### (47) Impoverishment rules for first person singular subjects (cont.)

d.  $[+V] \rightarrow \emptyset / [1, V = eszik$  'to eat', iszik 'to drink', ...]

e.  $[+V] \rightarrow \emptyset / [T = v = (1 \mid 2)]$ 

(47d) captures the forms of -ik-verbs and (47e) captures the fact that [+V] is deleted when the subject and the object are both first or second person. In (47e), "|" indicates

<sup>&</sup>lt;sup>14</sup> In present-day Hungarian, this class of verbs includes unergatives, unaccusatives, as well as transitive verbs; see Törkenczy (2002: 116) for a list of these verbs, and see Havas (2004) for some diachronic discussion with respect to medialisation.

<sup>&</sup>lt;sup>15</sup>Recall that the notion *inclusive reference* indicates that the referent of the subject is included in the referent of the object. In the cases at hand, this is because the subject is singular and co-indexed with a plural object. See den Dikken *et al.* (2001) and Rooryck (2006).

Note that these inclusive-reference forms obviously compete with reflexive pronouns and that they do not seem to be fully grammatical.

logical disjunction: the feature [+V] is deleted when the person features on T and v match, i.e. when both are either first or second person.

Second person singular suffixes are similar to first person singular suffixes in certain respects: the suffix *-Od*, indicating object agreement, also appears as the marker of a second person singular possessor, as shown in (49).

(49) *ház-ad* house-2sg.poss 'your house'

Given (49), the rule in (40) is too specific to allow for insertion of -Od in this case. Again, I will assume that the suffix -Ol/-sz that appears with intransitive predicates as well as indefinite objects is specified as [+V] and that deletion rules remove this feature in certain contexts. The revised spell-out rules are given in (50) and the deletion rules in (51). Note the parallel to the first person suffixes discussed above.

- (50) Vocabulary items involving second person singular subjects a.  $/-Ol/-sz/ \iff [2, sG] / [+V]$ b.  $/-Od/ \iff [2, sG]$
- (51) Impoverishment rules for second person singular subjects
  - a.  $[+V] \rightarrow \emptyset / [+N]$ b.  $[+V] \rightarrow \emptyset / [2, 3, s_G]$

(51a,b) show some overlap with the corresponding rules for first person subjects; indeed, some of these rules can be combined.

#### 4.4.2 Revised spell-out rules

The previous discussion leads to the following spell-out and deletion rules for Hungarian agreement suffixes.

#### (52) Vocabulary items involving first person subjects (final)

/-Ok/	$\leftrightarrow$	[1, sG] / [+V]
/-lAk/	$\leftrightarrow$	[1, 2, sg]
/ <i>-Uk</i> /	$\leftrightarrow$	[1, 3, pl]
/ <i>-Om</i> /	$\leftrightarrow$	[1, sg]
/-Unk/	$\leftrightarrow$	[1, pl]

#### (53) Vocabulary items involving second person subjects (final)

'-Ol/-sz/	$\leftrightarrow$	[2, sg] / [+V]
'-AAtOK/	$\leftrightarrow$	[2, 3, pl]
'-Od/	$\leftrightarrow$	[2, sg]
′-tOk∕	$\leftrightarrow$	[2, pl]

#### (54) Vocabulary items involving third person subjects (final)

/-a/-i/-e/	$\leftrightarrow$	[3, 3, sG]
/ <i>-AAk/-ik/</i>	$\leftrightarrow$	[3, 3, pl]
/-Ø/	$\leftrightarrow$	[3, sg]
/-nAk/	$\leftrightarrow$	[3, pl]
/ <i>-Uk</i> /	$\leftrightarrow$	[3, pl] / [+N]

#### (55) Impoverishment rules

- a.  $[+V] \rightarrow \emptyset / [+N]$
- b.  $[+V] \rightarrow \emptyset / [1, +PST]$
- c.  $[+V] \rightarrow \emptyset / [(1|2), 3, sG]$
- d.  $[+V] \rightarrow \emptyset / [T = v = (1 \mid 2), sG]$
- e.  $[+V] \rightarrow \emptyset / [1, V = eszik$  'to eat', iszik 'to drink', ...]

The syntactic derivations introduced in Section 4.3, together with the spell-out rules just introduced derive the full range of data discussed in this section.

I adopted Trommer's (2005) insight that some of the verb forms indicating object agreement are better analysed as being underspecified for the person of the object: this explains why they have a wider distribution than the verb forms expressing subject agreement only. While adding some complexity to the system by introducing some impoverishment rules, this approach explains the distribution of syncretic forms in

the paradigm and, as I show in the next section, it makes it possible to generalise the overlap of verbal and nominal morphology.<sup>16</sup>

#### 4.4.3 Verbal agreement and possessive suffixes

As briefly mentioned above, Szabolcsi (1994) notes that the overlap between possessive suffixes and verbal agreement suffixes seems to cut across the distinction between suffixes expressing subject agreement only and those expressing both subject *and* object agreement.

A welcome consequence of the analysis proposed here is that the overlapping distribution of verbal agreement suffixes and possessive suffixes can now be generalised:

#### (56) Agreement morphology and possessive suffixes

When agreement and possessive suffixes match for a given person  $\pi$ , they represent the least specific vocabulary item for that person, of the following form: /SUFFIX/  $\leftrightarrow [\pi, SG/PL]$ 

The generalisation in (56) states that verbal agreement and possessive suffixes match when the person of *one* argument is represented in a spell-out rule, but not when there are two. This gives rise to the impression that in the singular, possessive suffixes match verbal agreement suffixes that show object agreement, but in the plural, possessive suffixes match those that only indicate subject agreement (as suggested by Szabolcsi 1994).

But if the spell-out rules presented in Section 4.4.2 above are on the right track, -Om and -Od do not cross-reference the object after all. Given their insertion rules, they match intransitive contexts as well but they are only inserted when deletion of [+V] rules out insertion of -Ok and -sz/-l, respectively. (57), including the revised spell-out rules, shows the overlap between verbal agreement and possessive suffixes and the relevant spell-out rules for first and second person.

(57) a. 
$$l\acute{a}t$$
-om  $h\acute{a}z$ -am /-Om/  $\leftrightarrow$  [1, sG] see-1sG.OBJ house-1sG.POSS 'I see (it)', 'my house'

<sup>&</sup>lt;sup>16</sup> More is to be said about certain conditional and subjunctive forms that show syncretic forms in certain configurations. These are not relevant for present purposes, however.

b. lát-od ház-ad  $/-Od/ \leftrightarrow [2, sG]$ see-2sg.obj house-2sg.poss 'you (sg.) see (it)', 'your (sg.) house' lát-unk ház-unk  $/-Unk/ \leftrightarrow [1, PL]$ c. see-1PL.SBJ house-1PL.POSS 'we see', 'our house' lát-tok  $/-tOk/ \leftrightarrow [2, PL]$ d. ház-atok

see-2sg.овј house-2sg.poss 'you (pl.) see', 'your (pl.) house'

Consider the competing vocabulary items for the verb in (57a). After the syntactic derivation and the application of fusion as discussed in Section 4.3, the terminal v+T has the features [1, 3, sG, +V]. This feature specification triggers the impoverishment rule in (55c) above which deletes [+V]. Now the terminal is specified as [1, 3, sG] which correctly allows the insertion of the suffix *-Om*. This was shown in (48) above, repeated here.

#### (48) Impoverishment

$$\begin{bmatrix} +V & & & & \\ +V & & \\ u\phi & u\pi & 1,3 \\ & u\# & sG \end{bmatrix} \xrightarrow{\text{Impoverishment: (55c)}} \begin{bmatrix} +V & & & \\ +V & & & \\ u\phi & u\pi & 1,3 \\ & u\# & sG \end{bmatrix}$$

In the case of the nominal  $h\acute{a}z$ -am 'my house', the terminal node arguably lacks the specification for [+V] and therefore only -Om can be inserted, and not -Ok. The same reasoning holds for (58b). In (58c,d), there is no competition between spell-out rules.

The situation for third person suffixes is slightly more complex. First, in the plural, third person verbal agreement and third person plural possessors do not show the same form.

- (58) a. *lát-nak* see-3PL.SBJ 'They see.'
  - b. *lát-ják* see-3pl.OBJ 'They see it.'

- 4 Agreement with personal pronouns
  - c. *ház-uk* house-3pl.poss 'their house'

Since none of the suffixes in (58) match, third person plural suffixes are not in the scope of the generalisation in (56) (see Trommer 2005 for an alternative approach). In the third person singular, however, there is an overlap between verbal agreement and possessive suffixes:

- (59) a. *lát-ja* see-3sg.obj 'S/he sees (it).'
   b. *lát-t-a* see-PST-3sg.obj 'S/he saw (it).'
  - c. *ház-a* see-3sg.poss 'his/her house'

(59b) and (59c) in particular show matching verbal agreement and possessive morphology. However, when the possessed noun is in the plural, the suffix -a- seen in (59c) appears in all persons, not just third:

(60) a. ház-a-i-m house-POSS-PL.POSS-1SG.POSS 'my houses'
b. ház-a-i-d house-POSS-PL.POSS-2SG.POSS 'your (sg.) houses'
c. ház-a-i-Ø house-POSS-PL.POSS-3SG.POSS 'his/her houses'

I therefore analyse the suffix -a- in (59c) as a general possessive suffix that does not express the person of the possessor. If this is correct, the third person possessive suffix is  $\emptyset$ , as indicated in (60c) and the correct gloss of (59c) is as in (61):

(61) ház-a-Ø house-Poss-3sg.Poss 'his/her house'

This allows us to keep the generalisation in (56) intact. (62) shows the matching zero exponent of third person singular verb forms without object agreement and the  $\emptyset$ -suffix following a generic possessive suffix. (63) repeats the generalisation introduced in (56) above.

(62)  $l\acute{a}t-\varnothing \quad h\acute{a}z-a-\varnothing \quad /-\varnothing/\leftrightarrow [3, sG]$ see-3sG.SBJ house-Poss-3sG.Poss 's/he sees', 'his/her house'

#### (63) Agreement morphology and possessive suffixes

When agreement and possessive suffixes match for a given person  $\pi$ , they represent the least specific vocabulary item for that person, of the following form:  $/\text{SUFFIX}/ \leftrightarrow [\pi, \text{SG/PL}]$ 

#### 4.4.3.1 Interim summary

The syntactic analysis introduced in the previous section and the spell-out rules introduced here can express Szabolcsi's (1994) observation of the distribution of matching verbal agreement and possessive suffixes.

I adopted the gist of Trommer's (2005) analysis about the distribution of first and second person suffixes indicating object agreement, without adopting his suggestion that there is no portmanteau morphology in Hungarian (see the next section for discussion). I argued, following his proposal, that some suffixes that spell-out subject and object agreement are actually underspecified and do not include reference to the person of the object. Rather, their distribution follows from a number of impoverishment rules.

Note, finally, that this analysis serves as a synchronic account of the data but it does not shed light on the origin of the overlap between possessive and verbal morphology. For a number of possible explanations for Hungarian, see Havas (2004) (and see Malchukov 2013 for related phenomena in a number of unrelated languages).

#### 4.4.4 Further issues in verb morphology

Before concluding this section, I return to the question of compositionality in Hungarian verb morphology. I have analysed the suffixes -lAk (1sG/2), -a/-e/-i (3sG/3), -Uk (1PL/3), -AAtOk (2PL/3) and -AAk (3PL/3) as indicating both the person of the subject and the person of the object as portmanteau morphemes.

However, a large number of approaches to Hungarian object agreement and verb morphology draw a different conclusion and argue that (some of) the suffixes can transparently be segmented into separate subject and object markers.<sup>17</sup> In this section, I discuss the present approach relative to other analyses that assume a more "compositional" make-up of Hungarian verb morphology.

First, consider evidence for separate exponents of the person features of the subject and the object.<sup>18</sup> The forms of the verb *lát* are shown in Table 4.3 (cf. Table 2.1, p. 26 above). In the present tense, the element -j- is a candidate for an object agreement marker, as it appears in several forms that show third person object agreement.

	Subject agreement only (SBJ)		Subject and object agreement (OBJ)		
	Present tense	Past tense	Present tense	Past tense	
1sg	lát-ok	lát-t-am	lát-om	lát-t-am	
2sg	lát-sz	lát-t-ál	lát-od	lát-t-ad	
3sg	lát-∅	lát-ott	lát-ja	lát-t-a	
1pl	lát-unk	lát-t-unk	lát-juk	lát-t-uk	
2pl	lát-tok	lát-t-atok	lát-játok	lát-t-átok	
3pl	lát-nak	lát-t-ak	lát-ják	lát-t-ák	

 Table 4.3
 Present and past indicative forms of lát 'to see'

Ignoring 1/2sG for now, -*ja*- can be identified in 3sG, 1PL, 2PL and 3PL. In the past tense, however, -*j*- disappears, and the suffix -*t*- takes its place, presumably expressing

<sup>&</sup>lt;sup>17</sup> Trommer's (2005) paper is called "Hungarian has no portmanteau agreement" and he puts forth conceptual arguments against portmanteau agreement as well as the argument for underspecification reviewed above; see also Bartos (1999), É. Kiss (2002), den Dikken (2004, 2006) and Rocquet (2013) for approaches involving separate subject and object agreement suffixes.

<sup>&</sup>lt;sup>18</sup> For the sake of simplicity, I abstract away from morphophonological variation induced by vowel harmony and assimilation and I focus on the forms of a single verb, namely *lát* 'to see'. My general point holds across morphophonological variation.

past tense. This is shown in more detail in Table 4.4. While the contrast between the present tense forms points to *-j-* as an agreement marker, the differences between the agreeing present and past tense forms in Table 4.4 could suggest that *-j-* indicates present tense, like *-t-* indicates past (cf. also Rocquet 2013: 68f. for discussion).

	sBJ, present	овј, present	овј, past
3sg	lát-∅	lát- <b>j</b> -a	lát- <b>t</b> -a
1pl	lát-unk	lát- <b>j</b> -uk	lát- <b>t</b> -uk
2pl	lát-tok	lát- <b>j</b> -átok	lát- <b>t</b> -átok
3pl	lát-nak	lát- <b>j</b> -ák	lát- <b>t</b> -ák

Table 4.4 -*j*- and -*t*- in present and past tense forms of *lát* 'to see'

Another candidate for an object marker is *-a-*, as in the following pairs:

(64)	a.	lát-ott	b.	lát-t- <b>a</b>
		see-pst.3sg		see-pst-A-3sg
	b.	lát-t-atok see-pst-2pl	d.	<i>lát-t-<b>a</b>-atok</i> see-pst-A-2pl
	c.	lát-t-ak see-pst-3pl	e.	<i>lát-t-<b>a</b>-ak</i> see-pst-A-3pl

However, neither *-j-* nor *-a-* (or their several allomorphs) consistently distinguish a third person object agreement form from one that only appears with the subject. In addition, there are pairs of person suffixes that do not exhibit either of these: 1sG and 2sG in both present and past tenses, as well as arguably 1PL, do not show these exponents.

(65), adapted from Rebrus 2005: 64, shows a templatic structure for Hungarian verbs, where -j- and -a- are shown as the potential allomorphs of an object marker, and the two instances of  $\varphi$  indicate two positions of exponents of person and number features.

(65) stem + TENSE/MOOD/-j- + -a-/ $\phi$ (I) +  $\phi$ (II)

Tables 4.3 and 4.4 and the example in (64) showed verb forms that match different versions of the template in (65). (64d), for example, realises all of these exponents in

the template, while (64b) realises the first two suffixes but not the final one, and (64a) arguably realises the first one only. Rebrus (2005: 66, fn. 23) mentions that given the amount of variation in Hungarian verb morphology, it is questionable whether it is at all useful to assume that the morphemes in question are segmentable and that these segments are subject to morphophonological rules.

I have argued for an alternative, namely to treat the agreement morphemes as portmanteaus and I have attempted to show that this allows us to derive the spell-out of verb morphology and capture the overlap with possessive morphology. While this approach potentially glosses over certain details in verb morphology, it has several further advantages.

First, I provided both a syntactic and a morphological analysis that work together to derive the forms presented in this section. Second, by adopting the structure that involves a single agreement suffix, there is no need to assume null morphemes referring to object agreement in certain configurations (as Trommer 2005 and den Dikken 2006 do).

Trommer (2005) further argues against portmanteaus in Hungarian for conceptual reasons. Since vocabulary items are inserted into one terminal in DM, portmanteau morphology is difficult to model if a vocabulary item has to refer to properties of more than one terminal node (say a subject agreement probe T, and an object agreement probe v). However, I have argued that a single probe can carry features of two arguments when the subject's features are a proper superset of the object's features or when fusion applies. I have put forth a restrictive view of fusion of v and T: the two heads only fuse if they are sisters and their strongest features match. These assumptions are compatible with the existence of portmanteau morphology in Hungarian verb morphology.

Combining the syntactic and morphological structures proposed in this chapter results in a systematic and elegant explanation of the gaps in object agreement with personal pronouns, which has some advantages over other approaches. Both Trommer (2005) and den Dikken (2006) assume a range of null object agreement suffixes, but it is not clear why they should appear in exactly the contexts of first and second person pronouns, for example. Trommer assumes that in these contexts, object agreement features are deleted. Den Dikken (2006) goes further by assuming that in inverse contexts, first and second person objects are represented by null clitics. Second person pronouns, however, have non-null allomorphs that appear with first person subjects.

The present proposal derives the seemingly non-agreeing forms using a combination of syntactic processes and morphological rules, which explains the gaps in a systematic way.

#### 4.4.5 Interim summary

In this section, I provided an analysis of the morphological structure of verb forms in Hungarian that fits with the syntactic analysis provided in the preceding sections. I argued that there is one position in Hungarian verbs that spells out a full set  $\varphi$ -features. The  $\varphi$ -features are determined by the syntactic derivation and entailment relations among sets of person features.

Vocabulary insertion rules either directly spell-out the features on a terminal node in the syntax or are preceded by impoverishment rules. Following Trommer (2005), I have used impoverishment rules to derive the distribution of first and second person singular suffixes, which not only appear in verb morphology, but also as possessive suffixes.

I further proposed a generalisation that accounts for the overlap between possessive suffixes and verb morphology (see (56), p. 132). As Szabolcsi (1994) notes, verb morphology and possessive morphology overlap in different ways. When the subject or the possessor is singular, possessive morphology matches verb morphology expressing subject and object agreement. When the subject or the possessor is plural, possessive morphology matches verb morphology expressing subject agreement only. I showed that this pattern can be explained by the following generalisation: possessive suffixes and verb morphology match when both are the result of a vocabulary insertion rule that refers to one argument only.

#### 4.5 Comparison to other analyses of object agreement

Before concluding this chapter, I will position the current proposal with respect to existing proposals. Analyses of Hungarian object agreement, in particular with respect to agreement with personal pronouns, fall into two classes. Some researchers suggest that only third person arguments trigger object agreement, while others assume that

personal pronouns of other persons can trigger agreement as well - these approaches have to account for the gaps in agreement, and do so in different ways.

#### 4.5.1 Agreement with third person objects

Bartos (1999), É. Kiss (2002), den Dikken (2004), Coppock and Wechsler (2012), Coppock (2013) and Rocquet (2013) provide analyses of the former type: these authors focus on third person objects. Recall that Bartos (1999) suggests that only third person pronouns project a DP, but first and second person pronouns do not. Similarly, Coppock and Wechsler (2012) only attribute the feature [DEF] to third person pronouns. Coppock (2013) makes use of differences in the pronouns' referential properties: she suggests that anaphoric pronouns, like third person pronouns and reflexives, do trigger agreement, but indexical pronouns, like first and second person pronouns do not.

There are two obvious consequences of analysing Hungarian object agreement in this way. First, as mentioned above in much detail, the behaviour of second person pronouns with respect to object agreement requires a different explanation from object agreement with third person pronouns (and first person pronouns). I have attempted to show that second person pronouns clearly trigger object agreement with first person subjects and that it is possible (and thus arguably preferable) to account for object agreement with distinct pronouns by using a single mechanism. Obviously, this raises the question of why agreement is not always spelled out.

The present chapter provides an answer to this question. The syntactic mechanisms I used to explain the lack of agreement are well attested in other languages (cf. Béjar 2000; Bobaljik and Branigan 2006; Lochbihler 2008; Béjar and Rezac 2009; Keine 2010; Georgi 2012; Lochbihler 2012 and Chapter 5) and are compatible with a common syntactic structure of all personal pronouns. In addition, I showed that indexicality and anaphoricity do not influence object agreement, as both indexical and non-indexical pronouns can trigger agreement (see Section 3.2.1).

A second consequence of interpreting object agreement as being restricted to third person arguments is that it becomes more difficult to characterise the class of noun phrases that trigger object agreement. As discussed in Chapter 1, differential object marking tends to be triggered by direct objects that are high in definiteness, and/or animacy, or topicality. While definiteness correlates with object agreement even when object agreement is restricted to third person, this restriction itself is not typical across languages. By extending the potential to trigger object agreement to first and second person pronouns as well, it becomes possible to state a generalisation about object agreement in Hungarian that fits in well with DOM in other languages. This was suggested in (55) on p. 94 in Chapter 3 and is repeated here as (66):

#### (66) **Object agreement in Hungarian**

If a direct object is definite, it triggers object agreement.

The possibility of stating (66) is a consequence and advantage of the present analysis which takes the lack of object agreement with personal pronouns to be a superficial effect. This is a welcome result.<sup>19</sup>

#### 4.5.2 Agreement with first, second, and third person objects

The second kind of approach to object agreement in Hungarian acknowledges that not only third person arguments can trigger object agreement. Analyses of this type have been proposed by É. Kiss (2003, 2005), den Dikken (2006), É. Kiss (2013), Bárány (2015b) and É. Kiss (2015) — the present work also endorses this view. The analysis I presented here shares properties with these approaches, but I believe that it is also superior in certain respects. Let me first discuss É. Kiss's (2013) approach.

As mentioned above, É. Kiss (2013) characterises the Hungarian agreement patterns as follows (repeated from (6), p. 100; her animacy/person hierarchy is shown in (68)):

#### (67) Inverse agreement constraint for Hungarian

An object agreeing with a verb must be lower in the animacy hierarchy than the subject agreeing with the same verb, unless both the subject and the object represent the lowest level of the animacy hierarchy.

(É. Kiss 2013: 8)

(68) $1SG > 1PL/2 > 3$	(É. Kiss 2013: 8)
------------------------	-------------------

This is a descriptive statement that mostly fits the data. The underlying motivation for it is a fossilised constraint on agreement with topical direct objects in the history of

<sup>&</sup>lt;sup>19</sup>Note, however, that (66) does not imply that all noun phrases that trigger object agreement are definite.

Hungarian. Marcantonio (1985) suggests that object agreement in Hungarian used to be sensitive to the topicality of the direct object (see also Peredy 2009; cf. Nikolaeva 1999a,b; Dalrymple and Nikolaeva 2011 for similar suggestions in related languages).

É. Kiss (2013: 16) suggests that this sensitivity explains the constraint in (67): in the typical Uralic SOV clause, the subject would be a topic and the object would only trigger agreement if it were topical as well (a secondary topic; see Nikolaeva 2001; Dalrymple and Nikolaeva 2011; Bárány 2014a). The constraint blocks "that the secondary topic be more topical (in other words, more animate, more specific) than the primary topic. An object more animate, more salient than the subject can only be presented as a focus." (É. Kiss 2013: 16)

Consider a sentence with a second person subject and a (topical) first person object. According to É. Kiss (2013), there can be no object agreement in this case, because the direct object would be more topical than the subject – exactly what is ruled out by her constraint. While topicality historically did play a role in object agreement (Marcantonio 1985), its influence on object agreement in present-day Hungarian is not strong, if present at all (see Peredy 2009 for discussion). Most importantly, the distribution of object agreement with personal pronouns does not correlate with information structure, i.e. even objects that are less topical than their subjects are not agreed with when they are first and second person pronouns (and the subject is lower on the hierarchy). This suggests that É. Kiss's (2013) constraint can only be part of the grammar of presentday Hungarian as a constraint that specifically rules out agreement with first person pronouns and second person pronouns (when the subject is third person) because topicality does not influence present-day object agreement. While possible, this means that the distribution of object agreement in present-day Hungarian would not be stated as a generalisation, but merely as a descriptive statement that rules out certain combinations of the person of the subject and the object. In contrast, on the analysis I proposed here the distribution of agreement follows from syntactic principles.

A second problematic aspect of É. Kiss's (2013) analysis is her claim that being more topical correlates with being more animate and/or more specific, or simply higher in person. While there is arguably a general tendency for definite and animate entities to be more topical than indefinite and inanimate ones, personal pronouns are all definite and specific. However, personal pronouns can be topical or focused depending on the

discourse, independently of their (constant) property of being definite (see also Coppock and Wechsler 2010 for a similar argument).

Another analysis that assumes that not only third person pronouns trigger agreement is proposed by den Dikken (2004, 2006). Den Dikken's two analyses diverge in some respects; here, I will focus on den Dikken (2006). I will first summarise his approach to verb morphology and then highlight certain aspects of his analysis that contrast with the present proposal.

Den Dikken argues that Hungarian verb morphology includes inflectional suffixes as well as clitics, for both subject and objects. He identifies *-ja-* as a third person object clitic, and the suffixes *-om* and *-od* as subject clitics, for example:

(69) a. *lát-ja* see-3.0BJ.CL
b. *lát-om* see-1sg.sBJ.CL

Combining the two clitics in (69) would derive a hypothetical form *\*lát-ja-m* 'see-3.0BJ.CL-1SG.SBJ.CL' that does not exist. Den Dikken (2006) suggests that there is a co-occurrence restriction on subject and object clitics that rules out any verb form with a third person and a first or second person clitic:

# (70) Clitic co-cccurrence restriction (Proto-Uralic) A third person [object clitic] cannot co-occur with a first or second person [subject clitic].

(den Dikken 2006: 12)

While the constraint in (70) rules out forms like  $*l\acute{a}t$ -ja-m correctly, it is not clear why it would not give rise to a form like that in (71a). Since -k is not a clitic, it should not be subject to the restriction in (70). (71b) shows an analogous example that includes an object clitic and the first person inflectional suffix indicating the subject.

(71) a. \**lát-ja-k* see-3.0BJ-1sG.SBJ intended: 'I see him/her'

b. *lát-l-ak* see-2.0BJ-1SG.SBJ 'I see you'<sup>20</sup>

I want to highlight three more potential problems with den Dikken's (2006) analysis. First, while his clitic co-occurrence restriction resembles the person-case constraint (or PCC, see e.g. Béjar and Rezac 2003; Anagnostopoulou 2005; Adger and Harbour 2007; Rezac 2008b; M. Richards 2008 among many others), it actually seems to be more of an *ad hoc* tool to explain Hungarian verb morphology. The reason for this lies in the distinction den Dikken makes between inflectional suffixes and clitics. Clitics are said to be derived from Proto-Uralic pronouns, whereas inflectional affixes are presumably newer. But this cannot be the whole story: according to the data den Dikken presents, the element *-l-* that appears in the second person object agreement form *-lak* does not derive from an old pronoun. It is nevertheless treated as a clitic and therefore subject to the constraint in (70). The distinction between inflectional affixes and clitics is therefore unclear and not defined independently.

A second problem with den Dikken's analysis lies in the inventory of clitics. To explain why the second person object clitic *-l*- appears when the subject is first person, but not when the subject is third person, den Dikken (2006: 18) suggests that this clitic has a covert allomorph whose presence is triggered by a third person subject. Similarly, first person object clitics are also null. In other words, den Dikken (2006) assumes that while all personal pronouns trigger clitic doubling, many object clitics are null. These null clitics appear in inverse cases in den Dikken's analysis, but this is an accidental fact.

In the present analysis, on the other hand, the absence of agreement follows from spelling out a probe that has only agreed with the subject. I believe that the system introduced in Section 4.3 above provides a more principled analysis of the distribution of agreement with personal pronouns.<sup>21</sup>

<sup>&</sup>lt;sup>20</sup> Note that it is also unclear why there should not be a form like *\*lát-sz-ak* 'see-2.OBJ-1SG.SBJ' in which the second person suffix has the *-sz* allomorph that appears after stops like [t]. There is no allomorphy in the *-lAk* suffix (apart from obvious vowel harmony), indicating at least that the inventory of potential object agreement suffixes is restricted to and does not fully overlap with subject agreement suffixes.

<sup>&</sup>lt;sup>21</sup>Den Dikken (2006: 18, fn. 34) refers É. Kiss's (2005) proposal of deriving inverse agreement, whose essentials are similar to É. Kiss (2013). His main criticism is that É. Kiss (2005, 2013) relies on a "combined PERSON/NUMBER hierarchy" whose status in a Minimalist framework is questionable. The present approach avoids this issue and provides a syntactic alternative that derives the same effects.

A third problem for den Dikken's (2006) specific implementation of clitic doubling follows from his analysis of the structure of first and second person pronouns. Essentially, den Dikken (2004, 2006) assumes that the accusative first person pronoun *engem* and the accusative second person pronoun *téged* are actually possessive constructions, on analogy to possessed common nouns (see den Dikken 2006; Rocquet 2013 for detailed analyses of the role and nature of -g-):<sup>22</sup>

- a. en-g-em 1sG-g-1sG.POSS
  b. kez-em hand-1sG.POSS
  - c. *té-g-ed* 2sg-g-2sg.poss
  - d. *kez-ed* hand-2sg.poss

Den Dikken (2004, 2006) assumes that first and second person object clitics are generated in the left periphery of these possessed noun phrases or "pronouns" as their possessors. He analyses clitic doubling as movement of these possessors from the possessed NP onto the verb (recall the notion of possessor extraction discussed in Chapters 2 and 3). This resembles so-called bigDP analyses of clitic doubling, in which a clitic moves from an outer layer of a DP onto the verb, combining XP movement and head movement (Uriagereka 1995; Cecchetto 1999).

It is possible to test whether this kind of movement can happen. In a noun phrase that consists of two coordinated pronouns, movement of either a single possessor or two distinct possessors should be ruled out by the coordinate structure constraint. The following examples show that such combinations are grammatical, however. (73) shows a coordinated noun phrase whose constituents differ in person, and (74) shows a coordinated noun phrase whose constituents differ in number. The fact that these examples are grammatical makes it unlikely that each pronoun includes a movable possessor clitic.<sup>23</sup>

 $<sup>^{22}</sup>$  É. Kiss (2015) relates the analysis of personal pronouns along these lines to the lack of the usual accusative suffix *-t* on these pronouns and the possibility of dropping this suffix with nouns with first and second person possessors (see also Kamper 2006).

<sup>&</sup>lt;sup>23</sup> Den Dikken (2004) argues that the movement of the possessor clitic can explain the pattern of dative intervention in (i); however, his (2006) reanalysis of the structure of pronouns is no longer compatible

- 4 Agreement with personal pronouns
- (73) ... hogy egyedül csak a halál választ el engem és téged! that alone only the death separate.3sg.sbj vm I.Acc and you.sg.Acc
   '... also if anything but death parts me from you.'

(Ruth 1:17, http://szentiras.hu/KNB/Rút1 and English Standard Version)

 (74) ... hogy egyedül csak én választ-alak el téged és titeket! that alone only I separate-1sG>2 vM you.sG.ACC and you.PL.ACC
 '... only I part you (sg.) from you (pl.).'

Note also that third person pronouns are presumably not possessive NPs. It remains unclear where the third person object clitic, *-j-*, originates from, and why personal pronouns should have different syntactic structures.

#### 4.5.3 Inclusive reference, again

É. Kiss (2015) raises the data in (75) as problems for the analysis of agreement I proposed in Bárány (2015a,b). I will address this criticism below and show that the present proposal solves the issue in question.

(75)	a.	Te	titek-et	ajánl-od /	*ajánl-asz?	
	you.sg you.pl-Acc recommend-2sg.овј recommend-2sg.sвј 'Do you (sg.) recommend you guys?					
	b.	<i>Ti</i> you. 'Do y	<i>téged</i> PL you.sG.AC you guys rec	* <i>ajánl-játok /</i> c recommend-2pL. ommend you (sg.)'	<i>ajánl-otok?</i> овј recommend.2pl.sвј ?'	
					(É. Kiss 2015: 61	f. glosses adapted)

(75a) involves a case of inclusive reference. With a second person singular subject and a second person plural object, object agreement is preferable to subject agreement.(75b) shows that if the subject is plural and the object is singular, the pattern is reversed and subject agreement is preferred.

(i) Hagy-lak (\*János-nak) meg-látogat-ni (téged).
 let-1sG>2 János-DAT VM-visit-INF you.sG.ACC
 'I let you be visited (\*by János).'

(den Dikken 2006: 453)

with the earlier proposal. (i) still poses an interesting puzzle for which I do not have an explanation. See den Dikken (2004: 453) and den Dikken (2006: 16, fn. 31) for discussion.

Recall that É. Kiss's (2013) animacy/person hierarchy places first person singular higher than first person plural (repeated from (5) above):

(5) 
$$1 \text{SG} > 1 \text{PL}/2 > 3$$

The motivation for this split in first person pronouns comes from data like that in (76). As mentioned above, when the subject of a transitive verb is first person singular and the object is second person, the suffix *-lAk* appears, showing object agreement. When the subject is first person plural, however, the verb appears in the intransitive form, as in (76a,c). Object agreement with a third person pronoun is shown in (76b).

- (76) a. *Lát-unk*. see-1pl.sbj 'We see.'
  - b. *Lát-juk (ő-t).* see-1pl.obj s/he-ACC 'We see him/her.'
  - c. Lát-unk téged / titeket. see-1pl.sBJ you.sG.ACC you.pl.ACC 'We see you (sg./pl.).'

É. Kiss's (2013) hierarchy in (5) derives this because object agreement between first and second person arguments only arises when the person of the subject is higher than the person of the object. In my proposal, the fact that (76c) only shows subject agreement on the verb is a matter of spell-out. As argued in Section 4.3.1, the possibility of dropping objects with verb forms that show subject agreement only can be linked to the syntactic derivation in which v has entered an Agree relation with the object. The form *látunk* in (76c) on its own is in fact ambiguous between an intransitive reading and a reading with an implicit second person object.

É. Kiss's (2013) hierarchy also predicts the pattern of agreement in configurations involving two first person arguments.

(77) a. ?Én mink-et is belevesz-em / \*belevesz-ek a névsor-ba.
 I we-ACC too include-1sG.OBJ include-1sG.SBJ the namelist-ILL
 'I also include us into the list of names.'

(É. Kiss 2013: 6)

(77) shows a case of inclusive reference. But while the agreement pattern in (77) follows from the hierarchy in (5), example (75) including second person arguments does not and it constitutes a problem for É. Kiss (2013, 2015) as much as it does for Bárány (2015a,b). In that work, I attempted to explain these patterns by suggesting that first person singular pronouns were more richly specified than first person plural pronouns. If that were the case, the pattern in (77) would follow.

However, where É. Kiss (2013), Bárány (2015a,b) and É. Kiss (2015) succeed (or fail) in deriving the presence of object agreement, their approaches to not determine the *form* of object agreement and it remains unclear why first person plural objects would not trigger a form other than the one that also appears with third person objects (see the discussion of den Dikken *et al.* 2001 below for an approach along these lines).

The current proposal implements these patterns by impoverishment rules that bleed the insertion of the form showing subject agreement only when first and second person singular subjects have plural objects in the same person (see Section 4.4 above).

These impoverishment rules can arguably be motivated by the semantics of configurations involving inclusive reference. When the subject of the clauses in question is singular, its referent is part of the (plural) referent of the object. In this sense, these predicates are reflexive. When the subject is in the plural, however, and the object is in the singular, the subject's referent is not included in the object's referent and the predicate is not reflexive.

Consider actual reflexive pronouns. Reflexive pronoun objects seem impossible when the subject is plural and the object is singular, but they are possible when the subject is singular and the object is plural, as in (78a) and (78b), respectively (this is the same pattern as the one seen with inclusive reference involving personal pronouns).

- (78) a. \**Lát-unk / \*lát-juk magam.* see-1PL.SBJ see-1PL.OBJ myself intended: 'We see myself.'
  - b. Lát-om / \*lát-ok magunkat. see-1sG.OBJ see-1sG.SBJ ourselves 'I see ourselves.'

Suppose now that reflexives are ruled out when the referent of the antecedent does not include the referent of the reflexive, i.e. when the pronoun is not *fully* reflexive

because its antecedent has additional referents. This is the case in (78a). This would rule out singular reflexives with plural antecedents in general, but not singular antecedents of plural reflexives. Indeed, (78b) is more acceptable than (78a), as is its English translation.

The impoverishment rules in (55), repeated below, bleed the insertion of verbal morphology that only shows subject agreement in inclusive reference contexts. The mismatch between the reflexive interpretation of inclusive reference and form that only show subject agreement could motivate the rule in (79d). Note further that when the subject in (78) is plural, the present system correctly predicts that the verb only shows subject agreement.

#### (55) Impoverishment rules

- a.  $[+V] \rightarrow \emptyset / [+N]$
- b.  $[+V] \rightarrow \emptyset / [1, +PAST]$
- c.  $[+V] \rightarrow \emptyset / [(1|2), 3, s_G]$
- d.  $[+V] \rightarrow \emptyset / [T = v = (1 \mid 2), sG]$
- e.  $[+V] \rightarrow \emptyset / [1, V = eszik$  'to eat', iszik 'to drink', ...]

If correct, this provides a motivation for the pattern of agreement in contexts of inclusive reference. If not, it is no more stipulative than É. Kiss's (2013) account but has a wider empirical coverage because the hierarchy in (5) is not flexible enough to cover both first and second person inclusive reference.

#### 4.5.3.1 First person plural as third person

Using impoverishment rules to determine the spell-out of verb morphology in contexts that involve inclusive reference provides a solution to the puzzle of why the verb appears to show agreement with a third person object: the verb is actually underspecified for the person of the object and the least specific vocabulary item is inserted.

In this section, I briefly discuss an alternative approach, namely den Dikken *et al.*'s (2001) suggestion that first and second person plural pronouns can surface as *third* person pronouns and therefore trigger object agreement (I will only discus first person, but the reasoning extends to second person as well). Consider again the data in (79).

- 4 Agreement with personal pronouns
- (79) a. \*Én minket választ-ok meg. I we.Acc elect-1sg.sbj vм intended: 'I elect us.'
  - b. %Én minket választ-om meg. I we.acc elect-1sg.овј vм 'I elect us.'

(den Dikken et al. 2001: 140f.)

Den Dikken *et al.* (2001) write that for certain speakers of Hungarian, (79b) is acceptable. According to den Dikken *et al.*, the problem with (79) is that in general first and second person pronouns do not trigger object agreement, but the first person plural pronoun *minket*, in (79b), does. Den Dikken *et al.* (2001) argue that this can be explained by assuming that first person plural pronouns in Hungarian can be expressed using two different structures. Both of these are abstract comitative structures that include a first person singular *pro* and a comitative PP as in (80):

Den Dikken *et al.* explain the pattern in (79) by referring to binding theory. They argue that in (79a), if the structure of the personal pronoun is as shown in (80), the result is a violation of Principle B, since *pro* is bound by a first person singular pronoun in subject position. Basically, (79a) is ruled out for the same is reason (81) is ungrammatical.

(81) \*I see me.

To allow for agreement in (79b), den Dikken *et al.* suggest that there is an alternative structure for first person pronouns that is embedded under a null *third* person noun phrase, as shown in (82). The pronoun in (82) is embedded in an NP that "can essentially be paraphrased as 'those who are us'" when it appears as a direct object (den Dikken *et al.* 2001: 146).

(82) 
$$\nu/\text{AgrO} \dots [_{\text{VP}} \text{V} [_{\text{DP}} \text{D}_{[\text{def}]} [_{\text{SC}} \varnothing_i [_{\text{NP}} \text{`us'} (=[(80)])]_i]]]$$
  
(den Dikken *et al.* 2001: 146)

Clearly, this raises the question of why a first person plural subject would trigger agreement on the verb that differs from that of a third person subject. Den Dikken *et al.* 

(2001: 147) suggest that the answer lies in an economy constraint that only inserts the DP-form of the first person pronoun in (82), which is more complex than the pronominal structure in (80), if choosing the latter would crash the derivation by violating Principle B. In other words, a structure is chosen that does not give rise to a violation of binding theory.

The motivation for treating first person plural pronouns as abstract comitative structures comes from examples like (83).

(83) (*Mi*) a nővér-em-mel nem ment-ünk mozi-ba. we the sister-1sg.poss-com NEG go-1pL cinema-ILL 'my sister and I did not go to the cinema'

(den Dikken et al. 2001: 144)

Note that the comitative *a nővéremmel* 'with my sister' in (83) has a first person *singular* possessive suffix and that the meaning of (83) refers to at least the speaker and his/her sister (other people might be included); see Schwartz (1988) for discussion of similar constructions across languages.

Linking agreement in cases of inclusive reference to restrictions on binding is a promising approach, but den Dikken *et al.*'s (2001) analysis includes a lot of machinery and involves very intricate structures for (some) personal pronouns. Nevertheless, if correct, it is compatible with the analysis developed in this chapter.

#### 4.5.4 Interim summary

In this section, I discussed a number of alternative approaches to object agreement with Hungarian personal pronouns. I distinguished two types of analyses: those that only assume object agreement with third person objects, and those that assume that objects of any person can agree in principle. I argued that the analysis presented here derives the actual distribution of overt object agreement in a more systematic way than other analyses.

#### 4.6 Summary and conclusions

In this chapter, I extended the analysis of Hungarian object agreement to the domain of personal pronoun objects. The agreement patterns with personal pronouns pose an

interesting puzzle because it is not clear at first glance which syntactic or semantic properties of first person arguments should be incompatible with object agreement.

I argued that there is in fact no inherent difference that gives rise to the attested agreement patterns, but that the relative properties of the subject and the object determine the spell-out of agreement. All personal pronouns trigger object agreement, but object agreement is not always spelled-out.

The present analysis determines two kinds of syntactic derivations that depend on the configuration of the person features of the subject and the object. I provided spellout and impoverishment rules that apply to the result of the syntactic derivation and correctly determine the form of verb morphology in all possible contexts, including certain syncretic forms. In addition, I proposed a generalisation to capture the distribution of the overlap between possessive and verbal morphology as described by Szabolcsi (1994).

Finally, this chapter provided further evidence for the hypothesis proposed in Chapter 3 that Hungarian object agreement is triggered by person features. This chapter therefore also strengthens the hypothesis that person features can grammaticalise referential properties that determine differential case-marking and agreement patterns. I present more evidence for this in Chapter 5.

#### 4.A Appendix: Full derivations

In this appendix to Chapter 4, I repeat the full derivations shown in Section 4.3 including spell-out rules for easy comparison.

#### 4.A.1 Direct configuration

The derivation in (85) (repeated from (23) above) shows the derivation of a clause like (84), with a first person singular subject and a second person object, i.e. a direct configuration. (85a) and (85b) show the steps of the derivation while (86) shows the fusion operation.

(84) Lát-lak téged. see-1sG>2 you.sG.ACC 'I see you (sg.).'



After T and v fuse, as shown in (86), there is a single terminal that has the features shown in (87a). This matches the vocabulary insertion rule in (87b) (repeated from (53) in Section 4.4 above). Therefore, the suffix *-lAk* is inserted.

(87) a. 
$$\nu$$
+T: [1, 2, sG]  
b.  $/-lAk/ \leftrightarrow$  [1, 2, sG]

#### 4.A.2 Inverse configuration

The derivation in (89) (repeated from (26) above) shows the derivation of a clause like (88), with a third person singular subject and a second person object, i.e. an inverse configuration. (89a) and (89b) show the steps of the derivation.

(88) Lát (téged). see.3sg.sbJ you.sg.acc 'S/he sees you (sg.).'



(90b) shows the continuation of the derivation. T probes and finds the subject, and receives the values [3, sG]. The subject is assigned NOM. Note again that now no fusion takes place since the strongest features on v and T do not match.



In (89), v and T do not fuse and only T has a full set of  $\varphi$ -features, shown in (90a). The vocabulary item in (90b) (repeated from (54)) above is therefore inserted.

(90) a. T: [3, sG] b.  $/-\varnothing/ \leftrightarrow$  [3, sG]

#### 4.A.3 Two third person arguments

The derivation in (92) (repeated from (36) above) shows the derivation of a clause like (91), with a third person singular subject and a third person object. (92a) and (92b) show the steps of the derivation while (93) shows the fusion operation.

(91) Lát-ja ő-t. see-3sg.obj s/he-Acc 'S/he sees him/her.'

4 Agreement with personal pronouns



After T and *v* fuse, as shown in (93), there is a single terminal that has the features shown in (94a). This matches the vocabulary insertion rule in (94b) (repeated from (54) in Section 4.4 above). Therefore, the suffix -a/-i/-e is inserted.

4.A Appendix: Full derivations

(94) a.  $T+\nu: [3, 3, sG]$ b.  $/-a/-i/-e/ \leftrightarrow [3, 3, sG]$ 

### Part II

## Case and agreement in the grammar

# 5 Inverse agreement and global case splits

#### 5.1 Introduction

The goal of this chapter is to extend the analysis of inverse agreement in Hungarian to other languages and explore its cross-linguistic implications. In Chapter 4, I proposed that Hungarian object agreement relies on cyclic Agree and interprets person features as sets of sub-features.

In this chapter, I argue that the specific assumptions about the nature of Agree and entailment relations among person features are not simply *ad hoc* rules that serve to explain a quirk of Hungarian grammar. I show, following a.o. Béjar and Rezac (2009), Keine (2010) and Georgi (2012), that the agreement pattern found in Hungarian is only one possible expression of grammatical phenomena that can be characterised by the notions of "direct" and "inverse", i.e. certain relations between the  $\varphi$ -features of the external and the internal argument.

So-called *global case splits* (Silverstein 1976; de Hoop and Malchukov 2008; Malchukov 2008; Keine 2010; Georgi 2012) are a related phenomenon. The term *global* refers to the fact that the properties of more than one argument determine whether that argument shows case-marking or not. *Local* splits, on the other hand, depend on the properties of a single argument only. Kashmiri, discussed below, represents an example of a global case split. In this language, case-marking of the object depends on the properties of both the subject and the object.

As Georgi (2012) illustrates convincingly, such splits can pose a problem for derivational approaches to syntax. In a language in which the morphological case of the direct object shows a global case split, i.e. where its case depends on the properties of

#### 5 Inverse agreement and global case splits

both the subject and the object, v cannot assign Case upon entering an Agree relation with the direct object, as this would be too early in the derivation.

In this chapter, I propose that this problem of timing Case assignment finds a simple solution in the approach to cyclic Agree introduced in Chapter 4. If v can enter Agree relations as long as it has not exhausted its  $\varphi$ -probe and Case assignment follows Agree, the  $\varphi$ -features of both the external and the internal argument will be able to value a probe before Case assignment happens. Whether a language exhibits a global case split or not is then (in part) a consequence of the order of Agree and Case assignment on v (see Müller 2004a; Heck and Müller 2007; Müller 2009; Keine 2010; Müller 2010; Georgi 2014 on the order of syntactic operations initiated by functional heads).

Combining this approach with the view that person features can grammaticalise different properties across languages, including definiteness and animacy, I show that a small set of assumptions derives the agreement and case patterns of a wide range of unrelated languages.

To illustrate the phenomenon consider the following data from Kashmiri (Indo-European, Indo-Aryan; Wali and Koul 1997), to be discussed in more detail below. In (1a), the subject is a first person pronoun and the object is a second person pronoun. Both are unmarked for case (glossed as NOM). In (1b), however, the first person pronoun is an object, and now it surfaces in its dative form.

a. *bi chu-s-ath tsi parina:va:n* I.NOM be.M.SG-1SG.SBJ-2SG.OBJ you.NOM teaching 'I am teaching you.'
b. *tsi chu-kh me parina:va:n* you.NOM be.M.SG-2SG.SBJ I.DAT teaching 'You are teaching me.'

(Wali and Koul 1997: 155, glosses adapted)

(1) indicates that first person objects can surface as datives. As (2) shows, however, second person objects undergo a similar alternation. In (2a), the subject is a second person, and the object is a third person pronoun, both unmarked (NOM). When the subject is third person, as in (2b), and the object is second person, the object surfaces as dative.
(2) a. *tsi chi-h-an su parina:va:n* you.NOM be-2sG.SBJ-3sG.OBJ he.NOM teaching 'You are teaching him.'

b. *su chu-y tse parina:va:n* he.NOM be.M.SG-2SG.OBJ you.DAT teaching 'He is teaching you.'

(Wali and Koul 1997: 155, glosses adapted)

In brief, whether a second person object surfaces as dative or not depends on the person of the subject.

The pattern in (2) resembles the distribution of the Hungarian *-lak/-lek* suffix, discussed in the previous chapter: with a first person subject, the verb agrees with a second person object, but with a third person subject, it seemingly does not. As I have argued in Chapter 4 that the distribution of agreement in Hungarian can be likened to direct and inverse marking in other languages, I will suggest that data like (1) and (2) provide additional evidence for treating global case splits in a similar way.

What characterises these phenomena is that certain properties of the subject and the object, in this case their person features, have to be "compared" to determine the morphological form of the verb. I have argued in Chapter 4 that this comparison, and the distinction between direct and inverse configurations, can be read off the values of the  $\varphi$ -probes of T and  $\nu$ , respectively, after they have agreed with the subject and the object. Direct configurations are characterised by  $\nu$  having more than a single set of  $\varphi$ -features.

I will show that the same holds for global case splits, with the difference that the relevant exponent is the case morphology on one of the two arguments. Global case splits are therefore a dependent-marking counterpart to inverse agreement, which reflects the properties of two arguments on the verb, i.e. the head (Nichols 1986).

While the two phenomena appear to be two sides of the same coin, then, there is a crucial difference following from the dependent-marking nature of global case splits: the case morphology determined by the Agree relations with the subject and the object are not spelled out on the probe that is valued by the arguments, but on the dependents, i.e. the arguments providing the values. Yet in order to determine the form of case-marking, both the subject and the object will have had to value the head that determines case assignment. In other words, if the object's case depends on the

person features of the subject as well, case assignment to the object has to wait until the subject has agreed with the object's case assigner.

This chapter is structured as follows. In the following section, I discuss case splits in two languages (Kashmiri and Sahaptin) in more detail. In Section 5.3, I discuss a crucial aspect of the analysis of global case splits: the timing of case assignment and its interaction with Agree. In Section 5.4, I implement my analysis and show detailed derivations of case splits based on person and animacy in six unrelated languages, grouped in pairs. First, Kashmiri and Sahaptin show a differential object and subject marking, respectively. Second, Awtuw and Fore show evidence of animacy as a person feature, and third, Chukchi and Kolyma Yukaghir show agreement and case-marking patterns that cross the divide between direct and inverse. Section 5.5 concludes.

# 5.2 Case studies: Inverse agreement and global case splits

In this section, I will present data from two languages that show global case splits on different arguments. It will become clear that the patterns of case-marking and agreement that are morphologically expressed in these languages correspond closely to the "gaps" in Hungarian object agreement discussed in Chapter 4.

### 5.2.1 Kashmiri

As briefly mentioned above, Kashmiri (Indo-Aryan, Indo-European) has a split-ergative system based on aspect: in non-perfective aspect, the alignment of a clause is nominative-accusative, while it is ergative in perfective aspect. In non-perfective aspect, the features of both the subject and the direct object determine case-marking. Recall that so-called *inverse* configurations are those in which the person of the subject is "lower" than the person of the object, e.g. a second person subject and a first person object.

In such inverse contexts in non-perfective aspect, the object has an overt case that resembles the dative. This dative appears on the direct object in all inverse configurations, whereas the direct object is unmarked in direct configurations (note again, that this alternation does not appear in perfective aspects).

In the following pairs of examples (repeated from above), the direct object in the first example is unmarked (in direct contexts), whereas it is marked in the second example (in inverse contexts). (3) illustrates this for first and second person. The case-marking on the object is shown in (3b). The second person singular pronoun *tsi* 'you.NOM' is identical in (3a) and (3b), but the case-marking on the first person object changes depending on its grammatical role. (4) shows the analogous pattern for second and third person. (5) illustrates that two third person arguments also give rise to dative on the direct object.<sup>1</sup> The distribution of this dative is summarised in Table 5.1.

a.	<i>b</i> ғ І.пом 'I ат	<i>chu-s-ath</i> t be.м.sG-1sG.sв teaching you.'	t J-2sg.obj y	sŧ vou.noi	<i>parŧna:va:n</i> м teaching
b.	ts <del>i</del> you.N 'You a	<i>chu-kh</i> юм be.м.sg-2sc are teaching me	me g.sbj I.dat † e.'	<i>par<del>i</del>na</i> teachir	n:va:n ng
				(Wa	li and Koul 1997: 155, glosses adapted)
a.	<i>ts<del>i</del> you.</i> N 'You a	<i>chi-h-an</i> юм be-2sg.sвj- are teaching hir	<i>su</i> 3sg.овј he. n.'	р NOM te	ar <del>ı</del> na:va:n eaching
b.	<i>su</i> he.nc 'He is	<i>chu-y</i> ом be.м.sg-2sg. teaching you.'	tse овј you.da	<i>par</i> н т teacł	<i>na:va:n</i> ning
				(Wa	li and Koul 1997: 155, glosses adapted)
<i>su 1</i> he s 'He	<i>vuch-i</i> see-3sg will see	<i>təmis.</i> he.dat e him.'		(Wa	li and Koul 1997: 156, glosses adapted)
	a. b. a. b. su fhe s 'He	<ul> <li>a. bi</li> <li>I.NOM</li> <li>'I am</li> <li>b. tsi</li> <li>you.N</li> <li>'You a</li> </ul> a. tsi <ul> <li>you.N</li> <li>'You a</li> </ul> b. su <ul> <li>he.NC</li> <li>'He is</li> </ul> su vuch-i <ul> <li>he see-3sg</li> <li>'He will see</li> </ul>	<ul> <li>a. <i>bi chu-s-ath</i> I.NOM be.M.SG-1SG.SB 'I am teaching you.'</li> <li>b. <i>tsi chu-kh</i> you.NOM be.M.SG-2SG 'You are teaching me</li> <li>a. <i>tsi chi-h-an</i> you.NOM be-2SG.SBJ- 'You are teaching hin</li> <li>b. <i>su chu-y</i> he.NOM be.M.SG-2SG.4 'He is teaching you.'</li> <li><i>su vuch-i təmis.</i> he see-3SG he.DAT 'He will see him'</li> </ul>	<ul> <li>a. <i>bi chu-s-ath t</i> I.NOM be.M.SG-1SG.SBJ-2SG.OBJ Y 'I am teaching you.'</li> <li>b. <i>tsi chu-kh me</i> you.NOM be.M.SG-2SG.SBJ I.DAT f 'You are teaching me.'</li> <li>a. <i>tsi chi-h-an su</i> you.NOM be-2SG.SBJ-3SG.OBJ he. 'You are teaching him.'</li> <li>b. <i>su chu-y tse</i> he.NOM be.M.SG-2SG.OBJ you.DA 'He is teaching you.'</li> <li><i>su vuch-i tamis.</i> he see-3SG he.DAT 'He will see him.'</li> </ul>	<ul> <li>a. bi chu-s-ath tsi I.NOM be.M.SG-1SG.SBJ-2SG.OBJ YOU.NO 'I am teaching you.'</li> <li>b. tsi chu-kh me parina you.NOM be.M.SG-2SG.SBJ I.DAT teachin 'You are teaching me.'</li> <li>(Wa</li> <li>a. tsi chi-h-an su p you.NOM be-2SG.SBJ-3SG.OBJ he.NOM to 'You are teaching him.'</li> <li>b. su chu-y tse pari- he.NOM be.M.SG-2SG.OBJ YOU.DAT teach 'He is teaching you.'</li> <li>(Wa</li> <li>su vuch-i tamis. he see-3SG he.DAT</li> <li>'He will see him.'</li> </ul>

Note also that the dative case assigned to the object in these examples is a structural, and not an inherent Case (Béjar and Rezac 2009). Evidence for this comes from passivisation: the dative assigned to first and second person direct objects is not retained under passivisation, as shown in (6) for second person:

<sup>&</sup>lt;sup>1</sup> Note that second person arguments are always coded on the verb, while other persons do not have to be (Wali and Koul 1997: 246).

(6)	a.	su	kariy	tse	me	hava:l+			
		he.nc 'He w	ом do.fut.2so vill hand you	OBJ you.DA	t I.dat	r handover	•		
	b.	ts <del>i</del> you.N 'You y	<i>yikh</i> юм come.гол will be hande	r.2sg.obj.pa d over to m	<i>me</i> ss I.dat e by hi	<i>hava:l</i> + r handove: m.'	<i>karn<del>i</del></i> r do.inf.ai	<i>təm'sındi d</i> BL he.GEN b	ləs' y
							(Wal	i and Koul 19	97: 208)

In contrast to the nominative on the logical object in (6b), Wali and Koul (1997: 209) point out that indirect objects retain their dative under passivisation (see also Béjar and Rezac 2009).<sup>2</sup>

EA→IA	1	2	3
1	_		
2	DAT	-	
3	DAT	DAT	DAT

 Table 5.1
 Distribution of inverse dative in Kashmiri

The previous examples and Table 5.1 show that the distribution of this pronominal dative case is similar to the distribution of subject agreement with Hungarian personal pronouns. Note, however, that in Kashmiri, the bottom-right cell of Table 5.1 indicates that  $3\rightarrow 3$  counts as inverse as well, in contrast to Hungarian.<sup>3</sup>

Obviously, Kashmiri also differs from Hungarian in that the morphosyntactic exponent that is sensitive to the  $\varphi$ -features of the external and the internal argument is case morphology, and not verb morphology. Before turning to a full analysis of this pattern, I turn to a language in which case is assigned to the external argument rather than the internal argument in inverse contexts.

<sup>&</sup>lt;sup>2</sup> Wali and Koul (1997: 209) mention that this is not true for all varieties of Kashmiri. Dative can be retained under passivisation for some speakers. See also Bhatt (2007) on Hindi for a similar pattern.

<sup>&</sup>lt;sup>3</sup> This is a more "regular" behaviour, as a single probe cannot be valued by two third person arguments. For Hungarian, I have argued that the configuration  $3 \rightarrow 3$  patterns with direct configurations because v and T can fuse (see Section 4.3.2).

# 5.2.2 Sahaptin

Sahaptin (Sahaptian; Nez Perce is the other Sahaptian language; see Rigsby and Rude 1996; Zúñiga 2006; Deal 2010; Keine 2010) is similar to Kashmiri in that it shows differential case-marking in inverse contexts. However, whereas a global case split affects the direct object in Kashmiri, Sahaptin shows what Rigsby and Rude (1996) call the "inverse ergative" and the "obviative ergative". Both are realised on the subject, with the inverse ergative appearing with first and second person objects and the obviative ergative appearing with third person objects. In this language, case-marking of the external argument depends on properties of the subject and the direct or indirect object. The following examples from Rigsby and Rude (1996) illustrate the distribution and form of this case. The crucial difference lies in the person of the object: when the direct object is second person, as in (7b), the subject bears the inverse ergative suffix *-nim*.

(7)	a.	ɨwínš i-q́ínun-a yáamaš-na. man 3.noм-see-рsт mule deer-овј 'The man saw a/the mule deer.'	(Rigsby and Rude 1996: 673)
	b.	<i>ɨwínš-nɨm=nam i-qínu-ša.</i> man-ınv.erg=2sg 3.nom-see-ıpfv 'The man sees you.'	(Rigsby and Rude 1996: 677)

The obviative ergative appears when both the subject and the object are third person and it tracks the relative pragmatic status of the two arguments (Rigsby and Rude 1996; Zúñiga 2006). Zúñiga (2006: 146) characterises the the difference between the two arguments as "high-pragmatic" (HP) and "low-pragmatic" (LP), respectively. When an LP third person subject has a HP third person object, it will bear the obviative ergative suffix. An example is shown in (8).

(8)	ɨwínš-in pá-tuxฺnana yáamaš-na.	
	man-obv.erg 3inv-shot mule deer-obj	
	'The man shot a mule deer.'	(Rigsby and Rude 1996: 676)

The obviative ergative also correlates with a change in verb morphology. In (8), the verb shows the  $p\dot{a}$ - suffix rather than *i*- as in (7). Rigsby and Rude (1996) suggest that the suffix  $p\dot{a}$ - allows tracking the reference of salient arguments in a discourse, illustrating

with the pair in (9). In (9a), the subject of the first clause remains the subject of the second, conjoined clause. According to Rigsby and Rude (1996), the subject is the topic in both clauses.

- (9) a. iwínš i-qínun-a wapaanłá-n ku k<sup>w</sup>aaná i-2ίλiyawi-ya. man 3.NOM-see-PST grizzly-OBJ and that.OBJ 3.NOM-kill-PST
   'The man saw a grizzly and he killed it.'
  - b. iwínš i-qínun-a wapaanłá-n ku pá-?íλiyawi-ya.
    man 3.NOM-see-PST grizzly-OBJ and 3.INV-kill-PST
    'The man saw a grizzly and it killed him.' (Rigsby and Rude 1996: 677)

In (9b), on the other hand, the *object* of the first clause becomes the subject of the second clause and the verb shows the  $p\dot{a}$ -suffix, indicating a reversal of the roles of the first clause: the object of the first clause is topical in the second clause.<sup>4</sup>

As these examples illustrate, Sahaptin has a rich inventory of verbal marking in addition to case-marking. Pronominal enclitics co-reference first and second person arguments (participants), while the prefixes indicate third person arguments, as well as (certain) inverse, reflexive and reciprocal forms (Rigsby and Rude 1996: 675f.). In this chapter, I will only focus on describing the case-marking pattern, and in particular the distribution of the *inverse* ergative, i.e. case-marking on the subject when the object is first or second person (see Rigsby and Rude 1996 and Zúñiga 2006: 149ff. for discussion of verb morphology in Sahaptin in more detail, including the distribution of the prefix pá-). Table 5.2 summarises the distribution of ergative marking, and Table 5.3 summarises the distribution of inverse markers.

# 5.3 Case assignment and cyclicity

Recall the issue of timing Case assignment mentioned in the introduction to this chapter. When v agrees with the direct object in Kashmiri, it should not yet assign the direct object Case, because the Case of the direct object also depends on the person of the subject. Assuming that a post-syntactic impoverishment rule determines the spellout of the case on the direct object could be an option, but the context for such a rule

<sup>&</sup>lt;sup>4</sup> Note that the pattern of verb morphology in (9) resembles so-called *switch-reference* (see Finer 1985; Keine 2013). As I am interested on the case of the external argument in this section, I will not discuss this matter here.

# 5.3 Case assignment and cyclicity

EA→IA	1	2	3
1	_		
2		-	
3	inverse ergative	inverse ergative	obviative ergative

Table 5.2Distribution of the inverse ergative with singular subjects in Sahaptin (Rigsby and<br/>Rude 1996).

EA→IA	1	2	3
1	_		
2	pá-	_	
3	<i>i</i> -	<i>i</i> -	pá-

Table 5.3Distribution of inverse verb markers for singular subjects in Sahaptin (Rigsby and<br/>Rude 1996).

refers to a large syntactic domain and gives rise to an extremely powerful notion of impoverishment. Following Keine (2010), I will take the impoverishment rules in this chapter to only apply in the domain of a single functional head, although this head might be complex. This restricts the power of such rules.

(10) illustrates this argument (DAT in (10) represents the structural Case assigned to the direct object in Kashmiri). If v assigns DAT on agreeing with the direct object, it is unclear how later stages of the derivation can delete (by impoverishment) the DAT feature from the direct object if the configuration does not require this case on the object.<sup>5</sup>

<sup>&</sup>lt;sup>5</sup> From now on, I will use simplified feature matrices in derivations, as in (10). I will not indicate person, number and gender probes separately; they will be shown as  $u\phi$ . If a head only has a person probe, it is shown as  $u\pi$ , as before. Number and gender are only shown when relevant.

5 Inverse agreement and global case splits



The Kashmiri data introduced above show that the direct object is assigned dative in inverse configurations. A first step towards an analysis is to take this into account in the derivation. In the system of cyclic Agree argued for in Chapter 4, direct and inverse configurations, respectively, can be distinguished by the sets of person features on *v*.

Recall from the previous chapter that I assume a person specification like [1], [2] or [3] to refer to a set of features like SPEAKER, PARTICIPANT and  $\pi$  (Harley and Ritter 2002; McGinnis 2005; Béjar and Rezac 2009). Subset/superset relations referencing these sets give rise to entailment relations between persons and therefore model hierarchical effects. A "higher" set of person features is a proper superset of a "lower" set of person features (cf. (6), p. 8).

(11) 
$$[1] = \begin{cases} \text{SPEAKER,} \\ \text{PARTICIPANT,} \\ \pi \end{cases} \supset [2] = \begin{cases} \text{PARTICIPANT,} \\ \pi \end{cases} \supset [3] = \{\pi\} \end{cases}$$

Assuming further that unvalued features on v correspond to these sets, v can have the values shown in Table 5.4 after agreeing with the direct object and the subject. Table 5.4 clearly shows how direct and inverse configurations can be distinguished from each other: in inverse configurations, v only has a single set of person features. This is because the features of the subject on the second cycle of Agree cannot value v as they are not a proper superset of the object's features. The cells indicated by "—" refer to reflexives that are outside the scope of the present discussion.

In direct configurations, however, it is possible for v to be valued by two arguments: the features on the external argument are a (proper) superset of the features of the object and therefore v can be valued by the subject in addition to the object. v is

EA→IA	1	2	3
1	_	v: [1, 2]	v: [1, 3]
2	v: [1]	—	v: [2, 3]
3	v: [1]	v: [2]	v: [3]

Table 5.4Distribution of person features on v

valued by two sets of person features. In addition, the superset relation between the sets makes it clear that the stronger set always corresponds to the subject: whenever there are two sets of person features on v, the subject valued v second. This follows from the fact that any set of features on a probe that remains unvalued after a cycle of Agree has an additional person feature, i.e. if a second person argument values v, only v's unvalued first person feature set will be unaffected — the feature set referring to third person will be valued by entailment. Only a stronger set of features can therefore value a probe in a second cycle of Agree.

(12) illustrates a derivation taking these entailment relations among person features into account, showing a clause involving a third person subject and a second person object (and ignoring number and subject agreement, for now).

First, in (12a), v agrees with the direct object, which values v's  $\varphi$ -features. There is no Case assignment yet. Next, in (12b) v moves to T and probes again, because its  $\varphi$ -features have not been fully valued yet.<sup>6</sup> Since the subject has a [3] person feature which cannot value v again, v can now assign Case, as shown in (12c).

<sup>&</sup>lt;sup>6</sup> Note that movement of v to T is not strictly necessary: in the derivation in (12), it serves the purpose of allowing v to enter an Agree relation with the subject by agreeing downward.

For the sake of clarity, I adopt this approach to Agree throughout this chapter, but there are alternatives, like Zeijlstra's (2012) *Upward Agree* (see Preminger 2013 for a critique) and Wurmbrand's (2014) *Reverse Agree*. Finally, Béjar and Rezac (2009) and Georgi (2012) argue that if v's features percolate upwards in vP, v can also enter an Agree relation with the argument in SpecvP without moving.

5 Inverse agreement and global case splits



#### 5.3 Case assignment and cyclicity



Since v's only value is a [2] feature (or set of features, more precisely), it will assign DAT to its direct object. The derivation in (12) leaves two questions open, however:

- 1. How is Case assignment delayed?
- 2. How is the spell-out of Case determined on an argument?

The first of these two questions gets a simple answer. I assume that the order in which a probe discharges its features and gives rise to syntactic operations is fixed but can vary across languages. For v, the relevant head in (12), the order is one in which Agree precedes (<) the assignment of CASE features: [ $\varphi < CASE$ ]. This ordering means that Agree takes place before copying a CASE feature onto the direct object. Following the logic of cyclic Agree, Case assignment only takes place once the probe can no longer enter Agree relations.

This means that if v encounters a first person object, it will instantly be fully valued and assign Case to the direct object. With second and third person objects, however, it is possible that there is a second cycle of Agree before Case assignment can take place. On this second cycle, it is determined whether the external argument can value v or not, and this information can feed Case assignment, as desired. In this chapter, I locate these processes in syntax and I will discuss this choice and provide further evidence for it in Chapter 6.

Note that this system diverges from Chomsky's (2000, 2001) analysis of Agree, on which Case assignment and agreement in  $\varphi$ -features are two aspects of a single syntactic operation happening simultaneously. I have argued that a probe can attempt to enter several Agree relations to value its  $\varphi$ -features which influence which Case it assigns to an argument. Case can therefore be delayed until a  $\varphi$ -probe is no longer probing. In Chapter 6, I will discuss further evidence for dissociating Case and agreement from each other.

I now discuss how person features determine Case assignment.

# 5.3.1 CASE features, Case assignment and impoverishment

As we have seen, the direct object in Kashmiri can appear in nominative or in a form that resembles the dative of an indirect object (but behaves differently in syntax with respect to passivisation as shown in (6)). Assuming, as seems reasonable, that the spellout of direct object is determined by which features it has, there must be a way for v to determine which CASE feature to assign to the direct object.

One way of modelling this is to assume that CASE features are not privative, but complex (as I have been assuming for person features), consisting of sub-features like  $[\pm \text{subj}]$ ,  $[\pm \text{obj}]$  and  $[\pm \text{obl}]$ , for "subject", "object" and "oblique", respectively (see, e.g., Bierwisch 1967; Jakobson 1971 [1936]; Wiese 1999; Müller 2002; McFadden 2004; Müller 2004b; Keine and Müller 2008; Caha 2009; Keine 2010 and Chapter 6). On this perspective, v does not merely assign a single CASE feature, but a set of such features.

I will follow Müller (2005, 2006, 2007), Keine and Müller (2008) and Keine (2010) in assuming that these sets of CASE features can be modified during the syntactic derivation. Rather than assuming that impoverishment (cf. Chapter 1) can only apply after syntax, these authors suggest that feature structures can be subject to impoverishment as soon as the input conditions of an impoverishment rule are met. Figures 5.1 and 5.2 illustrate this.

Keine (2010) takes advantage of the model in Figure 5.2 by assuming that the result of an Agree operation, i.e. a feature matrix, can be subject to impoverishment rules *right away*. Impoverishment does not have to wait until the end of the syntactic de-



Figure 5.1 Standard view of the order of syntax and morphology (Keine 2010: 1)



Figure 5.2 Keine's proposed order of syntax and morphology (Keine 2010: 2)

rivation, but it can be triggered by certain configurations of features on a head during the derivation.

To illustrate this, consider a transitive clause with a first person subject and a second person object, such as the Kashmiri example in (3a), repeated as (13).

(13) bɨ chu-s-ath tsɨ parɨna:va:n
 I.NOM be.м.sG-1sG.sBJ-2sG.OBJ you.NOM teaching
 'I am teaching you.'

(Wali and Koul 1997: 155, glosses adapted)

I assume that NOM and DAT correspond to the features in (14). I treat non-dative object case as NOM here, as in the glosses above. (15) shows the relevant vocabulary insertion rules for Kashmiri.

(14) NOM = 
$$\begin{bmatrix} \\ \\ \end{bmatrix}$$
 DAT =  $\begin{bmatrix} +obl \end{bmatrix}$ 

(15) Vocabulary insertion rules  
a. 
$$\begin{bmatrix} \\ +obl \end{bmatrix} \leftrightarrow bt$$
 'I.NOM',  $tst$  'you.sg.NOM',  $su$  'he.NOM'  
b.  $\begin{bmatrix} +obl \end{bmatrix} \leftrightarrow me$  'I.DAT',  $tse$  'you.sg.DAT',  $tomis$  'he.DAT'

Finally, the impoverishment rule in (16) applies if v is valued by two sets of person features, shown as  $[\alpha, \beta]$ , and deletes the feature [+obl].

(16) 
$$[+obl] \rightarrow \emptyset / \nu = [\alpha, \beta]$$

If impoverishment can apply as part of the syntactic derivation, we can derive (13) as shown in (17) (following the earlier steps of the derivation as seen in (12)). The crucial point is step  $\bigcirc$ : the impoverishment rule in (16) deletes [+obl] if *v* is valued by sets of person features from two arguments. *v* therefore assigns an empty set of CASE features to the direct object: this is spelled out as nominative, according to (16).



On this perspective, v generally assigns DAT to its direct object, unless the feature [+obl] is deleted by the impoverishment rule in (16).

But why does this rule apply when v has been valued by two sets of person features in particular? An answer to this question comes from the nature of Agree, as discussed here. Following Béjar and Rezac (2009), I argued in Chapter 4 and summarised above that direct and inverse configurations can be distinguished by the content of v. (17) states that this impoverishment rule only applies in a direct configuration, i.e. when the subject's set of person features is a proper superset of the object's set of person features.<sup>7</sup>

<sup>&</sup>lt;sup>7</sup> While I do not necessarily endorse such a view, it is possible to give a functional answer to this question too. A direct configuration is more typical than an inverse one and therefore less likely to require

Since multiple valuation is only possible in such contexts, it corresponds exactly to direct configurations stated on a hierarchy like the one in (18):

(18) 1 > 2 > 3

[1, 2], [1, 3], and [2, 3] are direct, and distinguish the configurations in which the direct object in Kashmiri surfaces without case-marking from those in which the direct object bears dative (cf. also the agreement pattern with Hungarian personal pronouns and case-marking in Sahaptin). The analysis proposed here derives this split by relying on sets of person features, rather than hierarchies (see Section 5.5.1 for a brief comparison of this approach to other proposals).

In the rest of this chapter, I show how this approach successfully derives casemarking and agreement patterns in a number of unrelated languages and therefore presents a viable alternative to other approaches to global case splits.

# 5.4 Analysis

In this section, I will analyse three kinds of global case splits. First, I show how Kashmiri and Sahaptin are mirror images of each other in that the former has morphological case on the direct object in inverse configurations, whereas the latter has morphological case on the subject in the same contexts. In Section 5.4.2, I discuss the languages Awtuw and Fore, in which nominals are marked depending on the relative animacy of subject and object. I show how such a system can be captured in the same way as systems in which marking depends more directly on "person" features. Finally, in Section 5.4.3, I discuss Chukchi and Kolyma Yukaghir, which are similar in that the exponents of agreement (Chukchi) and case (Yukaghir) morphology do not adhere strictly to the distinction between direct and inverse. In Chukchi, a subset of inverse configurations shows the so-called *spurious antipassive*, whereas in Yukaghir the same case suffixes can appear in both direct and inverse contexts.

These languages are chosen because they illustrate the same kind of phenomena based on seemingly different properties, i.e. person and animacy. This serves to illustrate that these two notions can be thought of as one.

case-marking of the object. This is why the  ${\mbox{\tiny CASE}}$  features of the object are impoverished in direct configurations.

# 5.4.1 Global case splits on subject and object

I argued above that in order to derive the kind of case split exhibited by Kashmiri and Sahaptin, v only assigns Case once its  $\varphi$ -probe cannot enter any further Agree relations. For the sake of illustrating the analysis, I will first analyse both languages as having a single  $\varphi$ -probe on T and v, respectively (like Hungarian). I will return to indirect object agreement in Kashmiri in Chapter 6. For ease of exposition, derivations are split into several trees, as above.

# 5.4.1.1 Kashmiri

Starting with Kashmiri, consider again the data from the introduction to this chapter, repeated here.

(19)	a.	<i>b</i> ғ І.пом 'I am †	<i>chu-s-ath</i> be.m.sg-1sg.se teaching you.'	t J-2sg.obj y	ts <del>i</del> you.no	<i>parɨnaːva:n</i> ом teaching
	b.	<i>ts<del>i</del> you.n</i> 'You a	<i>chu-kh</i> ом be.м.sg-2sc re teaching me	me g.sbj I.dat e.'	<i>par<del>i</del>n</i> teachi	<i>a:va:n</i> ing
					(Wa	ali and Koul 1997: 155, glosses adapted)
(20)	a.	ts <del>i</del> you.n 'You a	<i>chi-h-an</i> ом be-2sg.sвj- re teaching hii	<i>su</i> 3sg.овј he n.'	í NOM 1.	<i>parŧna:va:n</i> teaching
	b.	<i>su</i> he.no 'He is	<i>chu-y</i> м be.м.sg-2sg. teaching you.'	tse овј you.da	<i>par</i> teac	<i>ana:va:n</i> Ching
					(Wa	ali and Koul 1997: 155, glosses adapted)
(21)	su he s	vuch-i see-3sg	<i>təmis.</i> he.dat			
	'He	'He will see him.'				ali and Koul 1997: 156, glosses adapted)

In order to derive all of the possible forms of personal pronoun objects in (21), the following assumptions are necessary. v has unvalued  $\varphi$ -features and can assign the set of CASE features shown in (22).

(22) 
$$\nu: \begin{bmatrix} u\phi \\ CASE \begin{bmatrix} +obl \end{bmatrix} \end{bmatrix}$$

To derive the patterns in (19) to (21), the CASE features in (23) suffice. (24) shows the vocabulary insertion rules for the data discussed here. Recall that non-dative object case is spelled out like NOM.

(23) NOM = 
$$\begin{bmatrix} \\ \end{bmatrix}$$
 DAT =  $\begin{bmatrix} +obl \end{bmatrix}$ 

(24) Vocabulary insertion rules  
a. 
$$\begin{bmatrix} & & \\ & +obl \end{bmatrix} \leftrightarrow bt$$
 'I.NOM', tst 'you.sg.NOM', su 'he.NOM'  
b.  $\begin{bmatrix} +obl \end{bmatrix} \leftrightarrow me$  'I.DAT', tse 'you.sg.DAT', tomis 'he.DAT'

Finally, a single impoverishment rule is needed to derive the distribution of DAT on direct objects:

(25) 
$$[+obl] \rightarrow \emptyset / \nu = [\alpha, \beta]$$

There are only two types of transitive derivations with personal pronoun direct objects: direct configurations, in which v is valued twice, and inverse configurations, in which it is only valued once. In direct configurations, v will be valued by the subject and the object together and will have two sets of person features. This creates the context for the rule in (25) to apply and delete [+obl] from the set of case features assigned by v. This is illustrated in (26), which shows the derivation of a clause with a first person subject and a second person object.

First, *v*'s  $\varphi$ -features are valued by the direct object in (A). Then *v* moves to a position where it enters an Agree with the subject in (B) and is valued with an additional [1] feature in (C) (cf. footnote 6 on the direction of Agree).

5 Inverse agreement and global case splits



The two sets of  $\varphi$ -features on v match the context for the impoverishment rule in (25): impoverishment applies and deletes [+obl] in (D). This results in an empty set of CASE features being assigned to the direct object and the (post-syntactic) vocabulary insertion of the nominative form, as determined by (24a) above.



Inverse configurations differ in that v will only end up with a single set of  $\varphi$ -features, as shown in (27). This will bleed the context for the impoverishment rule in (25). The

derivation in (27a) differs from the one in (26) in step  $\bigcirc$ : in (27a), the subject's [2] feature set cannot value *v*, which has already been valued as [1].



As v has a single set of  $\varphi$ -features, no impoverishment takes place, and the feature [+obl] is assigned to the direct object. Given the insertion rule in (24b), this form will be spelled out as a dative pronoun.



The derivations so far have ignored Kashmiri's split-ergativity, mentioned at the beginning of this section. They are therefore only valid for non-perfective aspects. In perfective aspects, the subject is ergative-marked and the object is in the absolutive. Without going into any detail about the structure of the ergative clause in Kashmiri, it might simply come with a different v, which assigns inherent case to its specifier rather than to the direct object. (28) shows that the verb agrees with both the subject and the object even when the former is ergative and the latter absolutive.  $v_{perf}$  might also assign Case before entering an Agree relation with the object, bleeding the possibility of impoverishing the CASE features it assigns (see also Müller 2009; Assmann *et al.* 2015).

(28) *tse vuch-u-th-as bi* you.erg saw-m.sg-2sg.sbj-1sg.obj I.Abs 'You saw me.' (Wali and Koul 1997: 156, glosses adapted)

### 5.4.1.2 Sahaptin

As briefly introduced above, Sahaptin differs from Kashmiri in that the case morphology that is restricted to inverse contexts appears on the external argument. In addition, inverse case-marking in Sahaptin only appears on third person subjects.

The fact that the global case split affects the external argument raises the question where the inverse and obviative "ergatives" come from:<sup>8</sup> are they assigned by v as a kind of inherent Case (Woolford 1997, 2006; Legate 2008, 2012; Sheehan 2014a, 2015) or assigned by T, resembling a structural Case (see Deal 2010 on Nez Perce and Sahaptin, Rezac *et al.* 2014 on Basque). Note that independently of whether ergative in Sahaptin is inherent or structural, its form is determined by the person features of the subject and the object.

Arguments for treating the ergative as structural and originating from T come from the range of thematic roles that ergative-marked subjects can bear. Consider the following examples:

<sup>&</sup>lt;sup>3</sup> Recall the discussion in Section 5.2.2 about inverse and obviative ergative: the former is sensitive to person alone, while the latter appears with third person pronouns of different pragmatic salience (Rigsby and Rude 1996; Zúñiga 2006). I will simplify here and focus on how the presence of both can be derived.

(29)	a.	x <sup>₩</sup> -ísaat-in	pá-tuyayč-a	áswani-na.	
		old.man-овv 'The old man	.ERG INV-lecture-PS	sт boy-овј.sg	
	1			171	
	b.	huli-in wind-obv.er	<i>pa-wilapx<sup>w</sup>-sa</i> g inv-blow.up-ipfv	<i>faf-ҳna.</i> v dust-овј.sg	
		'The wind is	blowing up the dus	st.'	(Rigsby and Rude 1996: 677)

The subjects of (29a,b) express different thematic roles and different levels of volitionality, yet they can both appear in the obviative ergative (see also Deal 2010: 102f. on similar examples from Nez Perce, where the ergative-marked subject lacks "characteristic properties of agents, e.g. animacy and volition").

I will thus assume that v assigns Case to the object in Sahaptin, and T assigns Case to the subject. For implementing this pattern, the following CASE features suffice:

(30) 
$$\operatorname{ERG} = \begin{bmatrix} +\operatorname{subj} \\ +\operatorname{obl} \end{bmatrix}$$
  $\operatorname{OBJ} = \begin{bmatrix} +\operatorname{obj} \end{bmatrix}$ 

(31) Vocabulary insertion rules

a.  $\begin{bmatrix} +subj, +obl \end{bmatrix} \leftrightarrow -nim (INV.ERG)$ b.  $\begin{bmatrix} +subj \end{bmatrix} \leftrightarrow -in (OBV.ERG)$ c.  $\begin{bmatrix} +obj \end{bmatrix} \leftrightarrow -na (OBJ), ina 'I.OBJ', ...$ d.  $\begin{bmatrix} \end{bmatrix} \leftrightarrow \varnothing$ 

As the ergative is assigned to the external argument by T, by assumption, the relevant impoverishment rules target the CASE features on T (rather than v, as in Kashmiri, cf. (25) above). These rules are shown in (32).

#### (32) Impoverishment rules

a.  $[+obl] \rightarrow \emptyset / T = [3]$ b.  $[+subj, +obl] \rightarrow \emptyset / T = [PART]$ 

Rule (32a) specifies that if T is valued by a [3] argument only, its [+obl] feature is deleted and the external argument is assigned [+subj] only. (32b) states that if T is only valued by PARTICIPANT arguments, it will assign an empty set of CASE features to the external argument (correctly restricting inverse or obviative ergative marking to third person subjects).

These ingredients lead to the following derivations. I will illustrate the appearance of the inverse ergative in (34), the obviative ergative in (35), and null suffixes in (36). Examples for each of these types of clauses are given in (33) ((33a) repeated from (7b), (33b) from (29b); (32c) is from Rigsby and Rude 1996: 674).

- (33) a. *iwínš-nim=nam i-q́ínu-ša.* man-INV.ERG=2SG 3.NOM-see-IPFV 'The man sees you.'
  - b. *hulí-in pá-wilapx<sup>w</sup>-ša łáłx-na.* wind-OBV.ERG INV-blow.up-IPFV dust-OBJ.SG 'The wind is blowing up the dust.'
  - c. ín=aš á-qínu-ša payúwii-na łmáma-an
     I-1sg 3.Abs-see-IPFV sick-овј old.woman-овј
     'I see the sick old woman.'

(34a) illustrates the first steps of deriving (33a).<sup>9</sup>



In (34b), after T has agreed with the subject and its  $\varphi$ -features are valued by the subject  $\bigcirc$ , it enters an Agree relation with the object  $\bigcirc$ : T ends up with the values [3, 2]. This value does not trigger any of the impoverishment rules in (32) and therefore T

<sup>&</sup>lt;sup>9</sup> Note that neither *v*-to-T movement nor immediate Case assignment to the direct object are crucial ingredients to this derivation. The existence of second-position enclitics like *=nam* '2sg' in (33a) indicates that *v* might move to a high position in the course of the derivation.

5.4 Analysis

assigns its full set of CASE features to the subject. This will be spelled out as the inverse ergative.



(35) shows the derivation of a clause giving rise to the obviative ergative on the external argument. The first steps of the derivation are essentially as in (34a), so only the remaining steps are shown. Again, T agrees with the subject and is valued [3] in  $\bigcirc$ . Agreeing with the direct object does not change this value, and thus the features on T provide the right context for the impoverishment rule in (32a). [+obl] is deleted and the subject is assigned [+subj] only, which is spelled out as the obviative ergative.



185

Finally, (36) shows the relevant steps of the derivation of (33c), with a first person subject and a third person object. T's  $\varphi$ -features are fully valued after the Agree relation with the first person subject in  $\bigcirc$  and provide the right context for the impoverishment rule in (25b) to apply in  $\bigcirc$ , deleting all the features. The result is that the subject is unmarked for case.





In Sahaptin, indirect objects also affect the appearance of the inverse ergative. In ditransitive clauses, it is the *indirect* object's  $\varphi$ -features that determine whether the subject bears inverse ergative, or not. Consider (37). The subject and the direct object are both third person. This is not a configuration that gives rise to the inverse ergative. However, the first person indirect object *ina* (in its object form) triggers the inverse ergative -*nim* on the subject. The direct object remains unmarked.

(37) *łmáma-nim=š i-ní-ya ína kpitłimá.* old.woman-INV.ERG=1SG 3.NOM-give-PST me piece of beadwork 'The old woman gave me a piece of beadwork.'

(Rigsby and Rude 1996: 677)

(37) can be modelled by the structures in (38), involving a low applicative structure, in which the indirect object is introduced in the specifier of ApplP (Pylkkänen 2008). I assume that Appl assigns Case to the direct object (which remains unmarked) in (A) and that v enters an Agree relation with the indirect object and assigns it object Case in (B). The indirect object is thus treated as the primary object in such constructions (Dryer

1986); in other words, the alignment of the direct and indirect object is *secundative* (Haspelmath 2005) (see also Chapter 6).



T, as before, enters an Agree relation with the subject and gets valued as [3]; but it will enter a second Agree relation and find the indirect object which provides the second value [1] in  $\bigcirc$  and  $\bigcirc$ . These sets of  $\varphi$ -features do not create the context for any of the impoverishment rules in (32), and therefore the subject is assigned an untouched set of CASE features in  $(\boxdot)$ , to be spelled out as the inverse ergative.



# 5.4.1.3 Interim summary

Kashmiri and Sahaptin both express the inverse configurations in case morphology, albeit not on the same argument: in Kashmiri, the direct object surfaces as dative

in roughly the same situations in which the subject surfaces as ergative in Sahaptin. As the distribution of these case-markers is very similar to the distribution of object agreement with personal pronouns in Hungarian as discussed in Chapter 4, I applied the same mechanisms to determine direct and inverse configurations: when v is valued by more than one set of  $\varphi$ -features, a derivation is direct. In Kashmiri, the direct object surfaces as dative when v has been valued by a *single* argument, i.e. in inverse contexts only.

I argued that in Sahaptin the same approach can derive the assignment of ergative case to the subject from T. Note, however, that since T agrees with the subject first, inverse configurations will result in more than one set of  $\varphi$ -features on T.

A crucial assumption throughout this chapter was that the order of operations that are carried out by a probe can be such that Agree precedes case assignment. If a probe can enter into several Agree relations before it is fully valued, Case assignment can be delayed until the head has agreed with more than one argument. The heads v and T thus determine the Case of the internal and the external argument, respectively, in a parallel fashion.<sup>10</sup>

Exploiting the role of v and T to license object and subject Case, respectively, is a possible advantage over other approaches to global case splits, where comparing the features of the subject and the object has to happen on a single functional head with two probes, one each for the subject and the object, respectively (e.g. Keine 2010; Georgi 2012; see Section 5.5.1). Assuming a restrictive mapping between functional heads and probes, as I have done here, is motivated empirically. There is a crosslinguistic tendency for languages to only exhibit object agreement if they also exhibit subject agreement (see e.g. Siewierska 2013, D'Alessandro and Roberts 2008: 488, and Chapter 6). A way of modelling this trend is to only assume  $\varphi$ -probes on functional heads if there is evidence for agreement between that head and an argument. To capture the asymmetry between subject and object agreement, I will argue that v can only have a  $\varphi$ -probe if T also has one. I will return to this question in Chapter 6.<sup>11</sup>

<sup>&</sup>lt;sup>10</sup> If the direct object has been assigned Case, it might not be accessible for an Agree relation with T any more, in contrast to the derivation in (38b). If so, an alternative approach to the impoverishment rules in (32) would be to compare the  $\varphi$ -features on T and v.

<sup>&</sup>lt;sup>11</sup>Note that this suggestion has far-reaching consequences. Implicitly, this means that languages that never exhibit object agreement, like English, lack probes on *v*. If true, this raises an obvious question: how is an English object licensed? See Chapter 6 for discussion.

In the following section, I address similar case splits that are based on animacy and extend the analysis presented so far.

# 5.4.2 Global case splits and animacy

### 5.4.2.1 Awtuw

Awtuw, a Sepik language spoken in Papua New Guinea, is discussed in Malchukov (2008) in the context of local and global case splits (data from Feldman 1986, cf. P. de Swart 2007; de Hoop and Malchukov 2008; Malchukov 2008 for other analyses). Feldman (1986: 87) writes that even though the verb is final in the language, the order of subject and object is too varied to be useful in identifying grammatical relations.

Awtuw has differential object marking and the subject of the clause is always unmarked. The set of OBJ suffixes is -re/-te/-e (Feldman 1986: 107).<sup>12</sup> These markers obligatorily appear on some direct objects, may appear on others, and always appear on indirect objects. Since personal pronoun and proper name objects, as well as indirect objects always require case-marking independently of the relative animacy of the subject and the object, the domain of differential object marking does not extend to all objects. In other words, for personal pronouns and proper names, object marking is not differential. In consequence, the global case split only affects common noun objects.

Feldman indicates that the "empathy hierarchy" in (39) determines the case-marking of the object based on the relative position of the subject and the object on the hierarchy.

(39) Pronoun > Proper name > [+human] > [+animate] > [-animate]
 (Feldman 1986: 108, labels adapted)

Feldman (1986: 110) writes that when "the object is equal to or higher than the Subject in empathy, it must take the Object suffix … When two unmarked NPs co-occur in a clause, the one that is higher on the empathy hierarchy is again obligatorily interpreted

<sup>&</sup>lt;sup>12</sup> There are further allomorphs that are not relevant for the present discussion. *-re/-te* indicate unmarked and (optionally) feminine gender, respectively, whereas *-e* is the suffix that appears on personal pronouns. The vowel quality can change due to vowel harmony.

as the Subject." Malchukov (2008) cites the following examples as indicating the global nature of the Awtuw case split:<sup>13</sup>

- (40) a. *tey tale-re yaw d-æl-i* 3.F.SG woman-OBJ pig FA-bite-PST 'The pig bit the woman.'
  - b. tey tale yaw(-re) d-æl-i
    3.F.SG woman pig(-OBJ) FA-bite-PST
    'The woman bit the pig.', \*'The pig bit the woman.'

(Feldman 1986: 110, glosses adapted)

In (40b), the default mapping of arguments to grammatical relations is mapping the argument that is higher on the empathy hierarchy onto the subject function.<sup>14</sup> This can be overridden by adding the object marker *-re* to the argument to be interpreted as the direct object as in (40a).

Awtuw case-marking therefore combines aspects of local and global case-marking. As mentioned above, for personal pronouns and proper name direct objects (as well as indirect objects), object marking is obligatory (Feldman 1986: 89). Case-marking in these contexts can therefore be seen as local: there is no need to refer to the properties of the subject to determine whether the object will be case-marked or not.

This local aspect of case-marking is illustrated in (41), for example. In (41a), the direct object rey-e 'him' is not equal or higher on the empathy hierarchy than the subject, yet it has to be marked.

- (41) a. *wan rey-e du-k-puy-ey* 1sg 3.M.SG-OBJ FA-IPFV-hit-IPFV 'I'm hitting him.'
  - b. \*wan rey du-k-puy-ey
    1sG 3.M.SG FA-IPFV-hit-IPFV
    \*'I'm hitting him.', \*'He's hitting me.' (Feldman 1986: 109, glosses adapted)

Similarly, in (42a), the proper name *Kampo* is case-marked. Again, proper names require object marking; this explains why *Kampo* cannot be the object in (42b). The case

 $<sup>^{13}</sup>$  I have adapted the glosses in this section. In addition to the regular glosses, Feldman (1986) uses FA for 'factive'.

<sup>&</sup>lt;sup>14</sup>Note that (40b) is not ungrammatical, it just cannot give rise to the interpretation in which *yaw* 'the pig' is the subject.

of the object in (41) and (42) is determined locally, the properties of the subject do not matter.

- (42) a. *rey piyren Kampo-re d-æl-i* 3.M.SG dog Kampo-OBJ FA-bite-IPFV 'The dog bit Kampo.'
  - b. rey piyren Kampo d-æl-i
    3.M.SG dog Kampo FA-bite-IPFV
    'Kampo bit the dog.', \*'The dog bit Kampo.'

(Feldman 1986: 109f. glosses adapted)

The distribution of the object case-marker is shown in Table 5.5. Shaded cells indicate that the marker is optional. These cells indicate where case-marking is global, because it is sensitive to the properties of the object as well as the subject. Note further that since inanimate definites can also show case marking, there is no cell that completely rules out case-marking.

EA→IA	1	2	3	[+human]	[+anim.]	[–anim.]
1	-овј	-овј	-OBJ	-овј	-овј	-овј
2	-OBJ	-OBJ	-OBJ	-OBJ	-OBJ	-OBJ
3	-OBJ	-OBJ	-OBJ	-OBJ	-OBJ	-OBJ
[+human]	-OBJ	-OBJ	-OBJ	-OBJ	-OBJ	-OBJ
[+animate]	-OBJ	-OBJ	-OBJ	-OBJ	-OBJ	-OBJ
[-animate]	-OBJ	-OBJ	-OBJ	-овј	-овј	-ОВЈ

 Table 5.5
 Distribution of object case-marking in Awtuw

In the languages discussed so far, animacy did not play a role in determining agreement or case morphology. In order to analyse the distribution of object case in Awtuw, I will assume that animacy is grammaticalised in this language, and expressed as a person feature. Animate third person noun phrases therefore have a set [3]: inanimate noun phrases lack person features.

Since Awtuw distinguishes humanness from animacy, there are in fact three possible types of third person in Awtuw. I will refer to these as [3<sup>H</sup>], [3] and [], corresponding

to humans, animates and inanimates, respectively. In present terms, these entities correspond to sets of features. So far, I have argued that person features grammaticalise referential properties only. This approach can easily be extended to include reference to animacy and humanness, however. Note that first and second person are always animate and human, and only third person makes a distinction between animate and inanimates (see M. Richards 2008 for motivation; for the features used in (43) see Harley and Ritter 2002; McGinnis 2005; Béjar and Rezac 2009). The relevant sets of features are the following:

(43) 
$$[1] = \begin{cases} \text{SPEAKER,} \\ \text{PARTICIPANT,} \\ \text{HUMAN,} \\ \pi \end{cases} \supset [2] = \begin{cases} \text{PARTICIPANT,} \\ \text{HUMAN,} \\ \pi \end{cases} \supset [3^H] = \begin{cases} \text{HUMAN,} \\ \pi \end{cases} \supset [3] = \{\pi\}$$

On the assumption that v agrees with direct objects bearing sets of person features, linking humanness and animacy to person features in this way captures that human objects tend to be case-marked (Feldman 1986: 110). Since case-marking is never fully ruled out, it must be possible that the verb assigns Case to the object in all configurations.

To model this, the following assumptions are necessary. First, subjects agree with the verb independently of their animacy.<sup>15</sup> I will assume that since the number of the subject, but not of the object, is represented on the verb, subject agreement in number entails the valuation of the set [3]. Second, since case-marking of human objects is obligatory for proper names and pronouns and there is tendency to case-mark definite, human common nouns, I will simplify by conflating these two categories as [3<sup>H</sup>].

Third, the verb does not actually express differences in person in verb morphology. This means that v and T do not have to be sensitive to the whole range of person features – global case-marking is mostly determined by reference to  $[3^H]$  and [3] alone. Any proper name or personal pronoun will value  $[3^H]$  by virtue of having a superset of person features of the set  $[3^H]$ : first person pronouns, for example, are [1] and therefore value  $[3^H]$ . Proper names are  $[3^H]$ . Pronouns and proper names are therefore

<sup>&</sup>lt;sup>15</sup> The majority of transitive examples in Feldman (1986) have proper names or pronouns as subjects, but there are also examples with inanimate subjects (see e.g. Feldman 1986: 104).

still accounted for. I assume v and T to be specified as follows. v only has a person ( $\pi$ ) probe, but no number probe, while T has both.

(44) 
$$v: \begin{bmatrix} u & u^{3^{H}} \\ u & u^{3} \end{bmatrix}$$
, CASE  $\begin{bmatrix} +obj \end{bmatrix}$   $T: \begin{bmatrix} u & u^{3^{H}} \\ u & u^{3} \end{bmatrix}$   $U\phi = \begin{bmatrix} u^{3^{H}} & u^{3^{H}} \\ u^{3^{H}} & u^{3^{H}} \end{bmatrix}$ 

Since any direct object can be case-marked in Awtuw, it is necessary to assume that v can assign object case to *any* kind of object. Case-marking becomes optional when the subject has a superset of the features of the object. In terms of the features used here, such a configuration occurs when the subject is human, [3<sup>H</sup>] but the object is merely animate, [3]. (45) shows such an example, repeated from (40b) above.

(45) tey tale yaw(-re) d-æl-i
3.F.SG woman pig(-овј) FA-bite-PST
'The woman bit the pig.'

(45) is derived as follows: v attempts to enter an Agree relation with the direct object. Unless it is fully valued, it will agree again before assigning object Case to the direct object. After v has entered Agree relations with as many arguments as it can, an optional impoverishment rule can delete v's [+obj] feature.

As far as the syntax is concerned, this accounts for the fact that any direct object can be case-marked. The possible values on v and T are shown in Table 5.6. The derivation of (45) is shown in (46).

EA→IA	3 <sup>H</sup>	3	[]
3 <sup>H</sup>	v: [3 <sup>H</sup> ], T: [3 <sup>H</sup> ]	v: [3 <sup>H</sup> , 3], T: [3 <sup>H</sup> ]	v: [ ], T: [3 <sup>H</sup> ]
3	v: [3 <sup>H</sup> ], T: [3]	v: [3], T: [3]	v: [], T: [3]

Table 5.6Distribution of person features in Awtuw.Darkshaded cells indicate obligatorycase-marking,lightshaded cells indicate optional case-marking.



In (46a), v first enters an Agree relation with the direct object, which values it as [3] and it gets an additional set of features [3<sup>H</sup>] from the subject. This gives rise to the context for an impoverishment rule to apply in (46b) and delete v's [+obj] feature. This rule is shown in (47) (note the similarity to the Kashmiri impoverishment rule in (25) above).

(47) **Impoverishment rule**  $[+obj] \rightarrow \emptyset / v = [\alpha, \beta]$ 

This approach successfully captures that object marking is obligatory when both arguments are merely animate, but not human, for example, as shown in (48).

(48) *piyren-re yaw di-k-æl-iy* dog-овј pig FA-IPFV-bite-IPFV 'The pig is biting the dog.'

(Feldman 1986: 110, glosses adapted)

In (48), v will only be valued by the direct object's [3] feature and therefore the context for the impoverishment rule in (47) will never arise, and Case assignment is forced.

As for inanimate arguments, contrary to Malchukov's (2008) interpretation of the data that case-marking in Awtuw is forced when two arguments are equal in features, (49) suggests that even here, case-marking might be optional:

 (49) stua tawkway urunk-urunk də-kə-kow-ey shop tobacco three-three FA-IPFV-give-IPFV
 'This shop sells cigarettes in threes/three at a time.'

(Feldman 1986: 189, glosses adapted)

To account for (49), an additional impoverishment rule is needed that can apply when v has not been valued at all. Such a rule would apply before v assigns Case and without any input from the subject. This is exactly what the approach pursued here allows: the final column in Table 5.6 is completely marked as optional. Since v's  $\varphi$ -probe assigns an empty set of CASE features if it has not found an argument to agree with, the subject's features cannot influence whether an impoverishment rule applies or not.

Apart from the optionality of applying impoverishment rules in Awtuw, the system proposed for case splits based on person in Kashmiri and Sahaptin accounts for the same kind of split based on animacy in the same way. If, as hypothesised throughout this thesis, person is a way of expressing the grammaticalisation of animacy, this is a welcome consequence.

# 5.4.2.2 Fore

Fore, a Trans-New Guinean language, is spoken in Papua New Guinea, like Awtuw. The data reported here are from Scott (1978) (analyses of the distribution of the nominal marking discussed here can be found in P. de Swart 2007; de Hoop and Malchukov 2008; Malchukov 2008; Georgi 2012).

Fore has been compared to Awtuw by these authors because the relative properties of the subject and the object seem to determine case-marking on the subject. In this

sense, it might be a mirror image of Awtuw in the same way that Sahaptin and Kashmiri mirror each other: Awtuw assigns case to the object depending on the relative animacy of the subject and the object. In Fore, similar marking can appear on the subject. Consider the data in (50).<sup>16</sup>

(50)	a.	mási wa ágaye.
		boy man he.sees.him
		'The boy sees the man.', *'The man sees the boy.'
	b.	mási wá-má agaye.
		boy man-dln he.sees.him

'The man sees the boy.'

(Scott 1978: 115)

(50a) can only mean 'the boy sees the man' because both arguments are human and in such cases the first argument is interpreted as the subject. However, if the suffix  $-m\dot{a}$  is added to the second argument, as in (50b), the second linear argument is interpreted as the subject.

Scott (1978) calls this suffix the "delineator".<sup>17</sup> While de Hoop and Malchukov (2008), Malchukov (2008) and Georgi (2012) analyse it as a subject case-marker, Scott (1978: 100f.) rather likens its function to a determiner. I will also treat it as a determiner, but the distribution of the delineator provides some evidence for either analysis.

First, it distinguishes arguments similarly to the object case-marker in Awtuw did (as in (50), for example). Second, as Scott (1978: 103) writes, the delineator must appear on inanimate subjects of transitive clauses. Again, this resembles the case-marking patterns discussed above because an inanimate subject will give rise to inverse configurations in a language where person grammaticalises animacy.

However, there are also arguments against analysing this suffix as a case-marker. First, it can appear on the subject of an intransitive, arguably an unaccusative in (51).<sup>18</sup> This is unexpected if the delineator marks that the subject is lower in animacy than the object, since the predicate in (51) has a single argument.

<sup>&</sup>lt;sup>16</sup> In the glosses in this section, DLN stands for 'delineator'.

<sup>&</sup>lt;sup>17</sup> See Höhn (in progress) for a cross-linguistic comparison of delineators and similar items.

<sup>&</sup>lt;sup>18</sup> The delineator surfaces as  $-m\dot{a}$  on animates and -wama on inanimates.

5.4 Analysis

(51) yaga:-wama kana-y-e. pig-DLN come-it-IND 'The pig comes.'

(Scott 1978: 102)

Second, the delineator can occur on the transitive *object* as well, even together with the subject, although this is "exceedingly rare" (Scott 1978: 102). (52) shows an example of this.

(52) *yaga:-wama-N a-ka-y-e.* pig-DLN-OBL it-see-he-IND 'He sees the pig.'

(Scott 1978: 102)

Scott (1978) argues that in (52), it is in fact the oblique case suffix on the object -*N* that distinguishes the subject and the object from each other, and suggests that Fore is a "pure nominative-accusative type language" (p. 102). On the object, the semantic contribution of the suffix *-wama* is to express its "agentive potentiality"; Scott (1978: 102) states that this gives rise to a reading of (52) as 'He sees the pig (doing something)'.

The function of the delineator, Scott suggests, is to turn the noun it attaches to into a *potential agent*. The term *potential agent* covers a class of nominal elements including "any proper noun representing an animate being, any personal pronoun, any inalienably-possessed kin term, or any term [with a delineator]" (Scott 1978: 105). Part of the contribution of the delineator seems to be adding common nouns to a morphosyntactic class that they do not belong to without the delineator.

Its use resembles that of a case-marker because it helps with mapping arguments onto grammatical roles. This mapping depends on several factors in Fore, with one of them being the presence of the delineator. Other factors are word order and the relative animacy of arguments. Scott (1978: 114) suggests that, together with word order, the scales in (53a,b) determine the basic mapping of unmarked arguments onto grammatical roles.

- (53) a. Potential Agent > Human > Animate > Inanimate
  - b. Subject > Indirect object > Direct object

I would like to suggest then that the delineator is in fact a determiner rather than a case-marker, following Scott (1978), and *contra* P. de Swart (2007), Malchukov (2008)

and Georgi (2012). Moreover, if it is a determiner that expresses the property of being a potential agent, this can be taken to be an abstract, formal counterpart of the semantic property [+human]. As such, Fore is another example of a language in which "person" features do not merely refer to first, second and third person, but where they express the grammaticalisation of certain semantic properties.

While in Hungarian this property was referentiality, in Awtuw and Fore it is humanness. In Hungarian, verb morphology is sensitive to this property on the direct object, and I have argued in much detail in Chapter 3 that this property is connected to the D position as several determiners and quantifiers in Hungarian trigger object agreement.

Fore can be seen as exhibiting a similar phenomenon: the delineator is a D-like element that is the formal expression of a semantic property related to humanness and agentivity. As such, it can be used to "add" these properties to arguments that lack them, much like a definite determiner "adds" definiteness to a noun phrase that is indefinite.

Seen in this light, the distribution of the delineator is arguably more natural than if it were a case-marker. It is not restricted to subjects or objects, and can appear on either. Proper names and pronouns, however, never take a delineator (cf. the behaviour of proper names and pronouns and the definite determiner in English). Finally, the obligatory occurrence of a delineator on an inanimate transitive subject could be the animacy-based analogue of a *definiteness effect*. Some languages, for example Malagasy, only allow definite subjects (see e.g. Keenan 2008). The restriction to animate subjects in Fore might be an analogous type of *animacy effect*, then.<sup>19</sup> A inanimate transitive subject can be made to conform to the restriction by adding a delineator to it. Again, this resembles the contribution of a definite determiner. This view of the delineator provides further evidence for the idea that D expresses person which can grammaticalise different referential and semantic properties across languages (Longobardi 2008; M. Richards 2008; cf. also Höhn 2015).

Summing up, the distribution of the delineator  $-m\dot{a}/-wama$  in Fore resembles that of an ergative case-marker like the one in Sahaptin to some degree, seemingly suggesting that there is a global case split in Fore that mirrors the one in Awtuw. However, the fact that this suffix also appears on intransitive subjects as well as transitive objects casts doubt on its analysis as an ergative marker.

<sup>&</sup>lt;sup>19</sup> P. de Swart (2007: 79) mentions that Jacaltec, Japanese and Lakhota only allow animate subjects.
I have followed Scott (1978) in suggesting instead that it is a determiner and I have sketched how this is compatible on the perspective taken in this thesis: person features grammaticalise semantic properties. In Fore, the formal property coded by the delineator corresponds to humanness or agentivity but it is expressed as a determiner, and not in case morphology (as in Awtuw) or on the verb (as in Hungarian).

## 5.4.3 Crossing the distinction between direct and inverse

In this section, I want to discuss two further languages that have case and agreement systems resembling the ones discussed so far. First, I will turn to Chukchi, discussing data from Bobaljik and Branigan (2006). In the variety of Chukchi they study, some verbs appear in a form they call the "spurious antipassive", i.e. the verb shows an antipassive marker even though it is transitive. This spurious antipassive is triggered in a subset of inverse configurations.

In the second language I discuss in this section, Kolyma Yukaghir, there are several allomorphs of accusative that are assigned depending on the person features of both the subject and the direct object. What distinguishes Kolyma Yukaghir from the languages discussed so far, however, is that these allomorphs cut across the direct/inverse divide.

## 5.4.3.1 Chukchi

Chukchi has an inverse pattern in agreement (Comrie 1980; Bobaljik and Branigan 2006). Somewhat similarly to Hungarian, certain inverse configurations in this language give rise to what Bobaljik and Branigan (2006) call the "spurious antipassive" (SAP). Verb forms in the SAP look like antipassives, as indicated by the infix *-ine-*, but they are semantically transitive and not antipassive. An example is given in (54b).

(54) a. *əryə-nan yəm ne-4?u-yəm* 3PL.ERG me.ABS 3.SBJ-see-1SG.OBJ 'They saw me.'

(Skorik 1977: 45, cited in Bobaljik and Branigan 2006: 48)

b. *ə-nan yəm Ø-ine-ł?u-y?i* he-ERG me.ABS 3SG.SBJ-**ANTIP**-see-3SG.SBJ 'He saw me.'

(Skorik 1977: 44, cited in Bobaljik and Branigan 2006: 49)

Note that both sentences in (54) have a third person subject and a first person singular object, yet only (54b) shows the spurious antipassive. Bobaljik and Branigan (2006) write that not all inverse configurations trigger the SAP, that it does not appear in all tenses and that exponents differ across dialects. Table 5.7 shows the configurations in which the SAP appears in the variety discussed by Bobaljik and Branigan (2006).

EA→IA	1	2	3
1	_		
2	SAP	_	
3	SAP (sg.)		

 Table 5.7
 Distribution of the SAP in Chukchi (Bobaljik and Branigan 2006)

Bobaljik and Branigan (2006) propose an analysis in which T licenses both the subject and the object. Certain configurations of features (the ones shown in Table 5.7) give rise to conflicting features on T which are resolved by deleting the object's features post-syntactically.

The "intransitive" nature of the SAP verb forms is explained by this deletion: the only  $\varphi$ -features available on T are those of the subject. Bobaljik and Branigan (2006) tie the appearance of the antipassive marker *-ine-* to the fact that the resulting structure resembles that of a true antipassive:

(55) a. Transitive clause (active)



(Bobaljik and Branigan 2006: 50)

b. True Antipassive clause



(Bobaljik and Branigan 2006: 50)

In a true antipassive, the marker *-ine-* is inserted because the object "remains in the domain of v" (Bobaljik and Branigan 2006: 66). The SAP arises when the object's features are deleted on T, as shown in (56). Bobaljik and Branigan (2006) interpret this

kind of structure as essentially the same as in (55), with the low position of **Obj** giving rise to SAP morphology.

(56) Spurious Antipassive



#### (Bobaljik and Branigan 2006: 51)

Bobaljik and Branigan's (2006) analysis can be restated in the terms of the analysis proposed here. I interpret their suggestion that the object's features move up to T as movement of v to T. The conflicting configurations that Bobaljik and Branigan argue are deleted can be removed by impoverishment rules targeting v and T. (57) shows a derivation that illustrates this.

In (57), the subject is third person singular, and the object is first person, a configuration that gives rise to the SAP. (57a) shows v agreeing with the object before moving up to T. For simplicity, I am ignoring number features on the object and v. v cannot agree with the the subject again, since it is fully valued.



In (57b), T agrees with the subject, has its  $\varphi$ -features valued and assigns ergative to the subject. As sisters, v and T are local enough to create the context of an impoverishment rule that deletes v's features, giving rise to the spurious antipassive. The impoverishment rule that achieves this is shown in (58b).



- (58) Impoverishment rules a.  $v \rightarrow \emptyset / v = [\alpha]$ 
  - b.  $v \rightarrow \emptyset / v = [1], T = [3, sG]$

Note that rule (58a) would give rise to the SAP in all inverse contexts: those in which v has a single set of  $\varphi$ -features. But the SAP in Chukchi is more restricted and requires more specific rules like the one in (58b). In varieties in which the SAP also appears in some direct contexts, spell-out rules can target v that has more than one set of  $\varphi$ -features, as in (58c):

# (58) **Impoverishment rules** (continued) c. $\nu \rightarrow \emptyset / \nu = [\alpha, \beta]$

The purpose of this brief discussion of Chukchi was to show that the analysis developed above for both accusative and ergative languages straightforwardly provides the means to capture the idiosyncratic distribution of the spurious antipassive in Chukchi. This phenomenon is also of interest for the discussion of Hungarian in Chapter 4, because Chukchi is another language in which (some) inverse configurations are characterised by seemingly intransitive verb forms, while retaining their transitive semantics. This connection was made by É. Kiss (2003, 2005, 2013) who suggests that these similarities might be due to a *sprachbund* between certain languages spoken or originating in parts of Siberia, including Chukchi and Hungarian. The range of languages discussed in this chapter, and the common analysis I have proposed for them, suggest that the phenomenon is more wide-spread.

## 5.4.3.2 Kolyma Yukaghir

Kolyma Yukaghir is a southern Yukaghir language spoken in Eastern Russia (Lewis *et al.* 2013), linked to the Uralic family by (Maslova 2003: 1). It has a complex set of casesuffixes whose distribution depends on the  $\varphi$ -features of the arguments. What makes Kolyma Yukaghir similar to Chukchi is that the suffixes cut across the direct/inverse distinction and there is a fair amount of syncretism in the suffixes.

Considering third person objects first, they get the accusative suffix  $-gele^{20}$  when they are pronouns, proper names, possessive or definite noun phrases (Maslova 2003: 93). Indefinite noun phrases take instrumental case, -le (Maslova 2003: 104).

This only holds when the subject is third person, however. When the subject is first or second person (i.e. [PARTICIPANT]), third person objects are zero-marked, while first

<sup>&</sup>lt;sup>20</sup> Also -kele or -jle.

and second person objects bear what Maslova calls "pronominal accusative" case, *-ul* (Maslova 2003: 94f.). The distribution of case suffixes is shown in Table 5.8.

EA→IA	1	2	3	[]
1		-11		-0
2		<i>at</i>		Ø
3		-gele		-le

Table 5.8Objective case suffixes in Yukaghir (Maslova 2003, cf. also Keine 2010: 146).

As Table 5.8 shows, the suffix *-ul* cannot be analysed as a suffix that appears in inverse contexts only, but it is restricted to configurations involving [PARTICIPANT] arguments.

The suffix *-gele* is more "regular": it only appears in inverse contexts. Third person objects are not case-marked when the external argument is first or second person, and indefinite objects have the suffix *-le* when the external argument is *not* first or second person. Note that this last characteristic is similar to Hungarian where object agreement differs from subject agreement in that it divides "third" person into two subclasses. The following examples illustrate the four suffixes.<sup>21</sup>

(59)	a.	<i>met tet-ul kudede-t.</i> I you-Acc kill-FuT(TR.1sG) 'I will kill you.'	
	b.	<i>met-ul amde-l-get polde-mek.</i> me-ACC die-PFV-ANR-ABL save-TR.2SG 'You have saved me from death.'	(Maslova 2003: 95)
	c.	<i>tet kimnī met-kele kudede-m.</i> your whip me-асс kill-тк.3sg 'Your whip has killed me.'	(Maslova 2003: 93)
	d.	<i>čolhoro-le tudel šinel'-e ningō ik-čī-m.</i> hare-1Ns he snare-1Ns many get.caught-CAUS.ITER- <sup>7</sup>	rr.3sg
		'He caught lots of hares with his snare.'	(Maslova 2003: 104)

<sup>&</sup>lt;sup>21</sup> The special glosses used in this section are ITER "iterative" and ANR "action nominaliser".

e. *met mēmē iŋī.* I bear be.afraid(TR.1SG) 'I am afraid of the bear.'

(Maslova 2003: 89)

(59a,b) show that first and second person pronouns get the same suffix when the subject is also a participant. (59b,c) show the alternation in case marking on a first person pronoun object, depending on the person of the subject. (59d) shows an indefinite noun phrase that surfaces with the instrumental suffix, and (59e) shows a bare noun object, unmarked because the subject is first person.

Kolyma Yukaghir poses a problem for the analysis of global case splits and inverse agreement that I have proposed in this chapter. This problem is the timing of Case assignment. Even with a first person object, the case-marking on the object depends on the person of the subject. This was not the case in Kashmiri, where objects in inverse contexts are assigned dative across the board, or in Awtuw, where case-marking on objects is also consistent. In these languages, a first person object would be assigned Case by v after a first cycle of Agree since v cannot be valued any further.

A crucial difference between Kashmiri, Sahaptin and Awtuw, on the one hand, and Yukaghir, on the other hand, is that object case-markers in the latter cut across the inverse/direct divide. This suggests that a different analysis is called for.

One way of characterising Yukaghir is that T plays a bigger role in Yukaghir. Keine (2010: 145ff.) suggests that in this language, T can assign Case to two arguments, so v is not involved in licensing object Case at all. Rather, T agrees with the subject, and possibly feeds impoverishment rules applying to the set of CASE features that T assigns to the object. This set of CASE features can be impoverished again on the direct object itself.

I will adopt Keine's suggestion that T assigns two Cases in Kolyma Yukaghir, but I will suggest that there is some evidence for agreement between v and the object. Notice that the verbs in (59) have suffixes from what Maslova (2003) calls the transitive paradigm, glossed as TR. Maslova (2003: 141) relates the exponent -m(e)-, seen in (59b,c,d), to transitivity, contrasting with -j(e)- in the corresponding intransitive paradigm. Since transitivity is expressed on the verb, it is possible that there is a kind of object agreement, the expression of which does not reflect  $\varphi$ -features, however.

To derive the case-marking pattern, I will assume that this exponent of transitivity is in fact a  $\varphi$ -probe that can agree with the direct object but cannot assign it Case. When

v moves up to T, impoverishment can apply to the features on v and T and determine the Case assigned to the object. An advantage of this approach could lie in providing a context for a second set of CASE features on T to become active. Rather than suggesting that there are two types of T head, one transitive and one intransitive, there is a single one. While it can assign two Cases, it only does this when v adjoins to it.

(60) shows the CASE features (adapted from Keine 2010: 148) and (61) shows the relevant impoverishment rules for the data discussed here.

		+subj		-subj	
(60)	NOM:	–obj	ACC:	+obj	
		-obl		-obl	

(61) Impoverishment rules

a.	[–subj]	$\rightarrow \emptyset$	/ v = [+part],	T = [+PART]
b.	[–subj], [–obl]	$\rightarrow \emptyset$	/ v = [ ],	T = [-part]
c.	[α]	$\rightarrow \emptyset$	/ v = [-PART],	T = [+part]

Finally, (62) shows the vocabulary insertion rules that determine the spell-out of the CASE features.

## (62) Vocabulary insertion rules

a. 
$$-gele \leftrightarrow \begin{bmatrix} -\operatorname{subj} \\ +\operatorname{obj} \\ -\operatorname{obl} \end{bmatrix}$$
 b.  $-ul \leftrightarrow \begin{bmatrix} +\operatorname{obj} \\ -\operatorname{obl} \end{bmatrix}$  c.  $-le \leftrightarrow \begin{bmatrix} +\operatorname{obj} \end{bmatrix}$  d.  $-\emptyset \leftrightarrow \begin{bmatrix} \end{bmatrix}$ 

(63) and (64) illustrate two derivations with first person objects (cf. (59b) and (59c), repeated here). The derivations illustrate how the difference in the person of the subject influences the case-marker on the object.

- (59) b. *met-ul amde-l-get polde-mek.* me-ACC die-PFV-ANR-ABL save-TR.2SG 'You have saved me from death.'
  - c. *tet kimnī met-kele kudede-m.* your whip me-ACC kill-TR.3SG 'Your whip has killed me.'

(63a) shows a Agree relation between v, which by assumption agrees but cannot assign accusative Case. After agreement, v moves to T.



*v* is fully valued after agreeing with a first person object and does not probe again. In (63b), T probes and agrees with the subject in  $\bigcirc$  but does not assign it Case yet. Since *v* has moved to T, impoverishment can apply in  $\bigcirc$ . The presence of two PARTICIPANT features on *v* and T triggers the rule in (61a), deleting [-subj] from T's (second) set of CASE features. The reduced set of features is assigned to the direct object in  $\bigcirc$ , spelled out as *-ul*, as shown in (62b).



The derivation of (59c) only differs from (63) in the operations shown in (63b) - (64) illustrates. Here, the subject is a third person noun phrase. The features on v and T therefore do not provide the right context for any of the impoverishment rules in (61) to apply. T assigns a full set of CASE features to the direct object in (E). This is spelled out as *-gele*, as shown in (62a).



Other derivations work analogously. Note that like in Hungarian, the relation between v and third person objects distinguishes between definite and indefinite direct objects. Again, I assume that Yukaghir person features grammaticalise definiteness. In Table 5.8, this difference is represented as [3] and [], respectively. Moreover, the subject agrees with the verb in number, not just person. As above, I take this asymmetry lead to T's  $\varphi$ -features being valued as [3] when agreeing with the subject, independently of the subject's definiteness.

To summarise, the distribution of accusative in Kolyma Yukaghir cannot be derived in the same way as suggested for Kashmiri and Awtuw in previous sections. The difference lies in the fact that the same person on the object can give rise to different accusative suffixes, depending on the person of the subject. Crucially, while a first person object in Kashmiri is always spelled out as dative, the case of first person objects in Yukaghir still depends on the person of the subject.

This means that if accusative Case comes from v, it has to delay Case assignment until *after* T has agreed with the subject. It is not clear how this could be implemented as a property of v. I do not, however, take this to be a disadvantage of the analysis I proposed in the previous sections. The reason is that the distribution of case-marking in Kolyma Yukaghir does not correlate with direct and inverse configurations. To a certain degree, then, this language shows a slightly different phenomenon than Kashmiri and Awtuw.

I have, following Keine (2010), suggested that T can assign two Cases in Yukaghir. My analysis diverges from Keine's, however, in the role of v. I have taken the existence of a transitive verb paradigm as suggestive of v agreeing with the direct object and moving to T where it makes Case assignment to the direct object possible. This analysis exploits the different distribution of probes on functional heads in Kolyma Yukaghir to derive a distribution of case-suffixes that differs from the one found in Awtuw and Kashmiri.

In the following, concluding section, I will come back to questions regarding the distribution of  $\varphi$ -probes and CASE features on the heads T and *v*.

## 5.5 Discussion and conclusions

Before concluding this chapter, I will discuss the necessary ingredients of analyses of with global case splits and relate my analysis to other proposals in the literature.

## 5.5.1 Other approaches to global case splits

A first ingredient relates to the **locality of**  $\varphi$ -features. Since global case splits are defined by being sensitive to the properties of the subject and the object, there must be a way of comparing these features before determining which case to assign. This is where approaches to global case splits vary strongly.<sup>22</sup>

Deal (2010), discussing Nez Perce, but extending her analysis to Sahaptin, suggests that the locus of comparison of the features of the subject and the object is the subject itself. She argues that the object's  $\varphi$ -features are transmitted to the subject by v; it therefore has two  $\varphi$ -features. In Sahaptin, the inverse ergative is only assigned when the subject's own  $\varphi$ -features are third person, and the object's  $\varphi$ -features are first or second person.

<sup>&</sup>lt;sup>22</sup> I am focusing on analyses in a framework comparable to the one in this chapter. Other approaches include P. de Swart (2007), de Hoop and Malchukov (2008) and Malchukov (2008). I am focusing on the syntactic derivation of case splits here and I have referenced the analyses mentioned when discussing the specific languages in this chapter.

While this provides the correct results, it seems to be difficult to extend this analysis to similar splits on the object (as in Kashmiri or Awtuw) since there does not seem to be a way of transmitting the subject's  $\varphi$ -features to the object, in analogy to Deal's (2010) suggestion. In addition, the motivation to assigning inverse ergative in inverse situations does not follow from her analysis — in principle, any combination of features could give rise to case-marking. While this is not necessarily a problem, it might be missing a generalisation about direct and inverse configurations.

Keine (2010) and Georgi (2012) assume that the locality of the  $\varphi$ -features of the subject and the object is simply due to  $\varphi$ -probes on a single functional head that agree with both arguments in turn. Georgi (2012) suggests that v carries language-specific probes that are co-indexed with the subject and the object. These probes have certain expectations about what kinds of arguments they encounter, e.g. a third person direct object, and they are specified as such: if v encounters a second person argument, it will need an extra feature to agree with it. Georgi suggests that this extra feature comes from the probe that is co-indexed with the subject: it is *marauded*, so that v can agree with the object. She suggests that Case assignment to the subject or the object indicates that *Maraudage* has happened. While very powerful, this approach relies on very specific unvalued features on probes for a given language and raises questions about the role of T in determining agreement.

Keine (2010) also suggests that a single functional head can carry several probes that are co-indexed with their relevant arguments. On his approach, though, the probes are equal, and once they have been valued, impoverishment rules can modify the sets of CASE features on the probes. This approach also raises questions about the role of functional heads in Case assignment and Agree: if T agrees with both the subject and the object, and assigns both arguments Case, what is the role of v?

In this chapter, I have argued that it is possible to maintain the perspective that T agrees with and assigns Case to the subject, and v agrees with and assigns Case to the object. The  $\varphi$ -features of both arguments can be represented on a single head, however, because I have assumed that a probe can agree as long as it is not fully valued.

An advantage of this approach is that direct and inverse configurations fall out as natural classes: in direct configurations, v will be valued by both arguments and have two values, and in inverse configurations, it will only have a single value.

A second aspect of describing global case splits relates to the **evaluation of the**  $\varphi$ -features. How do the  $\varphi$ -features on a functional head determine Case assignment or verb morphology? In Georgi's (2012) approach, *Maraudage* is restricted to contexts in which the object's features are a superset of the subject's features — in a similar way to the present proposal, but arguably involving more machinery. Keine (2010) invokes person and animacy hierarchies to determine which features are targeted by impoverishment rules. This approach to hierarchies and how they motivate constraints against atypical configurations can also be found in Aissen (1999, 2003), P. de Swart (2007), de Hoop and Malchukov (2008), Keine and Müller (2008) and Malchukov (2008).

These authors adopt Optimality Theory (OT, Prince and Smolensky 2004) to derive the relevant splits from markedness hierarchies. In brief, the system works as follows: markedness hierarchies are assumed to be theoretical primitives, valid for all languages. These hierarchies give rise to constraint rankings which indicate that certain syntactic configurations are typologically marked: first person *objects*, for example, are more marked than first person *subjects* – constraints against such configurations are ranked higher than constraints against less marked configurations. Keine (2010), in particular, uses constraint hierarchies to derive impoverishment rules that determine which Case is assigned to an argument.

As I briefly pointed out in Chapter 1, this type of approach has been met with some criticism. Some authors argue that typological hierarchies have a questionable role in syntax (Jelinek and Carnie 2003; Carnie 2005a,b; M. Richards 2008). Haspelmath (2008) also suggests that locating such hierarchies in UG is not the right approach since their effects can be derived from functional considerations.

My position is different. I acknowledge the existence of hierarchical effects but I have implemented these effects as a consequence of several factors. First, person features are sets of features such that a set corresponding to first person is a proper superset of a set corresponding to second person.<sup>23</sup> Second, subjects are higher in the

(Valentine 2001: 270)

<sup>&</sup>lt;sup>23</sup> But note that this does not have to be the case. In some languages, second person is said to be more prominent than first, e.g. in Ojibwe (or Nishnaabemwin; Valentine 2001; Lochbihler 2008; Béjar and Rezac 2009; Lochbihler 2012). In Ojibwe, 1 > 2 counts as inverse, rather than direct, as shown in (i).

<sup>(</sup>i) g-waabm-in 2-see-INV(local) 'I see you.'

structure of the clause than objects. Third, v agrees with the direct object first, and whether it can also Agree with the subject depends on the sets of features of the two arguments. These assumptions derive the contexts in which impoverishment can apply in a language like Kashmiri: only when v is valued by the sets of person features of two arguments. Therefore, hierarchical effects are a "built-in" feature of the analysis of person.

Finally, a third aspect of analysing global case splits is the **timing of Case assignment**: Case must not be assigned too early, i.e. before the  $\varphi$ -features of both arguments have been evaluated. For approaches involving a single head that can assign several Cases, this is simply a matter of specifying the order of operations on that head (Keine 2010; Georgi 2012). Deal (2010) argues that the presence of  $\varphi$ -features of both the subject and the object on the subject determine the spell-out of its Case. In this sense, Case assignment is not delayed, but it is merely the spell-out of Case that is influenced by features. As mentioned above, it would be difficult to adapt this approach to object case-marking.<sup>24</sup>

In the present proposal, this question is solved by the possibility of probes agreeing as long as their  $\varphi$ -features are not fully valued. If a probe comes with an ordering statement like [ $\varphi < CASE$ ], meaning that Agree has to take place before the probe assigns Case, Case assignment is simply delayed as long as the probe can still enter Agree relations. This makes it possible to assign different Cases depending on whether the subject and object are in a direct or an inverse configuration: if a probe fails to agree with a second argument, the configuration is inverse. This solution maintains that T and v are separate heads that have  $\varphi$ -probes and can assign Case when case-marking is distributed along the lines of direct and inverse configurations.

This can be modelled by assigning the sets corresponding to [1] and [2] different contents than in the languages discussed in this chapter: if [2] includes ADDRESSEE and [1] lacks SPEAKER, the entailment relations among these sets change. This makes it possible for children to learn different "hierarchies" like 1 > 2 vs. 2 > 1 by positing different sets of features. A universal, UG-internal hierarchy 1 > 2 > 3 might be less flexible. See Harley and Ritter (2002), McGinnis (2005) and Lochbihler (2008, 2012) for further discussion.

 $<sup>^{24}</sup>$  Again, a detailed comparison to Silverstein (1976), P. de Swart (2007), de Hoop and Malchukov (2008) and Malchukov (2008) would take us too far afield — obviously, the question of timing Case assignment also arises in these approaches. Since these authors do not assume as tight a connection between agreement and case in their formalism as is done here, I again focus on analyses that are comparable to the one presented here. The possible functions of case-marking, e.g. ease of identifying and distinguishing arguments, that these authors discuss may answer the question of *why* we find global case splits; the present chapter implements an answer to the question of *how* such splits are realised.

The present approach combines these three aspects of global case splits, i.e. the locality of  $\varphi$ -features, their comparison and their evaluation, in a simple way: a single head can agree more than once; it can be read off its  $\varphi$ -features whether the subject and object are in a direct or an inverse configuration; and, finally, Case assignment follows these two steps as specified by an ordering statement on a functional head.

## 5.5.2 Functional heads and probes

An advantage of maintaining the separate roles of T and v in determining inverse agreement and global case splits is the potential to explain a cross-linguistic tendency about the relation of subject agreement and object agreement.

Consider languages in which a verb agrees with a single argument in a transitive clause. According to Siewierska (2013), in a large majority of languages, this argument is the subject.<sup>25</sup> One way of implementing this generalisation in the present framework is to make the presence of  $\varphi$ -features on v dependent on the presence of  $\varphi$ -features on T. This means that no languages would only have  $\varphi$ -probes on v but not on T. These assumptions restrict the number of possible grammars in line with what we find empirically: T can have a single  $\varphi$ -probe while v lacks one, or two  $\varphi$ -probes can be distributed between T and v.<sup>26</sup> I discuss the distribution of  $\varphi$ -probes across functional heads in more detail in Chapter 6.

This kind of dependency between the feature content of different functional heads is in line with the approach to parametric variation suggested by Roberts (2012) (see

<sup>&</sup>lt;sup>25</sup> Some of the exceptions are ergative languages. Here, "object" agreement can arise as the sole agreement relation in a clause if the subject's case-marking blocks agreement with the verb. This is the case in Hindi, for example. The finite verb agrees with the highest unmarked argument. Because of Hindi's split-ergative system based on aspect, in imperfective aspects, the verb agrees with the subject. In perfective aspects, which are ergative, the verb agrees with the object *if* it is also unmarked.

It follows that in a language that is consistently ergative, the verb might agree with the object because the subject cannot Agree. In the technical terms that are relevant for the present discussion, this kind of language could still have a single  $\varphi$ -probe on T, just like a language that lacks object agreement altogether. That  $\varphi$ -probe will simply agree with the most local accessible argument, which would be the internal argument in a transitive clause, if the subject is not accessible because of its Case marking. See Chapter 6 for further discussion.

<sup>&</sup>lt;sup>26</sup> The analysis I proposed for Yukaghir is compatible with this suggestion; the peculiarity of this language compared to the other languages discussed in this chapter lies in the distribution of accusative case. I have modelled this by suggesting that T can assign two Cases, but v nevertheless agrees with the object. Since the distribution of case in Yukaghir differs from the other languages, it motivates this different analysis.

also Sheehan 2014b, 2015 on variation in alignment across languages), to which I will return in Chapter 7.

Restricting the variation in the distribution of probes across functional heads, while maintaining an empirically adequate analysis of distinct agreement and case assignment patterns is a welcome consequence of the analysis in this chapter.

## 5.5.3 Conclusions

The aim of this chapter was to provide further evidence for the hypotheses made in the previous chapters. First, in Chapter 3, I adopted M. Richards's (2008) proposal that person features do not only encode "person" but can encode semantic properties like definiteness or animacy as formal features that affect the syntax. After having seen evidence for the role of referential properties as well as person in the agreement system of Hungarian in Chapter 4, in this chapter I showed how animacy can lead to the same kinds of morphosyntactic phenomena in case-marking that "pure" person does, e.g. in the analysis of Awtuw compared to Sahaptin and Kashmiri.

Both reference and animacy give rise to similar kinds of splits in case-marking and agreement across languages, which is exactly what we expect if they are expressed as person features. The case of Fore arguably also provides an interesting argument in favour of locating these features on D: its "delineator" seems to serve the purpose of an animate determiner, similar to a definite determiner in other languages.

Second, this chapter provided further evidence for the specific analysis of representing person as sets of person features and the implementation of Agree. I adopted Béjar and Rezac's (2009) suggestion that a probe can agree more than once, but I embedded this in a system that does not invoke *repair strategies* in the way they suggest.

Rather, once v cannot agree any further, T agrees and assigns Case; the two functional heads simply show their "natural" behaviour. This system, worked out in detail for Hungarian in Chapter 4, extends straightforwardly to a number of unrelated languages discussed in this chapter.

In the next chapter, I discuss the interaction of agreement and case in syntax and embed the analysis developed so far in the architecture of the grammar.

## 6.1 Introduction

In this chapter, I will embed the languages surveyed in the previous chapters in a wider typological context and establish parameters that model the variation of case and agreement across languages. In the previous chapter, I discussed the interaction of object agreement and subject agreement and their influence on case-marking. In this chapter, I focus on the influence of case-marking on agreement.

The main claims I am making are the following. First, I argue that the inventory of  $\varphi$ -probes on functional heads differs across languages, but that it is restricted: in a language with a single  $\varphi$ -probe, this probe is on T. When a language has more  $\varphi$ -probes, the additional probes will be located on lower functional heads. As we have seen in Chapter 5, if a functional head can assign Case and has a  $\varphi$ -probe, the order in which Case assignment and agreement happens can vary and this ordering determines whether a language has global case splits or not.

Second, I will show that cross-linguistic generalisations about the distribution of subject and object agreement and the distribution of case and agreement alignment across languages proposed by Bobaljik (2008) can be implemented in the framework discussed here. Reformulating Bobaljik's (2008) generalisations is empirically motivated: while Bobaljik suggests that morphological case determines agreement, the data discussed in Chapter 5 and some of the data discussed below show that it is necessary to take abstract Case into account to determine certain agreement patterns (Legate 2008).

The system proposed here can account for Case and agreement influencing each other in either order: Case can determine agreement, and agreement can determine Case.

This chapter is structured as follows. I will start by discussing the distribution of case-marking and agreement in Hindi, Nepali and Marathi, languages with a single  $\varphi$ -probe, in Section 6.2. These three languages differ in interesting details with respect to which types of noun phrases can agree with the verb: in Hindi and Marathi, only arguments without case-marking can agree with the verb, while in Nepali both ergative and unmarked arguments trigger agreement.

Based on these agreement patterns, I will discuss Bobaljik's (2008) proposal that agreement is determined by morphological case and his generalisations about the distribution of ergative-absolutive and nominative-accusative alignment systems in both case and agreement. In brief, Bobaljik's approach correctly predicts a typological gap, namely the lack of languages which have nominative-accusative case alignment and ergative-absolutive agreement alignment (I illustrate these terms in Section 6.2.2).

I will then review one of Legate's (2008) arguments against this proposal: Legate argues that the distribution of morphological case and agreement in Marathi provides a counterexample to Bobaljik's analysis. She highlights the need for abstract Case to explain the agreement pattern in this language. I concur, and I follow Legate (2008) and Keine (2010) in assuming that abstract Case can influence agreement and that both Case and agreement are determined in the syntax.

In Section 6.3, I discuss Caha's (2009) approach to Case as sets of features that are ordered by subset/superset relations. I show that by treating different Cases as distinct sets of features, Bobaljik's (2008) generalisations about the distribution of case and agreement alignment follow naturally. Discussing languages with two  $\varphi$ -probes, I further extend the generalisation about the lack of languages with nominative case alignment and ergative agreement alignment to ditransitives and I show that an analogous typological gap exists in this domain: I suggest that there are no languages which show secundative case alignment and indirective agreement alignment.

Section 6.4 concludes.

## 6.2 Case, agreement and Bobaljik's generalisations

The agreement patterns in Hindi, Nepali and Marathi differ in interesting details. All three languages are split-ergative: in the perfective aspect, the subject is ergative, and it

is unmarked in other aspects. Hindi and Marathi do not allow agreement with ergative noun phrases, but Nepali does. I will start by discussing Hindi.

## 6.2.1 Hindi and Nepali

In (1a), both the subject and the object are unmarked. The verb agrees with the masculine subject. In the perfective clause in (1b), the subject bears ergative and the verb now agrees with the feminine direct object *kitaab*, as indicated on the verb.

(1)	a.	Rahul kitaab paṛh-taa thaa.	
		R.M book.f read-HAB.M.SG be.PST.M.SG	
		'Rahul used to read a/the book.'	
	b.	Rahul-ne kitaab paṛh-ii thii.	
		R-erg book.f read-pfv.f be.f.sg	
		'Rahul had read the book.'	(Bhatt 2005: 759)

Hindi also has differential object marking: the direct object can appear with the casemarker *-ko* when it is animate and specific. When both arguments are marked, as in (2), the verb surfaces in a default third person masculine form. Note that none of the arguments in (2) are masculine.

(2)	Mona-ne	is	kitaab-ko paṛh-aa.	
	Mona.F-ER	g this.o	BL book.F-ACC read-PFV.M.SG	
	'Mona had	read th	is book.'	(Bhatt 2005: 768)

The verb therefore agrees with the structurally highest argument that does not have morphological case (Woolford 1997; Bhatt 2005; Anand and Nevins 2006; Bobaljik 2008; Legate 2008).

The situation in Nepali is different (Bickel and Yādava 2000; Bobaljik 2008). While the language is also split-ergative, it differs from Hindi in that agreement is controlled by the highest ergative or unmarked argument.

In (3a), the subject ma 'I.NOM' and the object  $patrik\bar{a}$  'newspaper' are both unmarked for case. Agreement on the verb follows the first person subject. In (3b), the subject bears ergative case and the object remains unmarked. The verb still agrees with the subject.

- 6 Case and agreement in the grammar
- (3) a. *ma* yas pasal-mā patrikā kin-ch-u. 1sg.NOM DEM.OBL store-LOC newspaper.NOM buy-NPST-1sg 'I buy the newspaper in this store.'
  - b. *maile yas pasal-mā patrikā kin-ẽ / \*kin-yo.* 1SG.ERG DEM.OBL store-LOC newspaper.NOM buy-PST.1SG buy-PST.3SG.M 'I bought the newspaper in this store.'

(Bickel and Yādava 2000: 348)

Bickel and Yādava (2000: 348) therefore suggest that, in Nepali, the verb agrees with "the highest s/A-argument" where s is the single argument of an intransitive (or a passive) and A the most agentive argument of a transitive.

The difference between Nepali and Hindi is that Nepali allows the verb to agree with an argument in ergative, whereas Hindi does not. Crucially, Nepali nevertheless allows the verb to agree with an unmarked argument as well, as shown in (3a).

## 6.2.2 Bobalijk's generalisations

Based on the data just discussed, Bobaljik (2008) suggests that agreement between and a predicate and its arguments is determined by morphological case. Referring to and building on earlier work by Edith Moravcsik, and Marantz's (1991) notion of dependent case, Bobaljik (2008: 306) suggests that the hierarchy in (4) determines which arguments a verb can agree with based on their morphological case.

(4) Unmarked Case > Dependent Case > Lexical/Oblique Case

The levels in (4) refer to the following different types of Case. First, unmarked Case refers to arguments of a verb that do not have overt case morphology. Second, "Dependent Case" refers to a type of Case assignment based on dependency relations between two arguments (Marantz 1991; Bittner and Hale 1996; McFadden 2004; see also Preminger 2014; Baker 2015). The idea behind dependent Case assignment is that two DPs in a certain syntactic domain, where DP<sub>1</sub> asymmetrically c-commands DP<sub>2</sub>, form a dependency relation. If such a dependency relation is established, one of the DPs will be marked with morphological case. In nominative-accusative languages, the lower of the two DPs will get accusative, whereas in ergative-absolutive languages, the higher one will get ergative. Since the dependency is absent in intransitives, the

single argument in such a clause will be unmarked in both nominative-accusative and ergative-absolutive languages. Both ergative and accusative are instances of *dependent Case* because their presence depends on the presence of another DP in the relevant syntactic domain.<sup>1</sup>

Finally, lexical or oblique case refers to case assigned based not on the structural properties of the clause but on lexical properties of a verb.

The hierarchy in (4) forms the basis of two striking cross-linguistic generalisations which I will refer to as Bobaljik's first and second generalisation, respectively. The first one relates to which arguments a verb in a given language can agree with and is shown in (5).

## (5) a. **Bobaljik's first generalisation**

If one type of Case on the hierarchy in (4) triggers agreement, all types of Case higher on the hierarchy will also trigger agreement.

For Hindi, the sole type of Case on a noun phrase that can give rise to agreement with the verb is unmarked Case, the highest level on (4). In Nepali, ergative arguments can also agree; according to (5a), this implies that unmarked arguments also must be able to agree. And they do, as shown in (3) above.

What I will call Bobaljik's second generalisation follows from (5a) and relates to the distribution of the *alignment* of Case and agreement. It is shown in (5b).

## (5) b. Bobaljik's second generalisation

Languages with nominative-accusative case alignment cannot have ergative-absolutive agreement alignment.

In the perfective aspect, both Hindi and Nepali show ergative-absolutive case alignment. This means that the single argument of an intransitive clause, s, and the object of a transitive clause, pattern together: they are both unmarked. The subject of a transitive clause, A, is ergative. This is shown in (6a).

McFadden (2004) and Baker (2015) assume that this syntactic domain is a *phase* (Chomsky 2000, 2001, 2008). Baker (2015) implements certain patterns of differential object marking by suggesting that only direct objects that move out of VP into the same phase as the external argument will be assigned accusative, as they are otherwise not in the relevant domain for dependent case assignment.

6 Case and agreement in the grammar



Hindi and Nepali differ in their agreement alignment, however. As we have seen, the verb in Hindi agrees with unmarked arguments, i.e. the single argument of an intransitive, or s, and the unmarked object of a transitive in the perfective aspect, P. Agreement is therefore distributed in the same way as case-marking, namely as in (6a).

Nepali is different. While its case alignment patterns with Hindi, its agreement alignment does not: the verb agrees with the subject, irrespective of its Case, i.e. s and A. This is nominative-accusative alignment, shown in (6b). A language like German shows nominative-accusative alignment in case-marking and agreement.

(5b) states that the fourth logical possibility is impossible: a language with nominative-accusative case alignment (like German) but with ergative-absolutive agreement alignment (like Hindi). In such a language, agreement would be controlled by the unmarked subject in an intransitive, but by the case-marked *object* in a transitive. This is shown in (7).

#### (7) \*NOM-ACC case, ERG-ABS agreement



The generalisation in (5b) rules this out: since unmarked arguments are able to control agreement, it is impossible for the verb to bypass the unmarked subject in a transitive clause and agree with the object.

While Bobaljik (2008) assumes that it is *morphological* case that determines agreement along the lines of (4), the generalisations in (5) are in principle independent of this specific approach: I will implement them based on abstract Case below.

The move from morphological case to abstract Case in determining agreement is motivated empirically, because there are counterexamples to Bobaljik's (2008) claim that morphological case, rather than abstract Case, determines agreement. I turn to one such counterexamples now.

## 6.2.3 Marathi

Legate (2008) argues that Bobaljik's (2008) account of agreement does not suffice to explain the cross-linguistic variation the interaction of Case and agreement gives rise to. Among other evidence in defense of the notion of abstract Case she provides, Legate's account of Marathi (and Punjabi) is directly relevant to the present discussion, because Marathi agreement patterns pose a problem for Bobaljik's (2008) analysis in terms of morphological case (see Legate 2008: 94ff. and Keine 2010: 51ff.).

At first glance, the distribution of agreement in Marathi is identical to the distribution of agreement in Hindi in that ergative arguments do not trigger agreement in either language. This is shown in (8). I gloss unmarked arguments as ABS (following Legate 2008).

(8)	a.	mulī	gāņī	mhaṇtāt.
		girl.3pl.f.Abs	sing.pst.3pl.f	
		'Girls sing so		
	b.	mulī-ne	gāņī	mhațlī.
		girl.3pl.f-erg song.3pl.n.Abs sing.pst.3pl.n		
		'The girls san	g songs.'	

(Pandharipande 1997: 284, cited by Legate 2008: 94, Keine 2010: 51)

(8a) shows that in a transitive clause with both an absolutive subject and an absolutive object, the verb agrees with the higher argument in person, number and gender. In (8a), the subject *mulī* controls agreement. In (8b), with an ergative marker on the subject, the absolutive object  $g\bar{a}n\bar{i}$  controls agreement. So far, this pattern of agreement in Nepali can be captured by the same generalisation as for Hindi above: the verb agrees with the highest unmarked argument.

There is an additional factor in Marathi that interacts with agreement and casemarking, however. In addition to being split-ergative, Marathi has another split in case-morphology, based on person. First and second person pronouns never take an

ergative suffix, even if the aspect of the clause would require it. Third person pronouns do not show this restriction and behave as expected; this is shown in (9a). Interestingly, however, agreement does not change in (9b), where the first and second person subjects surface unmarked. The verb still agrees with the object  $g\bar{a}n\bar{i}$  'songs', rather than the unmarked subject.

(9)	a.	tyā-ne /	′ ti-ne	gāņī	mhațlī.
		he-erg	she-erg	song.3pl.n.abs	sing.pst.3pl.n
		'S/he sar	ng songs.	,	

b.  $m\bar{i} / t\bar{u}$   $g\bar{a}n\bar{i}$   $mhațl\bar{i}$ . I.ABS you.ABS song.3PL.N.ABS sing.PST.3PL.N 'I / you (sg.) sang songs.'

(Pandharipande 1997: 131, cited by Keine 2010: 52)

Legate (2008) suggests that the correct generalisation for agreement in Marathi is not that it follows the highest unmarked argument, but that it follows the highest argument with structural absolutive. She argues that while they look like absolutive on the surface, the first and second person subjects in (9b) are actually ergative and block agreement for this reason (the gloss in (9b) therefore reflects the surface form).

Since Bobaljik (2008) relies on morphological case to determine agreement, this pattern does not follow from his account straightforwardly. The problem is that in (9b), the highest unmarked argument is the subject — but the verb nevertheless agrees with the object. Bobaljik (2008: 312) mentions that agreement tracks morphological case "in as far as zero exponents do not obscure [it]" but this raises the question of how distinct zero exponents differ so that some can block agreement while others cannot. Legate (2008: 95) thus suggests that there is syntactic Case and that "[t]o wait until the morphology to do case and agreement is to wait too long".

How can we account for the fact that certain Cases block agreement and that this is a point of variation across languages? There are different answers to this question; two proposals are shown in (10) (see also Keine 2010: 30f. and Preminger 2014: Ch. 8).

(10) a. A DP with  $\theta$ -related Case may not value a  $\varphi$ -probe. (Rezac 2008a: 83)

 b. The Visibility of Inherent-Case to Verbal Agreement (VIVA) Parameter A language will differ as to whether the verb can agree with an inherently case-marked DP. (Anand and Nevins 2006: 6) The suggestion in (10a) fails to account for the fact that the verb can agree with an ergative argument in Nepali, if ergative is a  $\theta$ -related case (Woolford 1999, 2006; Aldridge 2008; Legate 2008, 2012). Anand and Nevins (2006), on the other hand, suggest that their VIVA in (10b) is a microparameter that specifies whether a given language allows agreement with ergatives. In Hindi, inherently Case-marked DPs are not visible, whereas in Nepali they are.

In sum, some languages allow agreement with ergative (Nepali, Sahaptin), while others do not (Hindi, Marathi). The same holds for other Cases, like accusative: some languages, like Hungarian, allow agreement with these, but others, like German, do not. In addition, some languages allow agreement with dative arguments (like Basque, Sahaptin or Kashmiri), but others do not (like Hungarian). I will return to the question of how this information is specified for a given language in Section 6.3.1 below.

## 6.3 Analysis

We have just seen evidence that the Case of an argument determines its ability to agree with the verb in some languages. In Chapter 5, I showed that agreement in  $\varphi$ -features can determine the Case of an argument as well: in global case splits, the (sets of) person features of two arguments determine which Case is assigned to the subject or the object. These situations are difficult to state in a system in which morphological case influences agreement, since the order of Case and agreement is reversed.

In this section, I will show that the analysis I proposed in Chapter 5 can derive the agreement patterns introduced in the previous section and that Bobaljik's generalisations in (5) can be stated in this system as well. To do this, I will first lay out the assumptions I am making about Case and agreement before showing how the agreement patterns of Hindi, Nepali, and Marathi can be analysed involving abstract Case (following Legate 2008; Keine 2010).

I will then extend the discussion to ditransitive constructions in Section 6.3.4 in languages with two  $\varphi$ -probes. I show that the alignment of Case and agreement in ditransitives is parallel to the alignment of Case and agreement in monotransitives.

#### 6.3.1 Theoretical assumptions

First, as in Chapter 5, I will adopt aspects of Keine's (2010) syntactic framework, most importantly his suggestion that impoverishment rules can take place in the syntax (and not just post-syntactically, as in standard Distributed Morphology). This means that impoverishment can affect sets of CASE features which in turn can serve as the input for Agree (see 5.3.1 for more discussion).

Second, I will assume that Case assignment and agreement are independent of each other, i.e. that Agree does not simultaneously lead to the valuation of Case and  $\varphi$ -features. Instead, I will assume that the functional heads v and T (and possibly others, like Appl) can assign Case. Rather than relying on unvalued Case features which make noun phrases visible for Agree, I take Case assignment to be directly determined by the syntactic structure of the clause and the argument structure of predicates (cf. argument structure lists in Head-driven Phrase Structure Grammar or HPSG, Sag *et al.* 2003). I adopt the view that Case and agreement are dissociated, i.e. that the operation Agree does not simultaneously value the Case features of a goal and the  $\varphi$ -features of a probe (see also Bhatt 2005; Baker 2012, 2015 who argue that Case and agreement do not go hand in hand in Hindi and Amharic, respectively).

Third, in addition to assigning Case, functional heads like T and v (and again, others like Appl) *can* have  $\varphi$ -probes, but they do not have to. A specific precedent for this and the previous assumption comes from Bhatt (2005) who assumes a single  $\varphi$ -probe on T for Hindi, with both T and v being case assigners. The independence of Case assignment and agreement allows for cross-linguistic variation in  $\varphi$ -probes; I will argue that this variation is restricted, however. Concretely, v can only have a  $\varphi$ -probe if T has one, etc. I will show that this, together with the other assumptions derives Bobaljik's (2008) generalisations. I will relate this restrictive mapping of probes onto functional heads to Roberts's (2012) approach to parameters in Chapter 7.

Fourth, I will assume that if a functional head like T or v is both a Case assigner *and* has a  $\varphi$ -probe, the order in which Case is assigned and agreement is triggered is subject to cross-linguistic variation. This assumption builds on work by Müller (2004a), Heck and Müller (2007), Keine (2010), Müller (2010) and Georgi (2014) and I have argued in Chapter 5 that it provides an explanation for why some languages have global case splits, while others do not.

Fifth, and finally, I will adopt the assumption that Case can be modelled by sets of features. As mentioned in Chapter 5, I will assume that these sets can be modified during the syntactic derivation (Keine and Müller 2008; Keine 2010). In addition, they determine the spell-out of morphological case on arguments (Legate 2008). As they will be crucial in this chapter, I will lay out my assumptions about the relation between different Cases in more detail now.

## 6.3.2 Case and CASE features

Rather than treating "nominative" and "accusative" as Case features, I assume that they are labels for sets of features. Nominative and accusative in German can be represented as in (11), for example (following Bierwisch 1967: 246). I will continue to refer to these sub-features as "CASE" features.<sup>2</sup>

(11) NOM = 
$$\begin{bmatrix} -obl \\ -gov \end{bmatrix}$$
 ACC =  $\begin{bmatrix} -obl \\ +gov \end{bmatrix}$ 

The feature [-obl] is shared by both cases in (11), suggesting that neither is "oblique", but they differ in the feature [gov], for "governed". Bierwisch (1967: 246) suggests the use of *governed* because accusative arguments tend to be governed by verbs (and maybe prepositions), but nominative arguments do not.

Note that the labels adopted for CASE features differ in the literature. The reason for this is in part that one of the functions of these features is to distinguish one Case from another (cf. Caha 2009; see Dresher 2009 for discussion of contrastive features in phonology). An alternative label for [gov] could be [dep], for example, mnemonic of "dependent case": this feature could distinguish ergative and accusative from absolutive and nominative, for example.

Morimoto (2002: 305f.) adopts a similar approach in her analysis of DOM in Bantu languages. She uses the features  $[\pm hr]$  for "higher role" and  $[\pm lr]$  for "lower role" (see also references there). On this view, NOM/ABS are a represented by an empty set of features, but Acc is [+hr], indicating that there is a *higher role*, namely the nominative subject (note the similarity to dependent Case).

<sup>&</sup>lt;sup>2</sup> See Bierwisch (1967), Jakobson (1971 [1936]), Wiese (1999), Morimoto (2002), Müller (2002), McFadden (2004), Müller (2004b), Keine and Müller (2008), Caha (2009) and Keine (2010) for background and a number of empirical applications of treating Case as sets of features.

What [gov], [dep] and [hr/lr] therefore have in common is that their presence (or positive value) correlates with transitivity: these features can distinguish unmarked Case from the Case assigned to an object or a transitive subject, or accusative and ergative, respectively. By adding features, we can distinguish further Cases like dative, for example. Rather than representing NOM and ACC as in (11), they can be represented as shown in (12). The label [case] in (12) is inspired by Legate's (2008) notation of Case with a zero exponent (ABS or NOM; cf. Legate 2008: 59f.): it can arguably be thought of as a generic abstract Case feature. [dep] distinguishes ACC from NOM. Dative can be added to this system by adding a feature [obl].

(12) NOM = 
$$\begin{bmatrix} case \end{bmatrix}$$
 ACC =  $\begin{bmatrix} case \\ dep \end{bmatrix}$  DAT =  $\begin{bmatrix} case \\ dep \\ obl \end{bmatrix}$ 

Thinking of Case in these terms makes it possible to treat the sets of CASE features in (12) as a type of hierarchy like Bobaljik's Case hierarchy in (4), repeated in (13).<sup>3</sup>

(13) Unmarked Case > Dependent Case > Lexical/Oblique Case

In keeping with the recurring theme of remodeling hierarchies as relations among sets, another way of formulating (13) is by interpreting > as expressing the relation  $\subset$ , i.e. a proper subset relation on the sets in (12). The idea is that unmarked case has a proper subset of CASE features compared to dependent case, which in turn has a proper subset of CASE features compared to lexical case.

Caha (2009) proposes precisely such a way of representing Case and briefly discusses the possibility of combining his approach to case with Bobaljik's (2008) proposal. He shows that patterns of syncretism in case systems can be captured if Cases are assumed to be composed of sub-features. He argues for a "cumulative classification" (Caha 2009: 21) of Case: the Cases in (14) are sets of different cardinality where each set adds a feature to the previous one.

## (14) Cumulative classification

а. NOM = W

<sup>&</sup>lt;sup>3</sup> Note that I use [] in the derivations below rather than {} to indicate sets of CASE features, using the notation shown in Chapter 5 for reasons of consistency.

6.3 Analysis

Caha's "Case sequence" in (15) reflects this way of organising Case (cf. Caha 2009: 10; see his footnote 5, ibid., for discussion of ERG case). Note the similarities to Bobaljik's (2008) hierarchy in (13).

# (15) The Case sequence NOM/ABS - ACC/ERG - GEN - DAT - INS - COM

By combining (14) and (15), each Case is related to the next one in the sequence by the  $\subset -(16)$  illustrates this with the features in (12) above.

(16) 
$$\left\{ case \right\} \subset \left\{ case, \\ dep \right\} \subset \left\{ dep, \\ obl \right\} \subset \cdots$$

(16) matches Caha's (2009) Case sequence and provides a way of implementing the hierarchy and the generalisations proposed by Bobaljik (2008).

Recall that one aspect of Bobaljik's (2008) proposal is that languages differ in which Cases block agreement and which do not. We have seen several examples above: while ergative arguments are opaque for agreement in Hindi and Marathi, they are not in Nepali (cf. Anand and Nevins's 2006 VIVA discussed in Section 6.2.3 above). The potential of a Case to block agreement is naturally implemented in this feature-based system by identifying a feature that introduces opacity to agreement in a given language. Any Case that has this feature corresponds to the type of Case blocking agreement on Bobaljik's (2008) hierarchy. We can now reformulate Bobaljik's first generalisation as follows (see (5a) for comparison):

## (17) Bobaljik's first generalisation and CASE features

If a given set  $\kappa$  of CASE features includes a feature [ $\alpha$ ] which blocks agreement, any superset of  $\kappa$  will block agreement as well. Sets not including [ $\alpha$ ] do not block agreement.

On the assumption that the feature(s) giving rise to unmarked Case are a proper subset of those features giving rise to dependent Case (and of lexical Case) in all languages, (17) implements the same hierarchical effect as Bobaljik's (2008) first generalisation.

Assuming further that in languages with a single instance of agreement only T has a  $\varphi$ -probe, Bobaljik's (2008) second generalisation also follows. The unattested pattern is a language with nominative-accusative case alignment and ergative-absolutive agreement alignment.

Consider why: the unattested scenario would involve T agreeing with an accusative object by skipping a nominative subject. Since the nominative subject will be higher than the accusative object, and because nominative will necessarily be a proper subset of accusative, it is impossible that nominative has a feature that blocks agreement, while accusative does not.

I will now apply these assumptions to the set of data discussed in Section 6.2 above: the distribution of agreement in Hindi, Nepali and Marathi.

#### 6.3.3 Deriving agreement in Hindi, Nepali and Marathi

In this section, I illustrate how the Hindi, Nepali and Marathi agreement patterns introduced above are derived using abstract Case features, and in accordance with the reformulation of Bobaljik generalisation in (17). I will start with Hindi.

## 6.3.3.1 Hindi

As we have seen above, Hindi only allows agreement with unmarked arguments. The subject is unmarked in the imperfective aspect, but ergative in the perfective aspect. The direct object can be unmarked or, when animate and specific, marked with *-ko*, a case-marker that resembles the dative (I gloss it as ACC following Bhatt 2005). The following examples, repeated from Section 6.2.1, illustrate this.

(18)	a.	<i>Rahul kita</i> R.м boo 'Rahul useo	<i>aab paṛh-taa</i> k.ғ read-нав.м.s d to read a/the b	<i>thaa.</i> sg be.pst.m.sg ook.'	
	b.	<i>Rahul-ne</i> R-erg 'Rahul had	<i>kitaab parָh-ii</i> book.F read-PFV read the book.'	<i>thii.</i> F be.F.SG	(Bhatt 2005: 759)
(19)	<i>Мо</i> Мо	ona-ne is na.F-erg this	kitaab-ko s.овl book.f-acc	<i>paṛh-aa.</i> read-pfv.м.sg	
	'Mona had read this book.'				(Bhatt 2005: 768)

These data show that any overt case morphology blocks agreement in Hindi. As discussed in Chapter 1, Bhatt (2005) argues that agreement and case-marking are dissociated in Hindi, rather than being two aspects of a single Agree operation. Bhatt suggests that only T has a  $\varphi$ -probe in Hindi, and that v can assign Case independently.

Following the discussion of CASE features in the previous section, I suggest that any Case that includes the feature [dep] makes arguments opaque for agreement. The relevant Cases are specified as in (20), followed by the vocabulary items in (21):<sup>4</sup>

(20) NOM = 
$$\begin{bmatrix} case \end{bmatrix}$$
 ERG =  $\begin{bmatrix} case \\ dep \end{bmatrix}$  DAT =  $\begin{bmatrix} case \\ dep \\ obl \end{bmatrix}$ 

Since, by assumption, [dep] blocks agreement and both ERG and DAT (the object Case for differential object marking) include this feature, only unmarked arguments will be able to agree in Hindi. To derive DOM, I follow Keine (2010: 60) in assuming that an impoverishment rule deletes the CASE features on the direct object when it is specific and animate. A version of this rule adapted to the CASE features I assume here is shown in (22).

<sup>&</sup>lt;sup>4</sup> This section owes much to Keine's (2010) analysis of Hindi. The major point of departure is the inventory of CASE features that I assume in order to link the analysis to the present discussion. The two analyses are otherwise fully compatible.

(22) 
$$\begin{bmatrix} case \\ dep \\ obl \end{bmatrix} \rightarrow \emptyset / [-HUMAN, -SPECIFIC]$$

Keine (2010: 60) adds another impoverishment rule to derive split-ergativity in Hindi. (23) states the feature [dep] is deleted from the set of CASE features assigned to the subject in the imperfective aspect.

(23) 
$$\left[ dep \right] \rightarrow \emptyset / \left[ -PFV \right]$$

Finally, I will assume with Keine (2010) that v assigns inherent Case to the subject and structural Case to the direct object. Hindi has a single  $\varphi$ -probe on T (Bhatt 2005; Legate 2008).<sup>5</sup>

The following structures show how these ingredients interact in a full derivation. I will show the derivations of the data discussed above. (18a) shows agreement with an unmarked, masculine subject. (18b) shows agreement with an unmarked, feminine object when the subject is ergative. Finally, (19) shows default masculine agreement when both the subject and the object have overt case. Derivations of these examples are shown in (24a)-(24c), respectively.

(18a) is in the imperfective aspect, creating the context for the impoverishment rule in (23) to delete the feature [dep] from the set of CASE features assigned to the subject, and the subject only receives [case]. The object's CASE features are impoverished because the object is non-specific and inanimate in (18a). Since the subject's features do not include the feature [dep], T and the subject enter an Agree relation and T is valued as 3, SG, M, the subject's  $\varphi$ -features.

<sup>&</sup>lt;sup>5</sup> There are several different analyses of ergative assignment in Hindi. Bhatt (2005: 767), for example, suggests that nominative and ergative are both assigned by "Tense in association with v" depending on aspect. Anand and Nevins (2006) suggest that T assigns nominative, v assigns ergative, and two distinct additional heads assign null and overt object case (*-ko*), respectively. Legate (2008) also suggests that ergative is assigned by v. Since inherent Case is assigned in conjunction  $\theta$ -role, the Case assigned to the subject and the (structural) Case assigned to the object do not interfere, even if assigned by the same head (cf. also Aldridge 2008: 978f.).



(24b) differs from (24a) in that v assigns ergative to the subject because the perfective aspect in (18b) bleeds the application of the impoverishment rule in (23). The presence of [dep] then blocks agreement between T and SUBJ, but when T continues to probe, it finds the direct object. The direct object's features have been deleted by the impoverishment rule in (22). Agreement on T will therefore reflect the person, number and gender of the direct object.

6 Case and agreement in the grammar



Finally, (24c) shows the derivation of (19). Here, the object's CASE features are not deleted because it is specific and therefore the impoverishment rule in (22) cannot apply. T fails to agree with either the subject or the direct object, since both have the feature [dep] and are invisible; T's  $\varphi$ -features get the default value of third person singular masculine, shown in (E).


To sum up, in Hindi any overt morphological case renders an argument opaque for Agree. In Bobaljik's (2008) terms, this means that any dependent Case is invisible for agreement; I model this by suggesting that the feature [dep] in a set of CASE features renders an argument opaque (see also Keine 2010).

# 6.3.3.2 Nepali

Nepali differs from Hindi in that it allows agreement with ergative subjects. I implement this by assuming that [dep] does not render an argument opaque for agreement. Apart from this difference, I will analyse Nepali and Hindi in the same way. I will again follow Keine (2010) in suggesting that split-ergativity is derived by an impoverishment rule that deletes the subject's [dep] feature in the imperfective aspect, giving rise to unmarked Case on the subject, and I will assume that v assigns inherent Case to the subject and structural Case to its object. The relevant data illustrating the Nepali agreement pattern are shown in (25), repeated from (3). In both (25a) and (25b), the verb agrees with the subject — the ergative in (25b) does not block agreement.

(25)	a.	та	yas	pasal-mā	patrikā	kin-ch-u.		
		1sg.no 'I buy t	м дем.о the news	BL store-LOC	newspape store.'	er.nom buy-npst-1	lsG	
	b.	<i>maile</i> 1sg.ere 'I boug	<i>yas</i> G DEM.OF ht the ne	pasal-mā ; BL store-LOC ; ewspaper in t	<i>patrikā</i> newspaper his store.'	kin-ẽ / r.nom buy-pst.1sg	* <i>kin-yo.</i> в buy-pst.3sg.м	
						(Bickel a	and Yādava 2000:	: 348)

The sets of CASE features and the relevant vocabulary items for the data discussed (abstracting away from person features) here are shown in (26) and (26).

(26) NOM = 
$$\begin{bmatrix} case \end{bmatrix}$$
 ERG =  $\begin{bmatrix} case \\ dep \end{bmatrix}$   
(27)  $\begin{bmatrix} case \end{bmatrix} \leftrightarrow ma'I.NOM'$   $\begin{bmatrix} case \\ dep \end{bmatrix} \leftrightarrow maile'I.ERG'$ 

The impoverishment rule in (28) deletes the feature [dep] in the imperfective.

(28) 
$$\left[\operatorname{dep}\right] \to \emptyset / \left[-\operatorname{PFV}\right]$$

(29a) shows the (simplified) structure of a sentence like (25a), with a nominative (unmarked) subject ignoring earlier stages of the derivation in which the direct object is assigned Case by the verb. In (29a), [dep] is deleted because the context for the rule in (28) is matched. v therefore assigns only [case] to the subject. After CASE assignment, T probes: the subject in SpecvP values T's  $\varphi$ -features.



In (29b), corresponding to (25b), the clause is in the perfective aspect, and the subject is assigned ergative CASE features, i.e. including [dep] in step  $\bigcirc$ . However, [dep] does not block agreement in Nepali (in contrast to Hindi), and therefore T will agree with the subject as in (29a).

(29) b.



While Hindi and Nepali use the same sets of CASE features, the different agreement patterns result from [dep] turning arguments opaque for agreement in Hindi, but not in Nepali. Both languages have a single  $\varphi$ -probe on T that determines agreement with the verb.

(29) a.

#### 6.3.3.3 Marathi

The agreement pattern in Marathi is similar to that of Hindi: ergative arguments are opaque for agreement. However, as discussed in Section 6.2.3 above, ergative Case in Marathi is not always spelled out. The relevant data are repeated here. First, (30a) and (30b) indicate that agreement is controlled by unmarked arguments, the ABS subject in (30a) and the ABS object in (30b). As before, ergative Case is assigned in the perfective.

(30)	a.	<i>mulī</i> girl.3pl.f.Af 'Girls sing s	gāņī 85 song.3pl.N ongs.'	mhaņtāt. N.ABS sing.PST.3PL.F	
	b.	<i>mulī-ne</i> girl.3pl.F-EF 'The girls sa	g <i>āņī</i> RG song.3PL.1 Ing songs.'	mhațlī. N.ABS sing.PST.3PL.N	
		(Panc	lharipande	1997: 284, cited by Legate 2008: 94, Keine 2010:	51)

In addition to the split based on aspect, Marathi also has a person split in the perfective (Legate 2008; Keine 2010). First and second person pronouns appear in absolutive rather than ergative. Nevertheless, they are opaque for agreement. This is shown in (31).

(31)	a.	tyā-ne / t	i-ne	gāņī	mhațlī.	
		he-erg sl	he-er	G song.3pi	L.N.ABS sing.pst.3pl.n	V
		'S/he sang	song	s.'		
	b.	mī / tū	g	āņī	mhațlī.	
		I.abs you.	ABS S	ong.3pl.n.	ABS sing.PST.3PL.N	
		ʻI / you (sg	g.) sar	ng songs.'		
				(Τ	Dandharinanda 100	7.1

(Pandharipande 1997: 131, cited by Keine 2010: 52)

Independently of the overt case-marking, the verb still agrees with the object  $g\bar{a}n\bar{i}$  'songs', rather than the unmarked subject, in both (31a) and (31b). Legate (2008) thus argues that it is abstract Case that determines agreement with the verb, not overt morphological Case. Hindi and Marathi therefore show the same restriction: ergative arguments are opaque for agreement. I take this to mean that [dep] renders arguments opaque for agreement in both languages. The two languages differ, however, in how the spell-out of this Case is determined.

I will again follow Keine (2010) in implementing this pattern of agreement, but I use the same set of CASE features as in Hindi and Nepali. (32) shows the feature specification of ABS and ERG, and (33) shows the corresponding vocabulary insertion rules.

$$(32) \quad ABS = \begin{bmatrix} case \end{bmatrix} \quad ERG = \begin{bmatrix} case \\ dep \end{bmatrix}$$
$$(33) \quad \begin{bmatrix} \end{bmatrix} \leftrightarrow -\emptyset \quad \begin{bmatrix} case \\ dep \end{bmatrix} \leftrightarrow -ne$$

To derive the zero spell-out of ergative with first and second person subjects, I assume with Keine (2010) that an impoverishment rule modifies the set of CASE features on the subject. This rule is shown in (34). It deletes the feature [case] when the subject is first or second person.

(34) 
$$[case] \rightarrow \emptyset / \_ [\pi: 1 \lor 2]$$

I will only focus on the perfective here to illustrate the contrast between first and second person subjects, on the one hand, and third person subjects, on the other, as shown in (31).

The vocabulary insertion and impoverishment rules interact as follows. If the subject is third person, (34) cannot apply and the subject's CASE features include both [case] and [dep], which gives rise to ERG-marking. This is shown in the derivation in (35a), corresponding to (31a). Since the subject's set of CASE features includes [dep], T will not enter an Agree relation with it.

6 Case and agreement in the grammar



In (35b), the subject is first person, and therefore the impoverishment rule in (34) applies, deleting the feature [case] on the subject. This means that the ergative suffix *-ne* cannot be inserted since its vocabulary item specifies a superset of the features on the subject. Nevertheless, [dep] is still present on the subject and makes it opaque for agreement.



#### 6.3.3.4 Interim summary

The Hindi, Nepali, and Marathi examples shown here have in common that they involve a single  $\varphi$ -probe on T and the same set of CASE features. But these languages differ in the particular feature that makes an argument opaque for agreement and the spell-out of CASE features.

Recall that while Hindi, Nepali and Marathi show ergative-absolutive case alignment in the perfective, they differ in their agreement alignment (cf. Section 6.2.2 above). Hindi and Marathi have ergative-absolutive agreement alignment: unmarked subjects and objects agree, but ergative subjects do not. Nepali agreement is different in that the verb agrees with the subject whether it is unmarked or ergative. This is nominativeaccusative agreement alignment.

So Hindi and Marathi have ergative-absolutive alignment in both case and agreement and German, for example, has nominative-accusative alignment in both case and agreement. Nepali combines two systems, however: it has ergative-absolutive case alignment (like Hindi and Marathi), but nominative-accusative agreement alignment (like German).

The fourth logical possibility is unattested: languages with nominative-accusative case alignment, but ergative-absolutive agreement alignment. As mentioned above, this follows from Bobaljik's (2008) generalisation and its reformulation in the present framework, repeated here:

#### (36) **Bobaljik's first generalisation and CASE features**

If a given set  $\kappa$  of CASE features includes a feature [ $\alpha$ ] which blocks agreement, any superset of  $\kappa$  will block agreement as well. Sets not including [ $\alpha$ ] do not block agreement.

(36) relates to the analysis presented here as follows: the languages discussed above have a single  $\varphi$ -probe on T. When there is an unmarked subject, T will agree with this subject. If the subject is ergative and has a feature [ $\alpha$ ] that makes it opaque for agreement, T will agree with the object (as in Hindi and Marathi). If the subject is ergative, but is not opaque for agreement, the verb will agree with it (as in Nepali).

What is impossible, however, is for T to bypass an unmarked subject and agree with the object. This is because in a language with nominative-accusative case alignment, the set of CASE features on the subject are a proper subset of the features on the object. It is impossible that the subject's set contains a feature that renders the subject, but not the object, opaque for agreement.

The alignment patterns of the languages discussed here are summarised in Table 6.1. For German, Nepali, Hindi, and Marathi, it shows their case alignment and indicates the assigner of subject case, their inventory of  $\varphi$ -probes, and the agreement alignment originating from that probe. The final column indicates which Case is opaque for agreement in each language. Note that the entries for Nepali, Hindi and Marathi indicate the perfective aspect only. The final row shows the unattested pattern of nominativeaccusative case alignment and ergative-absolutive agreement alignment.

By combining insights from Bobaljik (2008), Legate (2008), Caha (2009) and Keine (2010), it is possible to derive a wider empirical coverage than on Bobaljik's approach while retaining his cross-linguistic predictions about typological gaps in alignment systems.

I now turn to case and agreement alignment in languages with two  $\varphi$ -probes, one each on T and *v*.

	Case alignment	$\phi$ on	Agreement alignment	Case opacity
German	Т: пом	Т	T: nom	ACC+
Nepali	v: ERG	Т	T: nom	DAT+
Hindi	v: ERG	Т	T: erg	ERG+
Marathi	v: ERG	Т	T: erg	ERG+
*	T: NOM	Т	T: ERG	

Table 6.1 Alignment and  $\varphi$ -probes of languages discussed in this section

#### 6.3.4 Languages with two $\varphi$ -probes

We have just seen that it is possible to implement the cross-linguistic generalisations about case and agreement alignment proposed by Bobaljik (2008) without adopting his conclusion that morphological case determines agreement.

The languages discussed so far showed one instance of agreement. I suggested that Hindi, Nepali, Marathi, and German have a single  $\varphi$ -probe, on T, which will agree with an argument based on that argument's abstract Case features.

In this section, I will discuss languages with two  $\varphi$ -probes. As mentioned in Section 6.3.1, I assume that such languages have a  $\varphi$ -probe on v, in addition to the probe on T. Paralleling the discussion about case and agreement alignment in languages with a single probe, I will show that there is an analogous generalisation about v and two internal arguments in ditransitive constructions. The analysis proposed so far predicts that not all combinations of case and agreement alignment in ditransitives should be possible: no language should have secundative case alignment and indirective agreement alignment. I will show that this prediction is borne out.

#### 6.3.4.1 Case and agreement alignment in ditransitive constructions

Languages differ in the way that case-marking and agreement are manifested in ditransitive constructions. Dryer (1986), among others, suggests that in analogy to ergative-absolutive vs. nominative-accusative alignment in monotransitive clauses, lan-

guages differ in their alignment in ditransitive clauses: I will refer to these two types of alignment as "indirective" and "secundative", following Haspelmath (2005).<sup>6</sup>

In analogy to the notions A, S, P which are used to describe nominative and ergative alignment, Haspelmath (2005) uses P, T, and R for ditransitive alignment systems. These are defined as follows:

- (37) a. **P** is the single internal argument of a monotransitive.
  - b. T is the theme- or patient-like internal argument of a ditransitive.
  - c. **R** is the recipient-like internal argument of a ditransitive.

Indirective and secundative alignment are then defined by whether it is T or R that patterns with P with respect to case-marking or agreement ("flagging" and "indexing", respectively, in Haspelmath 2005).



These types of alignment can be demonstrated using different predicates in English: *donate* can be monotransitive or ditransitive, and shows indirective alignment, because its single internal argument P is a T argument in the ditransitive construction.<sup>7</sup>

The verb *equip* (*with*), on the other hand, takes P as its single internal argument that patterns with R in the ditransitive construction. This is secundative alignment. (39) and (40) illustrate.

- (39) a. I donate [ $_{P}$  the book].
  - b. I donate  $[_{T}$  the book]  $[_{R}$  to the man].
- (40) a. I equip  $[P_{P} \text{ the man}]$ .
  - b. I equip  $[_{R}$  the man $] [_{T}$  with the book].

 $<sup>^{6}</sup>$  Haspelmath (2005) also mentions neutral alignment in which none of P, T, and R are overtly marked. I will set this aside.

<sup>&</sup>lt;sup>7</sup> These examples are not meant to make any deep statements about the nature of English ditransitives or double object constructions; they just illustrate the alignment patterns under discussion.

While English does not show much case-marking, these examples illustrate the difference in alignment. In indirective alignment, the P and T arguments appear unmarked, while the R argument appears with a preposition (or a prepositional dative). In secundative alignment, the single internal argument in the monotransitive (40a) patterns with the ditransitive R argument to the exclusion of T.

Other languages show similar alternations in agreement in ditransitive constructions. Some varieties of the Ob-Ugric languages Khanty (Nikolaeva 1999a,b, 2001; Dalrymple and Nikolaeva 2011) and Mansi (Virtanen 2012, 2014, 2015) can alternate between indirective and secundative agreement with objects. The following examples illustrate this.

(41a) and (41b) show secundative and indirective agreement alignment, respectively, in Eastern Mansi. In both examples, the verb shows object agreement, glossed as 'sg<1sg' and 'sg<3sg', respectively. The first 'sg' indicates the number of the object; the rest indicates the person and number of the subject.

The trigger of agreement is an R argument  $n\ddot{a}\ddot{a}$ -n 'you' in (41a) and a dropped T argument in (41b) (referring back to the *bowl full of blood*). The object that is not agreed with is expressed as an oblique in both cases.

(41) a. am nää-n tat-øs-løm nee-l. 1sG 2sG-sG.2sG bring-PST-sG<1sG woman-INS 'I brought you a wife.', lit. 'I brought you with a wife.'
b. moot sõõn-toågøl keeløp-mø wø-s-tø, kõõp-posøm-øt püw-øtään other bowl-full blood-ACC take-PST-sG<3sG boat-stern-LOC son-LAT.sG.3sG tow-mø-s-tø. VM-give-PST-sG<3sG 'He took the other bowl full of blood and gave it to his son ...' (Virtue or 2012, 125f)

(Virtanen 2012: 125f.)

Northern Khanty shows the same pattern. In (42a), the unmarked and agreed with argument is a T, *cup*, and the R argument is expressed with the postposition *e:lti* 'to'. In (42b), the agreeing object is R and the T argument is an oblique, expressed with locative case.

(42)	a.	ma <b>a</b> : <b>n</b> Pe:tra e:li	ti ma-s- <b>e</b> ː <b>m</b>	/ ma-s-əm.
		I cup Peter to	give-pst-obj.1sg.sbj	give-pst-1sg.sbj
		'I gave a/the cup to	o Peter.'	

b. *ma Pe:tra a:n-na ma-s-e:m* / \**ma-s-əm*. I Peter cup-LOC give-PST-OBJ.1SG.SBJ give-PST-1SG.SBJ 'I gave a/the cup to Peter.'

(Dalrymple and Nikolaeva 2011: 148)

Khanty and Mansi therefore allow both secundative and indirective alignment, and both alignment types pattern together in case-marking and agreement. This does not have to be the case, however.

Amharic ditransitives, for example, can show indirective case-alignment, but secundative agreement alignment. This means that the single internal argument in a monotransitive, P, patterns with the theme-like T argument in a ditransitive with respect to case-marking, but with the recipient-like R argument with respect to agreement. This is shown in (43).

(43)	a.	Ləmma wɨ∬a-w-ɨn j-aj-əw-al.					
		Ləmma dog-def-acc 3sg.m.sbj-see-3sg.m.obj-aux(3sg.m	i.sbj)				
		'Lemma sees the dog.'	(Baker 2012: 257)				
	b.	Ləmma l-Almaz məts'haf-u-n sət't'- <b>at</b> .					
		Lәтта.м DAT-Almaz.f book-DEF.м-ACC give-(3sg.м.sbj)-	3sg.f.obj				
		'Lemma gave the book to Almaz.'	(Baker 2012: 258)				

In (43a), a single internal argument gets accusative case and triggers agreement. In (43b), the T object *məts'haf-u-n* 'book-DEF.M-ACC' gets accusative, but the verb agrees with the R argument *l-Almaz* 'DAT-Almaz'.

The different alignments are illustrated in (44), where the dotted box shows the alignment of case-marking and the dashed box shows the alignment of agreement in (43b).

(44) Amharic: indirective alignment for case, secundative alignment for agreement

Р	
Indirective case T	R Secundative agreement

These patterns can be generalised as follows. Let us assume that the same Case is assigned by the same head, so v assigns Case to P, and in indirective alignment it also

assigns Case to T. A distinct head, e.g. Appl in Amharic and P (a postposition) in Khanty, assigns Case to R. In secundative alignment, v assigns Case to P and R, and T is assigned case by another head, e.g. P in Khanty and Mansi.

What about agreement? In a language with two  $\varphi$ -probes, the single internal argument of a monotransitive is accessible for agreement.<sup>8</sup> The Case of this object will be accusative (or absolutive). Accusative precedes dative on Caha's (2009) Case sequence shown in Section 6.3.1 above and therefore has a proper subset of CASE features compared to dative.

This means that if a verb can agree with a dative argument, none of the CASE features that constitute dative introduce opacity. Following the logic of subset/superset relations among Cases, this entails that a verb must also be able to agree with an accusative argument. The Amharic example above provided an example of this.

Under the assumption that an R argument is structurally higher than a T argument in a ditransitive construction and both are below v, this predicts the existence of the alignments in (45a)–(45c) but rules out the combination of alignment types in (45d).

(45)	a.	indirective case and indirective agreement	e.g. Hungarian
	b.	secundative case and secundative agreement	e.g. Khanty
	c.	indirective case and secundative agreement	e.g. Amharic
	d.	*secundative case and indirective agreement	

Consider why: if a P argument agrees in a monotransitive, then in secundative case alignment, the R argument that patterns with P also must be able to agree. The distinct indirective systems in (45a) and (45c) can be explained by the opacity of the case on the R argument: if it is opaque, as in Hungarian, it does not agree, but if it is accessible, as in Amharic, it does.

But it is impossible to derive (45d). In secundative case alignment, the  $\tau$  argument bears a Case that is lower on Bobaljik's (2008) hierarchy, or, in present terms, it has more features than the Case on the R argument which patterns with the monotransitive P. If R is structurally higher than  $\tau$ , it is impossible for v to ignore R when it probes as R will have to be accessible. The following structures illustrate this.

<sup>&</sup>lt;sup>8</sup> As I will discuss in Chapter 7, I assume that an acquirer would not posit a  $\varphi$ -probe on  $\nu$  if there were no agreement with internal arguments.

- 6 Case and agreement in the grammar
- (46) a. Indirective case and indirective agreement, cf. (45a); DAT opaque



b. Secundative case and secundative agreement, cf. (45b)



c. Indirective case and secundative agreement, cf. (45c); DAT visible



(46a) and (46c) differ in whether dative is opaque for agreement or not: if it is, agreement ignores it. If it is not opaque, v agrees with R bearing dative because it is more local than T bearing accusative. Finally, consider (46d), corresponding to (45d). R is accusative and therefore must be accessible for agreement. There is thus no way that v can agree with the less local T, irrespective of its Case.

(46) d. \*Secundative case and indirective agreement, cf. (45d)



This prediction is analogous to Bobaljik's (2008) second generalisation: while he considers ergative-absolutive vs. nominative-accusative case-alignments and agreement patterns found in languages with a single  $\varphi$ -probe, the reasoning extends straightforwardly to languages with *two*  $\varphi$ -probes and the domain of *v* involving two internal arguments.

Is this prediction borne out? Are there no ditransitive systems with secundative case and indirective agreement? In Haspelmath's (2005) survey of 100 ditransitive systems, none of the languages show this pattern (Haspelmath 2005: 15ff.), although it should be noted that there are few languages with secundative case-alignment in his sample (Haspelmath 2005: 7). I therefore, tentatively, take the following generalisation to be valid:

#### (47) Alignment in ditransitives

There are no languages with secundative case-alignment but indirective agreement alignment in ditransitives.

(47) follows from the interaction of case and agreement in the grammar as suggested in this section in the same way as Bobaljik's (2008) second generalisation about the lack of ergative agreement alignment in nominative-accusative-case languages.

In addition, Haspelmath's (2005) Generalisations 6 and 7 point to a similar conclusion. (48) states that v never agrees only with R. This follows because P must bear a Case that is at least as or more accessible than the Case on R. (49) states that both P and R are always indexed in secundative agreement-alignment. This follows because P and R have the same (accessible) Case; this blocks agreement with T.

#### (48) Generalisation 6

Indirective indexing is never achieved by indexing of R alone (only by indexing of P and T alone, or by differential indexing of R).

(Haspelmath 2005: 13)

#### (49) Generalisation 7

Secundative indexing is always achieved by indexing of the P and R alone, never by indexing of T alone or by differential indexing of T.

(Haspelmath 2006: 14)

To summarise, Bobaljik's (2008) generalisations about the distribution of nominativeaccusative and ergative-absolutive alignment types can be extended to cover the distribution of indirective and secundative alignment in ditransitive constructions. The logic of the argument is the same as before. Acc has a proper subset of features of OBL. If Acc can be agreed with in a monotransitive, then it will be impossible for v to skip an R-argument bearing Acc argument and agree with the T-argument bearing OBL in a ditransitive with secundative case alignment.

Table 6.2 extends Table 6.1 with the languages discussed in this section, and the languages from Chapter 5. These languages have two  $\varphi$ -probes, one on T and one on v. For each language, Table 6.2 indicates the case and agreement alignment for each  $\varphi$ -probes and also shows the unattested patterns that are ruled out by the reformulated version of Bobaljik's first generalisation, repeated here.

#### (50) **Bobaljik's first generalisation and CASE features**

If a given set  $\kappa$  of CASE features includes a feature [ $\alpha$ ] which blocks agreement, any superset of  $\kappa$  will block agreement as well. Sets not including [ $\alpha$ ] do not block agreement.

The final column of Table 6.2 shows another parameter determining cross-linguistic variation in Case and agreement, namely the order of Case assignment and agreement on a given head (as discussed this in Chapter 5). I argued there, following Müller (2004a, 2009), Keine (2010) and Georgi (2012, 2014), that the order of operations on a head can vary. In Sahaptin and Awtuw, specified as  $\varphi < CASE$ , this order gives rise to global case splits. When CASE assignment precedes agreement, as in Hungarian and Amharic, arguments are assigned the same Case independently of the  $\varphi$ -features of other arguments.

	Case alignment	$\phi$ on	Agreement alignment	Case opacity	φ, case
German	Т: пом	Т	T: nom	ACC+	
Nepali	v: ERG	Т	T: nom	DAT+	
Hindi	V: ERG	Т	T: erg	ERG+	
Marathi	v: ERG	Т	T: erg	ERG+	
*	Т: пом	Т	T: erg		
Sahaptin	T: erg, v: sec	Τ, ν	T: erg, $v$ : sec	OBL+	$\phi \prec case$
Awtuw	T: NOM, <i>v</i> : IND	Τ, ν	T: NOM, $v$ : IND	ACC/DAT+	$\phi \prec case$
Amharic	T: NOM, <i>v</i> : IND	Τ, ν	T: nom, <i>v</i> : sec	OBL+	case $\prec \varphi$
Hungarian	T: nom, v: ind	Τ, ν	T: nom, <i>v</i> : ind	DAT+	$case \prec \varphi$
Khanty	Т:	Τ, ν	Т: NOM, <i>v</i> : both	OBL+	case $\prec \varphi$
*	T:, <i>v</i> : sec	Τ, ν	T:, <i>v</i> : ind		

Table 6.2 Alignment and  $\varphi$ -probes of languages discussed in this section

Before concluding this chapter, I will briefly discuss Case and agreement in two more languages. First, I turn to Senaya. Kalin (2014, 2015) and Kalin and van Urk (2015) suggest that this language has a single  $\varphi$ -probe on T in the perfective, and two  $\varphi$ -probes, on T and Asp, respectively, in the imperfective. Second, I will return to Kashmiri as an example of a language with three  $\varphi$ -probes, on T, v and Appl, respectively.

#### 6.3.5 Splits between one and two probes

Kalin (2014, 2015) and Kalin and van Urk (2015) discuss split-ergativity in Neo-Aramaic, in particular the language Senaya. In Senaya, the perfective and the imperfective show different agreement patterns. In the perfective aspect, there is subject agreement, but no object agreement. Subject agreement is expressed by "L-suffixes" (the "L" refers to the morphology of these suffixes).

In the imperfective, there is subject agreement and differential object agreement. Interestingly, *object* agreement is expressed by L-suffixes and subject agreement is expressed using a separate set of suffixes that Kalin and van Urk call S-suffixes.<sup>9</sup> (51) and (52) show examples of perfective and imperfective aspect, respectively.

(51)	<i>axnii dmex-lan.</i> we sleep.prF-L.1pL 'We slept.'	(Kalin and van Urk 2015: 662)
(52)	<i>ooya molp-a-lan.</i> she teach.IPFV-S.3.F.SG-L.1PL 'She teaches us.'	(Kalin and van Urk 2015: 662)

These examples illustrate "partial agreement reversal" (Kalin and van Urk 2015: 662): the L-suffix agrees with the subject in (51), but with the object in (52). Kalin and van Urk propose that agreement reversal follows if there is a single  $\varphi$ -probe on T in the perfective aspect, but a  $\varphi$ -probe on both T and Asp<sub>IPFV</sub> in the imperfective aspect, shown in (53):

<sup>&</sup>lt;sup>9</sup> As Kalin and van Urk (2015: 666) point out, the language shows nominative-accusative agreement alignment in both aspects because the transitive (A) and the intransitive subject (s) are expressed by the same suffixes in the respective aspects: L-suffixes for A, s in perfective, and S-suffixes for A, s in imperfective. In the perfective aspect, the object does not trigger agreement at all, and in the imperfective aspect, it is marked with an L-suffix.



(Kalin and van Urk 2015: 661)

The distribution of these  $\varphi$ -probes above *v*P explains the distribution of the suffixes: agreement with T is expressed by L-suffixes, and agreement with Asp by S-suffixes. In the perfective, T will find the subject, and in the imperfective, T will bypass the subject, which has already agreed with Asp.

Kalin (2015) focuses on the pattern of differential object agreement in Senaya (see also Kalin 2014: Ch. 4). While subjects always agree, only specific (and more definite) direct objects trigger agreement with the verb — spelled out as L-suffixes in imperfective aspect. Kalin implements this differential agreement by suggesting that nonspecific arguments do not need to be licensed, i.e. they are exempt from the Case Filter. Differential object agreement is illustrated in (54). Only the definite object in (54b) triggers agreement, spelled out with an L-suffix.

- (54) a. *axnii xa ksuuta kasw-ox.* we one book write.IPFV-S.1PL 'We write a book (fem.).'
  - b. *axnii oo ksuuta kasw-ox-laa.* we that book write.IPFV-s.1PL-L.3.F.SG 'We write that book (fem.).'

Technically, this works as follows: all nominals come with *unvalued* Case features, but only some nominals come with both uninterpretable and unvalued Case features. Kalin writes the former as [Case: \_\_] and the latter as [*u*Case: \_\_] (following Pesetsky and Torrego 2007). Kalin (2015) assumes that only nominals that have an uninterpretable *and* unvalued Case feature require licensing. In Senaya, specificity introduces an uninterpretable Case feature. Non-specific nominals escape licensing because they lack

an uninterpretable Case feature. Evidence for this comes from the perfective aspect which restricts the appearance of specific objects (Kalin 2015: 10f.).

However, an unvalued Case feature can still undergo agreement and be valued: since there is always at least one  $\varphi$ -probe which will find the subject, any subject will trigger agreement, even non-specific ones. While such subjects do not require licensing, they are nevertheless accessible for agreement. Representative derivations are shown below. First, (55) shows the derivation of subject agreement with a non-specific subject, which comes with an unvalued Case feature (but not an uninterpretable one). Asp and the subject agree, Asp values the subject's Case feature (shown as [CASE A ASP]) and in turn the subject values Asp's  $\varphi$ -features. The derivation is essentially the same if the subject is specific and comes with an unvalued and uninterpretable Case feature.



So how is differential object agreement derived? Kalin (2015) suggests that the additional  $\varphi$ -probe on T only appears when needed, i.e. when there is a specific object that requires licensing by a higher head. Kalin (2015) thus treats Agree as valuing a Case feature on a goal and  $\varphi$ -features on the probe simultaneously, but modifies the scope of the Case Filter: only nominals with both uninterpretable and unvalued Case features are subject to it. But not all nominals come with uninterpretable Case features. This is shown in the following derivations. First, (56) shows a clause with a non-specific object, which lacks an uninterpretable Case feature and therefore does not need licensing. T does not have a probe.



(57) shows a structure with a specific direct object: it has an unvalued and uninterpretable Case feature which requires licensing. Kalin (2015) suggests that therefore T comes with a  $\varphi$ -probe which serves to license the direct object.



255

#### 6.3.5.1 An alternative analysis for Senaya

The Senaya data can be accounted for in the present framework as well. The crucial difference lies in the nature of Agree: I have assumed throughout this thesis that referential properties of noun phrases can be grammaticalised as person features. Instead of assuming that specific noun phrases introduce an uninterpretable Case feature, I suggest for Senaya that only specific nominals come with a person feature, whereas non-specific noun phrases do not. Specific direct objects thus trigger agreement because they are visible for the  $\varphi$ -probe on T, but non-specific direct objects do not.

However, subject agreement is possible with non-specific arguments, just like in Hungarian (see Chapter 4). For Hungarian, I have argued that the additional number probe on T entails a person feature on any kind of subject – Hungarian v lacks a number probe and therefore does not agree with objects lacking person features.

In Senaya, the probe that agrees with the subject can either be Asp or T. Both have number probes, and neither is sensitive to the specificity of the subject, so this option does not work for this language. A (stipulative) alternative is to assume that v requires its specifier to have certain formal features, which make it visible for agreement but do not necessarily introduce a specific interpretation (cf. N. Richards 2010: 81f. for a similar idea with respect to animacy).

This approach captures the distribution agreement and case-marking in Hungarian, as well as the agreement pattern of Senaya, whereas Kalin's (2014, 2015) analysis cannot derive the distribution of agreement in Hungarian. The reason for this is the mismatch between case and agreement: direct objects in Hungarian have overt accusative case, but they do not always trigger agreement. Kalin's account explains the lack of agreement by the lack of a licenser, but the presence of accusative is mysterious: I am not aware of evidence that case and agreement would come from distinct heads in Hungarian.

There is thus a trade-off between the two approaches: I believe that the approach endorsed here, dissociating Case and agreement, accounts for a wider range of data, while the analysis of Senaya proposed by Kalin (2014, 2015) and Kalin and van Urk (2015) accounts for the distribution of agreement across aspect and the subject-object asymmetry in agreement in a very systematic and elegant way. Before returning to Kashmiri, a note is in order about the distribution of  $\varphi$ -probes on functional heads in Senaya. I have suggested so far that if a language has a single  $\varphi$ -probe, it is on T, and that a second  $\varphi$ -probe will be on v. In light of the data discussed here, this hypothesis has to be modified, since Kalin (2014, 2015) and Kalin and van Urk (2015) argue convincingly that the lower  $\varphi$ -probe is on Asp in Senaya, rather than v.

There are then (at least) two types of languages with two  $\varphi$ -probes: the type with probes on T and *v* I discussed above, and the type with probes on T and Asp, exemplified by Senaya. I want to highlight two consequences of this. First, both types are compatible with the suggestion that a single  $\varphi$ -probe in any language is on T and gives rise to agreement with an unmarked argument (as follows from Bobaljik's 2008 generalisations). Second, the position of the second probe makes a clear prediction about the pattern of object agreement a language exhibits: Senaya shows agreement reversal, i.e. T agrees with the subject in the perfective, and with the object in the imperfective aspect. In the languages where  $\varphi$ -probes are on T and *v*, there is no such reversal: T always agrees with the subject and *v* agrees with the object.

I conclude that Senaya provides further evidence that languages differ in the number of  $\varphi$ -probes they have on functional heads, but that in addition, languages can differ in where the second  $\varphi$ -probe is:  $\nu$  or Asp.

#### 6.3.6 A language with three $\varphi$ -probes: Kashmiri

Before concluding this chapter, I will briefly discuss Case and agreement in a language with three  $\varphi$ -probes, namely Kashmiri (Wali and Koul 1997; another language with three  $\varphi$ -probes is Basque, cf. Arregi and Nevins 2012; Odria 2014).

While the presence of agreement markers ("pronominal suffixes" in Wali and Koul 1997) depends on aspect as well as the person of the arguments, certain verb forms cross-reference all three arguments in a ditransitive. An example is shown in (58) (Wali and Koul 1997 gloss *su* as Acc, but it is homophonous with nominative; I therefore gloss it as NOM).

(58) *bi chu-s-an-ay su tse hava:li kara:n.* I.NOM be-1sg.sbj-3sg.obj-2sg.ю he.NOM you.DAT hand over doing 'I am handing him over to you.'

(Wali and Koul 1997: 253, glosses adapted)

The auxiliary *chu-s-an-ay* references the subject, the direct object and the indirect object (in that order) with its suffixes. Restrictions on whether these suffixes can be overt are related to the Case and the person of the arguments (Wali and Koul 1997: 247). First, only pronominal objects can be cross-referenced as suffixes. Second, only second person pronouns can appear both as a suffix and as overt pronoun (as in (58)). Pronouns of other persons either appear as a DAT argument or as a suffix.

Interestingly, the same holds if the DAT-argument is not an indirect, but a direct object. As discussed in Chapter 5, the direct object is dative when the set of person features of the direct object is a superset of the person features of the subject. Wali and Koul (1997: 228f.) provide the following examples:

(59) a. tsi vuch-ih-am (\*me). you see-2sg.sbj-1sg.obj I.DAT 'You see me.'
b. su vuch-i-y (tse). he see-3sg.sbj-2sg.obj you.DAT 'He will see you.'

(Wali and Koul 1997: 228f.)

Following the reasoning in previous chapters, I suggest that Kashmiri has  $\varphi$ -probes on T and v, as well as on Appl, which agrees with the indirect object. For the Case assigned to the indirect object, however, the person of other arguments does not play a role, i.e. there is no global split involving the indirect object, yet there is no reason why Appl should not agree cyclically, as v does. I will assume that v and Appl assign the sets of CASE features shown in (60):

$$(60) \qquad \nu: \begin{bmatrix} case \\ dep \end{bmatrix} \qquad Appl: \begin{bmatrix} case \\ dep \\ obl \end{bmatrix}$$

The following vocabulary items spell out these sets of features (cf. (24) in Section 5.3.1). The Case assigned by v and Appl is then spelled out equivalently, but they differ in their feature structures (cf. the discussion of Marathi above).

(61) **Vocabulary insertion rules** b.

The derivation of the ditransitive shown in (58) above can be modelled as follows. I will follow Georgala et al. (2008) and Georgala (2012) in assuming a so-called "raising applicative" structure, in which the indirect object is generated in the specifier of VP and moves up to the specifier of the applicative phrase to satisfy Appl's EPP feature (not shown in (62)). Appl will first enter an Agree relation with the indirect object (IO), shown in (62a).





Since Appl's sets of features have not been fully valued, it will attempt to enter another Agree relation, and it finds the direct object. In (62b), Appl enters an Agree relation with the direct object and then assigns Case to the indirect object before moving to v.



Recall that I argued in Chapter 5 that the set of CASE features on v is impoverished when v is valued by two sets of  $\varphi$ -features. Since this rule was restricted to v, it will not apply on Appl. In other words, the set of CASE features Appl assigns will not be affected by the  $\varphi$ -features of its arguments and there is therefore no global case split in the domain of the indirect and the direct object.<sup>10</sup>

(62c) shows the subsequent steps of the derivation. After Appl moves to v, v probes and enters an Agree relation with the direct object. Crucially, v does not agree with the *indirect* object, but only with the direct object and the subject. This is possible if the feature [obl] on the indirect object renders it opaque for agreement. Note that [obl] does not prohibit Appl from agreeing with the indirect object, because agreement happens *before* Case assignment (an instance of so-called *counterbleeding*). Since v is valued by two sets of person features, its set of CASE features is impoverished is repeated in (63) (see Section 5.3.1 for further discussion).

(63)  $[dep] \rightarrow \emptyset / \nu = [\alpha, \beta]$ 

<sup>&</sup>lt;sup>10</sup> I have simplified the order of operations in (62). It is possible that Appl agrees in (A), then moves to v, from where it attempts to agree again before assigning Case to the indirect object. Since neither this order nor the order shown in (62) changes the result, I showed the order in (62) for ease of exposition.



The CASE features on the direct object and the indirect object in (62) are spelled out as follows. Since the set of features assigned to the object is impoverished, the direct object will be spelled out unmarked. The indirect object's set of CASE features, however, will be spelled out as dative. The relevant insertion rules are repeated in (64).

(64) Vocabulary insertion rules  
a. 
$$\begin{bmatrix} & & \\ & & & \\ & & \\ & & \\ & & \\ & & \\ & & \\$$

Summing up, I have followed the logic of previous sections in this chapter, and I have illustrated how a language with three  $\varphi$ -probes fits in to the typology of case and agreement developed here. Adding Kashmiri to the languages discussed before gives us Table 6.3. Note that while for Nepali, Hindi, and Marathi, the tables in this chapter show their case and alignment patterns in the perfective aspect, Table 6.3 shows the imperfective aspect for Kashmiri. This is because only the imperfective aspect shows the global case split discussed in Section 5.2.1 and just above.

	Case alignment	φ on	Agreement alignment	Case opacity	φ, case
German	Т: пом	Т	Т: пом	ACC+	
Nepali	v: ERG	Т	Т: пом	DAT+	
Hindi	v: ERG	Т	T: erg	ERG+	
Marathi	v: ERG	Т	T: erg	ERG+	
*	Т: пом	Т	T: erg		
Sahaptin	T: erg, $v$ : sec	Τ, ν	T: erg, $v$ : sec	OBL+	$\phi \prec case$
Awtuw	T: NOM, $v$ : IND	Τ, ν	T: NOM, $v$ : IND	ACC/DAT+	$\phi \prec case$
Amharic	T: nom, v: ind	Τ, ν	Τ: NOM, <i>ν</i> : SEC	OBL+	case $\prec \varphi$
Hungarian	T: nom, v: ind	Τ, ν	T: NOM, $v$ : IND	DAT+	case $\prec \varphi$
Khanty	Т:  мом, <i>v</i> : both	Τ, ν	Т:	OBL+	case $\prec \varphi$
*	T:, <i>v</i> : sec	Τ, ν	T:, ν: ind		
Kashmiri	T: nom, v: ind	T, v, Appl	T: NOM, $v$ : IND	DAT+	$\phi \prec case$

Table 6.3 Alignment and  $\varphi$ -probes of languages discussed in this chapter

# 6.4 Conclusions

In this chapter, I showed how the patterns of case-marking and agreement in the languages discussed in Chapters 4 and 5 relate to each other.

I argued for an architecture of the grammar in which Case and agreement do not depend on, but can influence each other. I implemented this by suggesting that functional heads that assign Case can, but do not have to, also have a  $\varphi$ -probe.

Combined with an approach to Case that relates different Cases in terms of subset/ superset relations, a restrictive view of the distribution of  $\varphi$ -probes across languages can derive two cross-linguistic generalisations suggested by Bobaljik (2008). The generalisations in question state that agreement with dependent Cases (ergative and accusative) always implies agreement with unmarked cases (absolutive and nominative), and that there are no languages with nominative-accusative alignment in their case system, but ergative-absolutive alignment in their agreement system.

I showed that both of these generalisations can be expressed in the present framework, repeated in (65), and I extended the second generalisation to indirective and secundative case and agreement alignment in ditransitive constructions.

# (65) Bobaljik's first generalisation and CASE features If a given set κ of CASE features includes a feature [α] which blocks agreement, any superset of κ will block agreement as well. Sets not including [α] do not block agreement.

In the following chapter, I turn to a specific implementation of variation in Case and agreement in terms of a parameter hierarchy.

# 7 Parameters of DOM

# 7.1 A hierarchical approach to parameters

In this chapter, I will embed the analysis and the results from the previous chapters in a specific framework of cross-linguistic variation: the idea that (some) morphosyntactic variation across languages can be modelled in terms of parameters relating to properties of functional heads in a given language. Baker (2008: 156) refers to this as the "Borer-Chomsky Conjecture".

In particular, I adopt the view that parameters are not independent of each other but that it is possible to establish hierarchies of dependent parameters, as suggested in recent work by Roberts (2007), Biberauer (2008), Roberts and Holmberg (2010), Roberts (2012), Biberauer, Roberts and Sheehan (2014), Sheehan (2014b), Biberauer and Roberts (2015a,b) and Sheehan (2015), who analyse the distribution of null subjects, diachronic change, and variation in alignment systems across languages.<sup>1</sup>

The following structure, from Sheehan (2014b: 400) illustrates this approach more concretely. The root node in (1) is a first parameter. The way this parameter is set will influence the parameters that follow: if the first parameter is set to N for "no", a language will not set any further parameters in a given hierarchy. The further down a language is located in (1), the more parameter settings it has (see Biberauer *et al.* 2014 for an attempt to link this to the complexity of grammars).

As Roberts (2012) points out, this approach to parametric variation addresses some of the criticisms of the notion of parameter put forth by Newmeyer (2005) and Boeckx (2014). The notion of parameter hierarchies was introduced by Baker (2001).

#### 7 Parameters of DOM

(1) Schematic parameter hierarchy



Roberts (2012), Biberauer *et al.* (2014) and Biberauer and Roberts (2015a,b) further suggest that hierarchies of the type shown in (1) also reflect paths of language acquisition. They assume that learners apply certain strategies to account for a given set of data. On the one hand, an acquirer minimises the number of features that are necessary to account for the data and, on the other hand, she generalises these features. These strategies are shown in (2). They are instantiated in hierarchies like the one shown in (1) and therefore link cross-linguistic variation to language acquisition (cf. Bazalgette 2015).

(2) a. Feature economy (FE)

Postulate as few formal features as possible to account for the input.

b. Input generalisation (IG) If a functional head F sets parameter  $P_j$  to value  $v_i$  then there is a preference for all functional heads to set  $P_j$  to value  $v_i$ .

(Biberauer and Roberts 2015b: 300)

These conditions reflect aspects of a learner's approach to the primary linguistic data she receives. Biberauer *et al.* (2014: 111) suggest that these conditions together shape a learning path, stated in (3) (with some simplifications; cf. also Dresher 2009; Bazalgette 2015):

- (3) a. **Hypothesis I** (prior to analysis of linguistic input) No head H has feature F (satisfies both FE and IG)
  - b. Hypothesis II (at least one occurrence of F in input)All heads have F (FE overridden by input, IG still satisfied)

# c. **Hypothesis III** (at least one non-occurrence of F in input) Some heads have F (both FE and IG overridden by input)

In this chapter, I apply this approach to parameters and parameter hierarchies to the variation I surveyed in Chapters 4–6 and I sketch how Case and agreement across languages can be modelled in this framework. In what follows, I focus on how to capture cross-linguistic generalisations rather than on how a particular system is acquired.

# 7.2 Cross-linguistic variation in DOM

Throughout this dissertation, I have discussed differential argument marking in a range of languages that had some aspects in common, but differed in others. For example, the referentiality of the direct object plays a role in determining object agreement in Hungarian, but the animacy of the object does not. In Awtuw, on the other hand, animacy does play a role (cf. Section 5.4.2.1). In addition, the two languages differ in how differential marking is expressed: in Awtuw, direct objects can appear with or without a case-marker, whereas in Hungarian, direct objects are always case-marked; in this language, only object agreement is differential.

Some languages also show a correlation between movement and differential object marking. In Northern Khanty, objects that trigger object agreement are in a higher structural position than objects that do not (Nikolaeva 1999a, 2001; Dalrymple and Nikolaeva 2011). The same has been suggested for Amharic (Baker 2012, 2015), Persian (Karimi 2003a), Spanish (López 2012), and Hindi (Bhatt and Anagnostopoulou 1996) (among others).

Other languages do not exhibit this kind of link between DOM and movement: the position of the direct object in Hungarian does not seem to influence agreement (or it is very difficult to determine this; see É. Kiss 2008). Danon (2006) also analyses DOM in Hebrew independently of the position of the direct object in the clause.

Of course, some languages without DOM show similar kinds of movement: scrambling and object shift are well-known phenomena that exist in languages that show neither differential case, nor differential object marking (Thráinsson 2001; M. Richards 2004; Vikner 2005). The correlation between a more specific interpretation and such movement operations has also been repeatedly pointed out in the literature, leading,

#### 7 Parameters of DOM

among others, Aissen (2003), Jelinek and Carnie (2003) and López (2012) to suggest that the two phenomena are connected. I agree, but I do not think that such movement is a parameter of DOM, but a more general property of some languages. The literature on these topics is vast; see i.a. Diesing (1992), Bhatt and Anagnostopoulou (1996), Sportiche (1996), Alexiadou and Anagnostopoulou (1997), Aissen (2003), Dayal (2003), Jelinek and Carnie (2003), Karimi (2003a), Kornfilt (2003), Hallman (2004), M. Richards (2004), Baker (2012), López (2012) and Baker (2015) for discussion of the interpretation of noun phrases with respect to scrambling, clitic doubling, object agreement and case-marking.

The point is that DOM is not easily described in terms of movement, or with reference to animacy or specificity alone, since both languages with and without DOM can have scrambling and reflexes of animacy and specificity in their grammar. It is rather a phenomenon that arises through the interaction of variation across a number of properties.

Here, I am going to address some aspects of the variation across the languages discussed in the previous chapter in terms of the following predictors:

**Number of**  $\varphi$ **-probes** I have argued in Chapters 5 and 6 that if the number of  $\varphi$ -probes varies across languages, we gain a natural explanation of the distribution of subject and object agreement across languages. I showed how this variation can capture the trend that object agreement in a given language implies the existence of subject agreement.

While I have argued for the existence of covert agreement in Hungarian in Chapter 4, I showed that there are clear morphosyntactic cues that lead an acquirer to hypothesise that object agreement exists even if it is not spelled out. I assume that such cues are necessary for an acquirer to posit the existence of a  $\varphi$ -probe on a given head (a view compatible with the emergentist approach of Biberauer and Roberts 2015a,b).

In the same vein, I suggested in Section 5.4.2.1 that even though the verb in Awtuw does not show overt object agreement, there is reason to assume that the verb does in fact agree with both the subject and the object: case-marking on the object is sensitive to the relative animacy of both the subject and the object. I argued that case-marking in Awtuw is determined analogously to case-marking in Sahaptin and Kashmiri, where we see overt subject and object agreement.

**Distribution of**  $\varphi$ **-probes** By adopting (and adapting) cyclic Agree (Béjar and Rezac 2009), I argued that it is possible to account for global splits (i.e. agreement or case-marking that is sensitive to properties of both object and subject) while retaining the idea that  $\varphi$ -probes are on T and *v*, rather than assuming several probes on a single head.

This allows for a strict one-to-one mapping between functional heads and  $\varphi$ -probes, i.e. locating a  $\varphi$ -probe that generally agrees with the subject on T, and locating a  $\varphi$ -probe that generally agrees with the direct object on *v*. These tendencies can be overridden, however. The distribution of agreement in Hindi and Marathi shows an example of this: in these languages, there is a single instance of agreement, originating from a  $\varphi$ -probe on T, that is sensitive to the Case of the arguments in the clause. In both Hindi and Marathi, T can agree with either the subject or the object, depending on which one is the highest absolutive argument.

I have followed Bobaljik's (2008) proposal that it is the Case of arguments rather than their grammatical function that determines the distribution of case-marking and agreement across languages: it is not the presence of object agreement that implies the presence of subject agreement in a language, but rather the presence of agreement with arguments bearing morphological case which implies the presence of agreement with unmarked arguments. I implemented this generalisation in terms of abstract Case in Section 6.3.

**Order of operations** I have assumed that v and T (as well as other heads like Appl, P, ...) assign Case. They may but do not necessarily have to have  $\varphi$ -probes. I argued that when they do, the order of Case assignment and agreement can vary.

This makes it possible to capture the difference between languages which have global case splits (like Awtuw, Sahaptin and Kashmiri) and those that do not (like Hungarian). Both types of languages have at least two  $\varphi$ -probes but they differ in the timing of Case assignment with respect to agreement.

#### 7.2.1 Modelling dependencies

The first two properties just mentioned can be illustrated in a parameter hierarchy of the type discussed above. Figure 7.1 illustrates the distribution of  $\varphi$ -probes on func-

#### 7 Parameters of DOM

tional heads that can assign Case in the languages discussed in this dissertation and includes further languages of other types.<sup>2</sup>

N Japanese, Mandarin, Malagasy	Y Kinyarwanda	Y P2: I	Do all heads h	ve φ-probes? s v have a φ-probe?	
		N English, Hindi, Nepali, Marathi	Y N Hungarian, Sahaptin, Awtuw, Yukaghir, Amharic	<b>P4</b> : Does Appl have a φ-probe? Y Kashmiri, Basque	

**P1**: Does any head have a φ-probe?

Figure 7.1Parameters of functional heads and φ-probes (for Japanese, Mandarin, Malagasy<br/>see Baker 2008: 221 and Siewierska 2013; for Basque see Arregi and Nevins 2012;<br/>Odria 2014; for Kinyarwanda, see Beaudoin-Lietz *et al.* 2004 and footnote 2)

Figure 7.1 shows a systematic approach to the cross-linguistic variation of subject and object agreement that is compatible with the architecture of the grammar laid out in Chapter 6. The hierarchy captures the dependency of object agreement on subject agreement in terms of the distribution of  $\varphi$ -probes on functional heads. Note that the parameters do not refer to grammatical function, but only to the distribution of probes

(Beaudoin-Lietz et al. 2004: 183)

See van der Wal (2015) for discussion of differential agreement in Bantu.

<sup>&</sup>lt;sup>2</sup> Some comments with respect to Kinyarwanda are in order: I have not specified in detail what "all heads" refers to; presumably, these are all Case assigning heads, therefore including P. I think that Kinyarwanda seems to fit this classification based on examples like (i), in which locative objects are also expressed as markers on the verb.

 <sup>(</sup>i) Umugoré a-ra-na-ha-ki-zi-ba-ku-n-som-eesh-eesh-er-er-eza.
 woman she-Foc-also-THERE-IT-IT-THEM-YOU-ME-read-CAUS-CAUS-APPL-APPL-IPFV
 'The woman is also making us read it (book) with them (glasses) to you for me (in the house).'
on functional heads. This makes the hierarchy both general and flexible enough to account for the data discussed in the previous chapters.<sup>3</sup>

Consider again Hindi: morphological case and agreement are in complementary distribution. The verb can agree with both the subject and the object, but not at the same time. Both kinds of agreement can be analysed by assuming a single  $\varphi$ -probe on T (Bhatt 2005; Chapter 6). In terms of the hierarchy in Figure 7.1, Hindi is therefore like English in that there is one instance of agreement in the clause. But Hindi differs from English in that the former language can have ergative subjects which are opaque for agreement.

The assumption that additional  $\varphi$ -probes, in languages with more than one probe, are located on heads lower than T reflects the cross-linguistic distribution of agreement (discussed in Chapter 6). I suggested in Section 6.3.4 how attested types of case and agreement alignment in ditransitives can be derived by assuming a  $\varphi$ -probe on v. When languages show agreement with two internal arguments, I assume that they have an additional probe on Appl, a lower head. I argued that this is the case in Kashmiri.

Not all aspects of variation that I mentioned above are included in Figure 7.1, however. First, the difference in whether an indirect object agrees or not is related to the question which cases in a given language count as opaque: in Hungarian, datives cannot trigger agreement, but in Amharic and Sahaptin they can. In other words, languages with two  $\varphi$ -probes, which are shown in the same node in Figure 7.1, nevertheless differ in certain properties.

Another example is the choice of the order of Case assignment and agreement for a given head. In Figure 7.1, Hungarian and Sahaptin are at the same node (they have two  $\varphi$ -probes), but only the latter has a global case split. As shown in Table 6.2 in Chapter 6, this difference can be analysed by assuming that in Hungarian v has the order [CASE <  $\varphi$ ], i.e. case assignment before agreement, while v in Sahaptin has the order [ $\varphi$  < CASE].

Finally, Figure 7.1 does not indicate that the nature of person features and their grammaticalisation is a similar aspect of variation: languages at every level of the hierarchy

Figure 7.1 is similar in some respects to Sheehan's (2014: 403) hierarchy of clausal alignment in that both address properties of v. It also relates to Roberts's (2012: 324) null argument hierarchy, which involves  $\varphi$ -features on the heads v and T. I will not attempt to combine these hierarchies here.

#### 7 Parameters of DOM

differ in how they grammaticalise person features, i.e. what the role of animacy or referentiality is in a given language.

But the variation with respect to Case opacity, the order of agreement and Case assignment on a given head, and the grammaticalisation of person features are parameters in their own right: while they appear to have "narrow scope" with respect to the hierarchy in Figure 7.1, they are no less important or secondary. The value for each language with respect to parameters of Case opacity, an ordering statement or the nature of person features can be thought of as a particular node on another hierarchy that cross-classifies the languages shown in Figure 7.1.

In sum, Figure 7.1 illustrates some of the dependencies that are found in Case and agreement systems across languages and that follow from the analysis proposed in Chapters 4 to 6. Other aspects of variation can be modelled on hierarchies that partly overlap with the hierarchy in Figure 7.1. I believe that this account of the variation in Case and agreement across languages provides a specific answer to the questions of whether and how parameters determine differential object and subject marking, to which I turn now.

### 7.3 Conclusions — Parameters of DOM?

The parameter hierarchy in Figure 7.1 does not make any reference to differential object marking *per se*. This is because differential object marking (or agreement) depends not just on the distribution of functional heads across languages, but on other properties as well, like the ordering of operations on a given head for example.

A second point of variation is the role of referential properties like specificity and definiteness, or semantic properties like animacy. Does a language treat these as purely semantic features that only affect the interpretation of utterances or are they represented in syntax formally (as person features)? In the latter case, we expect to see variation as described and analysed in Chapters 4 and 5.

I take this to mean that there is no single parameter that predicts for any given language whether is has DOM or not. Rather, parameters like the ones in Figure 7.1 determine general properties of Case and agreement in a language that can give rise to seemingly different kinds of DOM. This means that differential object case-marking and differential object agreement are two expressions of the same phenomenon. Differential subject marking does not differ in "deep" ways either: as we have seen in Chapter 5, Sahaptin and Kashmiri are very similar in that the expression of case morphology depends on the sets of  $\varphi$ -features on v. In Sahaptin, differential case is expressed on the subject, whereas in Kashmiri, it is expressed on the object.

Before concluding this chapter, I briefly turn to one more aspect in which languages with DOM might differ. Dalrymple and Nikolaeva (2011: 215), following de Hoop and Malchukov (2007), identify three distinct types of languages with differential object marking (or differential object agreement). Their type 1 languages have DOM based on information structure only. Dalrymple and Nikolaeva (2011) cite the Finno-Ugric languages Khanty and Mansi as examples. Type 2 languages are those in which only "semantic features" determine DOM (p. 215). They cite Hebrew as an example. Their type 3 refers to languages in which both semantic and information structural properties determine DOM, and they suggest that Hindi is a language of this type.

I have not discussed the role of information structure in differential object marking in any detail in this dissertation (for this, see Dalrymple and Nikolaeva 2011 and Iemmolo 2010, 2012), but I have discussed languages of types 2 and 3. Dalrymple and Nikolaeva's (2011) typology does not readily accommodate languages like Hungarian, however. I have argued in Chapters 2 to 4 that differential object agreement in Hungarian cannot be described with reference to the semantic properties of the direct object, and neither by the information structural properties of the direct object.

Instead, I have argued that semantic properties grammaticalised as person features in Hungarian determine object agreement. I further suggested that evidence for this claim comes from the interaction between person features and agreement which gives rise to a kind of "inverse agreement" system. The global splits in Sahaptin, Awtuw, and Yukaghir, discussed in Chapter 5 also depend on person features.

Danon (2006, 2011) suggests a similar approach to DOM in Hebrew. He argues that, rather than being determined by semantic properties of the direct object, e.g. definiteness, DOM is triggered by the "formal marking of definiteness which does not always correlate with semantic or pragmatic characterizations of definiteness" (Danon 2006: 1005).

This characterisation allows for certain mismatches between the reference and the syntactic properties of an expression. Recall Awtuw, in which the relative humanness and animacy of the subject and the direct object plays a role in determining the pres-

#### 7 Parameters of DOM

ence of object case. If this were the only factor, DOM would only depend on semantic properties. But according to Feldman (1986), pronouns and proper names are always case-marked: certain classes of noun phrase that form syntactic natural classes diverge from the *semantic* trend. This can be seen as an instance of *formal marking of animacy*, parallel to the situation found in Hebrew and Hungarian with respect to definiteness (cf. also the discussion of the "delineator" in Fore in Section 5.4.2.2).

Danon (2006) suggests a correlation between the types of properties triggering DOM in a language and the degree to which DOM has been grammaticalised. He assumes that the origin of DOM could lie in functional pressure but that, over time, DOM becomes grammaticalised and the original functional triggers are replaced by formal triggers, like formal features on DPs. This characterisation fits Hungarian diachrony well: as Marcantonio (1985) shows, earlier stages of Hungarian involved differential case marking of topical direct objects (see also É. Kiss 2012, 2013).

This again shows that DOM is not governed by a single parameter: DOM in a single language can evolve over time and can be expressed in different ways, with changes in DOM arising through the interaction with other aspects of grammar. Nevertheless, languages with DOM will be in one of the positions on the parameter hierarchy in Figure 7.1 and the particular expression of DOM in a given language will be constrained by the properties illustrated there, i.e. the number of  $\varphi$ -probes, as well as the other properties discussed above like the nature of person features and the order of Case assignment and agreement.

## 8 Conclusions

In this closing section, I will briefly summarise the main points in this dissertation and their relation to each other. First, I argued that Hungarian object agreement is an instance of differential object marking that is sensitive to person. Second, I adopted and tested M. Richards's (2008) hypothesis that person features can grammaticalise referential properties like definiteness and semantic properties like animacy, and I showed that Hungarian and the languages Sahaptin, Kashmiri, and Awtuw exhibit very similar phenomena that link person, definiteness and animacy. Finally, I argued for a dissociation of Case and agreement and for an architecture of the grammar that locates Case and agreement in syntax and allows for impoverishment of features in the syntactic derivation. I showed that this, coupled with a decomposition of Case features, allows for wide empirical coverage of phenomena involving the differential expression of Case and agreement. This architecture also captures cross-linguistic generalisations about the distribution of alignment in case and agreement.

In Chapters 2 to 4, I argued that what is often called the "objective" or "definite" conjugation should be analysed as differential object agreement. The main claim is that object agreement in Hungarian is sensitive to person features in general, and that it is not just a phenomenon that affects third person objects, as assumed traditionally and also in recent work (Bartos 1999; Coppock 2013).

Evidence for this proposal comes from the distribution of object agreement with personal pronouns in Hungarian, which I discussed in Chapter 4. Not all personal pronoun objects trigger object agreement: when the subject is not first person, and the object is second person, or when the object is first person, the verb does not show agreement with the direct object.

Following É. Kiss (2013), I argued that this distribution of agreement is best explained as a type of inverse agreement system. I adopted and modified Béjar and Rezac's (2009) version of Agree that distinguishes direct from inverse configurations

#### 8 Conclusions

in terms of the number of sets of  $\varphi$ -features a probe, e.g. v, can have. When the subject's  $\varphi$ -features are a superset of the object's, i.e. in direct configurations, v can have two values; in inverse configurations, it can only have one. I provided spell-out rules that also predict the overlap between verbal agreement morphology and possessive morphology in Hungarian.

Further evidence for the suggestion that Hungarian agreement is sensitive to person features comes from the cross-linguistic overview in Chapter 5. I related the observation that Hungarian differential agreement is influenced by the relative person features of the subject and the object to differential case-marking in other languages. Differential subject marking in Sahaptin, and differential object marking in Kashmiri and Awtuw arises in the same contexts as differential agreement in Hungarian.

In addition, the languages discussed in Chapter 5 support the claim that person features can be more than just "person". I explored M. Richards's (2008) suggestion that person features can also express referential properties like specificity or definiteness, or semantic properties like animacy. If this is the case, we expect to find languages in which the relative animacy of the subject and the direct object gives rise to the same phenomena as "person" features: this is exactly what we find in Awtuw.

Hungarian also provides evidence for another aspect of this hypothesis: locating person features on the head D. Person features trigger object agreement and object agreement correlates with a specific interpretation of the direct object; but the presence of person features is linked to D, in line with suggestions by Bernstein (2008), Longobardi (2008), M. Richards (2008), Danon (2011) and Höhn (2015), among others.

Finally, in Chapters 6 and 7, I argued that Case and agreement are part of the syntactic derivation rather than being post-syntactic phenomena. I showed how this can account for the variation described in the previous chapters, as well as powerful crosslinguistic generalisations proposed by Bobaljik (2008) about the distribution of alignment of case and agreement across languages. Chapter 7 relates this analysis to current work in comparative syntax which models variation through the interaction of parameters that organised hierarchically. I argued that the variation in the distribution of subject and object agreement across languages fits with the approach to syntactic variation and acquisition argued for by Roberts (2012), Biberauer *et al.* (2014), Biberauer, Roberts and Sheehan (2014) and Biberauer and Roberts (2015b).

Abbott, Barbara (2010). Reference. Oxford: Oxford University Press.

- Adger, David and Daniel Harbour (2007). "Syntax and syncretisms of the person case constraint". *Syntax* 10.1, 2–37. DOI: 10.1111/j.1467-9612.2007.00095.x.
- Aissen, Judith (1999). "Markedness and subject choice in Optimality Theory". *Natural Language & Linguistic Theory* 17.4, 673–711. DOI: 10.1023/A:1006335629372.
- Aissen, Judith (2003). "Differential object marking: iconicity vs. economy". *Natural Language & Linguistic Theory* 21.3, 435–483. DOI: 10.1023/A:1024109008573.
- Aldridge, Edith (2008). "Generative approaches to ergativity". *Language and Linguistics Compass* 2.5, 966–995. DOI: 10.1111/j.1749-818X.2008.00075.x.
- Alexiadou, Artemis and Elena Anagnostopoulou (1997). "Toward a uniform account of scrambling and clitic doubling". In: *German: Syntactic Problems Problematic Syntax*.
   Ed. by Werner Abraham and Elly van Gelderen. Tübingen: Max Niemeyer, 142–161.
- Anagnostopoulou, Elena (2005). "Strong and weak person restrictions: A feature checking analysis". In: *Clitic and Affix Combinations*. Ed. by Lorie Heggie and Francisco Ordoñez. Amsterdam/Philadelphia: John Benjamins, 199–235. DOI: 10.1075/1a.74.08ana.
- Anand, Pranav and Andrew Nevins (2006). "The locus of ergative case assignment: Evidence from scope". In: *Ergativity: Emerging issues*. Ed. by Alana Johns, Diane Massam and Juvenal Ndayiragije. Dordrecht: Springer, 3–25. DOI: 10.1007/1-4020-4188-8\_1.
- Arregi, Karlos and Andrew Nevins (2012). *Morphotactics: Basque auxiliaries and the structure of spellout*. Dordrecht: Springer.
- Assmann, Anke, Doreen Georgi, Fabian Heck, Gereon Müller and Philipp Weisser (2015). "Ergatives move too early: On an instance of opacity in syntax". *Syntax*. DOI: 10.1111/synt.12034.
- Baker, Mark C. (2001). *The atoms of language: The mind's hidden rules of grammar*. New York: Basic Books.
- Baker, Mark C. (2008). *The syntax of agreement and concord*. Cambridge: Cambridge University Press.
- Baker, Mark C. (2010). "Formal generative typology". In: *The Oxford Handbook of Linguistic Analysis*. Ed. by Bernd Heine and Heiko Narrog. Oxford: Oxford University Press, 285–312.
- Baker, Mark C. (2012). "On the relationship of object agreement and accusative case: Evidence from Amharic". *Linguistic Inquiry* 43.2, 255–274. DOI: 10.1162/LING\_a\_ 00085.

- Baker, Mark C. (2015). *Case: Its principles and its parameters*. Cambridge: Cambridge University Press.
- Bárány, András (2012a). "Hungarian conjugations and differential object marking". In: Proceedings of the First Central-European Conference for Postgraduate Students. Ed. by Balázs Surányi and Diána Varga. Piliscsaba: Pázmány Péter Catholic University, 3– 25. URL: http://cecils.btk.ppke.hu/cecils1proceedings (visited on 20/10/2015).
- Bárány, András (2012b). "On the relation between Hungarian verb paradigms and differential object marking". MA thesis. University of Vienna. URL: http://othes. univie.ac.at/18778/ (visited on 20/10/2015).
- Bárány, András (2014a). "Review of *Objects and information structure* by Mary Dalrymple and Irina Nikolaeva". *Acta Linguistica Hungarica* 61.2, 236–243. DOI: 10.1556/ALing.61.2014.2.4.
- Bárány, András (2014b). "What triggers the Hungarian objective paradigm? A structural and feature-based account". In: ConSOLE XXI: Proceedings of the 21st Conference of the Student Organization of Linguistics in Europe. Ed. by Martin Kohlberger, Kate Bellamy and Eleanor Dutton. Leiden: Leiden University Centre for Linguistics, 21-44.
- Bárány, András (2015a). "Hungarian object agreement with personal pronouns". In: *Proceedings of the 32nd West Coast Conference on Formal Linguistics*. Ed. by Ulrike Steindl, Thomas Borer, Huilin Fang, Alfredo García Pardo, Peter Guekguezian, Brian Hsu, Charlie O'Hara and Iris Chuoying Ouyang. Somerville, MA: Cascadilla Proceedings Project, 208–217.
- Bárány, András (2015b). "Inverse agreement and Hungarian verb paradigms". In: Approaches to Hungarian: Volume 14, Papers from the 2013 Piliscsaba Conference. Ed. by Katalin É. Kiss, Balázs Surányi and Éva Dékány. Amsterdam/Philadelphia: John Benjamins, 37–64. DOI: 10.1075/atoh.14.02bar.
- Bárány, András and Ádám Szalontai (2015). "Agreement with possessed direct objects in Hungarian: An experimential approach". Talk at SinFonIJA 8, University of Ljlubljana. URL: http://andras.barany.at/files/handout-sinfonija-sep15.pdf (visited on 20/10/2015).
- Bartos, Huba (1997). "The nature of object agreement in Hungarian". In: *Proceedings* of the 21st Annual Penn Linguistics Colloquium. Philadelphia, Penn.: University of Pennsylvania, 19–34.
- Bartos, Huba (1999). "Morfoszintaxis és interpretáció: A magyar inflexiós jelenségek szintaktikai háttere [Morphosyntax and interpretation: the syntactic background of Hungarian inflection]". PhD thesis. Budapest: ELTE.
- Bartos, Huba (2001). "Object agreement in Hungarian: A case for minimalism". In: *The Minimalist Parameter*. Ed. by Galina M. Alexandrova and Olga Arnaudova. Current Issues in Linguistic Theory 192. Amsterdam/Philadelphia: John Benjamins, 311–324.

- Bazalgette, Tim (2015). "Algorithmic Acquisition of Focus Parameters". PhD dissertation. University of Cambridge.
- Beaudoin-Lietz, Christa, Derek Nurse and Sarah Rose (2004). "Pronominal object marking in Bantu". In: Proceedings of the 4th World Congress of African Linguistics, New Brunswick 2003. Ed. by Akinbiyi Akinlabi and Oluseye Adesola. Cologne: Rüdiger Köppe.
- Beghelli, Filippo and Tim Stowell (1997). "Distributivity and negation: the syntax of *each* and *every*". In: *Ways of Scope Taking*. Ed. by Anna Szabolcsi. Dordrecht: Kluwer, 71–107.
- Béjar, Susana (2000). "Structural markedness in formal features: Deriving interpretability". *Revue québécoise de linguistique* 28.1, 47–72. DOI: 10.7202/603186ar.
- Béjar, Susana (2003). "Phi-Syntax: A Theory of Agreement". PhD thesis. University of Toronto.
- Béjar, Susana and Milan Rezac (2003). "Person licensing and the derivation of PCC effects: Theory and acquisition". In: *Romance linguistics*. Ed. by Ana-Teresa Pérez-Leroux and Yves Roberge. Amsterdam: John Benjamins, 49–62.
- Béjar, Susana and Milan Rezac (2009). "Cyclic agree". *Linguistic Inquiry* 40.1, 35–73. DOI: 10.1162/ling.2009.40.1.35.
- Benveniste, Émile (1971). "The nature of pronouns". In: *Problems in general linguistics*. Coral Gables, FL: University of Miami Press, 217–222.
- Bernstein, Judy B. (2008). "Reformulating the determiner phrase analysis". *Language and Linguistics Compass* 2.6, 1246–1270. DOI: 10.1111/j.1749-818x.2008.00091.x.
- Bhatt, Rajesh (2005). "Long distance agreement in Hindi-Urdu". *Natural Language & Linguistic Theory* 23.4, 757–807. DOI: 10.1007/s11049-004-4136-0.
- Bhatt, Rajesh (2007). Unaccusativity and Case Licensing. Ms., University of Massachusetts at Amherst.
- Bhatt, Rajesh and Elena Anagnostopoulou (1996). "Object Shift and Specificity: Evidence from *ko*-phrases in Hindi". In: *Papers from the Main Session of CLS 32*. Ed. by Lise Dobrin, Kora Singer and Lisa McNair, 11–22.
- Biberauer, Theresa, ed. (2008). *The Limits of Syntactic Variation*. Amsterdam/Philadel-phia: John Benjamins.
- Biberauer, Theresa, Anders Holmberg, Ian Roberts and Michelle Sheehan, eds. (2010). *Parametric Variation: Null Subjects in Minimalist Theory*. Cambridge: Cambridge University Press.
- Biberauer, Theresa, Anders Holmberg, Ian Roberts and Michelle Sheehan (2014). "Complexity in comparative syntax: The view from modern parametric theory". In: *Measuring Grammatical Complexity*. Ed. by Frederick J. Newmeyer and Laurel B. Preston. Oxford: Oxford University Press, 103–127.
- Biberauer, Theresa and Ian Roberts (2015a). "Rethinking formal hierarchies: A proposed unification". *Cambridge Occasional Papers in Linguistics* 7, 1–31.

- Biberauer, Theresa and Ian Roberts (2015b). "The clausal hierarchy, features, and parameters: The cartography of syntactic structures, Volume 10". In: *Beyond Functional Sequence*. Ed. by Ur Shlonsky. Oxford: Oxford University Press, 295–313.
- Biberauer, Theresa, Ian Roberts and Michelle Sheehan (2014). "No-choice parameters and the limits of syntactic variation". In: *Proceedings of the 31st West Coast Conference on Formal Linguistics*. Ed. by Robert E. Santana-LaBarge. Somerville, MA: Cascadilla Proceedings Project, 46–55.
- Bickel, Balthasar and Yogendra P. Yādava (2000). "A fresh look at grammatical relations in Indo-Aryan". *Lingua* 110.5. DOI: 10.1016/S0024-3841(99)00048-0.
- Bierwisch, Manfred (1967). "Syntactic features in morphology: General problems of socalled pronominal inflection in German". In: *To Honor Roman Jakobson*. Vol. 1. The Hague/Paris: Mouton, 239–270.
- Bittner, Maria and Ken Hale (1996). "The structural determination of case and agreement". *Linguistic Inquiry* 27.1, 1–68.
- Bobaljik, Jonathan David (2008). "Where's phi? Agreement as a postsyntactic operation". In: *Phi Theory: Phi-Features across Modules and Interfaces*. Ed. by Daniel Harbour, David Adger and Susana Béjar. Oxford: Oxford University Press, 295–328.
- Bobaljik, Jonathan David and Phil Branigan (2006). "Eccentric agreement and multiple case-checking: Emerging issues". In: *Ergativity: Emerging Issues*. Ed. by Alana Johns, Diane Massam and Juvenal Ndayiragije. Dordrecht: Springer, 47–77. DOI: 10.1007/1-4020-4188-8\_3.
- Boeckx, Cedric (2014). "What principles and parameters got wrong". In: *Linguistic Variation in the Minimalist Framework*. Ed. by M. Carme Picallo. Oxford: Oxford University Press, 155–178.
- Bossong, Georg (1985). Differentielle Objektmarkierung in den Neuiranischen Sprachen. Tübingen: Gunter Narr.
- Bossong, Georg (1991). "Differential Object Marking in Romance and Beyond". In: *New Analyses in Romance Linguistics*. Ed. by D. Kibbee and D. Wanner. FB Sprachwissenschaft: 16910: John Benjamins, 143–170.
- Brody, Michael and Anna Szabolcsi (2003). "Overt Scope in Hungarian". *Syntax* 6, 19–51. DOI: 10.1111/1467-9612.00055.
- Caha, Pavel (2009). "The nanosyntax of case". PhD thesis. University of Tromsø.
- Carnie, Andrew (2005a). "A phase-geometric approach to multiple marking systems". In: *Perspectives on Phases*. Ed. by Martha McGinnis and Norvin Richards. Vol. 49. MIT Working Papers in Linguistics, 87–102.
- Carnie, Andrew (2005b). "Some remarks on markedness hierarchies: A reply to Aissen 1999 and 2003". *Coyote Papers* 14, 37–50.
- Cecchetto, Carlo (1999). "A comparative analysis of left and right dislocation in Romance". *Studia Linguistica* 53.1, 40–67. DOI: 10.1111/1467-9582.00039.

- Chisarik, Erika (2002). "Partitive Noun Phrases in Hungarian". In: *Proceedings of the LFG02 Conference*. Ed. by Miriam Butt and Tracy Holloway King. CSLI Publications, 96–115.
- Chisarik, Erika and John Payne (2001). "Modelling possessor constructions in LFG: English and Hungarian". In: *Proceedings of the LFG01 Conference*. Ed. by Miriam Butt and Tracy Holloway King. Stanford: CSLI Publications, 33–46.
- Chomsky, Noam (1980). "On binding". Linguistic Inquiry 11.1, 1-46.
- Chomsky, Noam (1981). Lectures on government and binding. Dordrecht: Foris.
- Chomsky, Noam (1982). Some concepts and consequences of the theory of Government and Binding. Cambridge, MA: MIT Press.
- Chomsky, Noam (1995). The Minimalist Program. Cambridge, MA: MIT Press.
- Chomsky, Noam (2000). "Minimalist inquiries: The framework". In: *Step by Step: Essays* on minimalist syntax in honor of Howard Lasnik. Ed. by Roger Martin, David Michaels and Juan Uriagereka. Cambridge, MA: MIT Press, 89–155.
- Chomsky, Noam (2001). "Derivation by phase". In: *Ken Hale: A life in language*. Ed. by Michael Kenstowicz. Cambridge, MA: MIT Press, 1–52.
- Chomsky, Noam (2005). "Three factors in language design". *Linguistic Inquiry* 36.1, 1–22. DOI: 10.1162/0024389052993655.
- Chomsky, Noam (2008). "On phases". In: *Foundational Issues in Linguistic Theory: Essays in Honor of Jean-Roger Vergnaud*. Ed. by Robert Freidin, Carlos P. Otero and Maria Luisa Zubizarreta. Cambridge, MA: MIT Press, 133–166.
- Chomsky, Noam and Howard Lasnik (1995). "The theory of Principles and Parameters". In: Chomsky, Noam. *The Minimalist Program.* Cambridge, MA: MIT Press. Chap. 1, 13–127.
- Chung, Sandra and William A. Ladusaw (2004). *Restriction and saturation*. Cambridge, MA: MIT Press.
- Comrie, Bernard (1980). "Inverse verb forms in Siberia: evidence from Chukchee, Koryak and Kamchadal". *Folia Linguistica Historica* 1, 61–74.
- Comrie, Bernard (1989). *Language universals and linguistic typology*. 2nd edition. Chicago: University of Chicago Press.
- Coppock, Elizabeth (2013). "A semantic solution to the problem of Hungarian object agreement". *Natural Language Semantics* 21.4, 345–371. DOI: 10.1007/s11050-013-9096-7.
- Coppock, Elizabeth and Stephen Wechsler (2010). "Less-travelled paths from pronoun to agreement: the case of the Uralic object conjugations". In: *Proceedings of the LFG10 Conference*. Ed. by Miriam Butt and Tracy Holloway King. CSLI Publications, 165–185.
- Coppock, Elizabeth and Stephen Wechsler (2012). "The objective conjugation in Hungarian: Agreement without phi-features". *Natural Language & Linguistic Theory* 30.3, 699–740. DOI: 10.1007/s11049-012-9165-5.

- Croft, William (2003). *Typology and universals*. Second. Cambridge: Cambridge University Press.
- D'Alessandro, Roberta and Ian Roberts (2008). "Movement and agreement in Italian past participles and defective phases". *Linguistic Inquiry* 39.3, 477–491. DOI: 10.1162/ ling.2008.39.3.477.
- Dalmi, Gréte (1998). "Object agreement in Hungarian". *Working Papers in the Theory of Grammar* 5.2, 91–119.
- Dalrymple, Mary and Irina Nikolaeva (2011). *Objects and information structure*. Cambridge: Cambridge University Press.
- Danon, Gabi (2006). "Caseless nominals and the projection of DP". *Natural Language* & *Linguistic Theory* 24.4, 977–1008. DOI: 10.1007/s11049-006-9005-6.
- Danon, Gabi (2011). "Agreement and DP-internal feature distribution". *Syntax* 14.4, 297–317. DOI: 10.1111/j.1467-9612.2011.00154.x.
- Dayal, Veneeta (2003). "Bare Nominals: Non-Specific and Contrastive Readings under Scrambling". In: *Word Order and Scrambling*. Ed. by Simin Karimi. Malden, MA: Blackwell, 67–90. DOI: 10.1002/9780470758403.
- de Cuba, Carlos and Barbara Ürögdi (2009). "Eliminating factivity from syntax: Sentential complements in Hungarian". In: *Approaches to Hungarian: Volume 11, Papers from the 2007 New York Conference.* Ed. by Marcel den Dikken and Robert M. Vago. Amsterdam/Philadelphia: John Benjamins, 29–63. DOI: 10.1075/atoh.11.03cub.
- de Cuba, Carlos and Barbara Ürögdi (2010). "Clearing up the 'facts' on complementation". University of Pennsylvania Working Papers in Linguistics 16.1, 41–50.
- de Groot, Casper (2009). "Identifiability and verbal cross-referencing markers in hungarian". *Linguistics* 47.4, 997–1019. DOI: 10.1515/LING.2009.035.
- de Hoop, Helen and Andrej L. Malchukov (2007). "On fluid differential case marking: A bidirectional OT approach". *Lingua* 117.9, 1636–1656. DOI: 10.1016/j.lingua.2006.010.
- de Hoop, Helen and Andrej L. Malchukov (2008). "Case-marking strategies". *Linguistic Inquiry* 39.4, 565–587. DOI: 10.1162/ling.2008.39.4.565.
- de Swart, Peter (2007). "Cross-linguistic variation in object marking". PhD dissertation. Utrecht University.
- Deal, Amy Rose (2010). "Ergative case and the transitive subject: a view from Nez Perce". *Natural Language & Linguistic Theory* 28.1, 73–120. DOI: 10.1007/s11049-009-9081-5.
- Déchaine, Rose-Marie and Martina Wiltschko (2002). "Decomposing pronouns". *Linguistic Inquiry* 33.3, 409–442. DOI: 10.1162/002438902760168554.
- Dékány, Éva (2011). "A profile of the Hungarian DP: The interaction of lexicalization, agreement and linearization with the functional sequence". PhD dissertation. University of Tromsø.

- Dékány, Éva (2015). "The syntax of anaphoric possessives in hungarian". *Natural Language & Linguistic Theory*, 1–48. DOI: 10.1007/s11049-014-9278-0.
- den Dikken, Marcel (1999). "On the structural representation of possession and agreement". In: Crossing Boundaries: Advances in the Theory of Central and Eastern European Languages. Ed. by István Kenesei. Amsterdam/Philadelphia: John Benjamins, 137–178.
- den Dikken, Marcel (2004). "Agreement and 'clause union'". In: *Verb Clusters: A study of Hungarian, German and Dutch*. Ed. by Katalin É. Kiss and Henk van Riemsdijk. Amsterdam/Philadelphia: John Benjamins, 445–498. DOI: 10.1075/1a.69.24dik.
- den Dikken, Marcel (2006). *When Hungarians Agree (to Disagree)*. Ms., CUNY Graduate Center. New York.
- den Dikken, Marcel (2012). "On the strategies of forming long A'-dependencies: Evidence from Hungarian". Ms., City University of New York.
- den Dikken, Marcel, ed. (2013). *The Cambridge Handbook of Generative Syntax*. Cambridge: Cambridge University Press.
- den Dikken, Marcel, Anikó Lipták and Zsófia Zvolenszky (2001). "On inclusive reference anaphora: New perspectives from Hungarian". In: *WCCFL 20: Proceedings of the 20th West Coast Conference on Formal Linguistics*. Ed. by Karine Megerdoomian and Leora Anne Bar-el. Somerville, MA: Cascadilla Press, 137–149.
- Diesing, Molly (1992). Indefinites. Cambridge, MA: MIT Press.
- Dresher, B. Elan (2009). *The Contrastive Hierarchy in Phonology*. Cambridge: Cambridge University Press.
- Dryer, Matthew S. (1986). "Primary Objects, Secondary Objects, and Antidative". *Language* 62.4, 808–845. DOI: 10.2307/415173.
- É. Kiss, Katalin (1995). "Definiteness effect revisited". In: *Approaches to Hungarian: Volume 5.* Ed. by István Kenesei. Szeged: JATE, 63–88.
- É. Kiss, Katalin (2000). "The Hungarian noun phrase is like the English noun phrase". In: *Approaches to Hungarian: Volume 7, Papers from the Pécs conference.* Ed. by Gábor Alberti and István Kenesei. Szeged: JATEPress, 119–149.
- É. Kiss, Katalin (2002). *The syntax of Hungarian*. Cambridge: Cambridge University Press.
- É. Kiss, Katalin (2003). "A szibériai kapcsolat avagy miért nem tárgyasan ragozzuk az igét 1. és 2. személyű tárgy esetén". *Magyar Nyelvjárások* 41, 321–326.
- É. Kiss, Katalin (2005). "The inverse agreement constraint in Hungarian: A relic of a Uralic-Siberian sprachbund?" In: Organizing Grammar: Linguistic Studies in Honor of Henk van Riemsdijk. Ed. by Hans Broekhuis, Norbert Corver, Riny Huybregts, Ursula Kleinheinz and Jan Koster. Berlin: Mouton de Gruyter, 108–116. DOI: 10.1515/ 9783110892994.108.
- É. Kiss, Katalin (2008). "Free word order, (non)configurationality, and phases". *Linguistic Inquiry* 39.3, 441–475. DOI: 10.1162/ling.2008.39.3.441.

- É. Kiss, Katalin (2012). "Null pronominal objects in Hungarian: A case of exaptation". *Acta Linguistica Hafniensia* 44.2, 192–206. DOI: 10.1080/03740463.2013.779077.
- É. Kiss, Katalin (2013). "The inverse agreement constraint in Uralic languages". *Finno-Ugric Languages and Linguistics* 2.3, 2–21.
- É. Kiss, Katalin (2014). "Ways of licensing Hungarian external possessors". Acta Linguistica Hungarica 61.1, 45–68. DOI: 10.1556/ALing.61.2014.1.2.
- É. Kiss, Katalin (2015). The Person-Case constraint and the Inverse Agreement Constraint are manifestations of the same information-structural restriction. Talk at the 2015 GLOW Colloquium.
- Ebert, Cornelia and Stefan Hinterwimmer, eds. (2013). *Different Kinds of Specificity Across Languages*. Dordrecht: Springer.
- Embick, David and Rolf Noyer (2007). "Distributed morphology and the syntaxmorphology interface". In: *The Oxford Handbook of Linguistic Interfaces*. Ed. by Gillian Ramchand and Charles Reiss. Oxford: Oxford University Press, 289–324.
- Farkas, Donka (1990). "Two cases of underspecification in morphology". *Linguistic Inquiry* 21.4, 539–550.
- Farkas, Donka and Henriëtte de Swart (2003). *The semantics of incorporation: From argument structure to discourse transparency*. Stanford: CSLI Publications.
- Feldman, Harry (1986). *A grammar of Awtuw*. Canberra: The Australian National University.
- Filimonova, Elena (2005). "The noun phrase hierarchy and relational marking: Problems and counterevidence". *Linguistic Typology* 9.1, 77–113. DOI: 10.1515/lity.2005. 9.1.77.
- Finer, Daniel L. (1985). "The syntax of switch-reference". Linguistic Inquiry 16.1, 35-55.
- Freidin, Robert, Carlos P. Otero and Maria Luisa Zubizarreta, eds. (2008). Foundational Issues in Linguistic Theory. Essays in Honor of Jean-Roger Vergnaud. Cambridge, MA: MIT Press.
- Georgala, Effi (2012). "Applicatives in their structural and thematic function: A Minimalist account of multitransitivity". PhD dissertation. Cornell University.
- Georgala, Effi, Waltraud Paul and John Whitman (2008). "Expletive and thematic Applicatives". In: *Proceedings of the 26th West Coast Conference on Formal Linguistics*. Ed. by Charles B. Chang and Hannah J. Haynie. Somerville, MA: Cascadilla Proceedings Project.
- Georgi, Doreen (2012). "A local derivation of global case splits". In: Local Modelling of Non-Local Dependencies in Syntax. Ed. by Artemis Alexiadou, Tibor Kiss and Gereon Müller. Linguistische Arbeiten 547. Berlin/Boston: de Gruyter, 305–336. DOI: 10. 1515/9783110294774.305.
- Georgi, Doreen (2014). "Opaque Interactions of Merge and Agree: On the Nature and Order of Elementary Operations". PhD thesis. Universität Leipzig.

- Gervain, Judit (2004). "The syntax and semantics of the resumptive dependency in Hungarian focus-raising constructions". In: *Proceedings of the Dislocated Elements Workshop*. Ed. by Benjamin Shaer, Werner Frey and Claudia Maienborn. Berlin: ZAS, 111– 131.
- Gervain, Judit (2009). "Resumption in focus(-raising)". *Lingua* 119.4, 687–707. DOI: 10. 1016/j.lingua.2007.11.008.
- Givón, Talmy (1976). "Topic, pronoun, and grammatical agreement". In: *Subject and Topic*. Ed. by Charles N. Li. New York: Academic Press, 149–188.
- Gulya, János (1966). *Eastern Ostyak chrestomathy*. Bloomington: Indiana University Publications.
- H. Varga, Márta (2010). "Inkongruens jelenségek a magyarban [Incongruent phenomena in Hungarian]". *Hungarológiai Évkönyv* 11.1, 44–53.
- Halle, Morris and Alec Marantz (1993). "Distributed Morphology and the pieces of inflection". In: *The view from building 20*. Ed. by Kenneth Hale and Samuel Jay Keyser. MIT Press, 111–176.
- Halle, Morris and Alec Marantz (1994). "Some key features of Distributed Morphology". *MIT Working Papers in Linguistics* 21, 275–288.
- Hallman, Peter (2004). "NP-interpretation and the structure of predicates". *Language* 80.4, 707–747. DOI: 10.1353/lan.2004.0195.
- Harbour, Daniel, David Adger and Susana Béjar, eds. (2008). *Phi Theory: Phi-Features across Modules and Interfaces*. Oxford: Oxford University Press.
- Harley, Heidi and Rolf Noyer (2003). "Distributed Morphology". In: *The Second* Glot International *State-of-the-Article Book*. Berlin/New York: Mouton de Gruyter, 463– 496.
- Harley, Heidi and Elizabeth Ritter (2002). "Person and number in pronouns: A feature-geometric analysis". *Language* 78.3, 482–526. DOI: 10.1353/lan.2002.0158.
- Haspelmath, Martin (2005). "Argument marking in ditransitive alignment types". *Linguistic Discovery* 3.1, 1–21. DOI: 10.1349/PS1.1537-0852.A.280.
- Haspelmath, Martin (2006). "Against markedness (and what to replace it with)". *Journal* of Linguistics 42.1, 25–70. DOI: 10.1017/S0022226705003683.
- Haspelmath, Martin (2008). "Parametric versus functional explanations of syntactic universals". In: *The Limits of Syntactic Variation*. Ed. by Theresa Biberauer. Amster-dam/Philadelphia: John Benjamins, 75–107.
- Haspelmath, Martin (2009). "The best-supported language universals refer to scalar patterns deriving from processing cost". *Behavioral and Brain Sciences* 32.5, 457–458. DOI: 10.1017/S0140525X09990689.
- Havas, Ferenc (2004). "Objective conjugation and medialisation". *Acta Linguistica Hungarica* 51.1–2, 95–141. DOI: 10.1556/ALing.51.2004.1–2.5.
- Heck, Fabian and Gereon Müller (2007). "Extremely local optimization". In: Proceedings of the thirty-fourth Western Conference on Linguistics: WECOL 2006. Ed. by Erin

Bainbridge and Brian Agbayani. Fresno: Department of Linguistics, California State University.

- Höhn, Georg F. K. (2015). "Unagreement is an illusion: Apparent person mismatches and nominal structure". *Natural Language & Linguistic Theory*. DOI: 10.1007/s11049-015-9311-y.
- Höhn, Georg F. K. (in progress). "Non-possessive person in the nominal domain". PhD dissertation. University of Cambridge.
- Holmberg, Anders and Ian Roberts (2013). "The syntax-morphology relation". *Lingua* 130, 111–131. DOI: 10.1016/j.lingua.2012.10.006.
- Iemmolo, Giorgio (2010). "Topicality and differential object marking". *Studies in Language* 34.2, 239–272. DOI: 10.1075/sl.34.2.01iem.
- Iemmolo, Giorgio (2012). "Differential object marking". PhD thesis. University of Pavia.
- Ionin, Tania (2006). "This is definitely specific: Specificity and definiteness in article systems". *Natural Language Semantics* 14.2, 175–234. DOI: 10.1007/s11050-005-5255-9.
- Ionin, Tania (2013). "Pragmatic Variation Among Specificity Markers". In: Different Kinds of Specificity Across Languages. Ed. by Cornelia Ebert and Stefan Hinterwimmer. Dordrecht: Springer, 75–103.
- Irimia, Monica-Alexandrina (2005). "Types of secondary predication". *Toronto Working Papers in Linguistics* 25, 20–29.
- Jakobson, Roman (1971 [1936]). "Beitrag zur allgemeinen Kasuslehre: Gesamtbedeutungen der russischen Kasus". In: *Word and Language*. Vol. II. Selected Writings. The Hague/Paris: Mouton, 23–71.
- Jánosi, Adrienn, Jeroen van Craenenbroeck and Guido Vanden Wyngaerd (2014). "Long split focalization in Hungarian and the typology of A'-dependencies". *Lingua* 150, 117–136. DOI: 10.1016/j.lingua.2014.07.009.
- Jelinek, Eloise and Andrew Carnie (2003). "Argument hierarchies and the mapping principle". In: *Formal Approaches to Function in Grammar: In honor of Eloise Jelinek*. Ed. by Andrew Carnie, Heidi Harley and MaryAnn Willie. Amsterdam/Philadelphia: John Benjamins, 265–296. DOI: 10.1075/la.62.20jel.
- Johns, Alana, Diane Massam and Juvenal Ndayiragije, eds. (2006). *Ergativity: Emerging Issues*. Dordrecht: Springer. DOI: 10.1007/1-4020-4188-8.
- Kalin, Laura (2014). "Aspect and Argument Licensing in Neo-Aramaic". PhD dissertation. Los Angeles: UCLA.
- Kalin, Laura (2015). "Licensing and Differential Object Marking: The View from Neo-Aramaic". Ms., University of Connecticut.
- Kalin, Laura and Coppe van Urk (2015). "Aspect splits without ergativity: Agreement asymmetries in Neo-Aramaic". *Natural Language & Linguistic Theory* 33.2, 659–702. DOI: 10.1007/s11049-014-9262-8.

- Kálmán, László (1995). "Definiteness Effect Verbs in Hungarian". In: *Approaches to Hun*garian. Ed. by István Kenesei. Vol. 5. Szeged: JATE Szeged, 221–242.
- Kamper, Gergely (2006). "Differential Object Marking". ELTE SEAS Working Papers in Linguistics 7. URL: http://seas3.elte.hu/delg/publications/even/2006/06ka.pdf (visited on 20/10/2015).
- Karimi, Simin (2003a). "On Object Positions, Specificity, and Scrambling in Persian".
  In: Word Order and Scrambling. Ed. by Simin Karimi. Malden, MA: Blackwell, 91– 125. DOI: 10.1002/9780470758403.ch5.
- Karimi, Simin, ed. (2003b). Word order and scrambling. Malden, MA: Blackwell. DOI: 10.1002/9780470758403.
- Keenan, Edward L. (2003). "The definiteness effect: Semantics or pragmatics?" *Natural Language Semantics* 11.2, 187–216. DOI: 10.1023/A:1024400425028.
- Keenan, Edward L. (2008). "The definiteness of subjects and objects in Malagasy". In: *Case and Grammatical Relations: Studies in honor of Bernard Comrie*. Ed. by Greville G. Corbett and Michael Noonan. Amsterdam/Philadelphia: John Benjamins, 241–261. DOI: 10.1075/tsl.81.12kee.
- Keine, Stefan (2010). *Case and agreement from fringe to core: A minimalist approach*. Berlin/New York: De Gruyter.
- Keine, Stefan (2013). "Deconstructing switch-reference". *Natural Language & Linguistic Theory* 31.3, 767–826. DOI: 10.1007/s11049-013-9194-8.
- Keine, Stefan and Gereon Müller (2008). "Differential argument encoding by impoverishment". In: *Scales*. Ed. by Marc Richards and Andrej L. Malchukov. Linguistische Arbeitsberichte 86. Universität Leipzig, 83–136.
- Kenesei, István (1994). "Subordinate clauses". In: *The Syntactic Structure of Hungarian*. Ed. by Ferenc Kiefer and Katalin É. Kiss. Vol. 27. Syntax and Semantics. New York: Academic Press, 275–354.
- Kiefer, Ferenc (2003). "Alaktan [Morphology]". In: Új magyar nyelvtan [New Hungarian grammar]. Ed. by Katalin É. Kiss, Ferenc Kiefer and Péter Siptár. Budapest: Osiris Kiadó, 187–284.
- Kiefer, Ferenc and Katalin É. Kiss, eds. (1994). *The Syntactic Structure of Hungarian*. Vol. 27. Syntax and Semantics. New York: Academic Press.
- Koopman, Hilda and Anna Szabolcsi (2000). Verbal Complexes. Cambridge, MA: MIT Press.
- Kornai, András (1989). "A főnévi csoport egyeztetése [Agreement in the noun phrase]". *Általános nyelvészeti tanulmányok* 17, 183–211.
- Kornfilt, Jaklin (2003). "Scrambling, Subscrambling, and Case in Turkish". In: *Word Order and Scrambling*. Ed. by Simin Karimi. Malden, MA: Blackwell, 125–155. DOI: 10.1002/9780470758403.ch6.

- Laczkó, Tibor (2007). "Revisiting possessors in Hungarian DPs: A new perspective". In: *Proceedings of the LFG07 Conference*. Ed. by Miriam Butt and Tracy Holloway King. CSLI Publications, 343–362.
- Lazard, Gilbert (2001). "Le marquage différentiel de l'objet". In: Language Typology and Language Universals: An International Handbook. Ed. by Martin Haspelmath, Ekkehard König, Wulf Oesterreicher and Wolfgang Raible. Vol. 2. Berlin/Boston: De Gruyter Mouton, 873–886.
- Legate, Julie Anne (2008). "Morphological and abstract case". *Linguistic Inquiry* 39.1, 55–101. DOI: 10.1162/ling.2008.39.1.55.
- Legate, Julie Anne (2012). "Types of ergativity". *Lingua* 122.3, 181–191. DOI: 10.1016/j. lingua.2011.10.014.
- Lewis, M. Paul, Gary F. Simons and Charles D. Fennig, eds. (2013). *Ethnologue: Languages of the world*. Seventeenth edition. Dallas, Texas: SIL International. URL: http://www.ethnologue.com.
- Lochbihler, Bethany (2008). "Person Encoding in the Ojibwe Inverse System". In: *Scales*. Ed. by Marc Richards and Andrej L. Malchukov. Linguistische Arbeitsberichte 86. Universität Leipzig, 295–315.
- Lochbihler, Bethany (2012). "Aspects of Argument Licensing". PhD thesis. Montréal: McGill University.
- Longobardi, Giuseppe (1994). "Reference and proper names: A theory of n-movement in syntax and logical form". *Linguistic Inquiry* 25.4, 609–665.
- Longobardi, Giuseppe (2001). "The structure of DPs: Some principles, parameters and problems". In: *The Handbook of Contemporary Syntactic Theory*. Ed. by Mark Baltin and Chris Collins. Oxford: Blackwell, 562–603.
- Longobardi, Giuseppe (2005). "Toward a unified grammar of reference". Zeitschrift für Sprachwissenschaft 24.1, 5–44. DOI: 10.1515/zfsw.2005.24.1.5.
- Longobardi, Giuseppe (2008). "Reference to individuals, person, and the variety of mapping parameters". In: *Essays on Nominal Determination: From morphology to discourse management.* Ed. by Henrik Høeg Müller and Alex Klinge. Amsterdam/Philadelphia: John Benjamins, 189–211. DOI: 10.1075/slcs.99.11lon.
- López, Luis (2012). *Indefinite objects: Scrambling, choice functions, and differential marking.* Cambridge, MA: MIT Press.
- Lyons, Christopher (1999). Definiteness. Cambridge: Cambridge University Press.
- Malchukov, Andrej L. (2008). "Animacy and asymmetries in differential case marking". *Lingua* 118.2, 203–221. DOI: 10.1016/j.lingua.2007.02.005.
- Malchukov, Andrej L. (2013). "Verbalization and insubordination in Siberian languages". In: *Shared Grammaticalization: With special focus on the Transeurasian languages*. Ed. by Martine Robbeets and Hubert Cuyckens, 177–208. DOI: 10.1075/slcs.132.14mal.

- Maleczki, Márta (2001). "Indefinite Arguments in Hungarian". In: *Argument Structure in Hungarian*. Ed. by István Kenesei. Budapest: Akadémiai Kiádo, 157–199.
- Marantz, Alec (1991). "Case and licensing". In: ESCOL '91: Proceedings of the Eighth Eastern States Conference on Linguistics. Ohio State University, 234–253.
- Marcantonio, Angela (1985). "On the definite vs. indefinite conjugation in Hungarian: A typological and diachronic analysis". *Acta Linguistica Academiae Scientiarum Hungaricae* 35.3-4, 267–298.
- Maslova, Elena (2003). *A grammar of Kolyma Yukaghir*. Berlin, Boston: De Gruyter Mouton.
- McFadden, Thomas (2004). "The Position of Morphological Case in the Derivation: A Study on the Syntax-Morphology Interface". PhD thesis. Philadelphia, Penn.: University of Pennsylvania.
- McGinnis, Martha (2005). "On markedness asymmetries in person and number". *Language* 81.3, 699–718. DOI: 10.1353/lan.2005.0141.
- Mithun, Marianne (1999). *The languages of Native North America*. Cambridge: Cambridge University Press.
- Morimoto, Yukiko (2002). "Prominence Mismatches and Differential Object Marking in Bantu". In: *Proceedings of the LFG02 Conference, National Technical University of Athens*. Ed. by Miriam Butt and Tracy Holloway King. CSLI Publications, 292–314.
- Moulton, Keir (2015). "CPs: Copies and compositionality". *Linguistic Inquiry* 46.2, 305–342. DOI: 10.1162/ling\_a\_00183.
- Müller, Gereon (2002). "Remarks on nominal inflection in German". In: *More than Words: A Festschrift for Dieter Wunderlich.* Ed. by Ingrid Kaufmann and Barbara Stiebels. Berlin: Akademie Verlag, 113–145.
- Müller, Gereon (2004a). "Argument Encoding and the Order of Elementary Operations". Ms., IDS Mannheim.
- Müller, Gereon (2004b). "On decomposing inflection class features: Syncretism in Russian noun inflection". In: *Explorations in Nominal Inflection*. Ed. by Gereon Müller, Lutz Gunkel and Gisela Zifonun. Berlin: Mouton de Gruyter, 189–227. DOI: 10.1515/ 9783110197501.189.
- Müller, Gereon (2005). "Pro-Drop and Impoverishment". Ms., Universität Leipzig.
- Müller, Gereon (2006). "Extended Exponence by Enrichment: Argument Encoding in German, Archi and Timucua". Ms., Universität Leipzig.
- Müller, Gereon (2007). "Some consequences of an impoverishment-based approach to morphological richness and pro-drop". Ms., Universität Leipzig.
- Müller, Gereon (2009). "Ergativity, accusativity, and the order of Merge and Agree". In: *Explorations of Phase Theory*. Ed. by Kleanthes K. Grohmann. Berlin, Boston: De Gruyter Mouton, 269–308. DOI: 10.1515/9783110213966.269.
- Müller, Gereon (2010). "On deriving CED effects from the PIC". *Linguistic Inquiry* 41.1, 35–82. DOI: 10.1162/ling.2010.41.1.35.

- Næss, Åshild (2004). "What markedness marks: The markedness problem with direct objects". *Lingua* 114.9–10, 1186–1212. DOI: 10.1016/j.lingua.2003.07.005.
- Nash, Lea and Alain Rouveret (2002). "Cliticization as unselective attract". *Catalan Journal of Linguistics* 1, 157–199.
- Nevins, Andrew (2007). "The representation of third person and its consequences for person-case effects". *Natural Language & Linguistic Theory* 25.2, 273–313. DOI: 10. 1007/s11049-006-9017-2.
- Nevins, Andrew (2011). "Multiple agree with clitics: person complementarity vs. omnivorous number". *Natural Language & Linguistic Theory* 29.4, 939–971. DOI: 10.1007/ s11049-011-9150-4.
- Newmeyer, Frederick J. (2002a). "A rejoinder to Bresnan and Aissen". *Natural Language* & *Linguistic Theory* 20.1, 97–99. DOI: 10.1023/A:1014213624881.
- Newmeyer, Frederick J. (2002b). "Optimality and functionality: A critique of functionallybased optimality-theoretic syntax". *Natural Language & Linguistic Theory* 20.1, 43–80. DOI: 10.1023/A:1014290005775.
- Newmeyer, Frederick J. (2005). *Possible and probable languages*. Oxford: Oxford University Press.
- Nichols, Johanna (1986). "Head-marking and dependent-marking grammar". *Language* 62.1, 56–119. DOI: 10.1353/lan.1986.0014.
- Nikolaeva, Irina (1999a). "Object agreement, grammatical relations, and information structure". *Studies in Language* 23, 331–376. DOI: 10.1075/sl.23.2.05nik.
- Nikolaeva, Irina (1999b). Ostyak. München: Lincom Europa.
- Nikolaeva, Irina (2001). "Secondary topic as a relation in information structure". *Linguistics* 39.1, 1–49. DOI: 10.1515/ling.2001.006.
- Odria, Ane (2014). "Differential object marking and the nature of dative case in Basque dialects". *Linguistic Variation*, 289–317. DOI: 10.1075/lv.14.2.03odr.
- Ortmann, Albert and Doris Gerland (2014). "She loves you, *-ja -ja -ja*: Objective conjugation and pragmatic possession in Hungarian". In: *Meaning and Grammar of Nouns and Verbs*. Ed. by Doris Gerland, Christian Horn, Anja Latrouite and Albert Ortmann. Düsseldorf: dup, 269–313.
- Pandharipande, Rajeshwari V. (1997). *Marathi*. Descriptive grammars. London: Routledge.
- Peredy, Márta (2009). *A stochastic account for variation in Hungarian object agreement.* Ms., Research Institute in Linguistics of the Hungarian Academy of Sciences, Budapest.
- Pesetsky, David (1987). "*Wh*-in-situ: Movement and unselective binding". In: *The Representation of (In)definiteness*. Ed. by Eric J. Reuland and Alice ter Meulen. Cambridge, MA: MIT Press, 98–129.
- Pesetsky, David and Esther Torrego (2007). "The syntax of valuation and the interpretability of features: Syntactic derivation and interpretation". In honor of Joseph

E. Emonds. In: *Phrasal and Clausal Architecture*. Ed. by Simin Karimi, Vida Samiian and Wendy K. Wilkins. Amsterdam/Philadelphia: John Benjamins, 262–294. DOI: 10. 1075/la.101.14pes.

- Pete, István (2006). "A határozott tárgyas ragozásról [On the definite objective conjugation]". *Magyar Nyelvőr* 130.3, 317–324.
- Piñón, Christopher (2006). "Definiteness effect verbs". In: *Event Structure and the Left Periphery: Studies on Hungarian.* Ed. by Katalin É. Kiss. Dordrecht: Springer, 75–90.
- Preminger, Omer (2009). "Breaking agreements: Distinguishing agreement and clitic doubling by their features". *Linguistic Inquiry* 40.4, 619–666. DOI: 10.1162/ling. 2009.40.4.619.
- Preminger, Omer (2011). "Agreement as a Fallible Operation". PhD thesis. Cambridge, MA: MIT.
- Preminger, Omer (2013). "That's not how you agree: A reply to Zeijlstra". *The Linguistic Review* 30.3, 491–500. DOI: 10.1515/tlr-2013-0015.
- Preminger, Omer (2014). Agreement and its failures. Cambridge, MA: MIT Press.
- Prince, Alan and Paul Smolensky (2004). *Optimality theory: Constraint interaction in generative grammar.*
- Pylkkänen, Liina (2008). Introducing Arguments. Cambridge, MA: MIT Press.
- Rackowski, Andrea and Norvin Richards (2005). "Phase Edge and Extraction: A Tagalog Case Study". *Linguistic Inquiry* 36.4, 565–599. DOI: 10.1162/002438905774464368.
- Rácz, Endre (1968). "Mondattan [Syntax]". In: *A mai magyar nyelv [Present-day Hun-garian]*. Ed. by Endre Rácz. Budapest: Tankönyvkiadó, 205–457.
- Rákosi, György (2014). "Possessed by something out there: On anaphoric possessors in Hungarian". *Argumentum* 10, 548–559.
- Rebrus, Péter (2000). "Morfofonológiai jelenségek [Morphophonological phenomena]".
  In: *Strukturális magyar nyelvtan [Structural Hungarian grammar]*. Ed. by Ferenc Kiefer. Vol. 3. Morfológia [Morphology]. Budapest: Akadémiai Kiádo, 763–947.
- Rebrus, Péter (2005). "Hogyan inflektál a magyar? [How does Hungarian inflect?]" In: Az ezerarcú elme: Tanulmányok Pléh Csaba 60. születésnapjára. Ed. by Judit Gervain, Kristóf Kovács, Ágnes Lukács and Mihály Racsmány. Budapest: Akadémiai Kiádo, 56–67.
- Rédei, Károly (1965). Northern Ostyak chrestomathy. Bloomington: Indiana University Publications.
- Rezac, Milan (2008a). "Phi-agree and theta-related case". In: *Phi Theory: Phi-Features across Modules and Interfaces*. Ed. by Daniel Harbour, David Adger and Susana Béjar. Oxford: Oxford University Press, 83–129.
- Rezac, Milan (2008b). "The syntax of eccentric agreement: the person case constraint and absolutive displacement in Basque". *Natural Language & Linguistic Theory* 26.1, 61–106. DOI: 10.1007/s11049-008-9032-6.

- Rezac, Milan, Pablo Albizu and Ricardo Etxepare (2014). "The structural ergative of Basque and the theory of Case". *Natural Language & Linguistic Theory* 32, 1273–1330. DOI: 10.1007/s11049-014-9239-7.
- Richards, Marc (2004). "Object Shift, Scrambling, and Symmetrical Syntax". PhD thesis. University of Cambridge.
- Richards, Marc (2008). "Defective agree, case alternations, and the prominence of person". In: *Scales*. Ed. by Marc Richards and Andrej L. Malchukov. Linguistische Arbeitsberichte 86. Universität Leipzig, 137–161.
- Richards, Marc and Andrej L. Malchukov, eds. (2008). *Scales*. Linguistische Arbeitsberichte 86. Universität Leipzig.
- Richards, Norvin (2010). Uttering trees. Cambridge, MA: MIT Press.
- Rigsby, Bruce and Noel Rude (1996). "Sketch of Sahaptin, a Sahaptian language". In: *Handbook of North American Indians*. Ed. by Ives Goddard. Vol. 17: Languages. Washington: Smithsonian Institution, 666–692.
- Rizzi, Luigi (1982). Issues in Italian syntax. Dordrecht: Foris.
- Rizzi, Luigi (1986). "Null objects in italian and the theory of pro". *Linguistic Inquiry* 17.3, 501–557.
- Roberts, Ian (1988). "Predicative APs". Linguistic Inquiry 19.4, 703-710.
- Roberts, Ian (2007). Diachronic Syntax. Oxford: Oxford University Press.
- Roberts, Ian (2010). "A deletion analysis of null subjects". In: Parametric Variation: Null Subjects in Minimalist Theory. Ed. by Theresa Biberauer, Anders Holmberg, Ian Roberts and Michelle Sheehan. Cambridge: Cambridge University Press, 58–87.
- Roberts, Ian (2012). "Macroparameters and minimalism: A programme for comparative research". In: *Parameter Theory and Linguistic Change*. Ed. by Charlotte Galves, Sonia Cyrino, Ruth Lopes and Juanito Avelar. Oxford: Oxford University Press, 320–335.
- Roberts, Ian and Anders Holmberg (2010). "Introduction: Parameters in minimalist theory". In: *Parametric Variation: Null Subjects in Minimalist Theory*. Ed. by Theresa Biberauer, Anders Holmberg, Ian Roberts and Michelle Sheehan. Cambridge: Cambridge University Press, 1–57.
- Rocquet, Amélie (2013). "Splitting Objects: A Nanosyntactic account of direct object marking". PhD dissertation. Ghent: Ghent University.
- Rooryck, Johan (2006). "Binding into pronouns". *Lingua* 116.10, 1561–1579. DOI: 10. 1016/j.lingua.2005.08.005.
- Rounds, Carol (2002). *Hungarian: An essential grammar*. London and New York: Routledge.
- Sag, Ivan A., Thomas Wasow and Emily M. Bender (2003). *Syntactic Theory: A Formal Introduction*. Ms., CSLI.
- Schultze-Berndt, Eva and Nikolaus P. Himmelmann (2004). "Depictive secondary predicates in crosslinguistic perspective". *Linguistic Typology* 8.1, 59–131. DOI: 10.1515/ lity.2004.004.

- Schwartz, Linda (1988). "Asymmetric feature distribution in pronoun 'coordinations'".
   In: Agreement in Natural Language: Approaches, Theories, Descriptions. Ed. by Michael Barlow and Charles A. Ferguson. Stanford: CSLI Publications, 237–249.
- Scott, Graham (1978). *The Fore language of Papua New Guinea*. Canberra: The Australian National University.
- Seržant, Ilja A. and Alena Witzlack-Makarevich (2015). "Differential argument marking: an introduction". Ms., University of Konstany/Kiel. URL: https://www.staff.unimainz.de/iserzan/Introduction.pdf (visited on 20/10/2015).
- Sheehan, Michelle (2011). "A note on case assignment to CP". Snippets 24, 18-19.
- Sheehan, Michelle (2014a). "Case and Ergativity". Ms., University of Cambridge.
- Sheehan, Michelle (2014b). "Towards a parameter hierarchy for alignment". In: Proceedings of the 31st West Coast Conference on Formal Linguistics. Ed. by Robert E. Santana-LaBarge. Somerville, MA: Cascadilla Proceedings Project, 399–408.
- Sheehan, Michelle (2015). "Parameterising ergativity: An inherent case approach". lingbuzz/002464, to appear in Jessica Coon, Diane Massam and Lisa Travis (eds.), *The Oxford Handbook of Ergativity*.
- Sheehan, Michelle and Wolfram Hinzen (2011). "Moving towards the edge". *Linguistic Analysis* 37.3-4, 406–458.
- Siddiqi, Daniel (2010). "Distributed Morphology". Language and Linguistics Compass 4.7, 524–542. DOI: 10.1111/j.1749-818x.2010.00212.x.
- Siewierska, Anna (2013). "Verbal person marking". In: The World Atlas of Language Structures Online. Ed. by Matthew S. Dryer and Martin Haspelmath. Leipzig: Max Planck Institute for Evolutionary Anthropology. URL: http://wals.info/chapter/ 102 (visited on 20/10/2015).
- Silverstein, Michael (1976). "Hierarchy of features and ergativity". In: *Grammatical Categories in Australian Languages*. Ed. by R. M. W. Dixon. Canberra: Australian Institute of Aboriginal Studies, 112–171.
- Skorik, Piotr Ja (1977). Грамматика чукотского языка [Grammar of Chukchi]: Частъ II: глагол, наречпе, служебные слова [Volume II: verb, adverb and auxiliary words]. Leningrad: Nauka.
- Sportiche, Dominique (1996). "Clitic constructions". In: *Phrase Structure and the Lexicon*. Ed. by Johan Rooryck and Laurie Zaring. Dordrecht: Kluwer Academic Publishers, 213–276.
- Szabolcsi, Anna (1983). "The possessor that ran away from home". *The Linguistic Review* 3, 89–102.
- Szabolcsi, Anna (1986). "From the definiteness effect to lexical integrity". In: *Topic, Focus and Configurationality*. Ed. by Werner Abraham and Sjaak de Meij. Amsterdam: John Benjamins, 321–348.
- Szabolcsi, Anna (1987). "Functional categories in the NP". In: *Approaches to Hungarian: Volume 2, Theories and analyses.* Ed. by István Kenesei. Szeged: JATE, 167–189.

- Szabolcsi, Anna (1994). "The noun phrase". In: *The Syntactic Structure of Hungarian*. Ed. by Ferenc Kiefer and Katalin É. Kiss. Vol. 27. Syntax and Semantics. New York: Academic Press, 179–274.
- Szabolcsi, Anna, ed. (1997). Ways of Scope Taking. Dordrecht: Kluwer.
- Takahashi, Shoichi (2010). "The hidden side of clausal complements". *Natural Language & Linguistic Theory* 28.2, 343–380. DOI: 10.1007/s11049-010-9091-3.
- Thráinsson, Höskuldur (2001). "Object shift and scrambling". In: *The Handbook of Contemporary Syntactic Theory*. Ed. by Mark Baltin and Chris Collins. Oxford: Blackwell, 148–202.
- Törkenczy, Miklós (2002). Practical Hungarian grammar. Budapest: Corvina.
- Trommer, Jochen (2005). "Hungarian has no portmanteau agreement". In: *Approaches to Hungarian: Volume 9, Papers from the Düsseldorf Conference.* Ed. by Péter Siptár and Christopher Piñón. Budapest: Akadémiai Kiadó, 283–302.
- Uriagereka, Juan (1995). "Aspects of the syntax of clitic placement in Western Romance". *Linguistic Inquiry* 26.1, 79–123.
- Valentine, Randy (2001). *Nishnaabemwin reference grammar*. Toronto: University of Toronto Press.
- van der Wal, Jenneke (2015). "Bantu object clitics as defective goals". *Revue romaine de linguistique* 60.2-3, 277–296.
- Vergnaud, Jean-Roger (2008 [1977]). "Letter to Noam Chomsky and Howard Lasnik on "Filters and Control," April 17, 1977". In: *Foundational Issues in Linguistic Theory. Essays in Honor of Jean-Roger Vergnaud*. Ed. by Robert Freidin, Carlos P. Otero and Maria Luisa Zubizarreta. Cambridge, MA: MIT Press, 3–15.
- Vikner, Sten (2005). "Object shift". In: *The Blackwell Companion to Syntax*. Ed. by Martin Everaert and Henk van Riemsdijk. Vol. III. Oxford: Blackwell, 392–436.
- Virtanen, Susanna (2012). "Variation in three-participant constructions in Eastern Mansi". *Linguistica Uralica* 48.2, 120–130. DOI: 10.3176/lu.2012.2.04.
- Virtanen, Susanna (2014). "Pragmatic direct object marking in Eastern Mansi". *Linguistics* 52.2, 391–413. DOI: 10.1515/ling-2013-0067.
- Virtanen, Susanna (2015). "Transitivity in Eastern Mansi: An Information Structural Approach". PhD thesis. Helsinki: University of Helsinki.
- von Heusinger, Klaus (2011). "Specificity". In: *Semantics: An International Handbook of Natural Language Meaning*. Ed. by Klaus von Heusinger, Claudia Maienborn and Paul Portner. Vol. 2. Berlin: de Gruyter, 1024–1057.
- Wali, Kashi and Omkar N. Koul (1997). *Kashmiri: A cognitive-descriptive grammar*. New York: Routledge.
- Wiese, Bernd (1999). "Unterspezifizierte Paradigmen: Form und Funktion in der pronominalen Deklination". *Linguistik online* 4.3. URL: http://www.linguistik-online. com/3\_99/wiese.html (visited on 20/10/2015).

- Woisetschlaeger, Erich (1983). "On the question of definiteness in "An old man's book"". *Linguistic Inquiry* 14, 137–154.
- Woolford, Ellen (1997). "Four-way case systems: Ergative, nominative, objective and accusative". *Natural Language & Linguistic Theory* 15.1, 181–227. DOI: 10.1023/A: 1005796113097.
- Woolford, Ellen (1999). "Animacy hierarchy effects on Object Agreement". In: New Dimensions in African Linguistics and Languages. Ed. by Paul F. A. Kotey. Africa World Press, 203–216. URL: http://people.uMAedu/ellenw/Woolford%20Animacy% 20Hierarchy%20Effects.pdf.
- Woolford, Ellen (2006). "Lexical case, inherent case, and argument structure". *Linguistic Inquiry* 37.1, 111–130. DOI: 10.1162/002438906775321175.
- Wurmbrand, Susi (2014). "The Merge condition: A syntactic approach to selection". In: *Minimalism and Beyond: Radicalizing the interfaces*. Ed. by Peter Kosta, Steven L. Franks, Teodora Radeva-Bork and Lilia Schürcks, 130–166. DOI: 10.1075/lfab.11. 06wur.
- Zeijlstra, Hedde (2012). "There is only one way to agree". *The Linguistic Review* 29.3, 491–539. DOI: 10.1515/tlr-2012-0017.
- Zúñiga, Fernando (2006). *Deixis and alignment: Inverse systems in indigenous languages of the Americas*. Amsterdam/Philadelphia: John Benjamins. DOI: 10.1075/tsl.70.

### Author index

Abbott, Barbara, 44 Adger, David, 5, 47, 77, 79, 144 Aissen, Judith, 3, 6, 9, 115, 213, 268 Aldridge, Edith, 225, 232 Alexiadou, Artemis, 268 Anagnostopoulou, Elena, 144, 267, 268 Anand, Pranav, 219, 224, 225, 229, 232 Arregi, Karlos, 257, 270 Assmann, Anke, 182 Baker, Mark C., 13, 21, 22, 220, 221, 226, 246, 265, 267, 268, 270 Bárány, András, 3, 26, 27, 33, 46, 53, 63, 65, 71, 74, 141, 142, 146, 148 Bartos, Huba, 22, 26, 32, 35, 43, 44, 46, 48-52, 55, 56, 58, 61, 63, 65, 67-71, 75, 77, 84, 93, 94, 105, 115, 136, 140, 275 Bazalgette, Tim, 2, 266 Beaudoin-Lietz, Christa, 270 Beghelli, Filippo, 53 Béjar, Susana, 5, 7, 9, 14, 16, 33, 48, 54, 76, 100, 106, 107, 109, 114, 122, 140, 144, 161, 165, 166, 170, 171, 176, 192, 213, 216, 269, 275 Benveniste, Émile, 5 Bernstein, Judy B., 48, 54, 57, 76, 94, 276 Bhatt, Rajesh, 13, 17, 22, 166, 219, 226, 230-232, 267, 268, 271 Biberauer, Theresa, 2, 265, 266, 268, 276 Bickel, Balthasar, 22, 219, 220, 236 Bierwisch, Manfred, 174, 227 Bittner, Maria, 220

Bobaljik, Jonathan David, 2, 22, 140, 199 - 202,217 - 226,228 - 230, 235, 242, 243, 247, 249, 250, 257, 262, 269, 276 Boeckx, Cedric, 265 Bossong, Georg, 3 Branigan, Phil, 22, 140, 199-202 Brody, Michael, 38 Caha, Pavel, 7, 174, 218, 227-229, 242, 247 Carnie, Andrew, 8, 213, 268 Cecchetto, Carlo, 145 Chisarik, Erika, 65 Chomsky, Noam, 4, 10–12, 14, 17, 20, 80, 174, 221 Chung, Sandra, 57 Comrie, Bernard, 3, 6, 9, 22, 115, 199 Coppock, Elizabeth, 26, 27, 32, 38-40, 44, 46, 48, 49, 51-53, 55, 57, 71-73, 75, 77, 78, 81, 82, 87, 88, 94, 99, 105, 115, 140, 143, 275 Croft, William, 6 D'Alessandro, Roberta, 188 Dalmi, Gréte, 26 Dalrymple, Mary, 9, 22, 26, 142, 245, 246, 267, 273 Danon, Gabi, 48, 54, 69, 76, 94, 267, 273, 274, 276 Dayal, Veneeta, 268 de Cuba, Carlos, 42, 88

Author index

de Hoop, Helen, 3, 161, 189, 195, 196, 211, 213, 214, 273 de Swart, Henriëtte, 29, 30 de Swart, Peter, 3, 189, 195, 197, 198, 211, 213, 214 Deal, Amy Rose, 167, 182, 183, 211, 212, 214 Déchaine, Rose-Marie, 51, 60 Dékány, Éva, 35, 54, 58-60, 65, 66 den Dikken, Marcel, 8, 18, 26, 35, 39, 42, 43, 58, 61, 65, 67, 74, 88, 105, 127, 129, 136, 138, 140, 141, 143-146, 148-151 Diesing, Molly, 31, 268 Dresher, B. Elan, 227, 266 Dryer, Matthew S., 186, 243 É. Kiss, Katalin, 22, 26, 27, 31, 33, 35-38, 41, 42, 44, 51, 52, 54, 58, 59, 61-65, 67-69, 74, 75, 77, 84, 94, 100, 105, 106, 116, 120, 127, 136, 140-142, 144-149, 204, 267, 274, 275 Ebert, Cornelia, 44 Embick, David, 18, 19, 122 Farkas, Donka, 29, 30, 75 Feldman, Harry, 4, 5, 22, 189-192, 195, 274 Filimonova, Elena, 6, 115 Finer, Daniel L., 168 Georgala, Effi, 259 Georgi, Doreen, 33, 67, 108, 115, 140, 161, 162, 171, 188, 195-197, 212-214, 226, 251 Gerland, Doris, 26 Gervain, Judit, 40, 88 Givón, Talmy, 6 Gulya, János, 28 H. Varga, Márta, 26, 43, 63

Hale, Ken, 220 Halle, Morris, 18, 122 Hallman, Peter, 31, 268 Harbour, Daniel, 5, 47, 77, 79, 144 Harley, Heidi, 5, 7, 18, 54, 106, 122, 170, 192, 214 Haspelmath, Martin, 3, 6, 9, 115, 187, 213, 244, 249, 250 Havas, Ferenc, 129, 135 Heck, Fabian, 162, 226 Himmelmann, Nikolaus P., 117 Hinterwimmer, Stefan, 44 Hinzen, Wolfram, 89 Holmberg, Anders, 2, 116, 265 Höhn, Georg F. K., 48, 76, 196, 198, 276 Iemmolo, Giorgio, 3, 9, 26, 273 Ionin, Tania, 44 Irimia, Monica-Alexandrina, 117 Jakobson, Roman, 174, 227 Jánosi, Adrienn, 43, 88 Jelinek, Eloise, 8, 213, 268 Kalin, Laura, 251-257 Kálmán, László, 31 Kamper, Gergely, 145 Karimi, Simin, 267, 268 Keenan, Edward L., 31, 198 Keine, Stefan, xvii, 6, 20, 22, 33, 108, 115, 140, 161, 162, 167-169, 174, 175, 188, 205-207, 211-214, 218, 223-227, 231, 232, 235, 238, 239, 242, 251 Kenesei, István, 40-42, 87 Kiefer, Ferenc, 63 Koopman, Hilda, 26 Kornai, András, 53, 64 Kornfilt, Jaklin, 268 Koul, Omkar N., 22, 162, 163, 165, 166, 175, 178, 182, 257, 258

Laczkó, Tibor, 65 Ladusaw, William A., 57 Lasnik, Howard, 11 Lazard, Gilbert, 3 Legate, Julie Anne, 22, 182, 217-219, 223-225, 227, 228, 232, 238, 242 Lewis, M. Paul, 204 Lochbihler, Bethany, 78, 106, 108, 140, 213, 214 Longobardi, Giuseppe, 4, 48, 54, 57, 76, 77, 94, 198, 276 López, Luis, 57, 267, 268 Lyons, Christopher, 6 Malchukov, Andrej L., 3, 115, 135, 161, 189, 190, 195-197, 211, 213, 214, 273 Maleczki, Márta, 31 Marantz, Alec, 18, 122, 220 Marcantonio, Angela, 142, 274 Maslova, Elena, 22, 204-206 McFadden, Thomas, 174, 220, 221, 227 McGinnis, Martha, 5, 7, 170, 192, 214 Mithun, Marianne, 78 Morimoto, Yukiko, 227 Moulton, Keir, 89 Müller, Gereon, 6, 20, 116, 162, 174, 182, 213, 226, 227, 251 Nash, Lea, 109 Nevins, Andrew, 5, 14, 219, 224, 225, 229, 232, 257, 270 Newmeyer, Frederick J., 3, 8, 265 Nichols, Johanna, 163 Nikolaeva, Irina, 9, 22, 26, 142, 245, 246, 267, 273 Næss, Åshild, 9 Nover, Rolf, 18, 19, 122 Odria, Ane, 257, 270 Ortmann, Albert, 26

Pandharipande, Rajeshwari V., 22, 223, 224, 238 Payne, John, 65 Peredy, Márta, 142 Pesetsky, David, 34, 253 Pete, István, 26 Piñón, Christopher, 31 Preminger, Omer, 14, 17, 18, 80, 171, 220, 224 Prince, Alan, 9, 213 Pylkkänen, Liina, 186 Rackowski, Andrea, 88 Rácz, Endre, 63 Rákosi, György, 39 Rebrus, Péter, 27, 137, 138 Rédei, Károly, 28 Rezac, Milan, 5, 7, 9, 14, 16, 33, 48, 54, 76, 100, 106, 107, 109, 114, 122, 140, 144, 161, 165, 166, 170, 171, 176, 182, 192, 213, 216, 224, 269, 275 Richards, Marc, 1, 4, 5, 8, 47, 54, 57, 76-78, 94, 144, 192, 198, 213, 216, 267, 268, 275, 276 Richards, Norvin, 79, 88, 256 Rigsby, Bruce, 22, 167–169, 182–184, 186 Ritter, Elizabeth, 5, 7, 54, 106, 170, 192, 214 Rizzi, Luigi, 116 Roberts, Ian, 2, 116, 117, 188, 215, 226, 265, 266, 268, 271, 276 Rocquet, Amélie, 39, 43, 88, 99, 136, 137, 140, 145 Rooryck, Johan, 129 Rounds, Carol, 26-28, 39, 46, 63 Rouveret, Alain, 109 Rude, Noel, 22, 167–169, 182–184, 186 Sag, Ivan A., 226

#### Author index

Schultze-Berndt, Eva, 117 Schwartz, Linda, 151 Scott, Graham, 22, 195-197, 199 Seržant, Ilja A., 9 Sheehan, Michelle, 2, 89, 182, 216, 265, 271, 276 Siddiqi, Daniel, 18, 19 Siewierska, Anna, 188, 215, 270 Silverstein, Michael, 3, 6, 9, 33, 115, 161, 214 Skorik, Piotr Ja, 199, 200 Smolensky, Paul, 9, 213 Sportiche, Dominique, 268 Stowell, Tim, 53 Szabolcsi, Anna, 26, 29, 31, 35-38, 46, 52, 58, 60-65, 67, 70, 73, 81, 84, 88, 89, 126, 132, 135, 139, 152 Szalontai, Ádám, 63, 71, 74 Takahashi, Shoichi, 89 Thráinsson, Höskuldur, 267 Torrego, Esther, 253 Törkenczy, Miklós, 26-28, 46, 63, 123,

- 129 Trommer, Jochen, 27, 125–127, 131,
- 134–136, 138, 139

Uriagereka, Juan, 145 Ürögdi, Barbara, 42, 88 Valentine, Randy, 213 van der Wal, Jenneke, 48, 76, 270 van Urk, Coppe, 251-253, 256, 257 Vergnaud, Jean-Roger, 10 Vikner, Sten, 267 Virtanen, Susanna, 22, 245 von Heusinger, Klaus, 44, 48 Wali, Kashi, 22, 162, 163, 165, 166, 175, 178, 182, 257, 258 Wechsler, Stephen, 26, 27, 32, 38-40, 44, 46, 48, 49, 51-53, 55, 57, 71, 75, 81, 82, 87, 94, 105, 140, 143 Wiese, Bernd, 174, 227 Wiltschko, Martina, 51, 60 Witzlack-Makarevich, Alena, 9 Woisetschlaeger, Erich, 37 Woolford, Ellen, 10, 182, 219, 225 Wurmbrand, Susi, 171 Yādava, Yogendra P., 22, 219, 220, 236 Zeijlstra, Hedde, 171 Zúñiga, Fernando, 78, 167, 168, 182